**Pan-Canadian Assessment Program** 

# **PCAP** 2019

# **Contextual Report on** Student Achievement in Mathematics



Pan-Canadian Assessment Program

# PCAP 2019

# Contextual Report on Student Achievement in Mathematics

#### Authors

Kathryn O'Grady, Council of Ministers of Education, Canada
Yitian Tao, Council of Ministers of Education, Canada
Allison Chapman-Chin, Council of Ministers of Education, Canada
Ashley Rostamian, Council of Ministers of Education, Canada
Elizabeth Costa, Council of Ministers of Education, Canada



The Council of Ministers of Education, Canada (CMEC) was formed in 1967 by the provincial and territorial ministers responsible for education to provide a forum in which they could discuss matters of mutual interest, undertake educational initiatives cooperatively, and represent the interests of the provinces and territories with national educational organizations, the federal government, foreign governments, and international organizations. CMEC is the national voice for education in Canada and, through CMEC, the provinces and territories work collectively on common objectives in a broad range of activities at the elementary, secondary, and postsecondary levels.

Through the CMEC Secretariat, the Council serves as the organization in which ministries and departments of education undertake cooperatively the activities, projects, and initiatives of particular interest to all provinces and territories. One of the activities on which they cooperate is the development and implementation of pan-Canadian testing based on contemporary research and best practices in the assessment of student achievement in core subjects.

#### Note of appreciation

The Council of Ministers of Education (Canada) would like to thank the students, teachers, and administrators whose participation in the Pan-Canadian Assessment Program ensured its success. The quality of your commitment has made this study possible. We are truly grateful for your contribution to a pan-Canadian understanding of educational policy and practices in reading, mathematics, and science at the Grade 8/Secondary II level.

#### Acknowledgements

Dr. Jennifer G. Anderson (Memorial University of Newfoundland) contributed to the literature review related to student indices.
Dr. Pierre Brochu (Consultation MEP Consulting Inc.) provided psychometric support.

Council of Ministers of Education, Canada 95 St. Clair West, Suite 1106 Toronto, Ontario M4V 1N6 Telephone: (416) 962-8100 Fax: (416) 962-2800 E-mail: cmec@cmec.ca

© 2022 Council of Ministers of Education, Canada

ISBN 978-0-88987-526-5

Ce rapport est également disponible en français.

| Introduction: What Is the Pan-Canadian Assessment Program?              | 1         |
|---|-----------|
| PCAP contextual guestionnaires  | 1         |
| Applications of PCAP data   | 2         |
| Objectives and organization of the report                               | 2         |
|   |           |
| Chapter 1: Student Demographic and Socioeconomic Characteristics        | 4         |
| Gender  | 4         |
| Gender in PCAP 2019   | 4         |
| Gender and mathematics achievement                                      | 6         |
| Language  | 7         |
| Learning in Canada's official languages                                 | 8         |
| Classifying language use in PCAP contextual data                        | 8         |
| Students' first languages   | 8         |
| Students' language use in everyday life                                 | 11        |
| Second-language study   | 12        |
| Student socioeconomic status  | 15        |
| Parents' education  | 15        |
| Books in students' homes  | 18        |
| Immigrant status  | 21        |
| Indigenous self-identity  | 24        |
| Summary   | 26        |
| Chapter 2: A Profile of Student Attitudes and Engagement in Mathematics | 28        |
| Chapter 2. A Frome of Student Attitudes and Engagement in Mathematics   | 20        |
| Attitude toward methomotics   | 30        |
| Attitude toward mathematics   | 50        |
| Mathematics sen-enicacy   | 50        |
| Time management   | 45<br>50  |
| Students' loarning experiences  | 50        |
| Linderstanding the language of mathematics                              | 55        |
| Activities used to support learning in mathematics                      | 55        |
| Activities used to support learning in mathematics                      | 05        |
| Supporting student rearining  | / I<br>71 |
| Homowork activities   | / 1<br>72 |
| Assossment  | ۲۷<br>72  |
| Rubrics   | 75<br>7/  |
|   | /4        |

| Feedback and support  |     |
|---|-----|
| Students' sense of belonging                                    |     |
| Cognitive demand in mathematical tasks                          | 80  |
| Level 1 problems  | 80  |
| Level 3 problems  | 81  |
| Cognitive demand of questions used in mathematics class         |     |
| Summary   | 83  |
| Chapter 2. Characteristics of Classrooms and Taashors           | 96  |
| Chapter 3: Characteristics of Classrooms and Teachers           |     |
| Classroom characteristics                                       |     |
| Class size  |     |
| Meeting the needs of all students                               |     |
| Class composition   |     |
| Substitute teachers   |     |
| Lost instructional time   |     |
| leacher characteristics   |     |
| Gender  |     |
| leacher specialization  |     |
| Professional development and self-efficacy                      |     |
| Professional development  |     |
| leacher self-efficacy   |     |
| Classroom practices   |     |
| Resources used for mathematics instruction                      |     |
| Probing student understanding                                   |     |
| Curriculum knowledge and understanding of vertical articulation |     |
| leachers' nomework expectations                                 |     |
| Assessment  | 123 |
| Challenges in teaching mathematics                              |     |
| Summary   | 128 |
| Chapter 4: Characteristics of Schools                           |     |
| School demographics and grade configuration                     | 129 |
| School demographics   | 130 |
| Grade configuration   | 138 |
| Diversity of the school population                              | 142 |
| Second-language learners  | 142 |
| Indigenous students   | 145 |
| Factors influencing learning                                    | 147 |
| Instructional time management                                   | 147 |
| Enrichment and extracurricular activities                       | 153 |
| Challenges to teaching and learning                             | 155 |
| Challenges to providing instruction                             | 155 |
| Challenges to learning  | 156 |
| Summary   | 160 |

| Conclusion            |     |
|-----------------------|-----|
| A profile of students |     |
| A profile of teaching |     |
| A profile of schools  |     |
| Final statement       | 164 |
| References            |     |
| Appendix: Data Tables |     |

#### LIST OF FIGURES

| FIGURE 1.1         | Percentage of students by gender self-identification                              | 5  |
|--------------------|---|----|
| FIGURE 1.2         | Achievement in mathematics by gender  | 7  |
| FIGURE 1.3         | Relationship between students' first language and mathematics achievement, by     |    |
|                    | language of the school system   | 11 |
| FIGURE 1.4         | Percentage of students enrolled in language-immersion programs, 2016 and 2019     | 13 |
| FIGURE 1.5         | Mathematics achievement by second-language learning status and language of the    |    |
|                    | school system   | 14 |
| FIGURE 1.6         | Percentage of students by their parents' education as reported by students        | 16 |
| FIGURE 1.7         | Relationship between parents' education and mathematics achievement               | 18 |
| FIGURE 1.8         | Percentage of students by the number of books in their home                       | 19 |
| FIGURE 1.9         | Relationship between the number of books in the home and mathematics              |    |
|                    | achievement   | 21 |
| FIGURE 1.10        | Percentage of students by immigrant status  | 22 |
| FIGURE 1.11        | Relationship between immigrant status and mathematics achievement                 | 23 |
| FIGURE 1.12        | Trends in mathematics achievement for Indigenous students, 2010–2019              | 26 |
| FIGURE 2.1         | Percentage of students by their responses to questionnaire items constituting     |    |
|                    | the attitude toward mathematics index   | 31 |
| FIGURE 2.2         | Results for the attitude toward mathematics index                                 | 32 |
| FIGURE 2.3         | Relationship between the attitude toward mathematics index and mathematics        |    |
|                    | achievement   | 33 |
| FIGURE 2.4         | Percentage of students by their responses to questionnaire items constituting the |    |
|                    | confidence with mathematical processes index                                      | 37 |
| FIGURE 2.5         | Results for the confidence with mathematical processes index                      | 37 |
| FIGURE 2.6         | Relationship between the confidence with mathematical processes index and         |    |
|                    | mathematics achievement   | 38 |
| FIGURE 2.7         | Percentage of students by their responses to questionnaire items constituting the |    |
|                    | confidence using technology in mathematics index                                  | 41 |
| FIGURE 2.8         | Results for the confidence using technology in mathematics index                  | 41 |
| FIGURE 2.9         | Relationship between the confidence using technology in mathematics index and     |    |
|                    | mathematics achievement   | 42 |
| FIGURE 2.10        | Percentage of students by their responses to questionnaire items constituting the |    |
|                    | student effort index  | 46 |
| FIGURE 2.11        | Results for the student effort index  | 46 |
| <b>FIGURE 2.12</b> | Relationship between the student effort index and mathematics achievement         | 47 |
| FIGURE 2.13        | Amount of time spent on homework each week  | 51 |
| FIGURE 2.14        | Relationship between mathematics homework effort and mathematics                  |    |
|                    | achievement   | 51 |
| FIGURE 2.15        | Student-reported school absences  | 52 |
| FIGURE 2.16        | Relationship between student absences and mathematics achievement                 | 52 |
| FIGURE 2.17        | Student-reported skipping and tardiness   | 53 |

| FIGURE 2.18 | Relationship between students' skipping and tardiness and mathematics             |
|-------------|---|
|             | achievement53   |
| FIGURE 2.19 | Time spent by students per week on activities outside of school hours             |
| FIGURE 2.20 | Percentage of students by their responses to questionnaire items constituting the |
|             | knowledge of general terms in mathematics index57                                 |
| FIGURE 2.21 | Results for the knowledge of general terms in mathematics index                   |
| FIGURE 2.22 | Relationship between the knowledge of general terms in mathematics index and      |
|             | mathematics achievement58   |
| FIGURE 2.23 | Percentage of students by their responses to questionnaire items constituting the |
|             | knowledge of geometry and measurement terms index61                               |
| FIGURE 2.24 | Results for the knowledge of geometry and measurement terms index                 |
| FIGURE 2.25 | Relationship between the knowledge of geometry and measurement terms index        |
|             | and mathematics achievement   |
| FIGURE 2.26 | Percentage of students by their responses to questionnaire items constituting the |
|             | teacher-directed mathematics activities index67                                   |
| FIGURE 2.27 | Results for the teacher-directed mathematics activities index67                   |
| FIGURE 2.28 | Relationship between the teacher-directed mathematics activities index and        |
|             | mathematics achievement68   |
| FIGURE 2.29 | Percentage of students who reported using what they learned in mathematics in     |
|             | other school subjects72   |
| FIGURE 2.30 | Types of homework assigned in mathematics class, as reported by students73        |
| FIGURE 2.31 | Types of assessment in mathematics classes, as reported by students74             |
| FIGURE 2.32 | Use of scoring rubrics in mathematics, as reported by students75                  |
| FIGURE 2.33 | Relationship between the frequency with which teachers use scoring rubrics and    |
|             | mathematics achievement75   |
| FIGURE 2.34 | Types of teacher feedback and support given in mathematics class, as reported by  |
|             | students76  |
| FIGURE 2.35 | Relationship between the frequency with which teachers provide feedback and       |
|             | support and mathematics achievement77   |
| FIGURE 2.36 | Students' sense of belonging in school78  |
| FIGURE 2.37 | Relationship between students' sense of belonging in school and mathematics       |
|             | achievement   |
| FIGURE 2.38 | Relationship between and students' response to the statement "I like school" and  |
|             | mathematics achievement79   |
| FIGURE 2.39 | Cognitive demand of questions used in mathematics82                               |
| FIGURE 2.40 | Relationship between the cognitive demand of mathematics questions and            |
|             | mathematics achievement83   |
| FIGURE 3.1  | Relationship between class size and mathematics achievement                       |
| FIGURE 3.2  | Percentage of teachers indicating that other adults were present in their         |
|             | mathematics class90   |
| FIGURE 3.3  | Relationship between the presence of another adult in the mathematics classroom   |
|             | and mathematics achievement91   |
| FIGURE 3.4  | Accommodations and modifications used in mathematics classrooms92                 |

| FIGURE 3.5         | Relationship between the frequency of accommodations and modifications for       |     |
|--------------------|--|-----|
|                    | students and mathematics achievement   | 93  |
| FIGURE 3.6         | Number of grade levels in mathematics classrooms                                 | 94  |
| FIGURE 3.7         | Mathematics achievement in single-grade and multi-grade classrooms               | 95  |
| FIGURE 3.8         | Number of days in the current school year taught by a substitute teacher         | 96  |
| FIGURE 3.9         | Relationship between the number of days taught by a substitute teacher and       |     |
|                    | mathematics achievement  | 96  |
| FIGURE 3.10        | Reasons for lost instructional time  | 97  |
| FIGURE 3.11        | Relationship between lost instructional time and mathematics achievement         | 98  |
| <b>FIGURE 3.12</b> | Relationship between gender of teacher and mathematics achievement               | 100 |
| <b>FIGURE 3.13</b> | Academic credentials of Grade 8/Secondary II mathematics teachers                | 101 |
| FIGURE 3.14        | Relationship between teacher education and mathematics achievement               | 102 |
| FIGURE 3.15        | Number of postsecondary mathematics courses teachers completed                   | 102 |
| FIGURE 3.16        | Areas of study during formal education and/or training                           | 103 |
| FIGURE 3.17        | Years of teaching experience   | 104 |
| FIGURE 3.18        | Relationship between years of teaching experience and mathematics                |     |
|                    | achievement  | 105 |
| <b>FIGURE 3.19</b> | Percentage of teachers' schedule assigned to mathematics                         | 106 |
| <b>FIGURE 3.20</b> | Relationship between percentage of teachers' schedule assigned to mathematics    |     |
|                    | and mathematics achievement  | 107 |
| <b>FIGURE 3.21</b> | Percentage of teachers identifying themselves as mathematics specialists         | 108 |
| <b>FIGURE 3.22</b> | Relationship between teacher specialization and student mathematics              |     |
|                    | achievement  | 108 |
| <b>FIGURE 3.23</b> | Number of days of professional development activities related to teaching        |     |
|                    | mathematics in the past five years   | 110 |
| <b>FIGURE 3.24</b> | Types of professional development activities for teachers in the past five years | 111 |
| <b>FIGURE 3.25</b> | Types of collaboration between mathematics teachers                              | 113 |
| <b>FIGURE 3.26</b> | Teachers' confidence in their ability to perform mathematics tasks               | 114 |
| <b>FIGURE 3.27</b> | Teachers' confidence in their ability to help their students understand          |     |
|                    | mathematics subdomains   | 114 |
| <b>FIGURE 3.28</b> | Resources used in mathematics instruction  | 117 |
| <b>FIGURE 3.29</b> | Teachers' use of opportunities for students to show understanding in             |     |
|                    | mathematics  | 118 |
| FIGURE 3.30        | Teachers' understanding of vertical articulation in mathematics curricula        | 119 |
| FIGURE 3.31        | Amount of time teachers expect students to spend on mathematics homework         |     |
|                    | per week   | 120 |
| FIGURE 3.32        | Relationship between the amount of mathematics homework assigned per week        |     |
|                    | and mathematics achievement  | 121 |
| FIGURE 3.33        | Types of homework assigned by mathematics teachers                               | 122 |
| FIGURE 3.34        | Teacher monitoring of students' homework   | 123 |
| FIGURE 3.35        | Types of questions teachers use on mathematics tests                             | 124 |
| FIGURE 3.36        | Relationship between frequency of extended-response items requiring multi-step   |     |
|                    | solutions and mathematics achievement  | 124 |
| FIGURE 3.37        | Types of questions teachers use in mathematics assessments to measure different  |     |
|                    | levels of thinking   | 125 |

| FIGURE 3.38        | Challenges in mathematics teaching, as reported by teachers                        | 126 |
|--------------------|--|-----|
| FIGURE 3.39        | Teachers' responses to questionnaire items related to the challenges to teaching   |     |
|                    | index  | 127 |
| FIGURE 3.40        | Relationship between the challenges to teaching index and mathematics              |     |
|                    | achievement  | 127 |
| FIGURE 4.1         | Size of communities in which schools are located                                   | 131 |
| FIGURE 4.2         | Relationship between the size of the community in which schools are located and    |     |
|                    | mathematics achievement  | 133 |
| FIGURE 4.3         | Percentage of public and private schools   | 134 |
| FIGURE 4.4         | Relationship between school governance and mathematics achievement                 | 135 |
| FIGURE 4.5         | Total enrolment in schools   | 136 |
| FIGURE 4.6         | Relationship between school size and mathematics achievement                       | 137 |
| FIGURE 4.7         | Number of grade levels in schools  | 138 |
| FIGURE 4.8         | Relationship between the number of grade levels in schools and mathematics         |     |
|                    | achievement  | 140 |
| FIGURE 4.9         | Grade 8/Secondary II enrolment in schools  | 140 |
| FIGURE 4.10        | Relationship between Grade 8/Secondary II enrolment and mathematics                |     |
|                    | achievement  | 142 |
| FIGURE 4.11        | Percentage of second-language learners in schools                                  | 143 |
| <b>FIGURE 4.12</b> | Percentage of students who identify as Indigenous in schools                       | 145 |
| FIGURE 4.13        | Percentage of semestered and full-year mathematics classes                         | 147 |
| FIGURE 4.14        | Minutes of mathematics instruction offered each week                               | 149 |
| FIGURE 4.15        | Relationship between minutes of mathematics instruction each week and              |     |
|                    | mathematics achievement  | 151 |
| FIGURE 4.16        | Percentage of schools offering daily mathematics instruction                       | 152 |
| FIGURE 4.17        | Relationship between daily mathematics instruction and mathematics                 |     |
|                    | achievement  | 153 |
| FIGURE 4.18        | Enrichment and extracurricular activities offered by schools                       | 154 |
| FIGURE 4.19        | Relationship between the enrichment and extracurricular activities index and       |     |
|                    | mathematics achievement  | 155 |
| FIGURE 4.20        | Challenges to providing instruction, as reported by school principals              | 156 |
| <b>FIGURE 4.21</b> | Percentage of schools by principals' responses to questionnaire items constituting | g   |
|                    | the challenges to learning index   | 157 |
| FIGURE 4.22        | Relationship between the challenges to learning index and mathematics              |     |
|                    | achievement  | 158 |
| FIGURE 4.23        | Percentage of students absent on a typical day for reasons other than school-      |     |
|                    | sponsored activities   | 159 |

#### LIST OF TABLES

| TABLE 1.1         | Percentage of students by gender self-identification, by language of the school system   | 6   |
|-------------------|--|-----|
| TABLE 1.2         | Percentage of students by their first language   | 9   |
| TABLE 1.3         | Percentage of students by their first language and language of the school system   | .10 |
| TABLE 1.4         | Percentage of students by the language they use in their everyday life   | .12 |
| TABLE 1.5         | Percentage of students enrolled in second-language programs, by language of the school system                                    | .14 |
| TABLE 1.6         | Percentage of students by their parents' education as reported by students, and language of the school system                    | 17  |
| TABLE 1.7         | Percentage of students by the number of books in their home and language of the school system                                    | 20  |
| TABLE 1.8         | Percentage of students by immigrant status and language of the school system   | 23  |
| TABLE 1.9         | Percentage of students by self-reported Indigenous identity  | .24 |
| <b>TABLE 1.10</b> | Percentage of students by self-reported Indigenous identity and language of  | 25  |
| TABLE 2.1         | Comparison of Canadian and provincial results, attitude toward mathematics   | 25  |
|                   | index  | 34  |
| TABLE 2.2         | Comparison of Canadian and provincial results by language of the school system, attitude toward mathematics index                | .34 |
| TABLE 2.3         | Summary of provincial results by language of the school system, attitude toward mathematics index                                | .35 |
| TABLE 2.4         | Comparison of Canadian and provincial results by gender, attitude toward mathematics index                                       | .35 |
| TABLE 2.5         | Summary of provincial results by gender, attitude toward mathematics index   | .36 |
| TABLE 2.6         | Comparison of Canadian and provincial results, confidence with mathematical  | 39  |
| TABLE 2.7         | Comparison of Canadian and provincial results by language of the school system, confidence with mathematical processes index.    | .39 |
| TABLE 2.8         | Summary of provincial results by language of the school system, confidence with mathematical processes index                     | .39 |
| TABLE 2.9         | Comparison of Canadian and provincial results by gender, confidence with mathematical processes index                            | .40 |
| TABLE 2.10        | Summary of provincial results by gender, confidence with mathematical processes index  | .40 |
| <b>TABLE 2.11</b> | Comparison of Canadian and provincial results, confidence using technology in mathematics index                                  | .43 |
| TABLE 2.12        | Comparison of Canadian and provincial results by language of the school system, confidence using technology in mathematics index | .43 |
| <b>TABLE 2.13</b> | Summary of provincial results by language of the school system, confidence using technology in mathematics index                 | .44 |
| TABLE 2.14        | Comparison of Canadian and provincial results by gender, confidence using technology in mathematics index                        | .44 |

| <b>TABLE 2.15</b> | Summary of provincial results by gender, confidence using technology in mathematics index | лл         |
|-------------------|---|------------|
| TABLE 2 16        | Comparison of Canadian and provincial results, student effort index                       | 44         |
| TABLE 2.10        | Comparison of Canadian and provincial results by language of the school system            |            |
|                   | student effort index  | 48         |
| TABLE 2 18        | Summary of provincial results by language of the school system student effort             | 40         |
|                   | index   | 49         |
| TABLE 2 19        | Comparison of Canadian and provincial results by gender, student effort index             | <u>ر</u> ب |
| TABLE 2.19        | Summary of provincial results by gender student effort index                              | <u>م</u>   |
| TABLE 2.21        | Principal component analysis of mathematics terms   | 56         |
| TABLE 2.22        | Comparison of Canadian and provincial results, knowledge of general terms in              |            |
|                   | mathematics index   | 59         |
| <b>TABLE 2.23</b> | Comparison of Canadian and provincial results by language of the school system,           |            |
|                   | knowledge of general terms in mathematics index   | 59         |
| <b>TABLE 2.24</b> | Summary of provincial results by language of the school system, knowledge of              |            |
|                   | general terms in mathematics index  | 60         |
| <b>TABLE 2.25</b> | Comparison of Canadian and provincial results by gender, knowledge of general             |            |
|                   | terms in mathematics index  | 60         |
| <b>TABLE 2.26</b> | Summary of provincial results by gender, knowledge of general terms in                    |            |
|                   | mathematics index   | 60         |
| <b>TABLE 2.27</b> | Comparison of Canadian and provincial results, knowledge of geometry and                  |            |
|                   | measurement terms index   | 64         |
| <b>TABLE 2.28</b> | Comparison of Canadian and provincial results by language of the school system,           |            |
|                   | knowledge of geometry and measurement terms index   | 64         |
| IABLE 2.29        | Summary of provincial results by language of the school system, knowledge of              | <b>C A</b> |
| T                 | geometry and measurement terms index  | 64         |
| IABLE 2.30        | Comparison of Canadian and provincial results by gender, knowledge of geometry            | сг         |
| TABLE 2.21        | Summary of provincial results by gondor knowledge of geometry and                         | 05         |
| IABLE 2.51        | measurement terms index   | 65         |
| TABLE 2 32        | Questionnaire items about activities in mathematics class                                 | 66         |
| TABLE 2.32        | Comparison of Canadian and provincial results, teacher-directed mathematics               | 00         |
|                   | activities index  | 69         |
| <b>TABLE 2.34</b> | Comparison of Canadian and provincial results by language of the school system.           |            |
|                   | teacher-directed mathematics activities index   | 69         |
| <b>TABLE 2.35</b> | Summary of provincial results by language of the school system, teacher-directed          |            |
|                   | mathematics activities index  | 70         |
| <b>TABLE 2.36</b> | Comparison of Canadian and provincial results by gender, teacher-directed                 |            |
|                   | mathematics activities index  | 70         |
| <b>TABLE 2.37</b> | Summary of provincial results by gender, teacher-directed mathematics activities          |            |
|                   | index   | 70         |
| TABLE 3.1         | Percentage of teachers by size of Grade 8/Secondary II mathematics classes                | 87         |
| <b>TABLE 3.2</b>  | Percentage of teachers by size of Grade 8/Secondary II mathematics classes and            |            |
|                   | language of the school system   | 89         |
| TABLE 3.3         | Percentage of mathematics teachers by gender  | 99         |
| TABLE 3.4         | Percentage of teachers by their perceptions of the impact of professional                 |            |
|                   | development activities on student learning  | .112       |

| TABLE 3.5           | Percentage of teachers by level of agreement with statements about factors      |       |
|---------------------|---|-------|
|                     | influencing student performance in mathematics                                  | 115   |
| TABLE 3.6           | Percentage of teachers by level of agreement with statements about attitudes    |       |
|                     | related to teaching and learning mathematics                                    | 116   |
| TABLE 4.1           | Definitions of communities by population size                                   | 130   |
| TABLE 4.2           | Percentage of schools by size of communities in which they are located and      |       |
|                     | language of school system   | 132   |
| TABLE 4.3           | Percentage of public and private schools, by language of the school system      | 135   |
| TABLE 4.4           | Percentage of schools by total enrolment and language of the school system      | 137   |
| TABLE 4.5           | Percentage of schools by number of grade levels and language of school system   | 139   |
| TABLE 4.6           | Percentage of schools by enrolment of Grade 8/Secondary II students and         |       |
|                     | language of the school system   | 141   |
| TABLE 4.7           | Percentage of schools by proportion of second-language learners and language    |       |
|                     | of the school system  | 144   |
| TABLE 4.8           | Percentage of schools by proportion of students who identify as Indigenous and  |       |
|                     | language of the school system   | 146   |
| TABLE 4.9           | Percentage of semestered and full-year mathematics classes, by language of      | -     |
|                     | the school system   | 148   |
| <b>TABLE 4.10</b>   | Percentage of schools by minutes of mathematics instruction offered each week   |       |
|                     | and language of the school system   | 150   |
| <b>TABLE 4.11</b>   | Percentage of schools offering daily mathematics instruction, by language of    |       |
|                     | the school system   | 152   |
| TABLE 4.12          | Questionnaire items related to school-level challenges to learning              | 157   |
| <b>TABLE 4.13</b>   | Percentage of schools by proportion of students absent on a typical day for     | 4 - 0 |
|                     | reasons other than school-sponsored activities and language of school system    | 159   |
| TABLE A.1.1         | Percentage of students by gender self-identification                            | 180   |
| TABLE A.1.2         | Percentage of students by gender self-identification and language of school     | 100   |
| T                   | System  | 101   |
| TABLE A.1.3         | Achievement in mathematics by gender, Canada                                    | 101   |
| TABLE A.1.4         | Percentage of students by first language spoken                                 | 101   |
| IABLE A.1.5         | Percentage of students by first language spoken and language of school system   | 182   |
| IABLE A.1.6         | Wathematics achievement by students' first language and language of the school  | 400   |
| T                   | system, Canada  | 182   |
| IABLE A.1.7         | Percentage of students by language used in everyday life and language of school | 183   |
| IABLE A.1.8         | Percentage of students by language used in everyday life and language of school | 18/   |
|                     | Percentage of students enrolled in language-immersion programs                  | 125   |
| TABLE A.1.3         | Percentage of students by enrolment in second-language programs and language    | 105   |
| TABLE A.I.IU        | of school system Canada   | 185   |
| <b>TABLE A.1.11</b> | Mathematics achievement by second-language learning status and language of      |       |
|                     | school system. Canada   | 186   |
| <b>TABLE A.1.12</b> | Percentage of students by their parents' education as reported by students      | 186   |
| <b>TABLE A.1.13</b> | Percentage of students by their parents' education as reported by students, and |       |
|                     | language of school system   | 187   |
| <b>TABLE A.1.14</b> | Mathematics achievement by parents' education as reported by students.          | •.    |
|                     | Canada  | 188   |
| <b>TABLE A.1.15</b> | Percentage of students by the number of books in their home                     | 188   |
|                     |   |       |

| <b>TABLE A.1.16</b>                          | Percentage of students by the number of books in their home and language of   |          |
|--|---|----------|
|  | school system   | .189     |
| <b>TABLE A.1.17</b>                          | Mathematics achievement by the number of books in the home, Canada  | 189      |
| <b>TABLE A.1.18</b>                          | Percentage of students by immigrant status  | .190     |
| <b>TABLE A.1.19</b>                          | Percentage of students by immigrant status and language of school system  | 190      |
| <b>TABLE A.1.20</b>                          | Mathematics achievement by immigrant status, Canada   | .191     |
| <b>TABLE A.1.21</b>                          | Percentage of students by their Indigenous identity as reported by students   | 191      |
| <b>TABLE A.1.22</b>                          | Percentage of students by their Indigenous identity as reported by students and language of school system           | 192      |
| TABLE A.1.23                                 | Mathematics achievement by Indigenous identity as reported by students,   | 102      |
|  | Distribution of students by questionnaire items constituting the attitude toward                                    | .195     |
| IABLE A.2.1.1                                | mathematics index. Canada   | 104      |
| Table A 2 1 2                                | Mathematics muex, Canada  | 105      |
| IABLE A.2.1.2           Table A.2.1.2        | Attitude toward mathematics index scores  | 105      |
| IABLE A.2.1.3                                | Attitude toward mathematics index scores by language of the school system   | .195     |
| IABLE A.2.1.4                                | Attitude toward mathematics index scores by gender  | .196     |
| TABLE A.2.1.5                                | Mathematics achievement by the attitude toward mathematics index, Canada  | 196      |
| TABLE A.2.2.1                                | Distribution of students by questionnaire items constituting the confidence with                                    |          |
|  | mathematical processes index, Canada  | .197     |
| <b>TABLE A.2.2.2</b>                         | Confidence with mathematical processes index scores   | 197      |
| <b>TABLE A.2.2.3</b>                         | Confidence with mathematical processes index scores by language of the school system                                | 198      |
| <b>TABLE A.2.2.4</b>                         | Confidence with mathematical processes index scores by gender   | 198      |
| TABLE A.2.2.5                                | Mathematics achievement by the confidence with mathematical processes index,  | 199      |
| TABLE A.2.3.1                                | Distribution of students by questionnaire items constituting the confidence using                                   | .135     |
|  | technology index Canada   | 199      |
| TABLE A.2.3.2                                | Confidence using technology index scores  | 200      |
|  | Confidence using technology index scores by language of the school system   | 200      |
| TABLE A.2.3.5<br>TABLE $\Delta$ 2 3 $\Delta$ | Confidence using technology index scores by ranguage of the school system   | 201      |
| TABLE A.2.3.4                                | Mathematics achievement by the confidence using technology index. Canada  | 201      |
| TABLE A 2 $/$ 1                              | Distribution of students by questionnaire items constituting the student effort                                     |          |
| IADLE A.2.4.1                                | index Canada  | 202      |
|  | Student effort index scores   | 202      |
| TABLE A.2.4.2<br>TABLE A $2/1/2$             | Student effort index scores by language of the school system  | 203      |
| TADLE A.2.4.3                                | Student effort index scores by ranguage of the school system  | 203      |
| TABLE A.2.4.4                                | Mathematics ashievement by the student effort index. Canada   | .204     |
| TABLE A.2.4.5                                | Distribution of students by hours non-wook sport on homowork. Conside   | .204     |
| IABLE A.2.4.0                                | Distribution of students by nours per week spent on nomework, Canada  | 205      |
| IABLE A.Z.4.7                                | Canada  | .205     |
| <b>TABLE A.2.4.8</b>                         | Distribution of students by the number of days absent or late for school, Canada                                    | 205      |
| <b>TABLE A.2.4.9</b>                         | Distribution and scores of students by the number of days absent or late for schoo<br>in the past two weeks. Canada | l<br>206 |
| <b>TABLE A.2.4.10</b>                        | Distribution of students by hours per week spent on activities outside of school                                    |          |
|  | hours. Canada   | .206     |
| <b>TABLE A.2.5.1</b>                         | Distribution of students by questionnaire items constituting the knowledge of                                       |          |
|  | general terms in mathematics index. Canada  | 207      |
|  |   | .207     |

| <b>TABLE A.2.5.2</b>  | Knowledge of general terms in mathematics index scores                            | .208 |
|-----------------------|---|------|
| <b>TABLE A.2.5.3</b>  | Knowledge of general terms in mathematics index scores by language of             |      |
|                       | the school system   | .208 |
| <b>TABLE A.2.5.4</b>  | Knowledge of general terms in mathematics index scores by gender                  | .209 |
| <b>TABLE A.2.5.5</b>  | Mathematics achievement by the knowledge of general terms in mathematics          |      |
|                       | index, Canada   | .209 |
| <b>TABLE A.2.6.1</b>  | Distribution of students by questionnaire items constituting the knowledge of     |      |
|                       | geometry and measurement terms index, Canada                                      | .210 |
| <b>TABLE A.2.6.2</b>  | Knowledge of geometry and measurement terms index scores                          | .210 |
| <b>TABLE A.2.6.3</b>  | Knowledge of geometry and measurement terms index scores by language of the       |      |
|                       | school system   | .211 |
| <b>TABLE A.2.6.4</b>  | Knowledge of geometry and measurement terms index scores by gender                | .211 |
| <b>TABLE A.2.6.5</b>  | Mathematics achievement by the knowledge of geometry and measurement              |      |
|                       | terms index, Canada   | .212 |
| <b>TABLE A.2.7.1</b>  | Distribution of students by questionnaire items constituting the teacher-directed |      |
|                       | mathematics activities index, Canada  | .213 |
| <b>TABLE A.2.7.2</b>  | Teacher-directed mathematics activities index scores                              | .214 |
| <b>TABLE A.2.7.3</b>  | Teacher-directed mathematics activities index scores by language of the school    |      |
|                       | system  | .214 |
| <b>TABLE A.2.7.4</b>  | Teacher-directed mathematics activities index scores by gender                    | .215 |
| <b>TABLE A.2.7.5</b>  | Mathematics achievement by the teacher-directed mathematics activities index,     |      |
|                       | Canada  | .215 |
| TABLE A.2.8           | Distribution of students who reported using what they learned in mathematics in   |      |
|                       | other subjects, Canada  | .216 |
| TABLE A.2.9           | Distribution of students by type of mathematics homework assigned, Canada         | .216 |
| <b>TABLE A.2.10.1</b> | Distribution of students by type of mathematics assessment method, Canada         | .217 |
| TABLE A.2.10.2        | Distribution of students by use of scoring rubrics in mathematics, Canada         | .217 |
| TABLE A.2.10.3        | Mathematics achievement of students by the frequency that teachers use rubrics,   |      |
|                       | Canada  | .217 |
| <b>TABLE A.2.11.1</b> | Distribution of students by teacher feedback and support in mathematics class,    |      |
|                       | Canada  | .218 |
| <b>TABLE A.2.11.2</b> | Mathematics achievement of students by teacher feedback and support in            |      |
|                       | mathematics class, Canada   | .218 |
| <b>TABLE A.2.12</b>   | Distribution and scores of students by students' sense of belonging in school,    |      |
|                       | Canada  | .219 |
| <b>TABLE A.2.13.1</b> | Distribution of students by type of problems encountered in mathematics class or  |      |
|                       | assignment, Canada  | .219 |
| <b>TABLE A.2.13.2</b> | Mathematics achievement of students by type of problems encountered in            |      |
|                       | mathematics class or assignment, Canada   | .220 |
| <b>TABLE A.3.1.1</b>  | Class size in Grade 8/Secondary II mathematics classes                            | .220 |
| TABLE A.3.1.2         | Achievement in mathematics by class size, Canada                                  | .221 |
| <b>TABLE A.3.1.3</b>  | Class size in Grade 8/Secondary II mathematics classes by language of the school  |      |
|                       | system  | .221 |
| TABLE A.3.2.1         | Percentage of teachers indicating that other adults were present in their         |      |
|                       | mathematics class   | .222 |
| TABLE A.3.2.2         | Achievement in mathematics by presence of other adults in the mathematics         |      |
|                       | classroom, Canada   | .222 |

| TABLE A.3.3           | Percentage of teachers and mathematics achievement by accomodations and           |   |
|-----------------------|---|---|
|                       | modifications used in mathematics classrooms, Canada22                            | 3 |
| <b>TABLE A.3.4.1</b>  | Percentage of teachers by number of grade levels in their mathematics             |   |
| T                     | Classrooms  | 4 |
| IABLE A.3.4.2         | Achievement in mathematics by number of grade levels, Canada                      | 4 |
| IABLE A.3.5.1         | Percentage of teachers by number of days taught by a substitute teacher22         | 4 |
| TABLE A.3.5.2         | Achievement in mathematics by number of days taught by a substitute teacher,      | 5 |
| TABLE A.3.6.1         | Percentage of teachers by reasons for lost instructional time Canada 22           | 5 |
| TABLE A.3.6.2         | Achievement in mathematics by reasons for lost instructional time. Canada 22      | 5 |
| TABLE A.3.7.1         | Percentage of teachers by gender self-identification                              | 6 |
| TABLE A.3.7.2         | Achievement in mathematics by gender self-identification of teacher Canada 22     | 6 |
| TABLE A.3.8.1         | Percentage of teachers and mathematics achievement by academic credentials        | Ŭ |
|                       | Canada  | 7 |
| <b>TABLE A.3.8.2</b>  | Percentage of teachers by courses taken during postsecondary studies, Canada22    | 7 |
| <b>TABLE A.3.8.3</b>  | Percentage of teachers by areas of study during formal pre-service education or   |   |
|                       | training, Canada  | 7 |
| <b>TABLE A.3.8.4</b>  | Percentage of teachers by years of teaching experience                            | 8 |
| <b>TABLE A.3.8.5</b>  | Achievement in mathematics by teacher's years of teaching experience, Canada22    | 8 |
| <b>TABLE A.3.8.6</b>  | Percentage of teachers by proportion of schedule assigned to mathematics          | 9 |
| <b>TABLE A.3.8.7</b>  | Achievement in mathematics by proportion of teacher's schedule assigned to        |   |
|                       | mathematics, Canada   | 9 |
| <b>TABLE A.3.8.8</b>  | Percentage of teachers by self-assessed specialization in mathematics             | 0 |
| <b>TABLE A.3.8.9</b>  | Achievement in mathematics by teacher's self-assessed specialization in           |   |
|                       | mathematics, Canada   | 0 |
| <b>TABLE A.3.9.1</b>  | Percentage of teachers by number of days of professional development activities   |   |
|                       | related to the teaching of mathematics in the past five years                     | 1 |
| <b>TABLE A.3.9.2</b>  | Percentage of teachers by types of professional development and their perceived   |   |
|                       | impact on student learning, Canada23  | 2 |
| <b>TABLE A.3.9.3</b>  | Percentage of teachers by types of collaboration with other mathematics teachers, |   |
|                       | Canada23  | 3 |
| <b>TABLE A.3.10</b>   | Percentage of teachers by confidence in their ability to do mathematics and to    |   |
|                       | help students understand mathematics, Canada23                                    | 3 |
| <b>TABLE A.3.11.1</b> | Percentage of teachers by level of agreement with statements about factors        |   |
|                       | influencing student performance in mathematics, Canada23                          | 4 |
| TABLE A.3.11.2        | Percentage of teachers by level of agreement with statements about attitudes      |   |
|                       | related to teaching and learning mathematics, Canada23                            | 4 |
| TABLE A.3.12.1        | Percentage of teachers by resources used in mathematics instruction, Canada23     | 5 |
| <b>TABLE A.3.12.2</b> | Percentage of teachers by use of variety of opportunities for students to show    |   |
|                       | understanding in mathematics, Canada23  | 5 |
| <b>TABLE A.3.12.3</b> | Percentage of teachers by amount of time in mathematics class spent on            |   |
|                       | mathematics subdomains, Canada23  | 6 |
| <b>TABLE A.3.12.4</b> | Percentage of teachers by understanding of vertical articulation, Canada23        | 6 |
| TABLE A.3.13.1        | Percentage of teachers by time they expect students to spend on mathematics       |   |
|                       | homework each week  | 7 |
| TABLE A.3.13.2        | Achievement in mathematics by time teachers expect students to spend on           |   |
|                       | mathematics homework each week, Canada23  | 7 |

| TABLE A.3.13.3        | Percentage of teachers by types of homework assigned, Canada                           | .238 |
|-----------------------|--|------|
| <b>TABLE A.3.13.4</b> | Percentage of teachers by frequency of monitoring student homework, Canada             | .238 |
| TABLE A.3.14.1        | Percentage of teachers by types of questions used on mathematics assessment,<br>Canada | 239  |
| TABLE A.3.14.2        | Achievement in mathematics by frequency of teachers' use of extended-response          | .200 |
|                       | items requiring multi-step solutions, Canada   | .239 |
| <b>TABLE A.3.14.3</b> | Percentage of teachers by mathematics assessment used to measure different             |      |
|                       | levels of thinking, Canada   | .239 |
| <b>TABLE A.3.15.1</b> | Percentage of teachers by challenges to mathematics teaching, Canada                   | .240 |
| TABLE A.3.15.2        | Relationship between the challenges to teaching index and mathematics                  |      |
|                       | achievement, Canada  | .241 |
| <b>TABLE A.4.1.1</b>  | Percentage of schools by size of communities in which they are located                 | .241 |
| <b>TABLE A.4.1.2</b>  | Percentage of schools by size of communities in which they are located and             |      |
|                       | language of school system  | .242 |
| <b>TABLE A.4.1.3</b>  | Achievement in mathematics by size of communities in which schools are located,        |      |
|                       | Canada   | .243 |
| <b>TABLE A.4.2.1</b>  | Percentage of public and private schools   | .243 |
| <b>TABLE A.4.2.2</b>  | Percentage of public and private schools and language of school system                 | .244 |
| <b>TABLE A.4.2.3</b>  | Achievement in mathematics by school governance, Canada                                | .244 |
| <b>TABLE A.4.3.1</b>  | Percentage of schools by total enrolment   | .244 |
| <b>TABLE A.4.3.2</b>  | Percentage of schools by total enrolment, by language of the school system             | .245 |
| <b>TABLE A.4.3.3</b>  | Achievement in mathematics by total enrolment, Canada                                  | .246 |
| <b>TABLE A.4.4.1</b>  | Percentage of schools by number of grade levels  | .246 |
| <b>TABLE A.4.4.2</b>  | Percentage of schools by number of grade levels and language of school system          | .247 |
| <b>TABLE A.4.4.3</b>  | Achievement in mathematics by number of grade levels, Canada                           | .248 |
| <b>TABLE A.4.5.1</b>  | Percentage of schools by total enrolment of Grade 8/Secondary II students              | .248 |
| <b>TABLE A.4.5.2</b>  | Percentage of schools by enrolment of Grade 8/Secondary II students and                |      |
|                       | language of the school system  | .249 |
| <b>TABLE A.4.5.3</b>  | Achievement in mathematics by total enrolment of Grade 8/Secondary II students,        |      |
|                       | Canada   | .250 |
| <b>TABLE A.4.6.1</b>  | Percentage of schools by proportion of second-language learners in schools             | .250 |
| <b>TABLE A.4.6.2</b>  | Percentage of schools by proportion of second-language learners and language of        |      |
|                       | the school system  | .251 |
| <b>TABLE A.4.7.1</b>  | Percentage of schools by proportion of students who identify as Indigenous in          | 252  |
| T                     | Schools  | .252 |
| IABLE A.4.7.2         | Percentage of schools by proportion of students who identify as indigenous and         | 252  |
|                       | Deresentess of schools by semester and full year classes                               | .253 |
| TABLE A.4.8.1         | Percentage of schools by semester and full year classes and language of school         | .254 |
| IABLE A.4.8.2         | system   | .254 |
| <b>TABLE A.4.9.1</b>  | Percentage of schools by minutes of mathematics instruction per week                   | .255 |
| <b>TABLE A.4.9.2</b>  | Percentage of schools by minutes of mathematics instruction offered each week          |      |
|                       | and language of the school system  | .256 |
| <b>TABLE A.4.9.3</b>  | Achievement in mathematics by minutes of mathematics instruction per week,             |      |
|                       | Canada   | .257 |
| TABLE A.4.10.1        | Percentage of schools by daily mathematics instruction                                 | .257 |
| TABLE A.4.10.2        | Percentage of schools by daily mathematics instruction and language of school          |      |
|                       | system   | .258 |
|                       |  |      |

| <b>TABLE A.4.10.3</b> | Achievement in mathematics by daily mathematics instruction, Canada2                | 58 |
|-----------------------|---|----|
| <b>TABLE A.4.11.1</b> | Percentage of schools by enrichment and extracurricular activities, Canada2         | 59 |
| <b>TABLE A.4.11.2</b> | Relationship between the enrichment and extracurricular activities index and        |    |
|                       | mathematics achievement, Canada2  | 59 |
| <b>TABLE A.4.12</b>   | Percentage of schools by challenges to providing instruction due to shortage or     |    |
|                       | inadequacy of resources, Canada2  | 60 |
| <b>TABLE A.4.13.1</b> | Percentage of schools by challenges to learning index, Canada2                      | 60 |
| TABLE A.4.13.2        | Relationship between the challenges to learning index and mathematics               |    |
|                       | achievement, Canada2  | 61 |
| <b>TABLE A.4.14.1</b> | Percentage of schools by proportion of students absent on a typical day for reasons |    |
|                       | other than school-sponsored activities2   | 61 |
| <b>TABLE A.4.14.2</b> | Percentage of schools by proportion of students absent on a typical day for reasons |    |
|                       | other than school-sponsored activities and language of school system                | 62 |
|                       |   |    |

The Pan-Canadian Assessment Program (PCAP) is a collaborative project that provides data on student achievement in Canadian provinces and territories.<sup>1</sup> It is part of the ongoing commitment of the Council of Ministers of Education, Canada (CMEC) to inform Canadians about how well their education systems are meeting the needs of students and society. Every three years, close to 30,000 Grade 8/Secondary II<sup>2</sup> students from across Canada are assessed with respect to their achievement of the curricular expectations common to all provinces and territories in three core learning domains: reading, mathematics, and science. The information gained from this pan-Canadian assessment provides ministers of education and other stakeholders with a basis for examining their provincial curriculum and other aspects of their school systems.

School programs and curricula vary from province to province and from territory to territory across the country, so comparing results in these domains is a complex task. However, young Canadians in different provinces and territories learn many similar skills in reading, mathematics, and science. PCAP has been designed to determine whether students across Canada reach similar levels of performance in these core disciplines at about the same age, and to complement existing provincial/ territorial assessments with comparative Canada-wide data on the achievement levels attained by Grade 8/Secondary II students. *PCAP 2019: Assessment Framework* (CMEC, 2020) provides the theoretical underpinnings, design principles, and performance descriptors that were used to develop test items in each of the three domains for the second cycle of PCAP (2016–23).<sup>3</sup>

Initial results from the PCAP 2019 assessment were released in *PCAP 2019: Report on the Pan-Canadian Assessment of Mathematics, Reading, and Science* (O'Grady, Houme, et al., 2021). In that report, results in mathematics, reading, and science were presented for Canada overall and for individual provinces. Results were further broken down by language of the school system and by gender.

The present report is the second of two reports providing results from PCAP 2019. While the first focused on the achievement results in the three domains assessed by PCAP, this report complements it by looking at contextual variables associated with mathematics achievement.

#### PCAP contextual questionnaires

Students participating in PCAP, and their teachers and school principals, complete questionnaires that are designed to provide all provinces and territories with contextual information to aid in the interpretation of the performance results. Researchers, policy-makers, and practitioners can use the information provided by these questionnaires to help them determine what factors influence learning outcomes. The content of the contextual questionnaires changes depending on which of the three domains is the primary focus of the PCAP assessment.

<sup>&</sup>lt;sup>1</sup> All ten provinces have participated in each PCAP administration. The three territories did not participate in PCAP 2019.

<sup>&</sup>lt;sup>2</sup> PCAP is administered to students in Secondary II in Quebec and Grade 8 in the rest of Canada.

<sup>&</sup>lt;sup>3</sup> During the first cycle of PCAP (2007–13), individual domains were the primary focus in a different year: reading in 2007, mathematics in 2010, and science in 2013. The pattern is being repeated during the second cycle (2016–23). Due to the global pandemic, the next administration of PCAP will be delayed from 2022 to 2023.

Because the primary domain of the 2019 PCAP assessment was mathematics, some contextual questions in the student questionnaire addressed factors that have been found in past studies to correlate with mathematics achievement. Some examples of these correlates include parental level of education, language spoken in the home, and the number of books in the home.

Contextual questionnaires completed by teachers cover questions about teaching and learning conditions, including teachers' homework expectations, assessment practices, areas of specialization, and years of teaching experience. The school questionnaire, completed by the principal, is the key source of information about each school, including the structure and organization of the school; school climate; school policies and practises; and curriculum and instruction.

The PCAP questionnaires are available on the CMEC website, at https://cmec.ca/697/PCAP\_2019. html. Access to the PCAP data set is available upon request.

#### Applications of PCAP data

PCAP is designed as a system-level assessment to be used primarily by provincial ministries of education to monitor and assess their respective educational systems. PCAP results are reported only at the pan-Canadian and provincial levels, by language of the school system, and by gender. They are not included in students' academic records, and no results for individual students, schools, or school boards/districts are reported by CMEC.

The goal of national (and international) large-scale assessments is to provide reliable information about academic achievement and insight into contextual factors associated with it. The data from studies such as PCAP provide policy-makers, administrators, teachers, and researchers with meaningful insights into the functioning of education systems and how they might be improved.

#### Objectives and organization of the report

This report presents the contextual results of the 2019 Pan-Canadian Assessment Program. It describes student, teacher, and school factors related to mathematics learning and teaching in Canada. Although the questionnaires cover many relevant areas, only a select number of results are presented here for illustrative purposes.

Results are reported at both pan-Canadian and provincial levels, with comparisons across participating provinces, and, where relevant, with other large-scale assessment surveys. Prince Edward Island and Newfoundland and Labrador did not oversample separately by language in order to examine difference between the performance of students in their English- and French-language school systems; consequently, results for only English-language schools are provided for these two provinces throughout this report; results for French-language schools are included only in the calculations for the overall Canadian and provincial totals and means.

The report includes four content chapters and a conclusion.

Chapter 1 presents data on five student demographic and socioeconomic characteristics: gender, language, socioeconomic status, immigrant status, and Indigenous identity.<sup>4</sup>

<sup>&</sup>lt;sup>4</sup> Only students attending schools under provincial jurisdiction participated in this study. Federally funded, on-reserve schools did not participate in PCAP.

Chapter 2 presents information on student indices that are correlated with mathematics performance.

Chapter 3 focuses on the learning context in Canadian classrooms, as well as the practices of Grade 8/ Secondary II teachers. It examines teachers' instructional strategies, tools, and activities and the relationship of those variables with achievement in mathematics. Assessment practices and their relationship to achievement in mathematics are also presented.

Chapter 4 explores issues surrounding time management in schools, including scheduled learning time, homework, out-of-class activities, and time lost to absenteeism. The chapter also provides an overview of Canadian schools that includes demographic information, factors influencing learning, and challenges to teaching and learning.

Major findings are summarized in the Conclusion. Finally, The appendix comprises tables with detailed data underpinning the findings discussed in this report.

A vast array of literature has illustrated that learning outcomes are affected by a student's individual and family demographic and socioeconomic characteristics. Academic achievement has been found to correlate, over time, with desirable social and personal outcomes, including better health, improved economic outcomes, political engagement, and overall well-being (Anderson & Winthrop, 2016; OECD, 2012; Onuzo et al., 2013). Family characteristics have been found to be a reliable predictor of a child's success at school, as well as their future career aspirations (Galliott et al., 2015; Hill et al., 2004).

This chapter presents the results of analyses of performance in the Pan-Canadian Assessment Program (PCAP) based on some background characteristics of participating Grade 8/Secondary II students. Five demographic and socioeconomic characteristics of students in Canada are examined for correlations with achievement in mathematics. These are gender, language, socioeconomic status, immigrant status, and Indigenous self-identity. Although these variables are reported separately in this report, many of them interact in producing observed patterns of academic achievement. For this reason, it is difficult to isolate the effects of any one variable on PCAP scores. Note that achievement results may differ slightly from those reported in the previous PCAP 2019 report (O'Grady, Houme, et al., 2021) because data for only those students who completed both the cognitive and questionnaire items are included in the present report.

#### Gender

Gender disparities in educational attainment are of considerable concern because they can have future consequences for personal and professional choices of young men and women. The gender gap in science, technology, engineering, and mathematics (STEM) fields has been well documented (see Kahn & Ginther, 2017, for a review of this literature). Such gender disparity is seen as detrimental because STEM careers are among the highest paying and fastest growing occupations. Although participation in STEM-related courses has been found to be similar among sexual majority and minority students in high school (Gottfried et al., 2015), little is known about the experiences of sexual minorities in STEM careers.

#### Gender in PCAP 2019

Inclusive education is valued in Canadian provinces and territories and has led to the development of policies and resources to support inclusion. One aspect of inclusive education relates to gender identity. In the PCAP 2019 student, teacher, and school questionnaires, the question about gender was expanded from the female/male choices of previous assessments to allow two additional choices, as shown in the box below.

| How do you identify yourself? |                                   |  |  |  |  |  |
|-------------------------------|-----------------------------------|--|--|--|--|--|
| 0                             | Male                              |  |  |  |  |  |
| 0                             | Female                            |  |  |  |  |  |
| 0                             | I identify myself in another way. |  |  |  |  |  |
| 0                             | l prefer not to say.              |  |  |  |  |  |

In Canada overall, 96.6 per cent of students identified themselves as female or male, with 48.3 percent identifying with each gender (Figure 1.1). A small proportion of students chose to identify themselves in another way (1.6 percent) or preferred not to say (1.8 percent). Similar proportions are observed in the provinces, with those who chose to identify themselves in another way ranging from 1.2 percent in Ontario to 2.5 percent in Alberta. The proportion of those who preferred not to say ranged from 1.1 percent in New Brunswick to 2.9 percent in Prince Edward Island; however, fewer than 30 students choose this option in 4 of the 10 provinces (Appendix A.1.1). Some variations are observed between anglophone and francophone school systems, as shown in Table 1.1 (Appendix A.1.2). Particularly for populations<sup>5</sup> with small sample sizes, such variations may be partly a result of the whole-class sampling process used in PCAP.



#### FIGURE 1.1 Percentage of students by gender self-identification

Note: Percentages may not add up to 100 due to rounding

<sup>&</sup>lt;sup>5</sup> "Population" refers to the respective official-language groups within each province.

|     | Anglophone school systems |      |  |                        | Francophone school systems |      |  |                        |
|-----|---------------------------|------|--|------------------------|----------------------------|------|--|------------------------|
|     | Female                    | Male | l identify<br>myself in<br>another way | l prefer<br>not to say | Female                     | Male | l identify<br>myself in<br>another way | l prefer<br>not to say |
| BC  | 47.6                      | 48.5 | 1.7                                    | 2.2                    | 51.7                       | 43.4 | 2.8‡                                   | 2.1‡                   |
| AB  | 50.2                      | 45.9 | 2.5                                    | 1.4                    | 47.3                       | 45.9 | 4.8‡                                   | 2.0‡                   |
| SK  | 49.9                      | 46.9 | 1.8                                    | 1.4                    | 40.7                       | 59.3 | 0.0                                    | 0.0                    |
| MB  | 49.4                      | 47.0 | 1.6                                    | 2.1                    | 45.3                       | 51.1 | 0.7‡                                   | 2.8‡                   |
| ON  | 48.9                      | 48.3 | 1.2                                    | 1.6                    | 43.4                       | 53.6 | 0.7‡                                   | 2.3                    |
| QC  | 48.1                      | 47.8 | 2.5‡                                   | 1.6‡                   | 46.3                       | 49.7 | 1.7                                    | 2.2                    |
| NB  | 49.2                      | 48.1 | 1.4‡                                   | 1.4‡                   | 47.4                       | 51.5 | 0.5‡                                   | 0.6‡                   |
| NS  | 46.4                      | 50.1 | 1.9                                    | 1.6‡                   | 45.2                       | 54.1 | 0.7‡                                   | 0.0                    |
| PE  | 44.0                      | 51.5 | 1.5‡                                   | 3.0‡                   |                            |      |  |                        |
| NL  | 48.4                      | 47.3 | 2.3‡                                   | 2.0‡                   |                            |      |  |                        |
| CAN | 48.9                      | 47.8 | 1.6                                    | 1.7                    | 46.1                       | 50.1 | 1.6                                    | 2.2                    |

# TABLE 1.1 Percentage of students by gender self-identification, by language of the school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

#### Gender and mathematics achievement

As shown in Figure 1.2, there was no difference between girls and boys in mathematics achievement at the pan-Canadian level, whereas students who identified themselves in another way or preferred not to say scored lower than those who identified themselves as female or male (Appendix A.1.3). The lack of difference between girls and boys in mathematics achievement is consistent with the results for Grade 8/Secondary II students in PCAP 2010 (CMEC, 2011), when mathematics was first the major domain. However, this result differs from the most recent international large-scale assessments in which Canada participated: boys outperformed girls in mathematics at the Grade 4 level in the 2019 administration of the Trends in International Mathematics and Science Study (TIMSS) (O'Grady, Rostamian, et al., 2021) and at age 15 in the 2018 administration of the Programme for International Student Assessment (PISA) (O'Grady, Deussing, et al., 2019).

#### FIGURE 1.2 Achievement in mathematics by gender



\* Significant difference compared to the *female* category

According to Cvencek, Meltzoff, and Greenwald (2011), a widespread cultural stereotype exists with respect to gender and mathematics, and their findings suggest that this stereotype gains acceptance early — as early as Grade 2 — and influences students' self-concept in relation to mathematics even before ages at which achievement gaps in mathematics emerge. Perceiving an academic field to be at odds with one's identify or self-concept has been shown to deter people from pursuing study in that field (Cheryan et al., 2009).

"Numeracy" is generally defined as the ability to understand and work with numbers; however, the definition of numeracy varies widely in academic literature and curriculum documents. Borgonovi, Choi, and Paccagnella (2018) have explored the evolution of gender gaps related to numeracy from childhood to adulthood in 23 countries, including Canada. Using data from TIMSS, PISA, and the Programme for International Assessment of Adult Competencies (PIAAC), an international household survey of adults aged 16 to 65, they have suggested that the gender gap favouring males in numeracy is smallest at age 10 and largest at age 27. Results from PIAAC reveal that females report using their numeracy skills less frequently, both at home and at work, compared to males (Arora & Pawlowski, 2017), and that gender differences in the use of numeracy skills at work are found even within the same occupational groups (Lindemann, 2015).

#### Language

Canada is a multilingual and multicultural country with various immigrant and Indigenous populations. In the 2016 Census, over 200 languages were reported as a mother tongue (Statistics Canada, 2017c). "Mother tongue," as used in Statistics Canada data reports, may be considered synonymous with "first language spoken." Canada's language groups may be classified into three distinct categories: official languages, non-official or heritage languages, and Indigenous languages (Duff & Becker-Zayas, 2017).

#### Learning in Canada's official languages

The two official languages of instruction in Canada are English and French, but the majority of students in Canada receive their first-language instruction in English. Canada's federal government and provincial and territorial governments, both in principle and practice, support opportunities for all Canadians to learn one or both of Canada's official languages (Government of Canada, 2017; Statistics Canada, 2016a). To ensure that all students have the opportunity to learn both of Canada's official languages, all school systems offer English or French as second language courses, and French immersion programs are offered in public education systems throughout Canada.<sup>6</sup> Some provinces also offer bilingual programs that combine instruction in an official language and a heritage language or an Indigenous language. As well, many schools offer second-language courses in languages other than French or English (Government of Canada, 2017).

PCAP samples are representative of both majority and minority official language groups<sup>7</sup> in the eight provinces that have sufficient numbers for valid statistical comparisons. Within anglophone school systems, although students in French immersion classes could opt, at the discretion of the school, to complete PCAP in either English or French, their results are reported with the English-language cohort.

#### Classifying language use in PCAP contextual data

*First language:* "First language" or "mother tongue" refers to the first language that the child learned in their family. In some families, children may have more than one language as their first language. In the PCAP 2019 student questionnaire, "first language" was explained as "the language you first learned and still understand."

*Language used in everyday life:* As students learn in school, expand their peer networks, and otherwise interact outside of their families, they may continue to use their first language, or they may come to adopt another language for most of their everyday communication. Some students maintain active fluency in more than one language. In the PCAP 2019 student questionnaire, students were asked about the languages they used outside of school (e.g., with family or friends, or in the community).

*Language of instruction:* Most Canadian students learn in one of Canada's two official languages. Some students learn in bilingual programs that combine instruction in a heritage language or an Indigenous language with one of Canada's official languages. In the student questionnaire, students were asked about enrolment in language-immersion programs, as well as second-language programs.

#### Students' first languages

Official languages and bilingualism are an important aspect of the Canadian identity. According to the 2016 Census, the proportion of population by mother tongue was 57 percent English, 21 percent French, and 22 percent other languages, of which over 215 were identified. When Canadians were asked about the first official language they spoke, the proportions were 75 percent English, 23 percent French, and 2 percent neither English nor French (Statistics Canada, 2019).

For a more detailed description of language policies in Canada, see the country chapter for Canada in the *PIRLS 2016 Encyclopedia* (Mullis, Martin, Goh, & Prendergast, 2017).

<sup>&</sup>lt;sup>7</sup> With respect to the two official languages in Canada, English is the majority language outside of Quebec — across the country, 64 percent of Canadians report speaking English most often at home. In Quebec, French is the majority language — 79 percent of people in Quebec report speaking French most often (Statistics Canada, 2020a).

PCAP results by province, and by the language of the school system, reflect the unique linguistic profiles of Canada's provinces. The majority of students who participated in PCAP 2019 spoke one of Canada's official languages as their first language. Canada is officially bilingual, and the highest rates of bilingualism are found in Quebec (45 percent) and New Brunswick (34 percent) (Statistics Canada, 2019).

Immigration also affects the first language of students: Canadian census data from 2016 show that 72.5 percent of immigrants have a first language other than French or English (Statistics Canada, 2017c). Provinces are differently impacted by immigration, with immigrants heavily concentrated in Canada's urban centres in British Columbia, Alberta, Ontario, and Quebec (Statistics Canada, 2015).

Students participating in PCAP 2019 were asked which language they considered to be their first language (the language first learned and still understood). Table 1.2 shows that, Canada-wide, 66 percent of participating students identified English as their first language and 18 percent identified French as their first language. In Canada overall, 17 percent of students reported that their first language was a language other than English or French; within the provinces, the proportions ranged from 4 percent in Newfoundland and Labrador and Prince Edward Island to 22 percent in British Columbia. At the pan-Canadian level, less than 1 percent of students stated that an Indigenous language was their first language. The highest proportions of students who reported an Indigenous language as their first language were in Manitoba and Quebec (1.4 percent) (Appendix A.1.4).

|     | English | French | Indigenous language | Other language |
|-----|---------|--------|---------------------|----------------|
| BC  | 76.8    | 1.0    | 0.7 ‡               | 21.5           |
| AB  | 78.9    | 1.7    | 0.8 ‡               | 18.6           |
| SK  | 84.6    | 0.6    | 1.3 ‡               | 13.5           |
| MB  | 78.5    | 2.1    | 1.4                 | 18.0           |
| ON  | 77.6    | 2.4    | 0.4‡                | 19.6           |
| QC  | 13.2    | 77.6   | 1.4                 | 7.8            |
| NB  | 70.0    | 24.9   | 0.4‡                | 4.6            |
| NS  | 92.3    | 2.7    | 0.3‡                | 4.7            |
| PE  | 93.8    | 2.1‡   | 0.4‡                | 3.6‡           |
| NL  | 95.4    | 0.7‡   | 0.8‡                | 3.1            |
| CAN | 65.6    | 17.5   | 0.7                 | 16.2           |

#### TABLE 1.2 Percentage of students by their first language

‡ There are fewer than 30 observations.

Note: Percentages may not add up to 100 due to rounding.

Table 1.3 presents the first language reported by students by language of the school system. At the pan-Canadian level, close to 80 percent of students reported that their first language is the same as the language of the school system — English in anglophone systems, and French in francophone systems. The proportion of students whose first language is other than English, French, or an Indigenous language is higher in the anglophone system (18 percent) compared to the francophone system (8 percent). At the provincial level, the majority of students in anglophone school systems speak English as their first language, while approximately 1 in 5 students in British Columbia, Alberta, Manitoba, and Ontario reported that another language was their first language. It should also be noted

that, in about half of the provinces, the majority of students in the francophone school system speak English as their first language. The proportion of French-first-language students in francophone school systems ranges from over 80 percent in Quebec and New Brunswick to approximately 30 percent in British Columbia, Alberta, and Nova Scotia. The proportion of English-first-language students in francophone school systems is above 40 percent in all provinces except Quebec and New Brunswick, where the proportions are less than 20 percent (Appendix A.1.5). Only a small number of students in both school systems reported that their first language was an Indigenous language; similarly, only a small minority of students in francophone school systems reported having a language other than English, French, or an Indigenous language as their mother tongue.

| Anglophone school systems |         |        |                        |                   | Francophone school systems |        |                        |                   |
|---------------------------|---------|--------|------------------------|-------------------|----------------------------|--------|------------------------|-------------------|
|                           | English | French | Indigenous<br>language | Other<br>language | English                    | French | Indigenous<br>language | Other<br>language |
| вс                        | 76.9    | 0.8 ‡  | 0.7 ‡                  | 21.6              | 59.6                       | 29.3   | 0.0                    | 11.1‡             |
| AB                        | 79.4    | 1.3 ‡  | 0.8‡                   | 18.6              | 45.6                       | 33.4   | 0.0                    | 21.0              |
| SK                        | 84.9    | U ‡    | 1.3 ‡                  | 13.5              | 54.9                       | 35.4   | 1.4 ‡                  | 8.3 ‡             |
| MB                        | 79.5    | 0.8 ‡  | 1.4                    | 18.3              | 40.8                       | 53.2   | 0.4 ‡                  | 5.7 ‡             |
| ON                        | 79.0    | 0.6 ‡  | 0.4 ‡                  | 20.0              | 49.9                       | 37.6   | 0.3 ‡                  | 12.2              |
| QC                        | 70.8    | 17.8   | 1.4 ‡                  | 10.0              | 6.6                        | 84.5   | 1.4                    | 7.6               |
| NB                        | 92.3    | 1.6‡   | 0.6 ‡                  | 5.5               | 15.0                       | 82.6   | 0.1 ‡                  | 2.3 ‡             |
| NS                        | 93.5    | 1.4‡   | 0.3 ‡                  | 4.8               | 66.3                       | 31.0   | U ‡                    | 2.3 ‡             |
| PE                        | 95.3    | 0.6‡   | 0.4 ‡                  | 3.7 ‡             |                            |        |                        |                   |
| NL                        | 95.5    | 0.6‡   | 0.8‡                   | 3.1               |                            |        |                        |                   |
| CAN                       | 79.8    | 1.2    | 0.6                    | 18.3              | 12.2                       | 78.7   | 1.2                    | 8.0               |

#### TABLE 1.3 Percentage of students by their first language and language of the school system<sup>8</sup>

<sup>‡</sup> There are fewer than 30 observations.

U Too unreliable to be published

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

Figure 1.3 shows differences in achievement scores in mathematics between anglophone and francophone schools. Both English- and French-first-language students in francophone schools achieved higher scores in mathematics than their peers in anglophone schools; however, scores were similar for students who reported languages other than French or English in both language systems (Appendix A.1.6). The results for students reporting an Indigenous language as their first language are not reported here because the number of students was too small to allow reliable reporting.

<sup>&</sup>lt;sup>8</sup> Data quality indicators presented in this report are adapted from data accuracy standards set forth by Statistics Canada (Statistics Canada, 2009); estimates based on fewer than 30 observations are flagged with the symbol ‡, and estimates with a coefficient of variation greater than 33.3 percent are considered to be too unreliable to be published and are therefore suppressed and represented by a "U."

### FIGURE 1.3 Relationship between students' first language and mathematics achievement, by language of the school system



\* Significant difference compared to same language in anglophone school systems

#### Students' language use in everyday life

Linguistic diversity is on the rise in Canada. More that 7.5 million Canadians reported speaking a language other than English and French at home, an increase of 14.5 percent between 2011 and 2016 (Statistics Canada, 2017e). Students may master several languages, and the language of the school may not be the same as the one(s) they use outside the school (e.g., with family or friends, or in the community). Students who speak a language other than French or English as their first language are exposed to one or both of Canada's official languages when they enter the school system, and they tend to adopt an official language in their daily interactions (Duff & Becker-Zayas, 2017). When the language(s) of fluency are different from the language of instruction, school achievement may be impacted (Bruckauf, 2016; OECD, 2010, 2016b). Immigrant status, considered later in this chapter, influences the likelihood that a student will speak a different language at home than that in which they are taught.

As shown in Table 1.4, at the pan-Canadian level, over half of students speak predominantly English, more than 1 in 10 students speak predominantly French, and 7 percent of students report speaking English and French equally in their everyday lives. The highest proportion of students who reported using other languages indicated that they also spoke English outside of school (16 percent) (Appendix A.1.7). As expected, the majority of students who reported French as their everyday language reside in Quebec (60 percent) and New Brunswick (13 percent); student in these provinces also reported the highest level of bilingualism in the two official languages in Canada (18 percent). More than 20 percent of Ontario students reported speaking English and another unofficial language, and more the 10 percent of British Columbia students speak predominantly other languages outside of school. The proportion of students that speak mostly an Indigenous language in their everyday lives is 1 percent or less; however, in every province, fewer than 30 students selected this choice. Further breakdown by language of the school system can be found in Appendix A.1.8.

|     | English only<br>or mostly<br>English | French only<br>or mostly<br>French | English<br>and French<br>equally | English and<br>a language<br>other than<br>French | French and<br>a language<br>other than<br>English | Mostly an<br>Indigenous<br>Ianguage | Mostly<br>other<br>languages |
|-----|--------------------------------------|------------------------------------|----------------------------------|---|---|-------------------------------------|------------------------------|
| ВС  | 65.3                                 | 0.6                                | 2.1                              | 19.7  | 0.3 ‡   | 0.5 ‡                               | 11.6                         |
| AB  | 69.1                                 | 0.5 ‡                              | 2.4                              | 17.9  | U ‡   | 0.3 ‡                               | 9.4                          |
| SK  | 77.1                                 | 0.8 ‡                              | 2.4                              | 13.2  | 0.5 ‡   | 1.2 ‡                               | 4.8                          |
| MB  | 68.5                                 | 0.9                                | 4.6                              | 15.0  | 0.2 ‡   | 1.2 ‡                               | 9.4                          |
| ON  | 61.5                                 | 1.3                                | 4.1                              | 23.0  | 0.6   | 0.5 ‡                               | 9.0                          |
| QC  | 7.5                                  | 60.1                               | 18.0                             | 3.7   | 6.5   | 0.6 ‡                               | 3.5                          |
| NB  | 62.1                                 | 12.9                               | 18.0                             | 4.0   | 0.4 ‡   | 0.4 ‡                               | 2.3                          |
| NS  | 83.4                                 | 1.7                                | 6.4                              | 4.7   | 0.5 ‡   | 0.3 ‡                               | 2.9                          |
| PE  | 85.9                                 | 1.2 ‡                              | 6.3                              | 3.9 ‡   | 0.3 ‡   | 0.4 ‡                               | 2.1 ‡                        |
| NL  | 90.2                                 | 0.2 ‡                              | 2.5 ‡                            | 4.2   | 0.1 ‡   | 0.5 ‡                               | 2.4 ‡                        |
| CAN | 54.0                                 | 13.0                               | 6.7                              | 16.3  | 1.7   | 0.5                                 | 7.8                          |

#### TABLE 1.4 Percentage of students by the language they use in their everyday life

‡ There are fewer than 30 observations.

U Too unreliable to be published

#### Second-language study

A substantial minority of Canadian students are enrolled in immersion programs to learn a second language. Between 2016 and 2019, the proportion of students in second-language programs in Canadian public schools increased by approximately 3 percent, while those in French immersion programs increased by nearly 6 percent (Statistics Canada, 2021a). Enrolment in French immersion programs outside of Quebec has been growing in popularity, increasing by almost 45 percent since 2003 (Government of Canada, 2017).

Research has indicated that second-language learners face particular difficulties in problem solving in mathematics. Although some of these difficulties can be attributed to language transfer and reading skills (Farnia & Geva, 2011; Han, 2012), a cohort study on the components of working memory that underlie the acquisition of mathematical calculation skills in elementary students in the United States did not find an achievement gap in mathematics between English-language learners (ELL) and monolingual children over the three years of the study (Swanson et al., 2019). The authors concluded that working memory tasks, which are highly associated with the control and focus of attention, play an important role in the development of computation abilities in both ELL and monolingual children.

In the PCAP 2019 student questionnaire, an immersion program was described as a program in which the majority of a student's courses are taught in a second language. As shown in Figure 1.4, at the pan-Canadian level, one-quarter of students reported that they have been enrolled in an English or French immersion program to learn a second language (Appendix A.1.9). For descriptive purposes, data from PCAP 2016 are also shown in the figure. These indicate that student-reported enrolment in second-language programs has increased between PCAP 2016 and 2019 for both official languages; by contrast, the proportion of students who reported enrolment in immersion programs for other languages remained about the same. For PCAP 2019, 2 percent of students reported that they had

been enrolled in immersion programs for an Indigenous language. This question about Indigenouslanguage immersion was asked for the first time in 2019 and may provide interesting data in future administrations of PCAP. The results related to enrolment in all language-immersion programs should be treated with caution. In the student questionnaire, students were asked about enrolment in languageimmersion programs and enrolment in second-language programs. Students enrolled in programs to learn a second language may have misinterpreted the definition of an immersion program, leading to higher-than-expected proportions of students reporting enrolment in language-immersion programs.



FIGURE 1.4 Percentage of students enrolled in language-immersion programs, 2016 and 2019

Note: Information on Indigenous languages immersion programs was not collected in 2016.

As noted above, the student questionnaire also asked about second-language programs — programs that usually focus on speaking, listening, reading, and writing in the second language. As shown in Table 1.5, about 1 in 4 students in both language systems reported being enrolled in such programs at the time of the assessment (Appendix A.1.10). Compared to the previous administration of PCAP in 2016, the percentage of students currently enrolled in French-as-a-second-language programs in anglophone schools has increased from 23 to 25 percent, while the percentage of students currently enrolled in English-as-a-second-language programs in francophone schools has increased from 13 to 23 percent. The percentage of students who reported that they were previously enrolled in such programs has remained stable in anglophone systems while increasing in francophone systems (O'Grady, Fung, Brochu, et al., 2019, Table 1.7). Although students were also asked about second-language programs in Indigenous and other languages, the response rate was too low for the results to be valid.

## TABLE 1.5 Percentage of students enrolled in second-language programs, by language of the school system

|                                     | Anglophone s          | school systems         | Francophone school systems |                        |  |
|-------------------------------------|-----------------------|------------------------|----------------------------|------------------------|--|
|                                     | Currently<br>enrolled | Previously<br>enrolled | Currently<br>enrolled      | Previously<br>enrolled |  |
| English-second-language<br>program* | 18                    | 11                     | 23                         | 18                     |  |
| French-second-language<br>program** | 25                    | 12                     | 13                         | 6                      |  |

\* These include English-language-learner (ELL), English-as-an-additional-language (EAL), and English-as-a-second-language (ESL) programs.

\*\* These include extended and intensive French.

Figure 1.5 shows that, in Canada overall, students in francophone school systems who were currently enrolled or had never been enrolled in a second-language program achieved statistically similar mathematics scores; however, the scores of students who previously had been enrolled in such programs were lower than these other two groups. In anglophone school systems, students who were enrolled in a second-language program, either currently or in the past, achieved significantly lower mean scores in mathematics compared to students who had never been enrolled in such a program (Appendix A.1.11).

# FIGURE 1.5 Mathematics achievement by second-language learning status and language of the school system



\* Significant difference compared to the *never enrolled* category within each school system

#### Student socioeconomic status

Socioeconomic status (SES), broadly understood and measured as a combination of parental educational attainment and family income, is one of the strongest predictors of academic achievement (Bruckauf, 2016; OECD, 2012; Perry & McConney, 2010). SES, which comprises both cultural and economic factors, is difficult to measure and understand because it is a complex cluster of variables that include parents' occupations and educational attainment, learning resources in the home, and how parents communicate the value of education to their children, among other variables (Crowe, 2013; Chevalier et al., 2013). It is also difficult to isolate the effects of SES from those of other factors like geography, genetic endowment, school characteristics, and immigrant status (Causa et al., 2009; OECD, 2016b).

A consequence of SES and home environment is that educational attainment tends to have an intergenerational correlation: that is, highly educated parents are more likely to have children who attain more education, while parents with less education are more likely to have children who attain lower levels of education (Causa et al., 2009; Chevalier et al., 2013; Onuzo et al., 2013). Because educational attainment is a central component of social mobility (i.e., the relationship between the socioeconomic status of parents and that of their offspring when they become adults), policy-makers have a strong interest in improving educational outcomes for all students, regardless of their socioeconomic backgrounds (Chevalier et al., 2013). Fortunately, evidence suggests that well-structured policy interventions, such as income support policies, have a particularly strong positive effect on the most disadvantaged children and families (Causa et al., 2009; Merry, 2013).

In this report, two measures serve as proxies for socioeconomic status: parents' education and the number of books in the home. Both of these factors have consistent correlations with students' academic achievement, but these correlations should be interpreted carefully, with a recognition that many factors influence a family's ability to support their child's learning.

#### Parents' education

Students were asked about the highest level of education completed by one of their parents — specifically, the parent with the highest level of education. To better reflect current family structures, students were instructed to answer this question by thinking either about their own parent or the person who is most like a parent to them.

At the pan-Canadian level, 61 percent of students reported that one of their parents had a college diploma or university degree. Provincially, this proportion ranged from 51 percent in Manitoba to 65 percent in Ontario. The distribution of credentials below a degree is relatively consistent across the provinces. At the same time, over 20 percent of students did not know the educational attainment of their parents, both in Canada overall and in all provinces except Prince Edward Island, where 15 percent of students could not provide information on parental education (Figure 1.6, Appendix A.1.12).



#### FIGURE 1.6 Percentage of students by their parents' education as reported by students

\* "Some postsecondary" refers to any kind of education after high school.

Within provinces and at the pan-Canadian level, parents of students in both French- and Englishlanguage school systems had similar levels of educational attainment, although more students in the francophone system did not know about their parent's education (Table 1.6, Appendix A.1.13).
|     | University<br>degree(s) | College or<br>cégep diploma | Some<br>postsecondary* | Completed<br>high school | Did not<br>complete high<br>school | l don't know |
|-----|-------------------------|-----------------------------|------------------------|--------------------------|------------------------------------|--------------|
|     |                         |                             | Anglophone school      | systems                  |                                    |              |
| BC  | 48                      | 13                          | 4                      | 8                        | 3                                  | 24           |
| AB  | 45                      | 13                          | 3                      | 9                        | 4                                  | 26           |
| SK  | 40                      | 14                          | 5                      | 11                       | 5                                  | 24           |
| MB  | 38                      | 13                          | 3                      | 14                       | 5                                  | 26           |
| ON  | 47                      | 18                          | 4                      | 6                        | 3                                  | 22           |
| QC  | 50                      | 13                          | 7                      | 8                        | 3                                  | 20           |
| NB  | 45                      | 17                          | 5                      | 13                       | 4                                  | 18           |
| NS  | 45                      | 18                          | 3                      | 9                        | 4                                  | 20           |
| PE  | 45                      | 16                          | 2 ‡                    | 15                       | 7 ‡                                | 14           |
| NL  | 42                      | 17                          | 3                      | 9                        | 3                                  | 26           |
| CAN | 46                      | 16                          | 4                      | 8                        | 3                                  | 23           |
|     |                         |                             | Francophone school     | systems                  |                                    |              |
| BC  | 50                      | 4 ‡                         | 2 ‡                    | 5 ‡                      | 1‡                                 | 37           |
| AB  | 47                      | 12 ‡                        | U ‡                    | U ‡                      | 3 ‡                                | 35           |
| SK  | 53                      | 9‡                          | 1‡                     | 6‡                       | 2 ‡                                | 29 ‡         |
| MB  | 49                      | 8 ‡                         | 2 ‡                    | 7 ‡                      | 2 ‡                                | 33           |
| ON  | 49                      | 13                          | 2                      | 4                        | 1                                  | 31           |
| QC  | 44                      | 13                          | 3                      | 9                        | 4                                  | 28           |
| NB  | 39                      | 14                          | 2 ‡                    | 6                        | 3 ‡                                | 35           |
| NS  | 50                      | 9 ‡                         | U ‡                    | 9 ‡                      | 2 ‡                                | 30           |
| CAN | 44                      | 13                          | 3                      | 8                        | 3                                  | 29           |

# TABLE 1.6 Percentage of students by their parents' education as reported by students, and<br/>language of the school system

\* "Some postsecondary" refers to any kind of education after high school.

‡ There are fewer than 30 observations.

U Too unreliable to be published

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

Student achievement correlates positively with the highest educational levels achieved by parents of students. There was a linear relationship between parental education and student achievement, with the highest mathematics scores attained by students whose parents had completed one or more university degrees. Achievement was significantly lower for students who reported that their parents finished their studies at the high school level (Figure 1.7, Appendix A.1.14).

#### FIGURE 1.7 Relationship between parents' education and mathematics achievement



\* Significant difference compared to the completed high school category

## Books in students' homes

The provision of enriched home environments is associated with families with higher SES, which in turn in associated with an increased likelihood that children will succeed in school (Evans et al., 2014). Families with higher SES are able to provide their children with social and cultural capital that increases the probability of success in school (Crowe, 2013; Huang & Liang, 2016; Lam & Ho, 2013). They are also likely to be more involved in their children's education and to have more learning resources like books, puzzles, games, and computers in their homes (Crowe, 2013; Shipley, 2011). The number of books in students' homes has been found to correlate with SES and student academic achievement. In the research literature, books in the home are regarded as both a source of academic knowledge and skills and a measure of the parents' commitment to the education of their children (Evans et al., 2014).

The PCAP student questionnaire asked students about the number of books in their homes. As shown in Figure 1.8, the results are similar across most of the provinces. In Canada overall and in all provinces except Quebec, at least 20 percent of students reported that there were more than 200 books in their home. The largest proportions of students have between 26 and 100 books in the home: this ranged from 30 percent in Saskatchewan to 37 percent in Newfoundland and Labrador (Appendix A.1.15). Across provinces, the distribution of the number of books in the home was similar in anglophone and francophone school systems (Table 1.7, Appendix A.1.16).



### FIGURE 1.8 Percentage of students by the number of books in their home

|                           | 0 to 10 books | 11 to 25 books | 26 to 100 books    | 101 to 200 books | More than<br>200 books |  |
|---------------------------|---------------|----------------|--------------------|------------------|------------------------|--|
| Anglophone school systems |               |                |                    |                  |                        |  |
| BC                        | 9             | 13             | 34                 | 21               | 24                     |  |
| AB                        | 8             | 13             | 33                 | 23               | 22                     |  |
| SK                        | 11            | 16             | 30                 | 22               | 21                     |  |
| MB                        | 12            | 15             | 31                 | 19               | 23                     |  |
| ON                        | 9             | 16             | 35                 | 20               | 20                     |  |
| QC                        | 8             | 13             | 30                 | 26               | 22                     |  |
| NB                        | 8             | 12             | 30                 | 22               | 27                     |  |
| NS                        | 9             | 14             | 31                 | 22               | 24                     |  |
| PE                        | 8‡            | 13             | 35                 | 18               | 27                     |  |
| NL                        | 9             | 14             | 37                 | 20               | 20                     |  |
| CAN                       | 9             | 15             | 34                 | 21               | 22                     |  |
|                           |               | Francoph       | one school systems |                  |                        |  |
| вс                        | 11‡           | 12             | 28                 | 23               | 26                     |  |
| AB                        | 12‡           | 12 ‡           | 25                 | 29               | 22                     |  |
| SK                        | 6‡            | 16‡            | 28‡                | 23‡              | 27‡                    |  |
| MB                        | 4‡            | 11             | 31                 | 28               | 27                     |  |
| ON                        | 10            | 17             | 33                 | 20               | 19                     |  |
| QC                        | 14            | 19             | 36                 | 17               | 14                     |  |
| NB                        | 18            | 20             | 33                 | 15               | 14                     |  |
| NS                        | 11‡           | 18‡            | 29                 | 23               | 20‡                    |  |
| CAN                       | 14            | 19             | 35                 | 17               | 15                     |  |

#### TABLE 1.7 Percentage of students by the number of books in their home and language of the school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

There is a clear positive relationship between the number of books in the home and achievement in mathematics (Figure 1.9, Appendix A.1.17). These results confirm data obtained in previous PCAP administrations.

# FIGURE 1.9 Relationship between the number of books in the home and mathematics achievement



\* Significant difference compared to the 0 to 10 books category

The pattern of higher achievement in mathematics in homes with more educational resources is consistent with the finding at the Grade 4 level in the TIMSS 2019 study. In that study, the home resources learning scale combined data on the number of books in the home, home study supports (e.g., internet connection, students having their own room), and parental education. That study found that, in Canada, 36 percent of students had many resources available at home, while 64 percent of students had some or few resources at home. Having many resources at home corresponded to students reporting more than 100 books in the home; the home study supports of both an internet connection and their own room; parents reporting more than 25 children's books at home; and at least one parent who had completed university and one parent who had a professional occupation. Students who reported having many resources for learning at home achieved high scores in mathematics (O'Grady, Rostamian, et al., 2021).

### Immigrant status

Canada has the second-largest foreign-born population in the world in proportion to its overall population, behind only Australia (CMEC, 2015; Duff & Becker-Zayas, 2017; Parkin, 2015). Research has found that children in immigrant families are more likely to be educationally disadvantaged (Andon et al., 2014; Bruckauf, 2016; OECD, 2010). Using data from earlier cycles of PISA, TIMSS, and the Progress in International Reading Literacy Study (PIRLS), Andon et al. (2014) have concluded that an achievement gap exists between immigrant and non-immigrant students in the three domains of reading, mathematics, and science across Member countries of the Organisation for Economic Co-operation and Development (OECD).

In Canada, immigrants are more likely than non-immigrants to fall into low-income categories (Collin & Jensen, 2009; CMEC, 2015). Despite this disadvantage, Canada is among the OECD countries that are more successful in closing the "immigrant achievement gap" (Parkin, 2015; Wech & Weinkam, 2016). Comparisons of average achievement between students who are immigrants and those who were born in Canada must be treated with caution, as scores may obscure important disparities among immigrant groups (Schnepf, 2008). Immigrant children and youth are not homogeneous (Andon et al., 2014; OECD, 2010; Parkin, 2015; Schnepf, 2008; Wech & Weinkam, 2016). They vary with respect to where they completed their previous education, at what age they

were immersed in schooling in one of Canada's official languages, and whether they already spoke English or French upon arriving in Canada (Bruckauf, 2016; OECD, 2016a). Like their domesticborn counterparts, immigrant children and youth also vary in the levels of education held by their parents.

In PCAP 2019, Manitoba had the largest proportion of students who were not born in Canada, followed by Alberta, British Columbia, Saskatchewan, and Ontario (Figure 1.10; Appendix A.1.18). The proportion of students not born in Canada is similar in the anglophone and francophone school systems in Canada overall and in most provinces (Table 1.8, Appendix A.1.19).



#### FIGURE 1.10 Percentage of students by immigrant status

|     | Anglophone     | school systems     | Francophone school systems        |
|-----|----------------|--------------------|-----------------------------------|
|     | Born in Canada | Not born in Canada | Born in Canada Not born in Canada |
| BC  | 82             | 18                 | 82 18                             |
| AB  | 82             | 18                 | 76 24                             |
| SK  | 84             | 16                 | 82 18 ‡                           |
| MB  | 80             | 20                 | 87 13                             |
| ON  | 85             | 15                 | 86 14                             |
| QC  | 92             | 8                  | 91 9                              |
| NB  | 91             | 9                  | 95 5                              |
| NS  | 92             | 8                  | 94 6‡                             |
| PE  | 93             | 7 ‡                |                                   |
| NL  | 95             | 5                  |                                   |
| CAN | 84             | 16                 | 90 10                             |

#### TABLE 1.8 Percentage of students by immigrant status and language of the school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

As shown in Figure 1.11, mathematics achievement of students born in Canada was higher than that of students with an immigrant background (Appendix A.1.20). This differs from the results found for Grade 4 students in TIMSS 2019 (O'Grady, Rostamian, et al., 2021) and for 15-year-olds in PISA 2018 (O'Grady, Deussing, et al., 2019), where no significance difference in achievement was found between students born in Canada and those with an immigrant background. This is also the opposite of the results in PCAP 2010 (where mathematics was also the major domain), in which students born outside of Canada achieved significantly higher scores in mathematics compared to those born in Canada (CMEC, 2012b).

#### FIGURE 1.11 Relationship between immigrant status and mathematics achievement



\* Significant difference compared to the born in Canada category

## Indigenous self-identity

Achievement gaps between Indigenous and non-Indigenous students are a persistent educational issue in Canada. High-school non-completion for Indigenous youth living off reserve compared to non-Indigenous youth is twice as high for males (20 percent versus 9 percent) and three times higher for females (16 percent versus 5 percent) (Uppal, 2017). The urgency of improving outcomes for Indigenous students is stressed in *Truth and Reconciliation Commission of Canada: Calls to Action* (Truth and Reconciliation Commission of Canada, 2012).

Based on data from the 2016 Canadian Census, there are a total of 1,673,785 Indigenous peoples in Canada, representing 4.9 percent of the total population. Outside the territories, the proportion of Indigenous peoples is highest in Saskatchewan (16 percent) and Manitoba (18 percent) (Statistics Canada, 2017a). Among school-aged children and youth, Indigenous populations are growing at much faster rates than the general population (Statistics Canada, 2017b).

In responding to the PCAP student questionnaire, students could identify themselves as First Nations, Inuit, or Métis. As shown in Table 1.9, the highest proportions of students who identified themselves as First Nations or Métis were in Saskatchewan (11 percent and 8 percent, respectively) and Manitoba (9 percent for both populations). The number of students who identified themselves as Inuit was very small, in both Canada overall and all provinces (Table 1.9, Appendix A.1.21), and in both language systems (Table 1.10, Appendix A.1.22). Although students could select more than one option with respect to Indigenous identity, only a very small proportion of students across the provinces chose to do so. It is important to note that, although Indigenous peoples make up a large proportion of the population in Nunavut (86 percent), Northwest Territories (51 percent), and Yukon (23 percent) (Government of Canada, 2020), the three territories did not participate in PCAP 2019.

|     | Not Indigenous | First Nations | Inuit | Métis |
|-----|----------------|---------------|-------|-------|
| вс  | 90.6           | 5.7           | U ‡   | 2.4   |
| AB  | 90.1           | 4.2           | U ‡   | 3.8   |
| SK  | 79.4           | 10.9          | U ‡   | 7.5   |
| MB  | 79.7           | 9.4           | U ‡   | 9.4   |
| ‡ON | 94.8           | 3.6           | U ‡   | 0.9   |
| QC  | 93.1           | 3.0           | 0.6 ‡ | 1.9   |
| NB  | 91.9           | 5.6           | 0.1 ‡ | 1.2 ‡ |
| NS  | 89.1           | 6.7           | 0.3 ‡ | 2.4   |
| PE  | 95.4           | 3.6 ‡         | 0.2 ‡ | 0.3 ‡ |
| NL  | 87.5           | 6.5           | 2.4 ‡ | 1.6 ‡ |
| CAN | 92.0           | 4.4           | 0.3   | 2.3   |

### TABLE 1.9 Percentage of students by self-reported Indigenous identity

‡ There are fewer than 30 observations.

U Too unreliable to be published

|     | Anglophone school systems |       | Francophone school systems |               | ystems |       |
|-----|---------------------------|-------|----------------------------|---------------|--------|-------|
|     | First Nations             | Inuit | Métis                      | First Nations | Inuit  | Métis |
| BC  | 5.7                       | U‡    | 2.4                        | 2.4‡          | 0.0    | 4.8‡  |
| AB  | 4.3                       | U‡    | 3.8                        | U‡            | U‡     | U‡    |
| SK  | 11.0                      | U‡    | 7.4                        | 0.0           | 0.0    | 12.8‡ |
| MB  | 9.6                       | U‡    | 9.1                        | 0.3‡          | 0.6‡   | 23.7  |
| ON  | 3.6                       | U‡    | 0.8‡                       | 2.5           | U‡     | 4.1   |
| QC  | 4.8                       | 0.8‡  | 1.2‡                       | 2.7           | 0.6‡   | 2.0   |
| NB  | 6.1                       | 0.0‡  | 0.6‡                       | 4.3           | 0.3‡   | 2.5‡  |
| NS  | 6.8                       | 0.3‡  | 2.3                        | 5.8‡          | 1.6‡   | 2.8‡  |
| PE  | 3.5‡                      | 0.2‡  | 0.2‡                       |               |        |       |
| NL  | 6.5                       | 2.4‡  | 1.6‡                       |               |        |       |
| CAN | 4.9                       | 0.2   | 2.2                        | 2.7           | 0.6 ‡  | 2.3   |

# TABLE 1.10 Percentage of students by self-reported Indigenous identity and language of the school system

‡ There are fewer than 30 observations.

U Too unreliable to be published

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

The PCAP sample is randomly drawn from school systems under the purview of the provincial ministry/department of education. Schools in remote locations and with a very small number of Grade 8/Secondary II students (generally fewer than 5) can be exempted from participating in PCAP. In all provinces, fewer than 30 students self-identified as Inuit. Because of these small numbers, the results for Indigenous students cannot be considered representative and are included here for descriptive purposes only.

As shown in Figure 1.12, among Indigenous students, students who identified as Métis achieved the highest scores in mathematics in PCAP 2019. Although a test for significance cannot be calculated for changes over time between 2010, when mathematics was first the major domain, and 2019, Figure 1.12 illustrates the pattern of change over time for mathematics (Appendix A.1.23; CMEC, 2012b, Chart 3.19).



#### FIGURE 1.12 Trends in mathematics achievement for Indigenous students, 2010–2019

\* Significant difference compared to First Nations category

## Summary

This chapter has presented PCAP 2019 data related to five student demographic and socioeconomic characteristics: gender, language, socioeconomic status, immigrant status, and Indigenous self-identification.

With respect to gender, no gender gap was found in mathematics in PCAP 2019. This finding is consistent with the results in PCAP 2010, when mathematics was first the major domain. However, these results differ from the most recent international large-scale assessments in which Canada participated. Boys outperformed girls in mathematics at the Grade 4 level in TIMSS 2019 and at age 15 in PISA 2018.

Canada-wide, both English- and French-first-language students in francophone schools achieved higher scores in mathematics than their peers in anglophone schools. Scores were similar in both language systems for students who reported first languages other than French or English.

In anglophone school systems, students who were enrolled in a second-language program, either currently or in the past, achieved significantly lower mean scores in mathematics compared to students who had never been enrolled in such a program. In francophone school systems, students who were previously enrolled in second-language programs had lower achievement than those who were never enrolled; however, the difference in achievement between students currently enrolled compared to those who were never enrolled in second-language programs was not statistically significant.

Two proxies for socioeconomic status are used in PCAP contextual reports: parents' educational levels and the number of books in students' homes. Both measures were clearly correlated with student achievement. Students with parents who have a university-level education achieved

significantly higher scores in mathematics than those whose parents have less education. Mathematics achievement was also highest among students with the greatest number of books in their home.

In PCAP 2019, students who were not born in Canada achieved scores in mathematics that were statistically lower than those of their Canadian-born counterparts. Among Indigenous students, student who identified themselves as Inuit or Métis achieved higher scores in mathematics compared to those who self-identified as First Nations.

This chapter examines the relationship between student variables and mathematics performance. It is organized around three broad themes: students' attitudes and beliefs, students' learning experiences, and supporting student learning. Within each theme, indices and variables related to responses to the PCAP 2019 student questionnaire and cognitive test are analyzed.

The first section explores students' attitudes toward and beliefs about mathematics, looking at students' self-efficacy, their effort in doing mathematics, and their time management. The chapter then turns to students' learning experiences, including their knowledge of mathematics terms and the activities used in the classroom to support learning in mathematics. The next section looks at how teachers support student learning, including the cross-curricular integration of mathematics, homework activities, assessment, and feedback, along with an examination of students' sense of belonging. The final section presents information on the cognitive levels of types of questions used in mathematics lessons and assessment.

Some groups of questions from the student questionnaire were subjected to principal component analysis, which allowed researchers to identify items that were related to a single construct; these items were then used to form indices (see the box below for more information on principal components analysis and indices). Most of the indices showed a significant relationship with mathematics performance; however, unless stated otherwise, only the indices with a correlation coefficient equal to or above .20 are included in this report.

In addition to the indices, this chapter analyzes other items from the student questionnaire, either individually or in groups. In some cases, items or groups of items that have either a weak or no correlation with achievement are included for descriptive purposes. As with all self-reported data, the questionnaire responses are based on students' perceptions of the construct being measured.

#### Statistical note on principal components analysis and index scores

**Principal component analysis (PCA)**. To reduce the complexity of the analysis and to obtain more stable measures of attitudes, values, and learning experiences, some groups of questions from the PCAP 2019 questionnaires were subjected to PCA with direct oblimin rotation (delta = 0). This technique is designed to determine if item responses cluster together in some meaningful way. If meaningful groupings can be found, PCA permits the construction of a number of indices that combine individual items. As an example of the efficiency of this technique, applying PCA to responses about student attitudes toward mathematics yielded one index, reduced from 10 individual questionnaire items.

*Index scores*. A score for each student on each index is derived from the PCA, in much the same way as a scaled mathematics score is derived from analyzing the mathematics test items. Index scores are typically computed in standard score form, with a mean of 0 and a standard deviation of 1. For convenience in presentation, and to avoid negative values on charts, the scores are transformed into a mean of 50 and a standard deviation of 10 for Canada as a whole. This is analogous to the transformation of mathematics scores to a mean of 500 and a standard deviation of 100. However, the index scale is deliberately different, in order to avoid confusing index scores with achievement scores. Mean index scores for groups such as provinces should be examined in relation to the Canadian mean of 50 and standard deviation of 10. For example, a mean score of 52 for a group implies that the group is 0.20 standard deviation units above the Canadian mean for that index. It is important to stress that index scores should not be interpreted as percentages. It should also be noted that the Canadian means may not be exactly at 50, due to the use of unweighted data during the computation of index scores. However, weights were used for all analyses of the index scores.

**Quarters**. In this report, the PCAP populations of interest are divided into four equal groups (quarters or quartiles), with regard to the value of the index under study. The mean score for each of these groups appears in applicable tables and/or figures. The bottom quarter represents numbers below the 25<sup>th</sup> percentile; the second quarter represents the 25<sup>th</sup> to the 49<sup>th</sup> percentile; the third quarter represents the 50<sup>th</sup> to the 74<sup>th</sup> percentile; and the top quarter represents the 75<sup>th</sup> percentile and above.

*Index names*. It should be noted that, although the names of indices in this report are somewhat arbitrary, they are intended to capture the main underlying idea represented by the items that constitute a specific index. Sometimes the index name conveys the construct being investigated; in other cases, the underlying idea is more generic. Included throughout the report are tables that identify questionnaire items with the corresponding index name. These are intended to convey a sense of how the indices have been labelled.

**Statistical significance**. This report discusses only variables that show statistically significant relationships (p < 0.05) with mathematics achievement and indices with a correlation coefficient equal to or above .20, unless otherwise noted. When comparing index scores, statistical significance is determined using the unrounded values for the index.

## Students' attitudes and beliefs

Students attitudes toward and beliefs about mathematics have been a topic of great interest for many years, in part because many societies look to fields related to science, technology, engineering, and mathematics (STEM) to bolster economic growth. Attitude (e.g., liking or disliking geometry, preferences for discovery learning) can be separated from beliefs (e.g., about mathematics, about themselves, about teaching, about social context) and emotions (e.g., joy or frustration in solving problems or aesthetic responses to mathematics) (McLeod, 1992).

Despite the need for STEM-related skills in work and everyday life, and the number of career opportunities in those fields, a majority of students do not view mathematics as personally beneficial (Middleton et al., 2016). Such attitudes could influence the amount of effort a student puts into learning mathematics. Disentangling the impact of attitudes on learning by further disaggregating constructs can help to define "variables and fuzzy definitions" (Batchelor et al., 2019) and identify attributes that contribute to achievement.

## Attitude toward mathematics

Attitudes have a profound impact on teaching and learning mathematics. Distinguishing between what students can do and what they will do can reveal important information about their learning (Hattie, 2009). Children are more likely to engage in an activity if they expect to do it well and if they value the activity (Eccles et al., 1993; Simpkins et al., 2006). According to the Western and Northern Canadian Protocol, *Common Curriculum Framework for K–9 Mathematics* (Western and Northern Canadian Protocol, 2006), "Students with positive attitudes towards learning mathematics are likely to be motivated and prepared to learn, participate willingly in classroom activities, persist in challenging situations and engage in reflective practices" (p. 3).

### **Description of the index**

In PCAP 2019, students were asked to respond to 10 items concerning their attitudes toward mathematics, as shown in Figure 2.1 (Appendix A.2.1.1). Through the process of principal component analysis, this set of items formed the *attitude toward mathematics index*. This index measures students' perception of their ability to do mathematics as well as their general attitudes toward mathematics. In Figure 2.1, questionnaire items for this index are presented in descending order by the percentage of students who either agreed or strongly agreed with each statement. Overall in Canada, 83 percent of students agreed or strongly agreed that they understood most of the mathematics that they were taught. At the Grade 8/Secondary II level, close to three-quarters of students recognized the importance of mathematics in preparing them for future job opportunities, and almost two-thirds believed that mathematics would be useful in their everyday lives. The pattern of responses for tasks related to learning mathematics was generally positive. However, more than half of students reported that they did not like estimating or explaining how they did mathematics, and more than three-quarters did not like mathematics questions that involve a lot of reading.

# FIGURE 2.1 Percentage of students by their responses to questionnaire items constituting the attitude toward mathematics index



### **Results for the index**

Pan-Canadian results for the attitude toward mathematics index are shown in Figure 2.2. Students in anglophone school systems and boys scored higher on this index than students in francophone school systems or girls (Appendices A.2.1.2, A.2.1.3, A.2.1.4).





\* Significant difference compared to the reference (Ref.) in each category

In this index, the top quarter represents students who tend to have a more positive attitude toward mathematics. These students were more likely to agree or strongly agree with the items in Figure 2.1, while students in the bottom quarter of the index were more likely to disagree or strongly disagree with the items.



32 PCAP 2019 Contextual Report

Figure 2.3 shows the relationship between the attitude toward mathematics index and mathematics achievement. Generally, increasingly positive attitudes and beliefs are associated with higher achievement scores. Of the categories shown in Figure 2.3, the largest gap between the top and bottom quarters of the index is among anglophone students, with a difference of 70 points (Appendix A.2.1.5).



FIGURE 2.3 Relationship between the attitude toward mathematics index and mathematics achievement

\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to top quarter) within each category

As shown in Table 2.1, students in Ontario, Alberta, Saskatchewan, Manitoba, and New Brunswick achieved results similar to or above the Canadian mean on this index, meaning that students in these provinces reported the most positive attitudes toward mathematics (Appendix A.2.1.2).

#### TABLE 2.1 Comparison of Canadian and provincial results, attitude toward mathematics index

| Above* the Canadian mean | Similar to the Canadian mean                      | Below* the Canadian mean   |
|--------------------------|---|--|
| Ontario                  | Alberta, Saskatchewan, Manitoba,<br>New Brunswick | British Columbia, Quebec, Nova<br>Scotia, Prince Edward Island,<br>Newfoundland and Labrador |

\*Denotes significant difference

In Ontario, students in both language systems scored above the respective Canadian means on this index, while French-language students in British Columbia, Alberta, Saskatchewan, Manitoba, New Brunswick, and Nova Scotia also scored above the Canadian French mean. Students in anglophone school systems in Saskatchewan and Manitoba scored at the Canadian English mean. The remaining provinces scored below the Canadian means in the respective language systems (Table 2.2, Appendix A.2.1.3).

### TABLE 2.2 Comparison of Canadian and provincial results by language of the school system, attitude toward mathematics index

|  | Anglophone school systems               |   |
|--|---|---|
| Above* the Canadian English mean   | Similar to the Canadian English<br>mean | Below* the Canadian English mean  |
| Ontario  | Saskatchewan, Manitoba                  | British Columbia, Alberta, Quebec,<br>New Brunswick, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |
|  | Francophone school systems              |   |
| Above* the Canadian French mean  | Similar to the Canadian French<br>mean  | Below* the Canadian French mean   |
| British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>New Brunswick, Nova Scotia |   | Quebec  |

\*Denotes significant difference

Within provinces in which students were sampled in sufficient numbers to provide reliable results by language of the school system, students in francophone schools scored higher than those in anglophone schools on this index in all provinces except Quebec and New Brunswick: in the former, there was no difference between the language systems; in the latter, students in anglophone schools scored higher than those in francophone schools (Table 2.3, Appendix A.2.1.3).

# TABLE 2.3 Summary of provincial results by language of the school system, attitude toward mathematics index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems |
|---|---|---|
| New Brunswick   | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Nova Scotia | Quebec  |

Table 2.4 shows that both girls and boys in Ontario, and girls in Saskatchewan and New Brunswick, had scores on this index that were higher than the respective Canadian means. Results similar to the Canadian means were found for boys in Saskatchewan and both girls and boys in Alberta and Manitoba (Appendix A.2.1.4).

# TABLE 2.4 Comparison of Canadian and provincial results by gender, attitude toward mathematics index

|   | Girls                                  |  |  |  |  |  |
|---|--|--|--|--|--|--|
| Above* the Canadian mean for<br>girls   | Similar to the Canadian mean for girls | Below* the Canadian mean for<br>girls  |  |  |  |  |
| Saskatchewan, Ontario,<br>New Brunswick | Alberta, Manitoba                      | British Columbia, Quebec,<br>Nova Scotia, Prince Edward Island,<br>Newfoundland and Labrador                   |  |  |  |  |
|   | Boys                                   |  |  |  |  |  |
| Above* the Canadian mean for<br>boys    | Similar to the Canadian mean for boys  | Below <sup>*</sup> the Canadian mean for<br>boys   |  |  |  |  |
| Ontario                                 | Alberta, Saskatchewan, Manitoba        | British Columbia, Quebec,<br>New Brunswick, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |  |  |  |  |

\*Denotes significant difference

Within provinces, girls reported a more positive attitude than boys toward mathematics in New Brunswick and Newfoundland and Labrador, while boys reported a more positive attitude than girls in British Columbia, Manitoba, Ontario, and Prince Edward Island. No significant gender difference was found in the remaining provinces (Table 2.5, Appendix A.2.1.4).

| Girls scored significantly higher<br>than boys | Boys scored significantly higher<br>than girls               | No significant difference between girls and boys |
|--|--|--|
| New Brunswick,<br>Newfoundland and Labrador    | British Columbia, Manitoba,<br>Ontario, Prince Edward Island | Alberta, Saskatchewan, Quebec,<br>Nova Scotia    |

#### TABLE 2.5 Summary of provincial results by gender, attitude toward mathematics index

## Mathematics self-efficacy

Self-efficacy is a person's belief in their own ability to successfully perform a task (Bandura, 1977). A more detailed definition is offered by the Standards for School Mathematics (National Council of Teachers of Mathematics, 2000): "When challenged with appropriately chosen tasks, students become confident in their ability to tackle difficult problems, eager to figure things out on their own, flexible in exploring mathematical ideas and trying alternative solution paths, and willing to persevere" (p. 21).

Research shows that there is a significant relationship between self-efficacy beliefs and academic performance across a wide variety of subjects, experimental designs, and assessment methods (Kilpatrick et al., 2001; Multon et al., 1991; Pajares & Kranzler, 1995; Pajares & Schunk, 2001). In a meta-analysis on gender differences and academic self-efficacy, Huang (2013) found that the subject was an important factor, with girls displaying higher self-efficacy in language arts, while boys displayed higher self-efficacy in mathematics, computer science, and social science. Sakellariou (2020), in studying the underachievement of girls in mathematics, found that most or all of the gap between girls and boys in that subject in OECD countries in Europe and the more affluent East Asian countries was explained by gender differences in self-belief among students. The relationship was less clear in other regions. Students with lower math self-efficacy have reported less emotional and social engagement during math class than students with higher self-efficacy (Martin & Rimm-Kaufman, 2015). Studies have also shown that children with high self-efficacy complete more problems correctly and rework more of the problems they missed (Collins, 1985; Schoenfeld, 1989).

In PCAP 2019, a set of ten items was designed to gather information on students' self-efficacy with respect to mathematics activities. Through the process of principal component analysis, nine of the items were organized into two components or indices: the *confidence with mathematical processes index* and the *confidence using technology in mathematics index*.

### **Confidence with mathematical processes**

*The Common Curriculum Framework* (Western and Northern Canadian Protocol, 2006) describes a number of mathematical processes (communication, connections, mental mathematics and estimation, problem solving, reasoning, technology, and visualization) that are foundational to mathematics curriculum across Canada (Alberta Education, 2016; Nova Scotia, 2019; Ontario, 2020–21). One aspect of mathematical processes is mental mathematics, which involves solving mathematical tasks through mental processes without paper and pencil or other material aids (Proulx, 2019). Mental math skills enable students to develop on-the-spot and economical ways of solving problems that are tailored to the tasks at hand (Davis et al., 1996; Lave, 1988; Proulx, 2019). Mathematical thinking is activated by contradiction, tension, and surprise, and is supported by learning environments that are rich with challenging questions and opportunities for reflection (Mason et al. 1982; Pólya, 1957). Providing challenging problems that are relevant to our world and to daily living increases student interest in mathematics (Weidemann, 1995).

### Description of the index

Figure 2.4 shows the six items that constitute the *confidence with mathematical processes index*. Students reported the highest level of confidence in doing pencil-and-paper calculations and in problem solving. Lower levels of confidence were reported for explaining problem-solving processes, doing mental math, and estimating (Appendix A.2.2.1).

# FIGURE 2.4 Percentage of students by their responses to questionnaire items constituting the confidence with mathematical processes index



### Results for the index

Pan-Canadian results for the confidence with mathematical processes index are shown in Figure 2.5. Students in anglophone school systems and boys reported significantly higher levels of confidence with mathematics processes than did students in francophone school systems or girls (Appendices A.2.2.2, A.2.2.3, A.2.2.4).





\* Significant difference compared to the reference (Ref.) in each category

The relationship between confidence with mathematical processes and mathematics achievement is shown in Figure 2.6. The bottom quarter of this index represents students with the least confidence in their ability to do these mathematics processes, while the top quarter represents students with the greatest confidence. The results show a positive relationship between confidence and achievement, with a significant difference in mathematics scores in all quartiles in each category. The greatest difference in mathematics scores between the top and bottom quarter of this index was found in anglophone school systems (102 points), while the smallest difference was found in francophone school systems (78 points) (Appendix A.2.2.5).





\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

Students in Ontario had the highest scores on the confidence with mathematical processes index. Students in Quebec, New Brunswick, and Newfoundland and Labrador had the lowest scores, all of which were below the Canadian mean (Table 2.6, Appendix A.2.2.2).

# TABLE 2.6 Comparison of Canadian and provincial results, confidence with mathematical processes index

| Above* the Canadian mean | Similar to the Canadian mean   | Below* the Canadian mean                            |
|--------------------------|--|---|
| Ontario                  | British Columbia, Alberta,<br>Saskatchewan, Manitoba,<br>Nova Scotia, Prince Edward Island | Quebec, New Brunswick,<br>Newfoundland and Labrador |

\*Denotes significant difference

As shown in Table 2.7, students in both anglophone and francophone school systems in Ontario and francophone school systems in British Columbia, Alberta, Saskatchewan, Manitoba, and Nova Scotia scored above the Canadian average on this index. Within provinces, higher levels of confidence in mathematics were reported by English-language students in Quebec and New Brunswick, compared to their francophone peers. French-language students in Nova Scotia had higher scores on this index than their anglophone counterparts (Table 2.8, Appendix A.2.2.3).

# TABLE 2.7 Comparison of Canadian and provincial results by language of the school system, confidence with mathematical processes index

| Anglophone school systems   |  |   |  |  |  |  |
|---|--|---|--|--|--|--|
| Above* the Canadian English mean  | Similar to the Canadian English<br>mean                  | Below* the Canadian English mean  |  |  |  |  |
| Ontario   | Alberta, Saskatchewan, Manitoba,<br>Prince Edward Island | British Columbia, Quebec,<br>New Brunswick, Nova Scotia,<br>Newfoundland and Labrador |  |  |  |  |
|   | Francophone school systems                               |   |  |  |  |  |
| Above* the Canadian French mean   | Similar to the Canadian French<br>mean                   | Below* the Canadian French mean   |  |  |  |  |
| British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Nova Scotia | New Brunswick  | Quebec  |  |  |  |  |

\*Denotes significant difference

#### TABLE 2.8 Summary of provincial results by language of the school system, confidence with mathematical processes index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems           |
|---|---|---|
| Quebec, New Brunswick   | Nova Scotia   | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario |

Compared to the Canadian means, both girls and boys in Ontario and girls in Prince Edward Island had the highest scores on the confidence with mathematical processes index, while the lowest scores were found for both genders in Quebec and for boys in New Brunswick, Nova Scotia, and Newfoundland and Labrador (Table 2.9). In all provinces, boys scored significantly higher than girls on this index (Table 2.10, Appendix A.2.2.4).

|                                      | Girls   |   |
|--------------------------------------|---|---|
| Above* the Canadian mean for girls   | Similar to the Canadian mean for girls  | Below* the Canadian mean for girls                                  |
| Ontario, Prince Edward Island        | British Columbia, Alberta,<br>Saskatchewan, Manitoba,<br>New Brunswick, Nova Scotia,<br>Newfoundland and Labrador | Quebec  |
|                                      | Boys  |   |
| Above* the Canadian mean for<br>boys | Similar to the Canadian mean for boys   | Below* the Canadian mean for<br>boys                                |
| Ontario                              | British Columbia, Alberta,<br>Saskatchewan, Manitoba,<br>Prince Edward Island                                     | Quebec, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador |

### TABLE 2.9 Comparison of Canadian and provincial results by gender, confidence with mathematical processes index

\*Denotes significant difference

# TABLE 2.10 Summary of provincial results by gender, confidence with mathematical processes index

| Girls scored significantly higher<br>than boys | Boys scored significantly higher than girls | No significant difference between girls and boys |
|--|---|--|
|  | All provinces                               |  |

### **Confidence using technology in mathematics**

The impact of technology on teaching and learning and the role of technology in the classroom are topics that have generated considerable research. Various studies have concluded that access to technology has only moderate effects on achievement (Escueta et al., 2017; Q. Li & Ma, 2010; OECD, 2015). In their synthesis of the research on technology in the classroom, Cheung and Slavin (2013) conclude that "educational technology is making a modest difference in learning of mathematics. It is a help, but not a breakthrough" (p. 102). Notwithstanding, technology use is a rapidly evolving field, and increased access to these technologies and changing global conditions surrounding their use are radically transforming classrooms (Devlin, 2019). Grasping all the benefits of technology is among the fundamental challenges that education faces today (OECD, 2018). PCAP 2019 was administered before the global pandemic required a rapid shift to online learning across Canada. In the future, it will be important to look closely at the challenges and opportunities presented by a greater focus on the use of technology in education.

### Description of the index

Figure 2.7 shows the three items that constitute the *confidence using technology in mathematics index*. Close to 80 percent of students reported that they are very confident using calculators in mathematics. The proportion decreases to 63 percent for students who are very comfortable with mathematics activities involving computers, while less than 20 percent are very confident in their coding or programming skills (Appendix A.2.3.1). It should be noted that, at the item level, correlations with achievement were found to be weak for two of the three items in this index.

# FIGURE 2.7 Percentage of students by their responses to questionnaire items constituting the confidence using technology in mathematics index



### Results for the index

At the pan-Canadian level, students in anglophone school systems were more confident using technology than were their counterparts in francophone school systems. Boys achieved significantly higher scores on this index than girls (Figure 2.8, Appendices A.2.3.2, A.2.3.3, A.2.3.4).

### FIGURE 2.8 Results for the confidence using technology in mathematics index



\* Significant difference compared to the reference (Ref.) in each category

Figure 2.9 presents the results by quarters of this index for Canada overall, as well as by the categories of language of the school system and gender. The top quarter of the index represents students with the highest level of confidence when doing mathematics activities involving technology. For each category, students in the top quarter of the index consistently had higher scores in mathematics achievement compared to the bottom quarter. There is also significantly higher achievement in all categories when comparing results in the second and bottom quarters as well as in the second and third quarters of the index. Achievement was lower for students in the top quarter compared to the third quarter for Canada overall and for girls; however, no difference was found between these two quarters for either English- or French-language schools or for boys (Appendix A.2.3.5).





\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

When provincial results are compared to the Canadian mean, students in Ontario reported the highest level of confidence in using technology in mathematics, with an index score above the Canadian mean. Scores on this index were similar to the Canadian mean in Alberta, Saskatchewan, Manitoba, and Prince Edward Island. Students in the remaining provinces reported the lowest levels of confidence using technology for activities in mathematics, with scores below the Canadian mean (Table 2.11, Appendix A.2.3.2).

### TABLE 2.11 Comparison of Canadian and provincial results, confidence using technology in mathematics index

| Above* the Canadian mean | Similar to the Canadian mean                             | Below* the Canadian mean  |
|--------------------------|--|---|
| Ontario                  | Alberta, Saskatchewan, Manitoba,<br>Prince Edward Island | British Columbia, Quebec,<br>New Brunswick, Nova Scotia,<br>Newfoundland and Labrador |

\*Denotes significant difference

Anglophone students in Ontario and francophone students in British Columbia reported the highest levels of confidence using technology for activities in mathematics. Levels of confidence similar to those at the pan-Canadian level were reported by both English- and French-language students in Alberta, Quebec, and New Brunswick. Students in anglophone systems in Saskatchewan and Prince Edward Island and in francophone systems in Manitoba, Ontario, and Nova Scotia also reported levels of confidence similar to the respective Canadian means (Table 2.12, Appendix A.2.3.3).

# TABLE 2.12 Comparison of Canadian and provincial results by language of the school system, confidence using technology in mathematics index

|                                  | Anglophone school systems  |  |
|----------------------------------|--|--|
| Above* the Canadian English mean | Similar to the Canadian English<br>mean                                  | Below* the Canadian English mean   |
| Ontario                          | Alberta, Saskatchewan, Quebec,<br>New Brunswick,<br>Prince Edward Island | British Columbia, Manitoba,<br>Nova Scotia,<br>Newfoundland and Labrador |
|                                  | Francophone school systems   |  |
| Above* the Canadian French mean  | Similar to the Canadian French<br>mean                                   | Below* the Canadian French mean  |
| British Columbia                 | Alberta, Manitoba, Ontario,<br>Quebec, New Brunswick,<br>Nova Scotia     | Saskatchewan   |

\*Denotes significant difference

Within provinces, the higher scores on this index were found in anglophone schools in Saskatchewan, Ontario, New Brunswick, and Nova Scotia, and in francophone schools in British Columbia. No significant difference between the language systems was found in the remaining provinces (Table 2.13, Appendix A.2.3.3).

# TABLE 2.13 Summary of provincial results by language of the school system, confidence using technology in mathematics index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems |
|---|---|---|
| Saskatchewan, Ontario,<br>New Brunswick, Nova Scotia                          | British Columbia  | Alberta, Manitoba, Quebec                           |

With respect to gender, the highest levels of confidence in using technology in mathematics were reported by girls in Prince Edward Island and boys in Ontario, while the lowest levels were reported by girls in British Columbia and boys in Manitoba, Quebec, New Brunswick, Nova Scotia, Prince Edward Island, and Newfoundland and Labrador (Table 2.14). Boys scored significantly higher on this index than girls in all provinces except Prince Edward Island, where girls expressed more confidence than boys (Table 2.15, Appendix A.2.3.4).

# TABLE 2.14 Comparison of Canadian and provincial results by gender, confidence using technology in mathematics index

|                                       | Girls  |  |
|---------------------------------------|--|--|
| Above* the Canadian mean for<br>girls | Similar to the Canadian mean for girls   | Below* the Canadian mean for<br>girls  |
| Prince Edward Island                  | Alberta, Saskatchewan, Manitoba,<br>Ontario, Quebec, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador | British Columbia   |
|                                       | Boys   |  |
| Above* the Canadian mean for<br>boys  | Similar to the Canadian mean for boys  | Below* the Canadian mean for<br>boys   |
| Ontario                               | British Columbia, Alberta,<br>Saskatchewan   | Manitoba, Quebec, New Brunswick,<br>Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |

\*Denotes significant difference

#### TABLE 2.15 Summary of provincial results by gender, confidence using technology in mathematics index

| Girls scored significantly higher than boys | Boys scored significantly higher<br>than girls  | No significant difference between girls and boys |
|---|---|--|
| Prince Edward Island                        | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Quebec, New Brunswick,<br>Nova Scotia, Newfoundland and<br>Labrador |  |

## Student effort

Many students have difficulties understanding the relevance of mathematics (Boaler, 2002; Godfrey Anderson, 2016; Onion, 2004) and do not acknowledge its value (Middleton et al., 2016). Such difficulties are particularly pronounced among girls (Frenzel et al., 2007; Ganley & Lubienski, 2016; Watt et al., 2012) and minorities (Matthews, 2018; Ng et al., 2018), and could influence the effort they put into learning mathematics. Although a decline in postsecondary graduation rates in science, mathematics, computer science, and engineering in Canada has been reported in the past (Conference Board of Canada, 2013; Natural Science and Engineering Research Council of Canada, 2017), there has been significant growth in the number of STEM graduates in Canada, with an increase of 51 percent from 2010 to 2018. Moreover, there has been a 35 percent increase in enrolment in postsecondary STEM programs during this same period (CMEC, 2021). A key predictor of students' interest in STEM-related careers at the end of high school is their interest in such careers at the start of high school (Sadler et al., 2012). It is concerning, then, that occupational intentions with regard to STEM-related careers decline dramatically between Grades 9 and 11 (Mangu et al., 2015). Math anxiety is considered a potential barrier to success in STEM-related fields and may be linked directly to avoidance of or underperformance in STEM domains in school (Daker et al., 2021). It is important to understand and address factors that contribute to success in STEMrelated fields, as these fields are at the forefront of innovation, and qualified people to fill jobs in these areas will continue to be in high demand.

### **Description of the index**

It is difficult to develop precise measures related to student effort because the amount of time and effort actually spent on learning tasks and the efficiency of learning are hard to determine. Despite the challenges associated with studying its effects, learning time remains important when considering students' opportunities to learn and the effort they put into their learning.

In order to characterize their effort in mathematics class, students were asked to indicate, using a four-point scale, the extent to which they agreed with six statements about their behaviour in class. Figure 2.10 presents these items in descending order by the percentage of students who agreed or strongly agreed with them. This set of items constitutes the *student effort index*, which was arrived at through principal component analysis.

Over 80 percent of students reported that they agree or strongly agree with statements about paying attention in class and being prepared for their assessments. Over half of students reported that they avoid distractions when studying; perhaps unsurprisingly, less than half of students reported practising concepts not taught in class (Figure 2.10, Appendix A.2.4.1).

# FIGURE 2.10 Percentage of students by their responses to questionnaire items constituting the student effort index



### **Results for the index**

Pan-Canadian results for the student effort index are shown in Figure 2.11. The scores on the index were higher in anglophone school systems than in francophone school systems, and girls had higher scores than boys (Appendices A.2.4.2, A.2.4.3, A.2.4.4).



### **FIGURE 2.11** Results for the student effort index

\* Significant difference compared to the reference (Ref.) in each category

Compared to students in the bottom quarter of this index, student in the top quarter reported a greater tendency to pay attention during lessons, be prepared and organized, and focus on understanding the material. For each category shown in Figure 2.12, a significant difference was found between the top and bottom quarters, ranging from 37 points in francophone school systems to 57 points in anglophone school systems. Fewer differences were found between the second and third quarters of the index, where the results were significantly different only for English-language students and boys. Results were similar between these quartiles in the other categories (Appendix A.2.4.5).



#### FIGURE 2.12 Relationship between the student effort index and mathematics achievement

\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

According to the scores for this index, Ontario students put more effort into their mathematics classes than did their counterpart in other provinces. Students in British Columbia, Alberta, Saskatchewan, Manitoba, and New Brunswick were found to have effort levels comparable to those of students in Canada overall, while the lowest effort in mathematics classes was found in the remaining provinces (Table 2.16, Appendix A.2.4.2).

| Above* the Canadian mean | Similar to the Canadian mean   | Below* the Canadian mean   |
|--------------------------|--|--|
| Ontario                  | British Columbia, Alberta,<br>Saskatchewan, Manitoba,<br>New Brunswick | Quebec,<br>Nova Scotia, Prince Edward Island,<br>Newfoundland and Labrador |

#### TABLE 2.16 Comparison of Canadian and provincial results, student effort index

\*Denotes significant difference

In both anglophone and francophone schools, Ontario students were found to have the highest scores on the effort in mathematics index. In English-language schools, students in British Columbia, Alberta, Saskatchewan, and Manitoba reported levels of effort similar to the Canadian English mean. In French-language schools, greater effort compared to the Canadian French mean was reported in all provinces except Quebec, where students scored below the Canadian mean on this index, and Alberta, where the index score was similar to the Canadian mean (Table 2.17, Appendix A.2.4.3).

# TABLE 2.17 Comparison of Canadian and provincial results by language of the school system, student effort index

|   | Anglophone school systems                            |   |
|---|--|---|
| Above* the Canadian English mean  | Similar to the Canadian English<br>mean              | Below* the Canadian English mean  |
| Ontario   | British Columbia, Alberta,<br>Saskatchewan, Manitoba | Quebec, New Brunswick, Nova<br>Scotia, Prince Edward Island,<br>Newfoundland and Labrador |
|   | Francophone school systems                           |   |
| Above* the Canadian French mean   | Similar to the Canadian French<br>mean               | Below* the Canadian French mean   |
| British Columbia, Saskatchewan,<br>Manitoba, Ontario, New Brunswick,<br>Nova Scotia | Alberta  | Quebec  |

\*Denotes significant difference

Within provinces, the index results show that students in francophone schools in Saskatchewan, Manitoba, Ontario, New Brunswick, and Nova Scotia reported putting more effort into their mathematics classes than did their counterparts in anglophone schools. No difference between the language systems was found in the remaining provinces (Table 2.18, Appendix A.2.4.3).

# TABLE 2.18 Summary of provincial results by language of the school system, student effort index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems |
|---|---|---|
|   | Saskatchewan, Manitoba, Ontario,<br>New Brunswick, Nova Scotia                | British Columbia, Alberta, Quebec                   |

Girls and boys in Ontario and girls in New Brunswick reported the highest levels of effort in mathematics, all of which were above the Canadian mean. Results on this index were similar to the respective Canadian means for both girls and boys in British Columbia, Alberta, Saskatchewan, and Manitoba, and for boys in New Brunswick. Index results were below the respective Canadian means for both girls and boys in the remaining provinces (Table 2.19). In all provinces, girls' scores on this index were higher than those of boys (Table 2.20, Appendix A.2.4.4).

#### TABLE 2.19 Comparison of Canadian and provincial results by gender, student effort index

|                                    | Girls  |  |
|------------------------------------|--|--|
| Above* the Canadian mean for girls | Similar to the Canadian mean for girls                                 | Below* the Canadian mean for<br>girls                                      |
| Ontario, New Brunswick             | British Columbia, Alberta,<br>Saskatchewan, Manitoba                   | Quebec, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |
|                                    | Boys   |  |
| Above* the Canadian mean for boys  | Similar to the Canadian mean for boys                                  | Below* the Canadian mean for<br>boys                                       |
| Ontario                            | British Columbia, Alberta,<br>Saskatchewan, Manitoba,<br>New Brunswick | Quebec, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |

\*Denotes significant difference

#### TABLE 2.20 Summary of provincial results by gender, student effort index

| Girls scored significantly higher<br>than boys | Boys scored significantly higher<br>than girls | No significant difference between girls and boys |
|--|--|--|
| All provinces                                  |  |  |

## Time management

Classroom learning is considered to be the core of student learning. While curricular policies and school resources often set the tone for learning, students' day-to-day classroom activities are likely to have a considerable impact on their mathematics learning. Students are engaged in learning not only inside the classroom but also as they take part in activities outside of school, including homework, sports, and social interactions. To explore issues surrounding students' time management, this section looks at data related to effort invested by students on homework, the loss of learning time resulting from student absenteeism, and time spent on out-of-school activities.

### Homework

Despite the emphasis that many teachers place on homework, students sometimes place a higher priority on their out-of-school activities than on completing their homework. Meanwhile, parents have to cope with the family stress that results from monitoring homework completion (Hoover-Dempsey et al., 2001; Warton, 2001). The majority of teachers, students, and parents believe that homework is a valuable, even essential, educational tool (e.g., Cooper et al., 1998; Xu, 2005). Yet it remains difficult to reach a definite conclusion about homework's value, despite the strong arguments the academic literature puts forward both for and against homework. For example, in a summary of research conducted in the United States between 1987 and 2003 on the effect of homework, Cooper, Robinson, and Patall (2006) concluded that there was generally consistent evidence for homework's positive influence on achievement, but they also acknowledged that all these studies had methodological issues. Empirical support for a positive relationship between homework and achievement is not unequivocal (e.g., De Jong et al., 2000; Trautwein & Köller, 2003). However, at least for mathematics in the middle grades, spending no time on homework is related to lower achievement; in contrast, for reading at the Grade 4 level, spending 15 minutes or less on homework is related to higher scores (CMEC, 2014).

The PCAP 2019 student questionnaire asked students how much time they spent every week on homework. In general, students reported spending more time on homework in each of their other subjects than in mathematics. Close to 75 percent of students reported spending an hour or less on mathematics homework each week (this includes the students who reported that no mathematics homework was assigned). Just over one in four students reported spending over an hour on mathematics homework. Approximately two-thirds of students reported spending an hour or less per week on homework for each of their other school subjects (including 11 percent who reported that no homework was assigned in these subjects). Thirty-seven percent reported spending more than an hour on homework in each of their other school subjects. Just over 10 percent of students reported spending two or more hours on mathematics homework per week, while 17 percent reported doing this amount of homework in each of their other subjects (Figure 2.13, Appendix A.2.4.6).

#### FIGURE 2.13 Amount of time spent on homework each week



The questionnaire measured homework effort by asking students how often they completed their mathematics homework, using a four-point scale ranging from *never* to *often*. Ten percent of students reported rarely or never completing their homework, while 67 percent reported often completing their homework. As shown in Figure 2.14, student who often complete their homework had higher achievement scores in mathematics than did students who completed their homework less often. Indeed, a difference of more than 50 points in mathematics scores was found between students who reported that they often complete their homework and those who rarely do so (Appendix A.2.4.7).



FIGURE 2.14 Relationship between mathematics homework effort and mathematics achievement

\* Denotes significant difference compared to the often category

### Loss of learning time

Although research in mathematics learning tends to focus on cognitive processes and strategies, students' motivation to learn is an important factor that can influence their academic success. Absence from classes in middle school and high school have been shown to affect both learning in the short term and educational attainment in the long term (Liu et al., 2021). While the negative impact of absenteeism on achievement has been found to be higher in reading than in mathematics (Holmes, 2020), it is clear that students who attend class regularly are more likely to have positive educational outcomes.

The PCAP student questionnaire asked respondents to report on their absences in the current school year and whether they were for non-school-related reasons (e.g., illness, appointments, travel, non-school sports activities, community events) or the result of school-related activities (e.g., field trips, school sports activities, music or cultural events). Approximately one-third of students reported having been absent from school for 10 days or more in that school year for non-school-related reasons, while 8 percent reported being absent for the same duration for school-related reasons. In general, students who reported being absent for five or fewer days were more likely to be away for school-related activities, whereas absences of six or more days tended to be more for non-school-related reasons (Figure 2.15, Appendix A.2.4.8).



#### FIGURE 2.15 Student-reported school absences

With respect to students' reports of absenteeism, the relationship between absence and mathematics achievement depends on the type of absence, as shown in Figure 2.16. There was a steady decline in mathematics scores as the number of absences for reasons that were not related to school increased. For school-related activities, results suggest that some involvement in activities such as field trips, sports, music, and so on, may be desirable, but that, at extreme levels, such absences may negatively influence achievement (Appendix A.2.4.8).



FIGURE 2.16 Relationship between student absences and mathematics achievement
Students may skip individual classes, or days of school, for a variety of reasons. Difficult home circumstances, poor relationships at school, substance abuse, and mental health problems are among the risk factors associated with chronic skipping of school.

Students were asked whether, over the two weeks preceding the PCAP 2019 assessment, they had skipped either full days of school or some classes or had been late for classes. As shown in Figure 2.17, the majority of students were never absent or late during that period. For students who were late or absent, tardiness occurred more often than skipping school. For the minority that skipped one or more times during the two-week period, it is noteworthy that this group was more likely to miss entire days of school rather than just individual classes. As shown in Figure 2.18, students who were never tardy and never skipped classes or school days achieved the highest scores in mathematics, with scores declining as the frequency of missing classes or school days or arriving late increased (Appendix A.2.4.9).

#### FIGURE 2.17 Student-reported skipping and tardiness



*Note*: Students provided information on the two weeks prior to the PCAP administration.





*Note*: Students provided information on the two weeks prior to the PCAP administration.

### **Out-of-school activities**

Student's use of time outside of the classroom reflects their interests and priorities. Time outside the classroom can also provide learning opportunities through structured out-of-school activities such as sports or community activities as well as social interactions and technology-based entertainment. Studies have found that structured extracurricular activities have a positive relationship with students' school engagement and academic achievement (Galloway et al., 2013; Knifsend & Graham, 2012).

Although out-of-school activities were not shown to have a significant relationship with achievement in mathematics in PCAP 2019, it is interesting to discover how students in this age group use their time when they are not in the classroom. Students were asked to report on a six-point scale, from no time to more than 6 hours, the number of hours in an average week that they usually spend doing a variety of activities. As Figure 2.19 shows, most students reported that they spend many hours using technology for personal reasons, and 45 percent of students reported that they engage in such activities for more than six hours a week. Close to one in three students reported spending more than six hours a week with friends or doing physical activities, and more than 60 percent of all students reported that they participated in such activities three or more hours per week. More than two-thirds of students were involved in extracurricular activities, such as clubs or music, and about 40 percent of students reported that they were involved with community service for at least some time each week. The activity that they engaged in the least frequently involved playing mathematics-related games or puzzles: only 14 percent of students engaged in this activity for one or more hours per week (Appendix A.2.4.10).



#### FIGURE 2.19 Time spent by students per week on activities outside of school hours

■ No time ■ Less than 1 hour ■ 1 to 2 hours ■ 3 to 4 hours ■ 5 to 6 hours ■ More than 6 hours

### Students' learning experiences

The classroom is where most formal learning takes place. Since students spend several hours each day in classes, PCAP focused on a number of factors that can influence classroom learning. This section explores two areas relevant to classroom learning, through indices related to understanding the language of mathematics and activities to support learning in mathematics.

### Understanding the language of mathematics

The ability to communicate mathematically is a fundamental process in curriculum across Canada (see, e.g., Alberta Education, 2016; Nova Scotia, 2019; Ontario, 2020–21). The communication process is a way of sharing ideas and clarifying understanding, and helps to build meaning and permanence for ideas (National Council of Teachers of Mathematics, 2000; Pimm, 1991). According to the Western and Northern Canadian Protocol (2006), "Communication can help students make connections among concrete, pictorial, symbolic, verbal, written and mental representations of mathematical ideas" (p. 6). Communicating mathematically allows students to understand mathematical ideas and take part in the process of mathematical thinking (Pimm, 1991; Stein et al., 1996).

In order to explore their knowledge of mathematics terms found in the Grade 8/Secondary II curricula in Canada, students were asked about their level of familiarity with 16 mathematics terms, which they ranked on a four-point scale, from *never heard of it* to *know this concept well*. Through the process of principal component analysis, these items were organized into three components, which are described in Table 2.21. Component 1 (general mathematics terms) and Component 2 (geometry and measurement terms) were found to have a correlation above .2 with mathematics achievement, which is the threshold value for reporting. Component 3 did not reach the correlation threshold for reporting. For information purposes, that component, along with four items that were loaded onto more than one component, are listed in Table 2.21, but they are not included in this analysis.

| TABLE 2.21 | Principal | component | analysis of | mathematics | terms |
|------------|-----------|-----------|-------------|-------------|-------|
|------------|-----------|-----------|-------------|-------------|-------|

| Analysis of items listed in question. How familiar are you with the following mathematical terms? |  |  |  |
|---|--|--|--|
| Results of<br>principal component analysis  | Mathematics terms  | Related subdomain(s)   |  |
| Component 1: General mathematics terms<br>Correlation: .4   | <ul> <li>Percentage</li> <li>Factors</li> <li>Square root</li> <li>Perimeter</li> <li>Equation</li> <li>Ratio</li> </ul> | <ul> <li>Numbers and operations</li> <li>Geometry and measurement</li> <li>Patterns and relationships</li> </ul> |  |
| Component 2: Geometry and<br>measurement terms<br>Correlation: .3                                 | <ul><li>Cartesian plane</li><li>Quadrant</li></ul>   | Geometry and measurement   |  |
| Component 3<br>Correlation: < .2<br>Not included in report  | <ul><li>Histogram</li><li>Median</li><li>Mode</li><li>Sampling</li></ul>   | <ul><li>Patterns and relationships</li><li>Data management and probability</li></ul>                             |  |
| Items loaded onto multiple components<br>Removed from analysis                                    | <ul><li>Percentage increase</li><li>Polygon</li><li>Integer</li><li>Quadrilateral</li></ul>                              | <ul><li>Numbers and operations</li><li>Geometry and measurement</li></ul>  |  |

### **Knowledge of general terms in mathematics**

#### Description of the index

Component 1 comprises six mathematics terms: percentage, factors, square root, perimeter, equation, and ratio. In this report, this component is referred to as the knowledge of general terms in mathematics index because the terms are found in three of the four subdomains described in the PCAP 2019 assessment framework: numbers and operations, geometry and measurement, and pattern and relationships (CMEC, 2020).

As shown in Figure 2.20, the majority of students reported that they know each of these concepts well. The most familiar terms to students were equation and perimeter, while the least familiar term was *factors*, with only about half of students reporting that they knew this term well. Approximately 10 percent of Grade 8/Secondary II students who participated in PCAP 2019 reported that they had never heard of these terms or had heard of them only a few times (Appendix A.2.5.1).





#### Results for the index

As shown in Figure 2.21, results for this index were similar for the two language groups, with no significant difference found between English- and French-language school systems at the pan-Canadian level. Some variation by language was evident at the provincial level (Appendix A.2.5.3). Girls had higher index scores than boys at the pan-Canadian level; this was also the case in all provinces except Ontario, where no gender gap was found (Appendix A.2.5.4).





Knowledge of general terms in mathematics index score

\* Significant difference compared to the reference (Ref.) in each category

Figure 2.22 shows the relationship between mathematics achievement and the knowledge of general terms in mathematics index. Students in the bottom quarter of the index were most likely to report that they had never heard of the six mathematics terms in this index or had heard of them only a few times. The top quarter of the index represents students who reported that they knew these terms well, which could reflect a greater understanding of mathematics achievement was found between the top and bottom quarters of the index. The largest difference was found for boys, with 104 point between the top and bottom quarters of the index, while the smallest difference was in francophone school systems, where 80 points separated students in the top and bottom quarters of this index (Appendix A.2.5.5).

#### FIGURE 2.22 Relationship between the knowledge of general terms in mathematics index and mathematics achievement



\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

Ontario students reported the highest level of knowledge of general mathematics terms, while students in Alberta, Saskatchewan, Quebec, and Nova Scotia reported levels of knowledge similar to that of Canadian Grade 8/Secondary II students as a whole. Students in the remaining provinces reported lower levels of knowledge of general mathematics terms than the Canadian average (Table 2.22, Appendix A.2.5.2).

 
 TABLE 2.22 Comparison of Canadian and provincial results, knowledge of general terms in mathematics index

| Above* the Canadian mean | Similar to the Canadian mean                  | Below* the Canadian mean  |
|--------------------------|---|---|
| Ontario                  | Alberta, Saskatchewan, Quebec,<br>Nova Scotia | British Columbia, Manitoba,<br>New Brunswick,<br>Prince Edward Island,<br>Newfoundland and Labrador |

\*Denotes significant difference

The results for this index by language of the school system are summarized in Table 2.23. Ontario students reported higher knowledge of general mathematics terms in both language systems, compared to the respective Canadian means, while francophone students in Manitoba and New Brunswick scored higher than the Canadian French mean. Results similar to the respective English and French means at the pan-Canadian level were obtained by students in both language system in Alberta and Nova Scotia, by English-language students in Saskatchewan and Quebec, and by French-language students in British Columbia.

## TABLE 2.23Comparison of Canadian and provincial results by language of the school system,<br/>knowledge of general terms in mathematics index

|                                  | Anglophone school systems                     |   |
|----------------------------------|---|---|
| Above* the Canadian English mean | Similar to the Canadian English<br>mean       | Below* the Canadian English mean  |
| Ontario                          | Alberta, Saskatchewan, Quebec,<br>Nova Scotia | British Columbia, Manitoba,<br>New Brunswick,<br>Prince Edward Island,<br>Newfoundland and Labrador |
|                                  | Francophone school systems                    |   |
| Above* the Canadian French mean  | Similar to the Canadian French<br>mean        | Below* the Canadian French mean   |
| Manitoba, Ontario, New Brunswick | British Columbia, Alberta,<br>Nova Scotia     | Saskatchewan, Quebec  |

\*Denotes significant difference

Within provinces, francophone students scored higher than anglophone students in British Columbia, Manitoba, and New Brunswick, while no significant difference was found between the two systems in the remaining provinces (Table 2.24, Appendix A.2.5.3).

## TABLE 2.24 Summary of provincial results by language of the school system, knowledge of general terms in mathematics index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems    |
|---|---|--|
|   | British Columbia, Manitoba,<br>New Brunswick                                  | Alberta, Saskatchewan, Ontario,<br>Quebec, Nova Scotia |

When scores on this index are examined by gender, only boys in Ontario scored above the overall mean score for Canada. Results similar to the respective Canadian means were found for both girls and boys in Alberta, Saskatchewan, and Quebec, and for girls in Manitoba, Ontario, Nova Scotia, and Newfoundland and Labrador (Table 2.25). At the provincial level, girls in all provinces except Ontario were more likely than boys to report that they knew general mathematics terms well (Table 2.26, Appendix A.2.5.4).

## TABLE 2.25 Comparison of Canadian and provincial results by gender, knowledge of general terms in mathematics index

|                                    | Girls  |  |  |  |
|------------------------------------|--|--|--|--|
| Above* the Canadian mean for girls | Similar to the Canadian mean for girls   | Below* the Canadian mean for<br>girls  |  |  |
|                                    | Alberta, Saskatchewan, Manitoba,<br>Ontario, Quebec, Nova Scotia,<br>Newfoundland and Labrador | British Columbia, New Brunswick,<br>Prince Edward Island   |  |  |
|                                    | Boys   |  |  |  |
| Above* the Canadian mean for boys  | Similar to the Canadian mean for boys  | Below* the Canadian mean for<br>boys   |  |  |
| Ontario                            | Alberta, Saskatchewan, Quebec  | British Columbia, Manitoba,<br>New Brunswick, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |  |  |

\*Denotes significant difference

## TABLE 2.26 Summary of provincial results by gender, knowledge of general terms in mathematics index

| Girls scored significantly higher<br>than boys   | Boys scored significantly higher<br>than girls | No significant difference between girls and boys |
|--|--|--|
| British Columbia, Alberta,<br>Saskatchewan, Manitoba, Quebec,<br>New Brunswick, Nova Scotia,<br>Prince Edward Island,<br>Newfoundland and Labrador |  | Ontario  |

### Knowledge of geometry and measurement terms

### Description of the index

The second component that was found to be correlated with mathematics achievement includes two mathematics terms: *Cartesian plane* and *quadrant*. The terms in Component 2 are included in the subdomain of geometry and measurement in Grade 8/Secondary II; thus, the index is called the *knowledge of geometry and measurement terms index*. Students who participated in PCAP 2019 reported less familiarity with these terms than with those in the knowledge of general terms index. Whereas more than 50 percent of students reported that they knew the concepts in the general mathematics terms index well (see Figure 2.20), less than 40 percent reported the same level of knowledge with respect to the two terms specific to geometry and measurement. Indeed, 30 percent of students reported that they have never heard of the term *Cartesian plane*, and 17 percent reported that they had never heard of the term *quadrant* (Figure 2.23, Appendix A.2.6.1). Given that the PCAP assessment is administered in April and May, it may be possible that at least some of these students had not yet been taught these topics in their mathematics class.

## FIGURE 2.23 Percentage of students by their responses to questionnaire items constituting the knowledge of geometry and measurement terms index



#### Results for the index

As with the previous index, students in the top quarter of the geometry and measurement terms index were most likely to report that they know these concepts, while students in the bottom quarter are most likely to report that they have never heard of the terms. As shown in Figure 2.24, students in francophone schools had much higher scores on this index than did their peers in anglophone schools, while there was no difference by gender. Results for this index at the provincial level can be found in Appendices A.2.6.2, A.2.6.3, and A.2.6.4.



#### **FIGURE 2.24** Results for the knowledge of geometry and measurement terms index

\* Significant difference compared to the reference (Ref.) in each category

The relationship between this index and mathematics achievement is clear: greater knowledge of these terms is associated with higher scores in mathematics, and there are significant increases in scores between all quartiles in all categories (Figure 2.25, Appendix A.2.6.5).

#### FIGURE 2.25 Relationship between the knowledge of geometry and measurement terms index and mathematics achievement



\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

Students in Alberta, Quebec, and Prince Edward Island had scores at or above the Canadian mean for the knowledge of geometry and measurement terms index (Table 2.27, Appendix A.2.6.2). Students in both language systems in Quebec and in anglophone school systems in Alberta and Prince Edward Island scored above the respective Canadian means on this index. Ontario anglophone students achieved results similar to their peers in English-language schools in Canada overall (Table 2.28).

#### TABLE 2.27 Comparison of Canadian and provincial results, knowledge of geometry and measurement terms index

| Above* the Canadian mean     | Similar to the Canadian mean | Below* the Canadian mean  |
|------------------------------|------------------------------|---|
| Quebec, Prince Edward Island | Alberta                      | British Columbia, Saskatchewan,<br>Manitoba, Ontario, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador |

\*Denotes significant difference

#### TABLE 2.28 Comparison of Canadian and provincial results by language of the school system, knowledge of geometry and measurement terms index

|  | Anglophone school systems               |  |
|--|---|--|
| Above* the Canadian English mean         | Similar to the Canadian English<br>mean | Below* the Canadian English mean   |
| Alberta, Quebec,<br>Prince Edward Island | Ontario                                 | British Columbia, Saskatchewan,<br>Manitoba, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador |
|  | Francophone school systems              |  |
| Above* the Canadian French mean          | Similar to the Canadian French<br>mean  | Below* the Canadian French mean  |
| Quebec                                   |   | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>New Brunswick, Nova Scotia             |

\*Denotes significant difference

As shown in Table 2.29, in all provinces that oversampled to obtain results by language, students in francophone systems had higher scores than their anglophone counterparts on this index (Appendix A.2.6.3).

## TABLE 2.29 Summary of provincial results by language of the school system, knowledge of geometry and measurement terms index

| Anglophone schools scored<br>significantly higher than<br>francophone schools                           | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems |
|---|---|---|
| British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Quebec, New Brunswick,<br>Nova Scotia |   |   |

When the results for this index are examined by gender, both girls and boys in Alberta, Quebec, and Prince Edward Island scored at or above the respective Canadian means, while the score for Ontario boys was similar to the Canadian mean for boys (Table 2.30).

#### TABLE 2.30 Comparison of Canadian and provincial results by gender, knowledge of geometry and measurement terms index

|                                       | Girls                                  |   |
|---------------------------------------|--|---|
| Above* the Canadian mean for<br>girls | Similar to the Canadian mean for girls | Below* the Canadian mean for<br>girls   |
| Quebec, Prince Edward Island          | Alberta                                | British Columbia, Saskatchewan,<br>Manitoba, Ontario, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador |
|                                       | Boys                                   |   |
| Above* the Canadian mean for<br>boys  | Similar to the Canadian mean for boys  | Below* the Canadian mean for<br>boys  |
| Quebec, Prince Edward Island          | Alberta, Ontario                       | British Columbia, Saskatchewan,<br>Manitoba, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador          |

\*Denotes significant difference

Within provinces, gender equity on this index was found in Alberta. Girls achieved higher scores than boys in Quebec and Prince Edward Island, while boys scored higher than girls in the remaining provinces (Table 2.31, Appendix A.2.6.4).

## TABLE 2.31 Summary of provincial results by gender, knowledge of geometry and measurement terms index

| Girls scored significantly higher<br>than boys | Boys scored significantly higher<br>than girls  | No significant difference between girls and boys |
|--|---|--|
| Quebec, Prince Edward Island                   | British Columbia, Saskatchewan,<br>Manitoba, Ontario, New Brunswick,<br>Nova Scotia,<br>Newfoundland and Labrador | Alberta  |

### Activities used to support learning in mathematics

Teachers are expected to engage students by creating classrooms that encourage positive social interaction, active engagement in learning, and self-motivation (Danielson, 2007). Research in social science in the past several decades has come to recognize that learning is a collaborative, social process. This process involves situated learning that leads to the acquisition of knowledge and skills (Cobb, 1986; Cobb & Yackel, 2011; Lave & Wenger, 1991). For example, Lave, Murtaugh, and de la Rosa (1984) argue that reasoning with and about quantities should be understood as an achievement jointly created by individuals and resources in the environment, not just as operations involving symbols and mental representations. Mathematics classrooms are environments where, ideally, students encounter engaging, worthwhile, and challenging tasks (National Council of Teachers of Mathematics, 2000) and where "students can share their ideas, improve on each other's ideas, challenge and be challenged, seek advice from peers, explain their thinking, provide evidence for their solutions, and explore evolving ideas and conjectures" (Learn Alberta, 2008). Yet in the report

on PISA 2012, the OECD (2013a) reported that only 25 percent of 15-year-olds help their friends with mathematics, and only 18 percent of students on average across the OECD reported that they regularly talk about mathematics problems with their friends (p. 106).

### **Description of the index**

In order to explore the types of activities used in classrooms to support the learning of mathematics, a set of 15 items, listed in Table 2.32, was presented to students. A four-point scale (*never, rarely, sometimes, often*) was used to collect information on how often student did these activities in mathematics class. Through the process of principal component analysis, these items were organized into three components; however, only one of these correlated with mathematic achievement above the .2 threshold.

### TABLE 2.32 Questionnaire items about activities in mathematics class

| Items listed in question "How often do you do the following in your mathematics class?"   |  |  |  |
|---|--|--|--|
| Correlation with achievement (≥.2)  | Weak or no correlation with achievement (<.2)  |  |  |
| <ul> <li>Component 1</li> <li>Pay attention to the teacher<br/>doing examples and giving<br/>explanations</li> <li>Observe teacher-guided problem<br/>solving and investigations</li> <li>Solve problems</li> <li>Copy notes</li> <li>Practise skills</li> <li>Work individually</li> </ul> | <ul> <li>Component 2</li> <li>Reflect on what was learned</li> <li>Justify your reasoning</li> <li>Use an alternative strategy to solve problems</li> <li>Create your own mathematical problems</li> </ul> | <ul> <li>Component 3</li> <li>Work in groups</li> <li>Share solutions with others</li> <li>Use concrete or virtual<br/>manipulatives (e.g., base-ten<br/>blocks, colour tiles, geometric<br/>solids)</li> <li>Use computer software</li> </ul> |  |

This component includes seven items, as shown in Figure 2.26, which together constitute the *teacher-directed mathematics activities index*. Results reveal that approximately two-thirds of students often use calculators, pay attention to the teacher doing examples and giving explanations, work individually, and solve problems. About half of students reported that they often practise skills, copy notes, or observe teacher-guided problem solving and investigations. (Correlations between individual items and achievement can be found in Appendix A.2.7.1.)

## FIGURE 2.26 Percentage of students by their responses to questionnaire items constituting the teacher-directed mathematics activities index



### **Results for the index**

Scores at the pan-Canadian level for the teacher-directed mathematics activities index are shown in Figure 2.27. French-language students scored higher on this index than English-language students, and girls obtained higher scores than boys (Appendices A.2.7.2, A.2.7.3, A.2.7.4).

#### **FIGURE 2.27** Results for the teacher-directed mathematics activities index



Teacher-directed mathematical activities index score

\* Significant difference compared to the reference (Ref.) in each category

Students were divided into four quarters, based on the self-reported frequency with which they engaged in teacher-directed activities in their mathematics classrooms. The top quarter represents students who indicated that they often did the seven activities listed in Figure 2.26; the bottom quarter represents students who were most likely to report that they rarely or never did such activities.

Figure 2.28 shows the relationship between the teacher-directed mathematics activities index and mathematics achievement. It shows a general pattern in which mathematics performance improves as the frequency of doing teacher-directed activities increases. Although there is a significant increase in mathematics achievement between most quarters in the various categories, the point difference between the bottom and second quarters is greater than the difference in points between the second and third as well as the third and top quarters of the index (Appendix A.2.7.5).

## FIGURE 2.28 Relationship between the teacher-directed mathematics activities index and mathematics achievement



\* Significant difference compared to the adjacent lower quarter (the bottom quarter is compared to the top quarter) within each category

Students in Quebec were most likely to report that they often did the activities that constitute this index, and their score on this index is above the Canadian mean. All other provinces were either at or below the Canadian mean (Table 2.33, Appendix A.2.7.2).

## TABLE 2.33 Comparison of Canadian and provincial results, teacher-directed mathematics activities index

| Above* the Canadian mean | Similar to the Canadian mean  | Below* the Canadian mean                         |
|--------------------------|---|--|
| Quebec                   | British Columbia, Alberta,<br>Saskatchewan, Ontario,<br>Nova Scotia,<br>Newfoundland and Labrador | Manitoba, New Brunswick,<br>Prince Edward Island |

\*Denotes significant difference

When scores are examined by language of the school system, index scores in Quebec were found to be above the Canadian mean in both anglophone and francophone schools. Results for the remaining provinces were similar to or below the respective Canadian means (Table 2.34, Appendix A.2.7.3).

# TABLE 2.34Comparison of Canadian and provincial results by language of the school system,<br/>teacher-directed mathematics activities index

| Anglophone school systems        |   |   |  |  |  |
|----------------------------------|---|---|--|--|--|
| Above* the Canadian English mean | Similar to the Canadian English<br>mean   | Below* the Canadian English mean        |  |  |  |
| Quebec                           | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Nova Scotia,<br>Newfoundland and Labrador | New Brunswick,<br>Prince Edward Island  |  |  |  |
| Francophone school systems       |   |   |  |  |  |
| Above* the Canadian French mean  | Similar to the Canadian French<br>mean  | Below* the Canadian French mean         |  |  |  |
| Quebec                           | British Columbia, Alberta,<br>Manitoba, Nova Scotia   | Saskatchewan, Ontario,<br>New Brunswick |  |  |  |

\*Denotes significant difference

Results on this index within provinces by language of the school system are shown in Table 2.35. Students in anglophone schools in Saskatchewan and in francophone schools British Columbia, Manitoba, and New Brunswick had higher scores on this index than their peers in the other language system. No significant difference between language systems was found for this index in the remaining provinces.

## TABLE 2.35 Summary of provincial results by language of the school system, teacher-directed mathematics activities index

| Anglophone schools scored<br>significantly higher than<br>francophone schools | Francophone schools scored<br>significantly higher than<br>anglophone schools | No significant difference between<br>school systems |  |
|---|---|---|--|
| Saskatchewan  | British Columbia, Manitoba,<br>New Brunswick                                  | Alberta, Ontario, Quebec,<br>Nova Scotia            |  |

With respect to gender, girls in Quebec had higher index scores than the Canadian mean, while girls in Alberta, Saskatchewan, Ontario, Nova Scotia, and Newfoundland and Labrador had scores similar to that mean. Results similar to the Canadian average for boys were found in British Columbia, Alberta, Saskatchewan, Manitoba, Ontario, Quebec, and Newfoundland and Labrador (Table 2.36, Appendix A.2.7.4). Interestingly, girls in all provinces were more likely than boys to report that they often did the activities that constitute this index (Table 2.37).

## TABLE 2.36 Comparison of Canadian and provincial results by gender, teacher-directed mathematics activities index

| Girls   |  |   |  |  |
|---|--|---|--|--|
| Above* the Canadian mean for girls            | Similar to the Canadian mean for girls   | Below* the Canadian mean for<br>girls                                 |  |  |
| Quebec  | Alberta, Saskatchewan, Ontario,<br>Nova Scotia,<br>Newfoundland and Labrador                           | British Columbia, Manitoba,<br>New Brunswick,<br>Prince Edward Island |  |  |
| Boys  |  |   |  |  |
| Above <sup>*</sup> the Canadian mean for boys | Similar to the Canadian mean for boys  | Below* the Canadian mean for<br>boys                                  |  |  |
|   | British Columbia, Alberta,<br>Saskatchewan, Manitoba, Ontario,<br>Quebec,<br>Newfoundland and Labrador | New Brunswick, Nova Scotia,<br>Prince Edward Island                   |  |  |

\*Denotes significant difference

## TABLE 2.37 Summary of provincial results by gender, teacher-directed mathematics activities index

| Girls scored significantly higher<br>than boys | Boys scored significantly higher than girls | No significant difference between girls and boys |  |
|--|---|--|--|
| All provinces                                  |   |  |  |

It could be useful to look at the individual activities that make up this index. Three activities showed a correlation with mathematics achievement above the .2 threshold: solving problems, practising skills, and working individually. Weak correlations, below the threshold of .2, were found for the remaining four activities. The positive relationship between achievement and doing these activities frequently may reflect the need for students to develop strong skills in mathematics through individual practice and problem solving.

### Supporting student learning

A supportive learning environment is important for fostering a love of mathematics in students. Research consistently shows that students who feel supported in their efforts to learn are most likely to become confident and competent learners. PCAP 2019 focused on a number of topics that can influence student learning. This section explores six areas relevant to classroom learning: crosscurricular integration of mathematics, homework activities, assessment, frequency of use of rubrics, provision of feedback and support, and students' sense of belonging.

### Cross-curricular integration of mathematics

The PCAP 2019 assessment framework describes common curricular concepts for Grade 8/ Secondary II mathematics in Canada (CMEC, 2020). In addition to the four content subdomains (numbers and operations, geometry and measurement, patterns and relationships, and data management and probability), five processes are recognized as important aspects of learning mathematics. The mathematical processes of problem solving, reasoning and proof, communication, connections, and representation highlight ways of acquiring and using content knowledge.

In mathematics, students are asked to make connections between different representations (written, graphical, numerical, and algebraic) of mathematical ideas. However, a broader application of this process of connection is to recognize and connect mathematical concepts and procedures to contexts outside of mathematics, such as other curricular areas, personal life, current events, sports, technology, arts and culture, and media. Teachers who intentionally bring math into other subject areas, or who bring other subjects into math, can increase the motivation of students to learn mathematics by highlighting some of its real-world applications. Students who participated in context-based learning during cross-curricular theme weeks reported high levels of enjoyment and interest, with increased engagement in their learning (Birchinall, 2013). Integration of art with science and mathematics curricula at the postsecondary level has been found to contribute to students' motivation, understanding, and ability to apply knowledge (Rachford, 2011).

In order to explore whether students experienced mathematics in cross-curricular ways, the student questionnaire asked students if they have used what they learn in mathematics in seven other subjects, as shown in Figure 2.29. Given the focus on STEM education, it is unsurprising that students reported that mathematics is most frequently used in science (86 percent) and technology (73 percent) classes. In addition, close to 40 percent of students reported that they have used what they learned in mathematics in art class, while about one in four students reported using their knowledge of mathematics in the remaining four subjects (Appendix A.2.8).

## FIGURE 2.29 Percentage of students who reported using what they learned in mathematics in other school subjects



### Homework activities

Teachers assign mathematics homework for various purposes, such as ensuring comprehension of the subject, knowledge retention, and use of the subject-related concepts in daily life. Homework is one of the most commonly used tools in classrooms (e.g., Trautwein et al., 2006) and has been shown to be useful in promoting students' agency and autonomy in their own learning (Ramdass & Zimmerman, 2011). Positive relationships have been found between homework completion and self-regulated learning skills (e.g., planning and monitoring) and self-efficacy beliefs in elementary and high school (Zimmerman & Kitsantas, 2005). A large body of research provides evidence of the value of homework for students' school engagement and academic success (e.g., Cooper, Robinson, et al., 2006; Fan et al., 2017; Núñez et al., 2013).

In the PCAP 2019 student questionnaire, 10 items were included in the question about the types of homework that was assigned in mathematics class. Students were asked to respond on a four-point scale (ranging from *never or almost never* to *daily or almost daily*). Figure 2.30 presents student responses. Students reported that the most common homework assigned was paper-and-pencil calculations, with 65 percent of students indicating that they were assigned this type of homework at least once per week. About half of students reported that word problems or studying for assessments were assigned for homework at least once per week. The least commonly assigned homework involved participation in online discussion forums or blogs and activities using concrete or virtual manipulatives (Appendix A.2.9). It is important to note that this assessment was administered in 2019 and does not reflect the shift to online learning that was a necessary safety precaution in response to the global pandemic.



#### FIGURE 2.30 Types of homework assigned in mathematics class, as reported by students

■ Never or almost never ■ 2 or 3 times per month ■ 1 to 3 times per week ■ Daily or almost daily

### Assessment

Teachers use assessment to gain insight into students' current ideas, gaps in understanding, and reasoning processes. This information can then be used to adapt instructional and assessment strategies to student needs. The power of assessment to reveal and support learning depends on how well students' responses authentically reflect their thinking and understanding (Shepard, 2005). Students were asked about how they were assessed, as well as about the use of rubrics and feedback in their classes.

To explore how they are assessed in mathematics, the student questionnaire asked students to rank the frequency (on a four-point scale from *never* to *often*) with which nine different types of assessments were used in their mathematics classes. As shown in Figure 2.31, at least 70 percent of students reported that they are sometimes or often assessed using homework or teacher-developed classroom assessments. Students are more likely to be assessed on individual, rather than group, assignments or projects; this approach may reflect the difficulty of fairly assessing individual contribution to work done by multiple students. The least common types of assessments used in mathematics class were student portfolios and/or journals, and peer assessment (Appendix A.2.10.1).



### FIGURE 2.31 Types of assessment in mathematics classes, as reported by students

### **Rubrics**

Assessment is integral to learning (Gronlund & Waugh, 2009); effective assessment requires a systematic, planned process to collect data and can result in improvements in both teaching and learning. To make the assessment of learning more transparent for students, and to communicate goals and expectations, teachers are increasingly providing them with information both before and after an assignment.

Rubrics provide students with a shared understanding of the specific, pre-established criteria that will be used to evaluate their work. Huba and Freed (2000) describe how "a rubric reveals, if you will, the scoring 'rules.' It explains to students the criteria against which their work will be judged ... [and] it makes public key criteria that students can use in developing, revising, and judging their own work" (p. 155).

Students were asked if they were familiar with a scoring rubric and if they used such a rubric when starting an assignment. Although about three-quarters of students reported knowing what a rubric is, only half of students reported that they sometimes use a rubric when starting an assignment in mathematics class (Figure 2.32, Appendix A.2.10.2). As shown in Figure 2.33, students who reported that their teachers often use a rubric when marking assignments have higher scores in mathematics than those in classrooms where rubrics were used less frequently (Appendix A.2.10.3).

#### FIGURE 2.32 Use of scoring rubrics in mathematics, as reported by students



Do you know what a scoring rubric is for marking assessments or assignments?

Do you sometimes use a rubric when you start an assignment in your mathematics class?





\* Significant difference compared to the often category

### Feedback and support

While a scoring rubric provides students with information about how their work will be assessed before beginning their assignments, feedback provides students with information afterwards to help them improve their learning. Researchers have suggested that providing feedback might foster improved learning, motivation, and achievement. However, to be effective, feedback has to be timely and focused on understanding and improvement or comparative performance, or both (Muis et al., 2013).

Students were asked about the frequency with which they received four types of feedback and support in mathematics class, as shown in Figure 2.34 (Appendix A.2.11.1). Over 75 percent of students reported that they sometime or often receive support in the form of information to help them solve a problem, details on marking criteria, and regular feedback to improve their learning.

## FIGURE 2.34 Types of teacher feedback and support given in mathematics class, as reported by students



Figure 2.35 presents the relationship between the frequency with which these types of supports are provided and achievement in mathematics (Appendix A.2.11.2). Increasing the frequency of providing hints or strategies to solve a problem had the strongest positive relationship with achievement, with a 23-point difference between the *never or rarely* and *often* categories. Providing details about marking was associated with higher mathematics scores only when it was done often. Surprisingly, increasing the frequency of regular feedback to improve learning was not associated with higher scores in mathematics. Providing students with opportunities to redo or resubmit work had a negative relationship with achievement, with a 22-point decrease in achievement between the *never or rarely* and *often* categories (Figure 2.35, Appendix A.2.11.2). It is important to exercise caution when interpreting this finding, as providing an opportunity to redo or resubmit work is not necessarily associated with lower achievement. More information about both the student and classroom contexts is needed to more fully understand this relationship.





\* Significance difference compared to the never or rarely category

### Students' sense of belonging

The relationship between students and teachers holds an important place within the research literature on students' sense of belonging or school connectedness. Students who experience school connectedness tend to like school, feel that they belong, believe teachers care about them and their learning, believe that education matters, have friends at school, believe that discipline is fair, and have opportunities to participate in extracurricular activities (Blum, 2005). A student's sense of belonging has a powerful influence over how they feel about school and whether they feel safe at school, and it is pivotal in producing academic success, emotional health and well-being, social development, and long-term productivity (Whitlock, 2003).

There is an abundance of research on the different impacts of student attitudes, confidence, and participation on their mathematics performance but, as Hattie (2009) points out, "correlates are not to be confused with the cause" (p. 3). Research exploring the contextual character of learning shifts the focus from individual, cognitive notions of ability to investigations into the importance of social interaction in learning situations (Lerman, 2000). More recently, research in mathematics is confronting issues of social justice (e.g., with respect to gender, ethnicity, class, linguistic background), recognizing that society can constrain student agency and participation (Black et al., 2009; Herbel-Eisenmann et al., 2012).

In PCAP 2019, students were asked to indicate the extent to which they agreed with statements about their sense of belonging in school. The overwhelming majority agreed or strongly agreed with all statements related to their sense of belonging (Figure 2.36). For example, over 90 percent of

students stated that they have a friend or friends at school whom they can trust, and approximately 80 percent of students reported that they make friends easily at school and feel that they belong. However, about one in three students reported that they disagreed with the statement "I like school" (Appendix A.2.12).



### FIGURE 2.36 Students' sense of belonging in school

The differences in mathematics achievement between students who strongly agree and those who strongly disagree with the respective statements related to sense of belonging in school are shown in Figure 2.37. The data show the largest difference for "At school, I am treated fairly" (60 points) and the smallest difference for "At school, I make friends easily" (20 points) (Appendix A.2.12).

### FIGURE 2.37 Relationship between students' sense of belonging in school and mathematics achievement



Figure 2.38 highlights the results for the item "I like school." More positive responses to this statement are associated with higher achievement in mathematics — a significant finding, as close to one-third of students reported that they disagreed with the statement. As school is a social endeavour, positive interactions with teachers and peers are an important component in motivating students to learn and to appreciate the value of mathematical knowledge both in school and in future career opportunities.

## FIGURE 2.38 Relationship between and students' response to the statement "I like school" and mathematics achievement



\* Significant difference compared to the adjacent lower quarter (and bottom quarter compared to top quarter) within each category

### Cognitive demand in mathematical tasks

There is a drive by policy-makers, educators, and assessment developers to increase the cognitive demands of tasks used in classrooms and assessed on large-scale assessments in order to support competencies required for postsecondary education and careers (e.g., Darling-Hammond et al., 2013; OECD, 2018). Webb's depth of knowledge (DOK) framework is a tool that can be used to organize questions into different levels of cognitive complexity (Webb, 2002). Webb's framework lays out descriptions for four DOK levels. In mathematics, these can be summarized as follows: Level 1, recall; Level 2, skills and concepts; Level 3, strategic thinking; and Level 4, extended thinking (Webb, 2002). Darling-Hammond et al. (2013), in their *Criteria for High-quality Assessments*, recommend that at least two-thirds of the items in an assessment be at Level 2 or higher, and that, in mathematics, at least one-third of assessments items should be at Level 3 or 4.

### Level 1 problems

Some mathematics problems describe scenarios and provide all the information required to solve them. The two examples given below are considered to be Level 1 questions.



#### Example 2:

The price of a video game increased from \$30 to \$45.

What is the percentage increase in the price of the video game?

### Level 3 problems

Some mathematics problems do not describe scenarios but require the use of mathematical knowledge to form conclusions. The two examples given below are considered to be Level 3 questions.



### Cognitive demand of questions used in mathematics class

In the PCAP 2019 student questionnaire, students were given the examples of the Level 1 and Level 3 questions shown above and asked how often they encountered these types of questions in their mathematics lessons and assessments. As shown in Figure 2.39, students reported that both levels of questions were given more frequently in lessons than in assessments. Responses indicate that 43 percent of students have often encountered Level 1 questions in lessons, while 39 percent have often encountered them on assessments. A lower proportion of students (about 30 percent) reported that they had often encountered Level 3 tasks, either during a lesson or on an assessment (Appendix A.2.13.1). These proportions reflect the range for Level 3 and 4 tasks recommended by Darling-Hammond et al. (2013) for the design of high-quality assessments; however, the findings do not reflect their recommendation for Level 1 tasks — as noted above, their research proposed that no more than one-third of tasks should be at Level 1.



#### FIGURE 2.39 Cognitive demand of questions used in mathematics

As shown in Figure 2.40, mathematics achievement increased for students who reported that their teachers assigned Level 1 cognitive tasks with increasing frequency in their mathematics lessons, with a 73-point difference between students who never or rarely encountered tasks at this level and those who often encountered such tasks. The difference was much smaller for tasks at Level 3: there was only a 20-point difference in mean scores between students who reported that Level 3 questions were often assigned in mathematics lessons and those who reported that they were rarely or never assigned, and no significant difference was found between the *never or rarely* and *sometimes* categories (Appendix A.2.13.2). These results seem to indicate that, although it is recommended that, in high-quality assessments, at least two-thirds of the items be at Level 2 or higher (Darling-Hammond et al., 2013), the more frequent use of Level 1 items was related to a greater gain in students' mathematics scores compared to more frequent use of Level 3 items. More information at the local level is needed to fully understand this pattern.

## FIGURE 2.40 Relationship between the cognitive demand of mathematics questions and mathematics achievement



\* Significance difference compared to the never or rarely category

While students need exposure to tasks at every level of cognitive demand, teachers need skills to evaluate what is and is not cognitively demanding, and they must be critical consumers of the mathematics resources available. Although low-level cognitive tasks are needed to scaffold student learning in mathematics, the highest-level problems require students to explore mathematical concepts. Such exploration develops competencies such as self-monitoring, critical thinking, and making connections, important skills that are required if students are to take advantage of future educational and professional opportunities.

### Summary

This chapter was organized around three broad themes related to students' learning and achievement in mathematics: attitudes and beliefs, students' learning experiences, and supporting student learning. Within each theme, indices and variables related to responses to the PCAP 2019 student questionnaire and cognitive test were analyzed.

The first theme, which explored students' attitudes and beliefs related to mathematics, comprised four topics: attitudes toward mathematics, mathematics self-efficacy, student effort, and time management. With regard to the first item, increasingly positive attitudes toward mathematics are linked to higher scores in the PCAP 2019 mathematics assessment. At the pan-Canadian level, students in anglophone schools reported more positive attitudes than their peers in francophone schools, and boys were more positive than girls about mathematics.

Mathematics self-efficacy involved two components: confidence with mathematical processes and confidence using technology in mathematics. For both indices, students with greater self-efficacy in mathematics had higher achievement scores in mathematics. At the pan-Canadian level, anglophone

students and boys were found to have higher confidence with respect to both mathematical processes and using technology in mathematics compared to francophone students or girls.

Students who had high scores on the student effort index were most likely to report paying attention in class and being prepared for their assessments. Index scores were higher in anglophone than in francophone school systems, and girls achieved higher scores than boys on this index.

Students' time management was explored by looking at data related to the effort students invested in homework, the loss of learning time resulting from student absenteeism, and students' out-ofschool activities. Over 10 percent of students reported spending two or more hours on mathematics homework per week, while close to 20 percent reported doing this amount of homework in each of their other subjects. Over 10 percent of students stated that they were not assigned homework in any subject. Those students who reported that they most often completed their homework had significantly higher achievement scores in mathematics compare to those who were less likely to complete their homework assignments.

With respect to absenteeism, about one-third of students reported having been absent from school for 10 days or more in that school year for non-school-related reasons, while 8 percent reported being absent for the same duration for school-related reasons. Most students neither skipped school nor were late for class in the two weeks prior to the assessment; for students who were late or did skip classes, tardiness occurred more often than skipping school. Students who had regular attendance tended to achieve higher scores in mathematics, and increasing frequency of missing or arriving late for classes is associated with poorer outcomes in mathematics.

When asked about their out-of-school activities, students reported that they spend many hours using technology for personal reasons: 45 percent of students reported that they engage in such activities for more than six hours a week. In addition, more than 60 percent of all students reported spending time with friends or doing physical activities three or more hours per week. More than two-thirds of students were involved with extracurricular activities, such as clubs or music, and about 40 percent of all students reported that they were involved with community service for at least some time each week.

The second theme of this chapter was students' learning experiences. This section looked at students' understanding of the language of mathematics and at activities used to support their learning in this subject. Unsurprisingly, students who had a good grasp of mathematical terms had better assessment outcomes than their peers. Although no difference was evident between language systems with respect to the knowledge of general terms in mathematics index, francophone students had much higher scores than anglophone students on the index of knowledge of terms related to geometry and measurement. At the pan-Canadian level, girls had higher scores than boys on the general terms in mathematics index, while no gender gap was found in the geometry and measurement terms index.

Students were asked about a broad range of activities used to help them learn mathematics, but many of these had little to no correlation with student achievement. Only those items that constitute the teacher-directed mathematics activities index were found to be correlated with student achievement, and students who had the highest scores on this index had the highest level of achievement on the mathematics assessment. The activities that make up this index tend to be teacher directed and to involve students practising their skills. Francophone students and girls scored higher on this index than did anglophone students and boys.

The third theme of this chapter was how student learning in mathematics was supported and assessed, and also how students' sense of belonging affected their learning. Teachers who intentionally bring math into other subject areas, or who bring other subjects into math, can increase the motivation of students to learn mathematics by highlighting its real-world applications. Given the focus on STEM education, it is unsurprising that students reported that mathematics is most frequently used in science (86 percent) and technology (73 percent) classes; yet close to 40 percent of students reported using what they had learned in mathematics in art class.

This section also analyzed what types of homework students reported being assigned. The most common form of mathematics homework was paper-and-pencil calculations, with two-thirds of students indicating that they were assigned this type of homework at least once per week. About half of students reported that word problems or studying for assessments were assigned for homework at least once per week. The least common homework assignments involved participating in online forums or blogs or using concrete or virtual manipulatives.

The student questionnaire asked about types of assessment used by teachers to determine student progress in mathematics. Students reported that teachers most frequently used homework or teacherdeveloped classroom assessments to assess learning. The least common type of assessments was that based on portfolios and journals. Students are more likely to be assessed on individual, rather than group, assignments or projects; this approach may reflect the difficulty of fairly assessing individual students' contribution to work done by multiple students. Students were also asked about the use of rubrics in assignments. Although about three-quarters of students know what a rubric is, only half of students reported that they use a rubric when starting an assignment in mathematics class. The frequent use of rubrics was associated with higher math achievement.

Over 75 percent of students reported that they sometime or often receive support from teachers in their mathematics class in the form of information to help them solve a problem, details on marking criteria, and regular feedback to improve their learning. An increase in the frequency with which teachers provided hints or strategies to solve a problem had the strongest positive relationship with achievement.

The relationship between students and teachers holds an important place within the research literature on students' sense of belonging or connectedness at school. The overwhelming majority of students who participated in PCAP 2019 agreed or strongly agreed with all statements related to their sense of belonging. For example, over 90 percent of students stated that they have a friend or friends at school whom they can trust, and approximately 80 percent of students reported that they make friends easily at school and feel that they belong. A sense of belonging or connectedness at school was positively related to student outcomes in mathematics. The more strongly that students agreed with the statement "I liked school," the higher their mathematics scores; however, about one in three students reported that they disagreed with that statement.

The chapter ended with information about the cognitive demands of tasks used in classrooms and on assessments. Generally, students reported that they have encountered a larger proportion of questions with lower, rather than higher, cognitive demands, both in their lessons and on assessments. Nonetheless, close to one-third of students reported that they often encounter questions at higher cognitive levels. This chapter presents findings on the characteristics of Canadian Grade 8/Secondary II teachers and classrooms, and highlights meaningful relationships between these characteristics and student achievement in mathematics. The drivers of educational attainment are becoming a significant focus of research, given the increasing importance of STEM skills in the labour force in this age of automation and artificial intelligence. With the goal of exploring some of these drivers, this chapter will explore four broad topics: classroom characteristics, teacher characteristics, classroom practices, and challenges to teaching.

### **Classroom characteristics**

Canadian classrooms vary in size and composition; there is no such thing as a "typical" or average classroom in Canada. Large schools may have many Grade 8/Secondary II classes, while smaller schools may have multiple grades within a single classroom. In order to better understand the complexities of classroom environments in Canada, the PCAP teacher questionnaire asked teachers about the characteristics of their classrooms.

### Class size

The effects of class size on achievement have been widely studied, from the perspective of both education and economics. Class size refers to the number of students in a class with a teacher. The issue of class size is of particular significance among educational stakeholders, as it is a key determinant of provincial education budgets. Class-size reduction is costly, as it requires hiring more teachers, which results in budget increases to cover their salaries and benefits. Parents prefer having their children in smaller classes, as they believe that having fewer students in a classroom leads to more personalized attention from teachers. Teachers, teachers' unions, and school administrators agree that students in smaller classes receive more individualized attention and that smaller class size has a positive impact of managing student behaviour and stabilizing teacher workload.

Class size has been a widely debated topic (overviews of the debate on the role of class size on achievement can be found in Hanushek (2003) and Rivkin et al. (2005)). Although decades of available evidence indicate that smaller class sizes have little-to-no impact on student achievement, other studies have reported that reduced class size has a positive impact in some settings (Chingos & Whitehurst, 2011). Either way, class size is not a singular determinant of high-quality instruction.

Although large-scale surveys such as PCAP can identify patterns in the relationship between class size and achievement, determining the reasons behind such relationships requires further study — often at the local level, where the effects of increased and decreased class size can be examined in a more meaningful way.

Table 3.1 shows the class sizes reported by mathematics teachers in schools sampled for PCAP 2019. It is important to note that, in the majority of provinces, fewer than 30 teachers reported classes with fewer than 15 students, 15 to 19 students, or 30 or more students. Based on teachers' responses, the most common class sizes ranged from 20 to 29 students (Appendix A.3.1.1). Although

a 56-point achievement gap was found in the PCAP 2019 mathematics assessment between the largest and smallest class-size categories (Figure 3.1), this information must be interpreted with caution, given the low number of teachers reporting in these categories. This limitation is due, in part, to the school-based sampling process used for PCAP. Typically, one class was selected per school, and the mathematics teacher for that class completed the teacher questionnaire.<sup>9</sup> However, for small populations, a representative sample size was sometimes drawn from a small number of schools. For example, although 165 anglophone schools were sampled in British Columbia, only 13 francophone schools were sampled to reach the same level of coverage of francophone students. Thus, for smaller populations, it was not always possible to have 30 observations in each category for class size.

|     | Fewer than 15 | 15 to 19 | 20 to 24 | 25 to 29 | 30 or more |
|-----|---------------|----------|----------|----------|------------|
| BC  | 3‡            | 10‡      | 36       | 40       | 11‡        |
| AB  | 4‡            | 10‡      | 19‡      | 39       | 28         |
| SK  | 14‡           | 22       | 24       | 27       | 12‡        |
| MB  | 9‡            | 21       | 39       | 23       | 8‡         |
| ON  | 7‡            | 5‡       | 30       | 43       | 14‡        |
| QC  | 5‡            | 9‡       | 15‡      | 54       | 17‡        |
| NB  | 8‡            | 16       | 29       | 35       | 11‡        |
| NS  | 6‡            | 18‡      | 36       | 37       | 2‡         |
| PE  | 8‡            | 8‡       | 56‡      | 27‡      | 0          |
| NL  | 20            | 8‡       | 18‡      | 47‡      | 7‡         |
| CAN | 7             | 9        | 27       | 42       | 15         |

#### TABLE 3.1 Percentage of teachers by size of Grade 8/Secondary II mathematics classes

<sup>‡</sup> There are fewer than 30 observations.

At the pan-Canadian level, the highest scores in mathematics were achieved by students in classes with 30 or more students. The largest gap in mathematics achievement was found between students in the smallest (fewer than 15) and largest (30 or more) class-size categories. No statistically significant differences were found between classes of fewer than 15 students and classes of 15 to 19 or 20 to 24 students (Figure 3.1, Appendix A.3.1.2). It is important to exercise caution when interpreting these findings, as class size may be confounded with many other factors, such as school size, school location, and school resources.

<sup>&</sup>lt;sup>9</sup> Student, teacher, and school response rates can be found in Appendix A in the PCAP 2019 public report (O'Grady, Houme, et al., 2021).



#### FIGURE 3.1 Relationship between class size and mathematics achievement

\* Significant difference compared to the *fewer than 15* category *Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

In PCAP 2019, as in earlier PCAP cycles, responses from teachers confirm the wide range of class sizes by province and by language of the school system. In Canada overall, the modal category of class size of Grade 8/Secondary II mathematics classes in 2019 was between 25 and 29 students, a figure consistent with the previous administration of PCAP in 2016 (O'Grady, Fung, Servage, et al., 2019). Table 3.2 presents class sizes by province for both language systems. In PCAP 2019, the proportion of classes with 25 or more students was higher in francophone schools than in anglophone school (69 percent and 55 percent, respectively). In Canada overall, the proportion of small classes (fewer than 15 students) was smaller in francophone schools than in anglophone ones (5 percent and 7 percent, respectively). At the provincial level, the trend was reversed: the proportion of classes with fewer than 15 students was higher in francophone schools compared to anglophone schools in all provinces except Quebec. Within anglophone school systems, Alberta has the highest proportion of classes with 30 or more students (28 percent); within francophone systems, the highest proportion was in New Brunswick (21 percent) (Appendix A.3.1.3).
|                           | Number of<br>participating<br>teachers* | Fewer than 15 | 15 to 19       | 20 to 24 | 25 to 29 | 30 or more |  |  |
|---------------------------|---|---------------|----------------|----------|----------|------------|--|--|
| Anglophone school systems |   |               |                |          |          |            |  |  |
| BC                        | 205                                     | 3‡            | 10‡            | 36       | 40       | 11‡        |  |  |
| AB                        | 140                                     | 4‡            | 10‡            | 19‡      | 39       | 28         |  |  |
| SK                        | 166                                     | 14‡           | 22‡            | 25       | 28       | 12‡        |  |  |
| MB                        | 162                                     | 9‡            | 21‡            | 39       | 23       | 8‡         |  |  |
| ON                        | 221                                     | 7‡            | 5‡             | 30       | 44       | 14‡        |  |  |
| QC                        | 70                                      | 19‡           | 14‡            | 27‡      | 35‡      | 5‡         |  |  |
| NB                        | 89                                      | 7‡            | 16‡            | 28‡      | 41‡      | 8‡         |  |  |
| NS                        | 116                                     | 6‡            | 18‡            | 37       | 36       | 2‡         |  |  |
| PE                        | 23                                      | 5‡            | 7‡             | 60‡      | 28‡      | 0          |  |  |
| NL                        | 116                                     | 20            | 8‡             | 18‡      | 47‡      | 7‡         |  |  |
| CAN                       | 1,308                                   | 7             | 9              | 29       | 40       | 14         |  |  |
|                           |   | Francop       | hone school sy | stems    |          |            |  |  |
| ВС                        | 16                                      | 22‡           | 33‡            | 22‡      | 22‡      | 0          |  |  |
| AB                        | 15                                      | 36‡           | 0              | 36‡      | 28‡      | 0          |  |  |
| SK                        | 7                                       | 57‡           | 29‡            | 14‡      | 0        | 0          |  |  |
| MB                        | 19                                      | 39‡           | 11‡            | 28‡      | 11‡      | 11‡        |  |  |
| ON                        | 123                                     | 10‡           | 6‡             | 37       | 32‡      | 15‡        |  |  |
| QC                        | 147                                     | 3‡            | 8‡             | 13‡      | 57       | 19‡        |  |  |
| NB                        | 62                                      | 11‡           | 19‡            | 32‡      | 18‡      | 21‡        |  |  |
| NS                        | 11                                      | 14‡           | 7‡             | 21‡      | 50‡      | 7‡         |  |  |
| CAN                       | 406                                     | 5             | 9              | 17       | 51       | 18         |  |  |

#### TABLE 3.2 Percentage of teachers by size of Grade 8/Secondary II mathematics classes and language of the school system

\* Numbers of participating teachers are from O'Grady, Houme, et al. (2021, Table A.4, p. 207).

<sup>‡</sup> There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

## Meeting the needs of all students

#### Presence of another adult

Inclusive classrooms aim to meet the needs of each student, irrespective of their ability and skills. Often, students and teachers in inclusive classrooms require the support of educational assistants to accomplish this goal. Class-size information alone may misrepresent the actual number of educators within a classroom. In Canada, many Grade 8/Secondary II mathematics classrooms have additional adults to support student learning: as presented in Figure 3.2, 12 percent of Grade 8/Secondary II mathematics teachers had another adult with them most or all of the time. However, more than half of teachers had no additional adults in their classroom. There is much variation across provinces: more than 60 percent of teachers in Ontario and Quebec reported never having another adult in their classrooms, compared to one-third of teachers in Nova Scotia. At least 1 in 5 teachers in British Columbia, Alberta, Saskatchewan, Manitoba, and New Brunswick stated that another adult was in their classroom most or all of the time, compared to fewer than 1 in 10 in the other provinces (Appendix A.3.2.1).



# FIGURE 3.2 Percentage of teachers indicating that other adults were present in their mathematics class

As shown in Figure 3.3, the average scores of students in mathematics classrooms where there was never another adult besides the classroom teacher were higher than those where another adult was present for part or all of the school day (Appendix A.3.2.2). As was the case with class size, caution is needed in interpreting this finding, as causal relationships cannot be drawn from large-scale survey data. With respect to this factor, schools may assign additional educational staff to classrooms in order to better support the needs of students with academic or behavioural challenges or special education needs, although further research would be needed within provinces and territories to explore this question.

#### FIGURE 3.3 Relationship between the presence of another adult in the mathematics classroom and mathematics achievement



\* Significant difference compared to the never category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

### Modifying teaching to meet student needs

Meeting all students' needs is a growing priority for ministries of education across the country. A growing challenge for teachers is to meet the diverse needs in student-centred classrooms. Differentiated instruction takes into consideration that all students must build on previous knowledge and skills, and that not all students in a classroom have the same foundation. A lesson related to a particular skill may be challenging and complex for some students but familiar, even boring, for others. Teachers strive to meet the needs of their students in a wide variety of ways, including using help from other professionals and modifying or adapting lessons and resources to accommodate the variety of needs they encounter (Tomlinson, 2017).

The PCAP teacher questionnaire asked teachers how often they used six types of accommodations to meet the needs of their students (Figure 3.4). The list was not intended to be exhaustive, and there was no attempt to link this information to specific classroom composition, but it serves to probe ways in which teachers strive to support their students. The two modifications that teachers were most likely to use often in their classrooms were allowing more time to accomplish a task (76 percent) and adapting teaching methods (59 percent). About half of teachers reported that they sometimes or often have help from an educational assistant in their classroom, use assistive technologies, or withdraw students from class for targeted intervention (Appendix A.3.3).





The data show that more frequent use of accommodations was associated with lower achievement in mathematics, as the average scores of students in classrooms where accommodations were never used tended to be higher than those in classrooms where accommodations were sometimes or often used (Figure 3.5, Appendix A.3.3). However, when interpreting such relationships, it is important to consider classroom composition at the school or student level, an analysis that is beyond the purview of large-scale assessment projects such as PCAP.

# FIGURE 3.5 Relationship between the frequency of accommodations and modifications for students and mathematics achievement



\* Significant difference compared to the never category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

### Class composition

Canada is recognized as one of the most diverse countries in the world. Canadian classrooms include students with varied socioeconomic and cultural backgrounds, languages, religions, and sexual orientations, as well as students with academic and behavioural challenges. Students within these classrooms have diverse learning needs, and the abilities of students in a given classroom may span several grade levels. The complexity of class composition can be compounded in classrooms that have multiple grades — that is, classrooms in which teachers are tasked to deliver grade-specific curricula to two or more grade levels within the same classroom. The additional challenge for teachers in such classrooms is that they need to be experts in two or more levels of curricula across many subject areas. Out of administrative necessity, small schools, rural schools, and dual-track immersion schools often opt for multi-grade classrooms to balance teacher-student allocations. Many studies have found that multi-grade classrooms are not detrimental to student achievement and offer a learning environment that is more similar to the real world (Burns & Mason, 2002; Abalde, 2014).

Figure 3.6 presents teachers' responses on the number of grades in their mathematics classrooms. At the pan-Canadian level, 30 percent of classrooms have multiple grades. Teachers in Saskatchewan, Manitoba, Ontario, and New Brunswick reported the most multi-grade classrooms, while teachers in Quebec, Nova Scotia, and Newfoundland and Labrador reported the fewest. Fifteen percent of teachers in Alberta, Manitoba, and New Brunswick reported having three or more grade levels within a classroom (Appendix A.3.4.1).



#### FIGURE 3.6 Number of grade levels in mathematics classrooms

The data from PCAP 2019 show that students who learn in a classroom with one grade achieved higher scores in mathematics than their peers in multi-grade classrooms. Results for students in classes with two grades and those in classes with three or more grades were similar to each other (Figure 3.7, Appendix A.3.4.2). Although there may be no achievement difference in mathematics in various multi-grade classrooms, the increased teacher workload associated with such configurations would be an important factor to consider when interpreting this information at the local level. In Canada, we value many kinds of diversity in classrooms, and teaching in a multi-grade classroom is but one example. But teachers with two or more grade levels within a classroom may lack specific training, classroom resources, or the time required to meet the various needs of a heterogeneous and multi-age group of students. Nonetheless, some positive aspects of multi-grade classroom configurations have been identified. In a study of Catholic schools facing declining enrolments in the United States, Proehl et al. (2013) found that, in multi-grade classrooms, "the students were more likely to nurture other students and be nurtured by them; assume shared responsibility and leadership in the classroom and at home; were involved in fewer disciplinary incidents; and were more respectful of their classmates" (p. 417). Although there have been efforts to promote schools without designated grade levels (e.g., Anderson, 1992) and to champion multi-age education (Pardini, 2005), the most common classroom configuration remains single grade.

#### FIGURE 3.7 Mathematics achievement in single-grade and multi-grade classrooms



\* Significant difference compared to the one grade only category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

### Substitute teachers

There are occasions when a teacher is absent and, consequently, a substitute teacher is necessary. Teachers may, among other things, need to attend personal and family appointments, stay home while ill, or engage in professional development opportunities. Yet such interruptions to regular instructional time may cause disruptions to students' regular routines and interfere with maintaining high academic expectations.

Substitute teachers have education credentials and licensing requirements within provinces to qualify them to fill in for regular classroom teachers who are not able to be present. Despite their qualifications, substitute teachers may be unfamiliar with the curriculum, the routines of the school, or the classroom routines of the teachers whom they are replacing. Moreover, qualified substitutes are challenging to find in some subject areas; for example, there continues to be a shortage of French-language teachers (Office of the Commissioner of Official Languages, 2019).

At the pan-Canadian level, the majority (53 percent) of teachers who participated in PCAP 2019 reported that their classes were taught by a substitute teacher for five or fewer days out of the school year (Figure 3.8). At the upper range, at least 10 percent of teachers in Alberta, New Brunswick, and Prince Edward Island reported that a substitute teacher had taught their class for 20 or more days in the 2018–2019 school year (Appendix A.3.5.1).



FIGURE 3.8 Number of days in the current school year taught by a substitute teacher

The data on the frequency of the use of substitute teachers show a non-linear relationship with achievement. Students in classrooms in which the teacher was absent between 10 and 19 days had the lowest achievement compared to the other three categories (Figure 3.9, Appendix A.3.5.2). Although it seems surprising that longer teacher absences may have limited impact on student achievement, this may be because substitute teachers are becoming more productive over the course of a longer assignment, or administrators may be seeking out more productive (or experienced) teachers for extended job assignments (Herrmann & Rockoff, 2010).

# FIGURE 3.9 Relationship between the number of days taught by a substitute teacher and mathematics achievement



\* Significant difference compared to the 10 to 19 category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

## Lost instructional time

Time is an important school resource, and the amount of time available for students to learn has a direct impact on teachers' opportunities to teach and students' opportunities to learn. Instructional time, and the continuity of learning time, may be disrupted by both planned events such as assemblies and unplanned events such as school closures due to bad weather.

In PCAP 2019, more than half of mathematics teachers said that instructional time was sometimes or often lost due to student misbehaviour or other disruptions such as announcements, assemblies, and visits. Discussion unrelated to the mathematics lesson was the least likely reason for loss of instructional time, as more than two-thirds of teachers said that such discussions rarely or never caused time to be lost in their classroom (Figure 3.10, Appendix A.3.6.1). This suggests that instructional time is most frequently lost for reasons that are outside the control of teachers.



#### FIGURE 3.10 Reasons for lost instructional time

As shown in Figure 3.11, the average mathematics scores of students in classrooms where instructional time was sometimes or often lost due to student misbehaviour or other disruptions were lower than those in classrooms where instructional time was never or rarely lost due to these reasons. Time lost when mathematics classroom discussion went off-track seems to have the weakest relationship with mathematics scores (Appendix A.3.6.2). Caution must be taken in interpreting these results, as classroom disruptions and student misbehaviours may be confounded with many other factors, such as classroom composition, school climate, students' socioeconomic status, and students' attitudes toward learning.

#### FIGURE 3.11 Relationship between lost instructional time and mathematics achievement



\* Significant difference compared to the never or rarely category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

## **Teacher characteristics**

In PCAP 2019, teachers were asked to self-report on several demographic traits, as well as their pedagogical practices, training, and attitudes. This section will explore a number of these teacher traits in order to support an understanding of student achievement in mathematics. Specifically, this section describes trends with respect to teachers' gender, years of experience, teaching assignment, understanding of vertical articulation, and confidence in teaching.

### Gender

As a response to achievement gaps by gender in a number of subjects, researchers have been considering the association between teachers' gender and student achievement. Andersen and Reimer (2019) have explored same-gender advantage, noting three broad theoretical explanations for this advantage: 1) teachers prefer students of the same gender, resulting in a "discrimination effect," with stronger expectations and evaluations for students of the same gender; 2) students respond more positively to teachers of the same gender through a "role-model effect"; and/or 3) teachers' pedagogical approaches and students' learning processes change as a result of a gendered "interaction effect" between the teacher and student. In their study of the relationship between teacher and student gender and student achievement in Denmark, the authors found evidence that students of the same gender as their teachers had higher performance in mathematics and literacy, and that this advantage was greater for girls than boys (Andersen & Reimer, 2019). Similarly, in their study of 10 francophone countries in West and Central Africa, Lee et al. (2019) found that girls with female teachers showed stronger mathematics and reading performance than those with male teachers, while no advantage was found for boys with male teachers. At the same time, some studies have noted reports of poorer mathematics results for girls than boys in elementary classrooms with female teachers. Some researchers have related these results to math anxiety among female teachers, which could reinforce gender stereotypes related to the proficiency of boys and girls in mathematics; however, other researchers argue that such gaps are related more to the level of background knowledge of the teacher than to gender (Antecol et al., 2012).

In Canada overall, 98 percent of Grade 8/Secondary II mathematics teachers who completed the PCAP teacher questionnaire identified themselves as female or male. A small proportion of teachers identified themselves in another way (0.03 percent) or preferred not to say (1.6 percent). Mathematics teachers at this level are more likely to be male than female (60 to 38 percent). This pattern differs from the general distribution of educators by gender in public elementary and secondary schools reported by Statistics Canada (2021b) in 2019–20, in which 75 percent of teachers were female and 25 percent were male. The pattern that mathematics teachers were more likely to be men held across all provinces, although percentages varied: Manitoba (65 percent) and Quebec (66 percent) reported the highest proportions of male teachers, while the smallest proportions were reported in British Columbia (52 percent) and Newfoundland and Labrador (51 percent). The proportion of teachers who chose to identify themselves in another way was generally too small to be reported reliably, while, in provinces with sufficient numbers to be reported reliably, the proportion of teachers who chose not to self-identify by gender ranged from 1 to 5 percent (Table 3.3, Appendix A.3.7.1).

|     | Female | Male  | Another way | I prefer not to say |
|-----|--------|-------|-------------|---------------------|
| BC  | 47.4   | 51.7  | 0           | 0.9‡                |
| AB  | 37.5   | 57.6  | 0           | 4.9‡                |
| SK  | 42.3   | 56.5  | 0           | 1.2‡                |
| MB  | 32.8   | 65.3  | 0           | 1.9‡                |
| ON  | 37.8   | 60.8  | 0           | 1.4‡                |
| QC  | 33.3   | 66.0  | 0           | 0.7‡                |
| NB  | 35.1   | 63.7  | 1.2‡        | 0                   |
| NS  | 37.1   | 61.8  | 0           | 1.1‡                |
| PE  | 44.3‡  | 55.7‡ | 0           | 0                   |
| NL  | 49.1   | 50.9  | 0           | 0                   |
| CAN | 38.0   | 60.3  | 0.0‡        | 1.6‡                |

#### TABLE 3.3 Percentage of mathematics teachers by gender

<sup>‡</sup> There are fewer than 30 observations.

In Canada overall, there is no relationship between achievement in mathematics and whether students are taught by a male or female teacher at the Grade 8/Secondary II level (Figure 3.12, Appendix A.3.7.2). Although mathematics scores were significantly different for students in classrooms whose teacher chose not to self-identify by gender, the proportion of teachers in these categories was too small to be able to draw reliable conclusions regarding student achievement.

#### FIGURE 3.12 Relationship between gender of teacher and mathematics achievement



\* Significant difference compared to the *female* category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

-- Data suppressed to ensure confidentiality of respondents

## Teacher specialization

There is widespread scholarly agreement that teachers should have a solid mastery of the content that they are teaching (Bolyard & Moyer-Packenham, 2008; Goldhaber & Brewer, 1996; Y. Li, 2008; Rice, 2010). According to an extensive review of the literature on the quality of science and mathematics teachers over the past 40 years (Bolyard & Moyer-Packenham, 2008), evidence points to a generally positive association between subject-matter preparation (as measured by subject-specific degree(s) and coursework) and student achievement. Badgett (2014) found a positive association between teachers who have completed a master's degree and student reading achievement; however, Horn and Jang (2017) noted that other studies did not find similar trends at most levels of education for student achievement in reading and mathematics. Yet another study found that, at the elementary level, an increased level of mathematics-content specialization for teachers was associated with more positive student outcomes (Y. Li, 2008).

Some research also suggests that teacher experience plays a role in student achievement. For example, in their investigation of middle-school mathematics and language teachers, Ladd and Sorensen (2017) found that students of teachers who had more years of teaching experience had higher test scores and better behaviour (seen, most prominently, in the reduction of student absenteeism). However, others have not found evidence of an association between increased classroom, emotional, or instructional supports and increased years of teacher experience. They have suggested, instead, that the more positive associations may be a result of less effective beginner teachers leaving the classroom to pursue other career paths (Graham et al., 2020).

In PCAP 2019, specialization in mathematics was measured through questions related to teachers' education, experience, and self-identification as a specialist. Teacher education includes the degrees or diplomas earned, the number of mathematics or mathematics-related courses completed during preservice training, and areas of study in pre-service education or training. Teacher experience includes years of teaching practice and the proportion of the teacher's assignment that is in mathematics. Teachers were also asked to indicate whether they considered themselves a specialist in teaching mathematics, and, if so, whether that was based on education, experience, or both.

#### **Teacher education**

In Canada, Grade 8/Secondary II teachers generally become qualified to teach by completing a bachelor of education (BEd) degree, either concurrently with an undergraduate degree or consecutively following the completion of an undergraduate degree from an accredited university. At least one supervised practicum in the field is required in any teacher education program. Its duration ranges from approximately two to six months, depending on the province or territory and accrediting institution. Some provinces and territories also require a qualifying examination, completion of a probationary teaching period, and/or completion of a mentoring or induction program that may provide another full year of professional support, including orientation, mentoring, and professional development in areas such as subject-specific content and processes, classroom management, and effective communication. Incentives may be offered at the provincial/territorial or school board/ district level, including higher salaries and promotions, for teachers to further their qualifications by acquiring additional academic credentials or specialist courses.<sup>10</sup>

In PCAP 2019, teachers were given a list of eight education attainment categories and asked to identify all that applied to them. As shown in Figure 3.13, close to 90 percent of teachers reported holding a BEd or equivalent (i.e., a form of teacher training). About one-third of teachers had a bachelor of arts (BA) degree, while one-quarter had a bachelor of science (BSc) degree. Nine percent had a master's degree in education, while another 9 percent had a master's degree in another specialization; 1 percent had completed studies at the doctoral level (Appendix A.3.8.1). In Canada overall, no relationship was found between teacher education and student achievement in mathematics (Figure 3.14, Appendix A.3.8.1).



#### FIGURE 3.13 Academic credentials of Grade 8/Secondary II mathematics teachers

*Note:* Teachers identified all degrees that they held; numbers represent percentages for each category.

<sup>&</sup>lt;sup>10</sup> Data are available for teacher education requirements for mathematics from the *Teacher Education and Development Study in Mathematics 2008*, in which teacher education was surveyed in 17 countries and several Canadian provinces (CMEC, 2010).



#### FIGURE 3.14 Relationship between teacher education and mathematics achievement

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

To examine the level of specialist training in mathematics, teachers were asked two questions related to their postsecondary studies: (1) the number of semester-long courses in mathematics or mathematics-related topics (content), and (2) the number of semester-long courses in mathematics teaching methodology (pedagogy). As shown in Figure 3.15, Grade 8/Secondary II teachers reported taking more courses in mathematics content than in pedagogy. One-third of teachers reported taking six or more courses related to mathematic content in their postsecondary training compared to less than 10 percent of teachers taking the same number of course related to mathematics pedagogy (Appendix A.3.8.2).





To further elucidate the level of specialist training among Grade 8/Secondary II teachers who participated in PCAP 2019, a six-item question asked about the extent to which the specific areas of study shown in Figure 3.16 were part of their formal education and/or training. Most teachers indicated that, during their education/training, they had received at least an introduction to or overview of mathematics (either as a discipline or in relation to pedagogy), assessment methods,

educational psychology, and special education. Over 40 percent of teachers indicated that mathematics as a discipline was an area of emphasis during their education/training; close to 50 percent of teachers indicated that teaching remedial mathematics was not covered at all.



#### FIGURE 3.16 Areas of study during formal education and/or training

#### **Teacher experience**

The effects of teacher experience are rather complex and are intertwined with a number of other factors. Teaching mathematics requires not only a strong understanding of mathematical content, but also specialized knowledge in the form of mathematical pedagogy, or mathematical knowledge for teaching (Ball et al., 2008), in order to support the needs of the diverse learners. Novice teachers may have had extensive exposure to mathematics content through their postsecondary studies and may be confident that they have the background knowledge to teach effectively. According to Thanheiser (2009), they may have mastered the procedural knowledge of mathematics, but they may not explicitly know or understand the mathematical constructs involved. Training for pre-service and novice teachers must include "addressing their beliefs about what it means to know mathematics, to learn mathematics, and to teach mathematics for *understanding*" (Suppa et al., 2020, p. 494). Santagata and Lee (2021) suggest that pre-service and induction programs for novice mathematics teachers should include a focus on individual teachers' pedagogical knowledge, the development of a student-centred vision of mathematics instruction, and tailored support for teaching mathematics during a teacher's first year.

In PCAP 2019, Grade 8/Secondary II teachers were asked how many years they had been teaching. As Figure 3.17 indicates, the length of teaching experience varies greatly. About 20 percent of teachers in Alberta, Saskatchewan, and Manitoba have fewer than five years of experience. On the other side of the scale, at least 30 percent of teachers in Quebec, Prince Edward Island, and Newfoundland and Labrador reported that they had been teaching for more than 20 years (Appendix A.3.8.4).



**FIGURE 3.17** Years of teaching experience

Figure 3.18 illustrates the relationship between students' mathematics achievement and the level of experience of their teachers. In Canada overall, there was no significant difference in students' achievement in relation to the number of years that their teacher had been teaching mathematics once the teacher exceeded 5 years of teaching experience. On the other hand, students in classes taught by a teacher with fewer than 5 years of teaching experience had mathematics achievement similar to that of students taught by a teacher with 5 to 10 years of teaching experience, but had lower achievement than students in classes taught by a teacher with 11 or more years of teaching experience (Appendix A.3.8.5). In PCAP 2010, when mathematics was first the focus of the study, there was a positive relationship between teacher experience and students' mathematics achievement up to the 11-to-15-year range and then a decline thereafter (CMEC, 2012b).



FIGURE 3.18 Relationship between years of teaching experience and mathematics achievement

*Notes:* Darker shade denotes significant difference compared to the *fewer than 5 years* category "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

#### **Teaching assignment**

Teaching assignments may also play a role in students' learning. In their investigation of the structural differences in elementary mathematics teaching assignments, Webel et al. (2017) found advantages and disadvantages in both teaching and learning mathematics that were associated with different teaching assignments. The authors highlight the difference between the "self-contained teacher," who is responsible for teaching all subjects, and "departmentalization," "in which teachers teach one or a few (but not all) subjects to multiple class groups" (Slavin, 1987, pp. 116–117). In their interviews with and observations of elementary mathematics teachers in a departmentalized structure, the authors found that teachers reported having more time to reflect on mathematics teaching and to review specific lessons between classes. This, however, came at the disadvantage of reduced flexibility in meeting the needs of particular classes, such as through extending lessons. The authors further specified models of departmentalization: the "team approach," in which teachers of different content areas collaborate to address student needs; "class swap," in which students switch their teacher for a single subject; and "grade-level mathematics teacher," which is similar to the approach at the secondary level.

The mathematics teachers of students writing the PCAP 2019 assessment were asked to indicate approximately what percentage of their total teaching time for the school year was assigned to mathematics. As shown in Figure 3.19, Quebec stands out as having the highest proportion of teachers whose assignment focused on mathematics: 85 percent of its mathematics teachers spend at least 70 percent of their teaching time teaching mathematics. By contrast, Saskatchewan reported the smallest proportion (7 percent) of teachers whose mathematics assignment made up at least 70 percent of their instructional time. In Canada, on average, approximately one-third of teachers reported having 20 to 39 percent of their total teaching assignment dedicated to mathematics

instruction, while another third reported being assigned to this subject for at least 70 percent of their time (Appendix A.3.8.6). Depending on the structure of a school and/or the organization of a school district, Grade 8/Secondary II teachers could be generalists teaching all subjects to a given class, or they could be considered specialists and assigned to teach mathematics to multiple grades and classes in a given school.



FIGURE 3.19 Percentage of teachers' schedule assigned to mathematics

The relationship between the proportion of teachers' schedules assigned to mathematics and student achievement is shown in Figure 3.20 (Appendix A.3.8.7). At the pan-Canadian level, the percentage of a teacher's schedule devoted to teaching mathematics was not significantly associated with student achievement in classes in which the teacher's mathematics assignment was less than 40 percent of their total teaching assignment. However, mean scores in mathematics were higher in classes in which teachers' mathematics assignment was between 40 and 69 percent of their total teaching assignment. Scores were even higher in classes in which the teacher's mathematics assignment was 70 percent or more of their total teaching assignment, a finding that is consistent with the results reported for mathematics in PCAP 2010 (CMEC, 2012b).

#### FIGURE 3.20 Relationship between percentage of teachers' schedule assigned to mathematics and mathematics achievement



\* Significant difference compared to the 20 to 39% category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

#### Teacher self-identification as a specialist

In the PCAP 2019 teacher questionnaire, teachers were asked whether they considered themselves a specialist in teaching mathematics and, if so, whether they were a specialist based on their education, experience, or both. In all provinces, the highest percentage of teachers who consider themselves specialists reported that they were specialists by both education and experience. Close to 70 percent of teachers identify themselves as specialists by education and experience in Quebec and Nova Scotia, while over 80 percent of teachers in Newfoundland and Labrador consider themselves specialists based on these two criteria. On the other hand, over half of Prince Edward Island teachers do not consider themselves specialists in mathematics (Figure 3.21, Appendix A.3.8.8).



#### FIGURE 3.21 Percentage of teachers identifying themselves as mathematics specialists

Students tended to achieve higher scores when they are taught by teachers who consider themselves specialists by both experience and education, compared with those who do not consider themselves to be specialists (Figure 3.22). However, no significant relationships were observed between the other categories of teacher specialization and student achievement (Appendix A.3.8.9).

#### FIGURE 3.22 Relationship between teacher specialization and student mathematics achievement



<sup>\*</sup> Significant difference compared to the not considered a specialist category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

## Professional development

A wide variety of professional development (PD) opportunities are accessible to teachers and can be pursued both individually and collaboratively, depending on teacher needs. Examples include enrolment in additional qualification programs or courses; attending workshops or conferences; engagement with professional learning communities (PLCs); working toward certificates and/or degrees; reading professional and/or academic literature; and participation in research, informal dialogue, mentoring, or peer observation. Most school districts/boards in Canada also schedule dedicated PD days that address specific school or district issues and initiatives.

Professional development provides opportunities for in-service teachers to upgrade their pedagogical skills and knowledge. Koellner and Jacobs (2015) identify a range of PD models, placing them on a continuum from highly adaptive (e.g., responsive to goals, resources, and context, and based on general guidelines) to highly specified models (e.g., predetermined experiences based on particular learning goals). Adaptive approaches are an increasingly common type of PD, and adaptive PD for mathematics educators provides ongoing long-term opportunities that facilitate incremental learning for teachers (Koellner & Jacobs, 2015).

In PCAP 2019, teachers were asked about the number of days in the past five years in which they had participated in professional development related to the teaching of mathematics. The largest proportion of teachers indicating that they had had nine or more days of PD dedicated to mathematics was in Prince Edward Island, where over 60 percent of teachers reported this level of participation. On the other end of the scale, approximately 50 percent of teachers in Saskatchewan and New Brunswick reported having had two or fewer PD days related to teaching mathematics in the past five years (Figure 3.23, Appendix A.3.9.1). There could be a variety of explanations for this low PD participation. It may imply that, as generalists, Grade 8/Secondary II teachers do not pursue content-specific PD or that they have a stronger need for development in other areas. Alternatively, teachers may not have easy access to PD opportunities focusing on mathematics, or PD on broader topics may be more readily available to them.





Teachers were also asked whether they had participated in the 16 different types of professional development shown in Figure 3.24. At the pan-Canadian level, the most common form of PD was attending workshops or conferences (92 percent). Around 80 percent of teachers had pursued PD to learn more about teaching strategies, integrating technology into teaching, and strategies to adapt instruction to students' interests and needs. Included in the 16 types of professional development are two types of PD specific to mathematics. Over half of the teachers who completed the teacher questionnaire have attended PD sessions focused on mathematics content knowledge, and 62 percent reported receiving support in mathematics teaching. The least popular forms of PD were online training and academic courses (Appendix A.3.9.2).

| FIGURE 3.24 | <b>Types</b> | of professional | development | activities for | teachers in | the past | five years |
|-------------|--------------|-----------------|-------------|----------------|-------------|----------|------------|
|-------------|--------------|-----------------|-------------|----------------|-------------|----------|------------|

| Workshops or conferences  |       | 92                | 8      |
|---|-------|-------------------|--------|
| Teaching strategies   |       | 81                | 19     |
| Integration of technology into teaching   |       | 79                | 21     |
| Differentiated instruction/resources to adapt to<br>students' interests and needs                       |       | 78                | 22     |
| Professional learning communities   | -     | 75                | 25     |
| Formative assessment (assessment for learning,<br>assessment as learning)                               | 7     | /3                | 27     |
| Development of common assessment items  | 69    | )                 | 31     |
| Implementation of new resources   | 65    |                   | 35     |
| Receiving support in mathematics teaching (e.g., lead teachers, coaches, mentors, numeracy specialists) | 61    |                   | 39     |
| Responding to assessment data (school, provincial, national, international)                             | 59    |                   | 41     |
| Mental health literacy/well-being   | 58    |                   | 42     |
| Mathematics content knowledge   | 56    |                   | 44     |
| Receiving instructional feedback from an administrator  | 53    |                   | 47     |
| Social-emotional learning/self-regulation   | 47    |                   | 53     |
| Online training (e.g., webinars, videos)  | 37    | 63                |        |
| Academic courses (e.g., university)   | 29    | 71                |        |
| (   | 0 20  | 40 60             | 80 100 |
|   | Perce | entage of teacher | S      |
|   | Yes   | 5 🔳               | No     |

Teachers were also asked about the impact of these professional development activities on student learning in their classrooms. Table 3.4 presents teachers' perceptions of the impact of each type of PD activity. The PD activity selected by the highest proportion of teachers as having some or a lot of impact on student learning was teaching strategies (85 percent). Eighty-one percent of teachers indicated that PD related to mathematics content knowledge had some or a lot of impact on student learning; three-quarters of teachers thought that receiving support in mathematics teaching from lead teachers, coaches, mentors, or numeracy specialists had a similar impact on student outcomes (Appendix A.3.9.2).

# TABLE 3.4 Percentage of teachers by their perceptions of the impact of professional development activities on student learning

| Professional development activity  | None | A little | Some | A lot |
|--|------|----------|------|-------|
| Teaching strategies  | 1‡   | 14       | 56   | 30    |
| Workshops or conferences   | 2‡   | 14       | 57   | 27    |
| Formative assessment (assessment for learning, assessment as learning)                                       | 1‡   | 18       | 46   | 35    |
| Mathematics content knowledge  | 1‡   | 18       | 49   | 32    |
| Academic courses (e.g., university)  | 2‡   | 18       | 48   | 32    |
| Implementation of new resources  | 1‡   | 20       | 51   | 28    |
| Development of common assessment items   | 2‡   | 19       | 53   | 26    |
| Professional learning communities  | 2‡   | 20       | 47   | 31    |
| Integration of technology into teaching  | 3‡   | 20       | 48   | 30    |
| Differentiated instruction/resources to adapt to students' interests and needs                               | 1‡   | 21       | 44   | 33    |
| Receiving support in mathematics teaching (e.g., from lead teachers, coaches, mentors, numeracy specialists) | 5‡   | 20       | 50   | 25    |
| Mental health literacy/well-being  | 3    | 29       | 48   | 19    |
| Social-emotional learning/self-regulation  | 3‡   | 29       | 49   | 19    |
| Receiving instructional feedback from an administrator   | 7    | 30       | 46   | 17    |
| Online training (e.g., webinars, videos)   | 6‡   | 33       | 49   | 12    |
| Responding to assessment data (school, provincial, national, international)                                  | 11   | 35       | 41   | 12    |

**‡** There are fewer than 30 observations.

Note: Shaded rows indicate activities that at least 80 percent of teachers reported as having some or a lot of impact.

#### Collaboration

Mathematics teachers were asked about the type of collaboration they had experienced with their colleagues. The majority of teachers reported that they never or almost never visit another classroom to learn about teaching or work together to score student work. However, most teachers reported that they engage in discussions with their colleagues at least two or three times a month on how to teach a particular topic (Figure 3.25, Appendix A.3.9.3).



#### **FIGURE 3.25 Types of collaboration between mathematics teachers**

## Teacher self-efficacy

Teachers' sense of efficacy — the belief that they can have a positive effect on student learning — has emerged as an important construct in teacher education. Teacher efficacy is grounded in Bandura's social cognitive theory (1977; 1997), which roots human agency in a sense of self-efficacy. According to Bandura, self-efficacy beliefs motivate people toward specific actions in all aspects of their lives, and therefore have predictive value. Bandura identified two dimensions of self-efficacy: personal self-efficacy and outcome expectancy. When applied to teaching, the personal self-efficacy factor is generally known as personal teaching efficacy (PTE). Teachers with a high level of PTE have confidence that their training and/or experience will allow them to help their students overcome obstacles to their learning (Bandura, 1977). The second factor, when applied to teaching, is commonly called general teaching efficacy (GTE) and is related to teachers' belief that they can influence a student's motivation and performance. According to Moslemi and Mousavi (2019), teachers with higher self-efficacy are likelier to accommodate different learner needs and ultimately remain in the profession.

### Confidence

As shown in Figure 3.26, nearly all teachers said they were somewhat confident or very confident in their ability to perform a variety of mathematical tasks. Relatively small proportions of teachers were not confident in their ability to use technology or online platforms for instruction, but the majority of teachers were not confident about their ability to do coding or programming (Figure 3.26, Appendix A.3.10).



#### FIGURE 3.26 Teachers' confidence in their ability to perform mathematics tasks

Teachers were also asked about their confidence in helping their students develop their understanding of the subdomains of mathematics. As shown in Figure 3.27, teachers expressed the most confidence in their ability to help students understand numbers and operations and the least confidence in teaching the data management and probability subdomain (Appendix A.3.10). Nonetheless, more than 80 percent of teachers were very confident about their ability to help students develop their understanding in each of the four domains.

#### FIGURE 3.27 Teachers' confidence in their ability to help their students understand mathematics subdomains



#### **Attitudes and beliefs**

In PCAP 2019, teachers were asked to identify factors they believed influenced student performance in mathematics. Over 95 percent of teachers agreed or strongly agreed that work ethic, learning from the errors they made, prior knowledge, and teaching influenced students' mathematics outcomes (Table 3.5, Appendix A.3.11.1).

| Students' performance in my mathematics class is mostly influenced by: |                      |          |       |                   |  |  |  |
|--|----------------------|----------|-------|-------------------|--|--|--|
|  | Strongly<br>disagree | Disagree | Agree | Strongly<br>agree |  |  |  |
| Work ethic   | 1‡                   | 1        | 46    | 52                |  |  |  |
| Learning from the errors they make                                     | 1‡                   | 3‡       | 60    | 36                |  |  |  |
| My high expectations for all students                                  | 1‡                   | 11       | 58    | 31                |  |  |  |
| Prior knowledge  | 0‡                   | 3        | 71    | 27                |  |  |  |
| Teaching   | 0‡                   | 3        | 74    | 23                |  |  |  |
| Parents/guardians  | 1‡                   | 15       | 65    | 19                |  |  |  |
| Peer influence   | 1‡                   | 17       | 66    | 15                |  |  |  |
| Availability of help outside the classroom                             | 3                    | 26       | 62    | 9                 |  |  |  |
| Natural ability  | 6                    | 39       | 49    | 6                 |  |  |  |

# TABLE 3.5 Percentage of teachers by level of agreement with statements about factors influencing student performance in mathematics

‡ There are fewer than 30 observations.

Teachers were asked to self-report on their attitudes related to teaching and learning mathematics. A large majority of teachers strongly agreed with a statement about the importance of practice for student learning in mathematics. A majority of teachers also strongly agreed that all students can succeed in mathematics, with a majority disagreeing or strongly disagreeing that success in mathematics requires natural ability (Table 3.6, Appendix A.3.11.2).

#### TABLE 3.6 Percentage of teachers by level of agreement with statements about attitudes related to teaching and learning mathematics

| To what extent do you agree with the following statements about mathematics?  |                      |          |       |                   |  |  |  |
|---|----------------------|----------|-------|-------------------|--|--|--|
|   | Strongly<br>disagree | Disagree | Agree | Strongly<br>agree |  |  |  |
| Practice is important for student learning in mathematics   | 0‡                   | 0‡       | 29    | 71                |  |  |  |
| It is possible for all students to succeed in mathematics   | 1‡                   | 7        | 40    | 52                |  |  |  |
| Success in mathematics requires hard work   | 0‡                   | 5        | 49    | 46                |  |  |  |
| Student success in mathematics requires good teaching   | 0‡                   | 2        | 56    | 42                |  |  |  |
| There is not enough emphasis on basic computational skills in the early grades  | 3                    | 23       | 41    | 33                |  |  |  |
| Students should be given the opportunity to engage<br>in computational thinking (e.g., programming, coding,<br>robotics) in the mathematics classroom | 2‡                   | 17       | 58    | 23                |  |  |  |
| By the time students reach Grade 8/Secondary II, the emphasis in mathematics teaching should be more on problem solving                               | 1‡                   | 28       | 56    | 16                |  |  |  |
| Students should not be allowed to use calculators until they have mastered basic computational skills   | 9                    | 44       | 36    | 12                |  |  |  |
| Success in mathematics requires natural ability   | 14                   | 65       | 19    | 2‡                |  |  |  |
| Because calculators are easily available, there is less<br>need to emphasize basic computational skills in teaching<br>mathematics                    | 38                   | 50       | 10    | 2‡                |  |  |  |

‡ There are fewer than 30 observations.

## **Classroom practices**

This section looks at data related to classroom practices of mathematics teachers in schools that participated in PCAP 2019 and their relationship to achievement, where relevant. The topics discussed in this section include teaching resources, probing student understanding, curriculum knowledge and understanding of vertical articulation, teachers' homework expectations, and assessment practices.

## Resources used for mathematics instruction

To examine the resources used in mathematics instruction, the teacher questionnaire listed 13 items along with a frequency of use scale that ranged from *never* to *often*. As Figure 3.28 shows, the majority of teachers reported often using calculators and mathematics curriculum documents in their mathematics instruction. Forty-five percent of teachers often use activities that they designed and about 40 percent often use interactive white boards, worksheets, and textbooks. The least frequently used resource was packaged instructional programs, with close to 80 percent of teachers never or rarely using them (Figure 3.28, Appendix A.3.12.1).



#### FIGURE 3.28 Resources used in mathematics instruction

### Probing student understanding

PCAP 2019 asked teachers how they provide students with opportunities to show their mathematical understanding. Over 80 percent of teachers reported using mathematical language to probe student understanding daily or almost daily. The majority of teachers also asked students daily or almost daily to justify their reasoning and to give oral and written explanations. Forty-three percent of teachers reported integrating technology in students' learning at least weekly to probe their mathematical understanding (Figure 3.29, Appendix A.3.12.2).

# FIGURE 3.29 Teachers' use of opportunities for students to show understanding in mathematics



■ Never or almost never ■ 2 or 3 times per month ■ 1 to 3 times per week ■ Daily or almost daily

## *Curriculum knowledge and understanding of vertical articulation*

Shulman (1986) distinguishes between *lateral curriculum knowledge*, defined as the teacher's ability to link content from one subject area to another (i.e., cross-curricular instruction), and *vertical curriculum knowledge*, defined as the teacher's familiarity with subject area content that has been taught in preceding years and/or will be taught in later years. Whereas *horizon knowledge* refers to the teacher's awareness of how content is connected to the broader mathematical landscape (Hill & Ball, 2009), *vertical articulation* is connected to teachers' understanding of learning progressions, which describe student outcomes throughout different grade levels and provide standards for assessments (Suh & Seshaiyer, 2015). Teachers' knowledge and skills are essential to supporting learning progressions, ultimately enabling the mathematical agenda to move forward (Suh & Seshaiyer, 2015).

To explore their understanding of vertical articulation in mathematic curricula, teachers were asked about their level of agreement with the following two statements:

- I have a deep understanding of the mathematics concepts taught in earlier grades and how they connect to the Grade 8/Secondary II mathematics curriculum.
- I have a deep understanding of the mathematics concepts taught in later grades and how they connect to the Grade 8/Secondary II mathematics curriculum.

As shown in Figure 3.30, 92 percent of teachers agreed or strongly agreed with the first statement, while 78 percent of teachers agreed or strongly agreed with the second (Appendix A.3.12.4).

#### FIGURE 3.30 Teachers' understanding of vertical articulation in mathematics curricula



## Teachers' homework expectations

In PCAP 2019, teachers were asked to report on the amount of time they expected students to spend on mathematics homework. In Newfoundland and Labrador, a quarter of teachers reported that they expected students to do more than two hours of mathematics homework per week. In contrast, only 1 percent of teachers in New Brunswick reported having the same expectation, while 35 percent of teachers in that province reported that they do not assign any mathematics homework. In Canada overall, approximately 60 percent of teachers assign between 30 minutes and two hours of homework in mathematics per week (Figure 3.31, Appendix A.3.13.1). From the student perspective, as reported in Chapter 2, close to 60 percent of students reported spending an hour or less on mathematics homework (this figure does not include those students who reported that their teacher did not assign homework in mathematics). Approximately one in four students spend over an hour on homework in this subject (Figure 2.13, Appendix A.2.4.6).



#### FIGURE 3.31 Amount of time teachers expect students to spend on mathematics homework per week

Figure 3.32 indicates that students whose teachers reported assigning more than 30 minutes of homework per week had higher mathematics achievement on average than students who were assigned less than 30 minutes of homework per week. However, there was no significant difference in mathematics achievement between students who were assigned 30 minutes to an hour of homework and those who were assigned more than an hour of homework (Appendix A.3.13.2). In order to benefit from homework assignments, students need to make an effort to complete those assignments. As shown in Chapter 2, a difference of more than 50 points in mathematics scores was found between students who reported that they often complete their mathematics homework and those who rarely do so (Figure 2.14, Appendix A.2.4.7).





*Notes:* Darker shade denotes significant difference compared to the *less than 30 minutes* category "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

Teachers were asked about the types of homework that they assigned in their mathematics classrooms. Figure 3.33 lists the types of homework in descending order by the proportion of teachers who assign them at least once a week. The most common types of homework were paper-and-pencil calculations and word problems, which were assigned at least weekly by two-thirds of teachers. Perhaps not surprisingly, few teachers reported assigning homework that was based on concepts not taught in class or that required students to participate in online discussions (Figure 3.33, Appendix A.3.13.3).





Teachers were asked to report on how they monitored their students' homework effort. Nearly three-quarters of teachers reported often having a class discussion on homework to gauge the level of effort student had put into it. A majority of teachers also reported often monitoring completion of homework and often having students correct their homework in class. Almost three-quarters of teachers reported that they never or rarely considered homework when determining students' marks (Figure 3.34, Appendix A.3.13.4).





### Assessment

Teachers use assessment to gain insight into students' current ideas, gaps in understanding, and reasoning processes. This information can then be used to adapt instructional and assessment strategies to student needs. The power of assessment to reveal and support learning depends on how well students' responses authentically reflect their thinking and understanding (Shepard, 2005). Questions about assessment are included in each of the three PCAP questionnaires. Questions for teachers look at assessment practices, including strategies and types of questions.

#### **Question types**

There is much debate on the use of multiple-choice tests versus constructed-response tests in which students must communicate their knowledge and understanding using methods such as text, graphs, or tables. Much of the debate has focused on equivalency related to difficulty, reliability, validity, and psychometrics. There are numerous research articles in favour of each type of test, in addition to mixed-methods tests, which use both types. It appears that a broad range of assessment tools is needed to capture important learning goals and processes and to directly connect assessment to ongoing instruction (Shepard, 2000).

Figure 3.35 presents types of questions in descending order by the proportion of teachers who sometimes or often use them on mathematics tests. Overall in Canada, over 90 percent of mathematics teachers reported that they sometimes or often use various types of written-response questions to assess their students' understanding, while approximately 60 percent of teachers use selected-response questions with the same frequency. The most common type of written-response question on mathematics tests requires multi-step solutions, with 71 percent of teachers reporting that they often used this type of question (Figure 3.35, Appendix A.3.14.1).

#### **FIGURE 3.35 Types of questions teachers use on mathematics tests**



Figure 3.36 shows that there is a positive relationship between the frequency of use of extendedresponse items requiring multi-step solutions in mathematics assessment and average mathematics achievement. It is important to note that less than 2 percent of teachers reported that these types of questions were never or rarely used (Appendix A.3.14.2). No relationship with achievement was found for other types of questions.

# FIGURE 3.36 Relationship between frequency of extended-response items requiring multi-step solutions and mathematics achievement



\* Significant difference compared to the *sometimes* category

*Note:* "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.
## Levels of thinking

Teachers use different types of questions to probe students' level of thinking during their interactions with mathematics concepts. As shown in Figure 3.37, over 80 percent of teachers often use questions that measure students' ability to apply knowledge and understanding. Smaller proportions of teachers reported that they often use questions that measure students' ability to explain, justify, evaluate, and generalize their thinking (60 percent) and students' knowledge of facts and concepts (59 percent) (Appendix A.3.14.3).

# FIGURE 3.37 Types of questions teachers use in mathematics assessments to measure different levels of thinking



## Challenges in teaching mathematics

In PCAP 2019, teachers were asked to report on challenges that they face in teaching mathematics. As shown in Figure 3.38, 41 percent of teachers reported that the range of student ability in their class presented a lot of challenges to their mathematics teaching. About one-quarter of teachers reported that they are often challenged by large class sizes, disruptive students, and diverse classroom composition. The challenges that were reported least frequently were concerns about safety and the teacher's limited background knowledge in mathematics (Appendix A.3.15.1).

#### FIGURE 3.38 Challenges in mathematics teaching, as reported by teachers

| The range of student abilities in the cla  | ass                       | 3 23 |          | 33        |          | 4     |    | 1  |      |     |
|--|---------------------------|------|----------|-----------|----------|-------|----|----|------|-----|
| Large class siz  | zes                       | 28   |          | 24        |          | 24    |    | 24 | 4    |     |
| Disruptive stude   | nts                       | 15   | 37       | ,         |          | 24    |    | 24 | 4    |     |
| Students coming from a wide variety<br>kgrounds (e.g., socioeconomic, linguist<br>cultural, et | of<br>ic <i>,</i><br>:c.) | 26   |          | 29        |          | 23    |    | 2  | 2    |     |
| Insufficient time for planni   | ing                       | 27   |          | 36        | 5        |       | 23 |    | 14   |     |
| Time of day for instructi  | on                        | 37   |          |           | 37       |       | 1  | L6 | 11   |     |
| Too much content in the curricul   | um                        | 35   |          |           | 33       |       | 22 | 2  | 10   |     |
| ortage of computer hardware or softwa  | are                       |      | 52       |           |          | 27    |    | 12 | 9    |     |
| of resources for mathematics instruction   | ion                       |      | 44       |           |          | 36    |    | 12 | 7    |     |
| Low morale in the sch  | loc                       |      | 54       |           |          | 28    | 3  | 12 | 2 6  |     |
| Inadequate resources for lesson plann  | ing                       |      | 55       |           |          |       | 33 |    | 8 4  | 1   |
| Insufficient professional developme  | ent                       |      | 49       |           |          | 34    |    | 1  | 12 4 | 1   |
| equate collegial support (e.g., mentori  | ng)                       |      | 66       |           |          |       | 22 |    | 84   | 1   |
| xternal assessments or standardized te   | sts                       |      | 58       |           |          |       | 27 |    | 11 4 | 4   |
| Pressure from parents/guardia  | ans                       |      | 56       |           |          |       | 32 |    | 9 3  | 3   |
| Inadequate physical facilit  | ies                       |      | 7        | 73        |          |       | 1  | 17 | 8    | 3   |
| rriculum inappropriate for the grade le  | vel                       |      | 67       |           |          |       | 23 | ;  | 8    | 2   |
| quate support from school administrate   | ors                       |      | 7        | 73        |          |       |    | 18 | 7    | 2   |
| ncerns for personal safety or the safety<br>studer   | of<br>nts                 |      |          | 82        |          |       |    | 1  | 33   | 2   |
| nitations in my background in the subj   | ect                       |      |          | 76        |          |       |    | 18 | 5    | 1   |
|  |                           | 0 20 | 0        | 40        | 6        | 50    | 8  | 0  | 1    | 100 |
|  |                           |      | Perc     | entag     | ge of te | achei | °S |    |      |     |
| Not at all   |                           | M    | ore thar | n a littl | e        |       |    | ot |      |     |

Large class size Disruptive stude Students coming from a wide variety backgrounds (e.g., socioeconomic, linguisti cultural, et Insufficient time for plann Time of day for instructi Too much content in the curricul Shortage of computer hardware or softwa Shortage of resources for mathematics instructi Low morale in the sche Inadequate resources for lesson planni Insufficient professional developme Inadequate collegial support (e.g., mentori External assessments or standardized te Pressure from parents/guardia Inadequate physical facilit Curriculum inappropriate for the grade level Inadequate support from school administrate Concerns for personal safety or the safety stude Limitations in my background in the subj

Principal component analysis was applied to these components to generate the *challenges to teaching index*, which comprises three of these items: the range of student abilities in the class, disruptive students, and students coming from a wide variety of backgrounds (e.g., socioeconomic, linguistic, cultural, etc.) (Figure 3.39). The top quarter of the index represents students whose teachers were most likely to report that these three factors presented challenges to their mathematics teaching, while the bottom quarter represents those whose teachers reported such challenges least frequently. Unsurprisingly, students have higher achievement in mathematics in classrooms in which teachers reported fewer challenges to teaching, with a 44-point difference in PCAP mathematics scores in Canada overall between the top and bottom quarter of this index. There was no difference in achievement between student in the top and third quarters of the index (Figure 3.40, Appendix A.3.15.2).

# FIGURE 3.39 Teachers' responses to questionnaire items related to the challenges to teaching index



## FIGURE 3.40 Relationship between the challenges to teaching index and mathematics achievement



\* Significant difference compared to the adjacent lower quarter; the bottom quarter is compared to top quarter Note: "Mean score in mathematics – teacher level" refers to the mean student score in a teacher's classroom. The scores shown are the Canadian averages of those means.

## **Summary**

This chapter has presented information on the characteristics of Canadian Grade 8/Secondary II classrooms and on the practices of teachers, based on data gleaned from the PCAP 2019 teacher questionnaire.

Teachers' instructional practices and beliefs and their relationship with achievement in mathematics were examined. According to PCAP 2019, the majority of Canadian teachers have confidence in their ability to teach mathematics and believe that they can positively influence student outcomes. They believe that all students can succeed in mathematics and that this success is the result of quality teaching and student effort rather than natural ability. Higher mathematics achievement was associated with large classes (25 or more students) that had only one grade level and those where teachers' mathematics assignment was 40 percent or more of their total teaching assignment. Lower achievement was found in classrooms where another adult was present most of the time, where accommodations were often used to meet the needs of students, and where teachers reported greater challenges to teaching caused by class composition related to the range of student ability, diverse backgrounds, and disruptive students. Large achievement gaps were found in classrooms in which program modifications were most frequently used and where students were most often given more time to accomplish tasks.

Mathematics teachers use a variety of resources in their classrooms, with the most common resources being curriculum guides and calculators. In terms of providing opportunities for students to demonstrate their understanding, the majority of teachers ask students to use mathematical language, justify their reasoning, and give oral and written explanations daily or almost daily. More than 40 percent of teachers reported integrating technology in students' learning at least weekly to probe their mathematical understanding. Better mathematics outcomes were found in classrooms where teachers assigned 30 minutes or more of mathematics homework per week and often assessed their students using extended-response items requiring multi-step solutions. Mathematics teachers more frequently assessed their students' learning by asking them to demonstrate their understanding by constructing responses rather than by answering selected-response questions.

The majority of Canadian teachers hold a bachelor of education or equivalent (i.e., a form of teacher training). They have pursued professional development opportunities, predominantly though workshops and conferences and through activities focused on teaching strategies, integrating technology into teaching, and strategies to adapt instruction to students' interests and needs. Most teachers reported that they have a deep understanding of the mathematics concepts taught in both earlier grades and later grades and how they connect to the Grade 8/Secondary II mathematics curriculum. Higher mathematics achievement was found in classrooms where teachers considered themselves specialists by both education and experience compared to teachers who did not consider themselves specialists.

Excellence in education depends not only on the motivation of students, the quality and dedication of school personnel, and parental support, but also on the characteristics of and resources available in the school environment. Decisions made at one level of a school system that affect these characteristics and the allocation of these resources are affected by decisions made at other levels. Classroom-level decisions are influenced by school-level decisions, which, in turn, are influenced by decisions related to resources, policies, and practices made at the level of the school district, or even the province/ territory.

All schools experience challenges as they strive to provide the best educational opportunities possible for their students. Nonetheless, from a pan-Canadian and provincial/territorial perspective, the quality of education in Canada is very high. According to a report relating provincial/territorial profiles to achievement equity (CMEC, 2012a), Canada has achieved both high performance levels and a relatively high degree of equity among students across provinces. And, unlike the findings in many other countries that participate in PISA, in Canada the performance of students is only weakly related to their socioeconomic status (OECD, 2019). Yet it is apparent that socioeconomically advantaged students in Canada attend the most successful schools and have access to greater resources; meanwhile, students attending schools with disciplinary problems may experience difficulty with academic performance (Mullis, Martin, Foy, et al., 2012).

This chapter examines four aspects of the schools that participated in PCAP 2019: demographic background and grade configuration; diversity of the school population; factors influencing learning; and challenges to learning and teaching.

Over 1,500 schools from all provinces participated in PCAP 2019. Schools were randomly selected, with the probability of selection proportional to the size of the school, based on the list of all schools with Grade 8/Secondary II students under the purview of the ministry/department of education in each province. Schools were selected separately in the anglophone and francophone school systems; students in French immersion programs were considered part of the anglophone population. Each of the participating schools received a school questionnaire, to be answered by the school principal. The following sections analyze the information provided in response to this questionnaire.

## School demographics and grade configuration

Ongoing efforts by educational researchers have tried to discern the most effective learning environments, ones in which all students can thrive. This section describes environmental factors of the schools in which participating Grade 8/Secondary II students were enrolled. These include demographic features such as the size of the communities in which the schools are located, the type of governance (public versus private), and the number of students enrolled. This section also examines grade configurations in participating schools.

## School demographics

Many factors influence the extent to which a school achieves efficiency and effectiveness. As complex, open systems, schools are always changing in response to the actions of people and institutions both within and outside of school walls. Students and teachers are at the core of schooling, but their classroom environment is situated within a complexity of institutions, resource networks, policies, and practices. Provinces determine the statutory contexts of schooling, and school districts are responsible for developing policies that apply these statutes to best meet the unique needs of the schools and families within their communities.

## **Community size**

The size of the community in which a school is located impacts the allocation of resources and influences options for grade configurations within the school. Over the past few decades, many small rural schools have been closed, and, overall, rural communities have had greater challenges than urban ones in keeping their schools open. Rural school districts often cover large geographical areas, resulting in some unique operational challenges such as the logistics of student transportation networks, maintaining modern facility infrastructure, offering a variety of courses to students, and providing teachers with professional development opportunities. In PISA 2015, a rural disadvantage in academic performance was observed when the performance of students attending schools in communities with populations of less than 3,000 was compared to that in communities with a population greater than 100,000 (Echazarra & Radinger, 2019). Yet, rural communities believe that their schools offer advantages for students and families. With lower teacher-student ratios, these schools may offer students more personalized support, as teachers have opportunities to adapt to student needs and tailor lessons to suit learners.

All schools strive to offer high-quality instruction to students, regardless of their geographical locations. New technologies offer innovative ways to close the rural-urban gaps in education by connecting professional learning networks and expanding distance-learning opportunities for students (OECD, 2013b).

The schools that participated in PCAP 2019 were located in communities ranging from rural areas and small towns to large cities. The population ranges used in this study to categorize different communities are given in Table 4.1.

| Rural/small town        | Medium town     | Small city       | Medium city        | Large city     |
|-------------------------|-----------------|------------------|--------------------|----------------|
| Fewer than 5,000 people | 5,000 to 24,999 | 25,000 to 99,999 | 100,000 to 499,999 | Over           |
|                         | people          | people           | people             | 500,000 people |

## TABLE 4.1 Definitions of communities by population size

At the pan-Canadian level, schools are distributed fairly equally among four of the five categories of communities listed in the school questionnaire. The exception is small cities, which have a lower proportion of schools compared to both smaller and larger communities. Provincially, the proportion of schools by community size varies according to provincial populations: the largest numbers of rural schools are found in the Atlantic and Prairie provinces, which have the smallest populations, while smaller proportions of rural/small-town schools are found in Canada's most populous provinces (Figure 4.1, Appendix A.4.1.1).



#### FIGURE 4.1 Size of communities in which schools are located

In Canada overall, 23 percent of schools in anglophone systems and 26 percent of schools in francophone systems were located in a rural area or small town. In anglophone school systems, the proportion of rural schools varies from 13 percent in Ontario to 81 percent in Prince Edward Island. In francophone school systems, the highest percentage of rural schools was found in Manitoba (67 percent); in British Columbia, none of the francophone schools that participated in PCAP were rural. Across Canada, 21 percent of schools in anglophone systems and 17 percent of schools in francophone systems were located in large cities. At the provincial level, the highest proportions of large-city schools participating in PCAP were located in both the anglophone and francophone school systems in Alberta (39 and 40 percent, respectively) and in the anglophone system in Manitoba (36 percent) (Table 4.2, Appendix A.4.1.2).

|     | Rural/small<br>town | Medium town | Small city     | Medium city | Large city |
|-----|---------------------|-------------|----------------|-------------|------------|
|     |                     | Anglophone  | school systems |             |            |
| BC  | 18‡                 | 10‡         | 25             | 25          | 22‡        |
| AB  | 25‡                 | 24‡         | 11‡            | 1‡          | 39         |
| SK  | 48                  | 8‡          | 6‡             | 37          | 0‡         |
| MB  | 47                  | 13‡         | 4‡             | 1‡          | 36         |
| ON  | 13‡                 | 27‡         | 12‡            | 28          | 20         |
| QC  | 17‡                 | 25‡         | 17‡            | 17‡         | 25‡        |
| NB  | 51                  | 18‡         | 26‡            | 5‡          | 0          |
| NS  | 46                  | 33          | 6‡             | 15‡         | 1‡         |
| PE  | 81‡                 | 10‡         | 10‡            | 0           | 0          |
| NL  | 76                  | 14‡         | 2‡             | 9‡          | 0          |
| CAN | 23                  | 22          | 12             | 22          | 21         |
|     |                     | Francophone | school systems |             |            |
| BC  | 0                   | 9‡          | 64‡            | 18‡         | 9‡         |
| AB  | 20‡                 | 10‡         | 10‡            | 20‡         | 40‡        |
| SK  | 33‡                 | 0           | 33‡            | 33‡         | 0          |
| MB  | 67‡                 | 7‡          | 7‡             | 0           | 20‡        |
| ON  | 30                  | 19‡         | 19‡            | 14‡         | 18‡        |
| QC  | 18‡                 | 25‡         | 24             | 13‡         | 20‡        |
| NB  | 58                  | 27‡         | 13‡            | 2‡          | 0          |
| NS  | 40‡                 | 30‡         | 10‡            | 20‡         | 0          |
| CAN | 26                  | 23          | 21             | 13          | 17         |

# TABLE 4.2Percentage of schools by size of communities in which they are located and<br/>language of school system

<sup>‡</sup> There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

Figure 4.2 shows the relationship between the size of the community in which schools are located and mathematics achievement. Mean scores were similar in smaller communities (rural/small towns and medium towns); however, scores for schools located in cities of any size were higher than those in smaller centres (Appendix A.4.1.3). Such a finding needs to be considered in the context of regional differences, school resources, and other broader underlying factors impacting rural communities and smaller towns.

# FIGURE 4.2 Relationship between the size of the community in which schools are located and mathematics achievement



\* Significant difference compared to the *rural/small town* category *Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## School governance

The popularity of private schools has been on the rise in Canada, with enrolments steadily increasing in recent years (Statistics Canada, 2020b). The relative strengths and weaknesses of public and private education have long been debated. Parents choose private schools for a host of reasons. Private schools offer parents a choice that is believed to provide a learning environment with high-quality teachers within a caring environment where students will thrive. However, some observers have argued that private schools contribute to greater stratification of educational opportunities and outcomes (OECD, 2012). In OECD countries, including Canada, higher academic achievement in private schools has been found to be associated primarily with the higher socioeconomic status and education levels of the families they serve (OECD, 2012; Frenette & Chan, 2015).

As indicated by principals' responses to the PCAP 2019 school questionnaire, 90 percent of participating schools across Canada were publicly funded. This is similar to the proportion reported by Statistics Canada (91.8 percent) for the year prior the PCAP study (Statistics Canada, 2020b). The highest proportions of private schools that participated in PCAP were located in British Columbia and Quebec (29 percent), while only public schools were sampled in New Brunswick, Nova Scotia, and Prince Edward Island (Figure 4.3, Appendix A.4.2.1).



FIGURE 4.3 Percentage of public and private schools

In Canada overall, 91 percent of participating schools in anglophone school systems were publicly funded, compared with 83 percent of schools in francophone school systems. The highest proportions of private anglophone schools were found in British Columbia (30 percent) and Quebec (28 percent), while private francophone schools were reported only in Quebec (29 percent) (Table 4.3, Appendix A.4.2.2).

|     | Anglophone s | school systems | Francophone s | chool systems |
|-----|--------------|----------------|---------------|---------------|
|     | Public       | Private        | Public        | Private       |
| BC  | 70           | 30‡            | 100‡          | 0             |
| AB  | 88           | 12‡            | 100‡          | 0             |
| SK  | 98           | 2‡             | 100‡          | 0             |
| MB  | 86           | 14‡            | 100‡          | 0             |
| ON  | 94           | 6‡             | 100           | 0             |
| QC  | 72           | 28‡            | 71            | 29‡           |
| NB  | 100          | 0              | 100           | 0             |
| NS  | 100          | 0              | 100‡          | 0             |
| PE  | 100‡         | 0              |               |               |
| NL  | 97           | 3‡             |               |               |
| CAN | 91           | 9              | 83            | 17‡           |

#### TABLE 4.3 Percentage of public and private schools, by language of the school system

<sup>‡</sup> There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

As shown in Figure 4.4, the average mathematics score of students in private schools was higher than that of students in public schools (Appendix A.4.2.3). As noted in the literature, such difference in achievement should be considered in the context of the socioeconomic status and parental education levels of the students attending these schools; moreover, given the small proportion of private schools in the sample (10 percent), any conclusions should be drawn with caution.



### FIGURE 4.4 Relationship between school governance and mathematics achievement

\* Significance difference compared to *public* category

*Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## **School size**

Although some studies have not found a connection between the number of students in a school and student achievement (Jones & Ezeife, 2011), others have found a non-linear relationship (Borland & Howsen, 2003). Regardless, the size of a school may be related to several factors that have been found to have a relationship with learning, including socioeconomic status, the number of grade levels in a school, teacher instructional practices, the school environment (Jones & Ezeife, 2011; Gershenson & Langbein, 2015), and opportunities for teacher collaboration and professional development (Abalde, 2014).

The school questionnaire asked principals about their schools' enrolment. In Canada overall, close to two-thirds of schools participating in PCAP 2019 had 101 to 500 students, while about one-quarter of schools had enrolments of 501 to 1,000 students. The highest proportion of small schools, with 100 students or fewer, was in Newfoundland and Labrador (23 percent), while the highest proportion of large schools, with more than 1,000 students, was in Quebec (27 percent) (Figure 4.5, Appendix A.4.3.1).



## **FIGURE 4.5** Total enrolment in schools

When the data are explored by language of the school system, the majority of participating schools in both language systems had between 101 and 500 students, with the exception of anglophone schools in Alberta and francophone schools in Quebec. The greatest proportion of large schools (more than 1,000 students) in anglophone systems was in British Columbia (17 percent) and Quebec (15 percent). The greatest proportion of small schools (under 101 students) was in francophone systems in British Columbia (27 percent) and Saskatchewan (33 percent) (Table 4.4, Appendix A.4.3.2).

|     | Anglophone school systems |            |                 |                    |                 | Francophone school systems |                 |                    |  |  |
|-----|---------------------------|------------|-----------------|--------------------|-----------------|----------------------------|-----------------|--------------------|--|--|
|     | 100<br>or fewer           | 101 to 500 | 501 to<br>1,000 | More than<br>1,000 | 100 or<br>fewer | 101 to 500                 | 501 to<br>1,000 | More than<br>1,000 |  |  |
| BC  | 2‡                        | 53         | 28              | 17                 | 27‡             | 55‡                        | 18‡             | 0                  |  |  |
| AB  | 15‡                       | 48         | 35              | 1‡                 | 10‡             | 90‡                        | 0               | 0                  |  |  |
| SK  | 9‡                        | 83         | 7‡              | 0‡                 | 33‡             | 67‡                        | 0               | 0                  |  |  |
| MB  | 5‡                        | 81         | 14              | 1‡                 | 7‡              | 80‡                        | 13‡             | 0                  |  |  |
| ON  | 4‡                        | 67         | 27              | 3‡                 | 18‡             | 65                         | 14‡             | 3‡                 |  |  |
| QC  | 8‡                        | 53‡        | 25‡             | 15‡                | 1‡              | 41‡                        | 28              | 30                 |  |  |
| NB  | 6‡                        | 70         | 23‡             | 0                  | 11‡             | 76                         | 13‡             | 0                  |  |  |
| NS  | 3‡                        | 79         | 18‡             | 0                  | 0               | 80‡                        | 20‡             | 0                  |  |  |
| PE  | 10‡                       | 67‡        | 24‡             | 0                  |                 |                            |                 |                    |  |  |
| NL  | 22‡                       | 67         | 11‡             | 0                  |                 |                            |                 |                    |  |  |
| CAN | 6                         | 66         | 25              | 3                  | 7               | 53                         | 22              | 18                 |  |  |

#### TABLE 4.4 Percentage of schools by total enrolment and language of the school system

<sup>‡</sup> There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

Figure 4.6 shows that higher results in mathematics were obtained in larger schools (more than 500 students) compared to schools with 500 or fewer students (Appendix A.4.3.3). However, as very few schools that participated in PCAP 2019 had an enrolment of 100 or fewer students, comparisons of mathematics achievement of students in these schools with that of students in schools with larger enrolments are not conclusive.

### FIGURE 4.6 Relationship between school size and mathematics achievement



\* Significant difference compared to the 101 to 500 category

*Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## Grade configuration

Grade configuration determines the number of times a student will move to a new school and the age at which such transitions occur. Numerous studies have explored grade configurations, and these have noted a trend toward merging junior high schools with their elementary counterparts to create schools that include Kindergarten through Grade 8 (Schwartz et al., 2011). It appears that the relationship between grade configuration and achievement is not uniform across countries. An Ontario school board found that various grade configurations (K-6, K-8, and 6-8) did not have a significant long-term effect on Grade 6 EQAO (Education Quality and Accountability Office) assessments within that board once other school-level predictors such as school socioeconomic status, suspension rates, and the proportion of students with specials needs within schools were included in the model (Toronto District School Board Research and Information Services, 2011). Similarly, a study based in Sweden found that various grade configurations (1-6, 1-9, and 7-9) had no significant effect on grade point average, high school graduation, high school program selected (i.e., academic versus vocational), and enrolment in postsecondary programs (Holmlund & Böhlmark, 2019). These studies contrast with work completed in the United States, where Schwartz and colleagues (2011) found that students who attended K-8 schools or K-4/5-8 schools had higher achievement in Grade 8 compared to students who transitioned to middle schools later (i.e., K-5/ 6-8 or K-6/7-8).

In the school questionnaire, principals were asked about the number of grade levels in their schools. The most common grade configuration was nine or more grades, and close to 90 percent of participating schools in Saskatchewan and Ontario had that configuration. Quebec had the highest proportion of schools with five to eight grades (72 percent), while Nova Scotia has the highest proportion of schools with one to four grades (35 percent) (Figure 4.7, Appendix A.4.4.1).



### FIGURE 4.7 Number of grade levels in schools

At least three-quarters of participating schools in anglophone systems in Canada overall, Saskatchewan, and Ontario had nine or more grades; similar proportions were found in francophone school systems in British Columbia, Manitoba, and New Brunswick. Approximately one-third of anglophone schools in the Maritime provinces had four or fewer grades, while about 1 in 10 francophone schools in Alberta and Quebec were this size (Table 4.5, Appendix A.4.4.2).

|     | Anglop        | phone school sys | tems                | Franco        | ophone school sys | stems            |
|-----|---------------|------------------|---------------------|---------------|-------------------|------------------|
|     | 1 to 4 grades | 5 to 8 grades    | 9 or more<br>grades | 1 to 4 grades | 5 to 8 grades     | 9 or more grades |
| BC  | 24            | 40               | 36‡                 | 0             | 18‡               | 82‡              |
| AB  | 21            | 19‡              | 60                  | 10‡           | 40‡               | 50‡              |
| SK  | 1‡            | 10‡              | 89                  | 0             | 33‡               | 67‡              |
| MB  | 16            | 14‡              | 71                  | 0             | 7‡                | 93‡              |
| ON  | 6‡            | 7‡               | 87                  | 4‡            | 35                | 61               |
| QC  | 4‡            | 58               | 38‡                 | 13‡           | 74                | 13‡              |
| NB  | 32‡           | 17‡              | 51                  | 7‡            | 9‡                | 84               |
| NS  | 39            | 18‡              | 43                  | 0             | 40‡               | 60‡              |
| PE  | 33‡           | 5‡               | 62‡                 |               |                   |                  |
| NL  | 19‡           | 19‡              | 62                  |               |                   |                  |
| CAN | 11            | 13               | 76                  | 9‡            | 55                | 36               |

#### TABLE 4.5 Percentage of schools by number of grade levels and language of school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

As shown in Figure 4.8, in Canada overall, average mathematics achievement in schools with five to eight grades was higher than that in schools with nine or more grades (Appendix A.4.4.3). However, given that, at the pan-Canadian level, less than 20 percent of schools that participated in PCAP 2019 had between five and eight grades, comparisons of mathematics achievement of students in those schools with that of students in schools with other grade configurations should be made with caution.

# FIGURE 4.8 Relationship between the number of grade levels in schools and mathematics achievement



\* Significant difference compared to the 5 to 8 grades category

*Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## Grade 8/Secondary II enrolment

Enrolments in each grade affect how schools choose to configure classes, as they strive to optimize resources and effectively meet students' learning needs. Principals of schools participating in PCAP 2019 reported the number of Grade 8/Secondary II students in their schools. Across Canada, Quebec had the largest proportion of schools (60 percent) with more than 100 students in the PCAP target grade. More than half of the schools in Saskatchewan, Prince Edward Island, and Newfoundland and Labrador had 25 or fewer students in Grade 8 (Figure 4.9, Appendix A.4.5.1).



#### FIGURE 4.9 Grade 8/Secondary II enrolment in schools

In Canada overall, close to 60 percent of English-language schools and over 40 percent of Frenchlanguage schools had fewer than 50 students at the Grade 8/Secondary II level. The highest proportions of schools with the largest enrolment at this grade level (200 or more students) were found in the anglophone school systems in British Columbia (19 percent) and in the francophone school systems in Quebec (30 percent) (Table 4.6, Appendix A.4.5.2).

|     | 25 or fewer | 26 to 50 | 51 to 75         | 76 to 100 | 101 to 200 | More than<br>200 |
|-----|-------------|----------|------------------|-----------|------------|------------------|
|     |             | Ang      | lophone school s | ystems    |            |                  |
| BC  | 26‡         | 7‡       | 11‡              | 12‡       | 26         | 19               |
| AB  | 30‡         | 16‡      | 19‡              | 15‡       | 16         | 4‡               |
| SK  | 55          | 34       | 9‡               | 0‡        | 2‡         | 0‡               |
| MB  | 43‡         | 21‡      | 15               | 10‡       | 11‡        | 0‡               |
| ON  | 19‡         | 39       | 24               | 9‡        | 7‡         | 2‡               |
| QC  | 26‡         | 21‡      | 11‡              | 9‡        | 21‡        | 11‡              |
| NB  | 31‡         | 25‡      | 13‡              | 12‡       | 18‡        | 1‡               |
| NS  | 18‡         | 27‡      | 14‡              | 19‡       | 19‡        | 3‡               |
| PE  | 48‡         | 10‡      | 19‡              | 0         | 24‡        | 0                |
| NL  | 52          | 18‡      | 6‡               | 6‡        | 10‡        | 7‡               |
| CAN | 27          | 31       | 19               | 10        | 10         | 3                |
|     |             | Fran     | cophone school s | systems   |            |                  |
| BC  | 73‡         | 18‡      | 9‡               | 0         | 0          | 0                |
| AB  | 50‡         | 30‡      | 10‡              | 10‡       | 0          | 0                |
| SK  | 50‡         | 50‡      | 0                | 0         | 0          | 0                |
| MB  | 67‡         | 27‡      | 0                | 0         | 0          | 7‡               |
| ON  | 51          | 23‡      | 9‡               | 4‡        | 11‡        | 2‡               |
| QC  | 11‡         | 8‡       | 4‡               | 11‡       | 35         | 30               |
| NB  | 51‡         | 27‡      | 9‡               | 8‡        | 4‡         | 2‡               |
| NS  | 60‡         | 10‡      | 10‡              | 20‡       | 0          | 0                |
| CAN | 29          | 15       | 6‡               | 9‡        | 24         | 19               |

# TABLE 4.6 Percentage of schools by enrolment of Grade 8/Secondary II students and language of the school system

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

As can be seen in Figure 4.10, the relationship between mathematics achievement and enrolment figures of Grade 8/Secondary II students was not linear. Compared to small schools with 25 or fewer students at this grade level, students in schools with 51 to 75 and more than 100 students achieved higher scores in mathematics. No significant differences were found for the other enrolment categories (Appendix A.4.5.3).





\* Significant difference compared to the 25 or fewer category Note: "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## Diversity of the school population

Two indicators of school populations' diversity were included in the PCAP 2019 school questionnaire: the proportion of students in English- or French-second-language programs and the proportion of students who identify as Indigenous.

## Second-language learners

Principals were asked to report the percentage of students in their schools who identified as Englishlanguage learners in anglophone schools and French-language learners in francophone schools. These students are, or have been, in special classes for those whose first language is not the language of the school system. Although these programs are available to all Canadian families who send their children to schools in an official language that is different than their home language, they are often associated with immigrant students.

Nova Scotia and Newfoundland and Labrador stand out as having the largest proportion of schools in which more than 50 percent of students are in second-language programs: more than half of the student body are second-language learners in 22 percent of Nova Scotia schools and 28 per cent of Newfoundland and Labrador schools. This compares to only 9 percent of schools in Canada overall. Generally, across provinces, principals most frequently reported that 5 percent or less of their students are in second-language programs (Figure 4.11, Appendix A.4.6.1).



#### FIGURE 4.11 Percentage of second-language learners in schools

In at least one in five anglophone schools in Quebec, Nova Scotia, and Newfoundland and Labrador and the same proportion of francophone schools in Alberta and Nova Scotia, more than 50 percent of students are second-language learners. In about half of the provinces, the percentage of schools where 5 percent or less of students are in second-language programs is similar between the two language systems (Table 4.7, Appendix A.4.6.2).

|     | 0%  | 1 to 5% | 6 to 10%        | 11 to 25%   | 26 to 50% | More than 50% |
|-----|-----|---------|-----------------|-------------|-----------|---------------|
|     |     | /       | Anglophone scho | ool systems |           |               |
| BC  | 8‡  | 58      | 18‡             | 11‡         | 1‡        | 4‡            |
| AB  | 14‡ | 26‡     | 25‡             | 17‡         | 10‡       | 8‡            |
| SK  | 20‡ | 42      | 9‡              | 17‡         | 7‡        | 5‡            |
| MB  | 16‡ | 40      | 9‡              | 11‡         | 13‡       | 11‡           |
| ON  | 15‡ | 46      | 12‡             | 13‡         | 5‡        | 8‡            |
| QC  | 21‡ | 23‡     | 6‡              | 13‡         | 15‡       | 23‡           |
| NB  | 28‡ | 40      | 9‡              | 5‡          | 5‡        | 12‡           |
| NS  | 30  | 40      | 8‡              | 0           | 2‡        | 20‡           |
| PE  | 19‡ | 57‡     | 5‡              | 5‡          | 0         | 14‡           |
| NL  | 54  | 15‡     | 1‡              | 2‡          | 0         | 28‡           |
| CAN | 16  | 43      | 13              | 13          | 6         | 9             |
|     |     | F       | rancophone sch  | ool systems |           |               |
| BC  | 27‡ | 9‡      | 9‡              | 27‡         | 9‡        | 18‡           |
| AB  | 50‡ | 10‡     | 20‡             | 0           | 0         | 20‡           |
| SK  | 40‡ | 60‡     | 0               | 0           | 0         | 0             |
| MB  | 29‡ | 29‡     | 21‡             | 21‡         | 0         | 0             |
| ON  | 30  | 20‡     | 9‡              | 12‡         | 11‡       | 18‡           |
| QC  | 58  | 20      | 3‡              | 8‡          | 1‡        | 8‡            |
| NB  | 54  | 19‡     | 4‡              | 5‡          | 2‡        | 16‡           |
| NS  | 33‡ | 33‡     | 0               | 0           | 0         | 33‡           |
| CAN | 49  | 20      | 6‡              | 9‡          | 4‡        | 12            |

## TABLE 4.7 Percentage of schools by proportion of second-language learners and language of the school system

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

## Indigenous students

As mentioned in Chapter 1, in Canada overall, 4.4 percent of students who participated in PCAP 2019 identified themselves as First Nations, 2.3 percent as Métis, and 0.3 percent as Inuit (Table 1.9, Appendix A.1.21). It is important to note that federally funded schools do not participate in PCAP, and only those students of Indigenous identity who were attending schools under provincial jurisdiction are reported here.

Principals were asked to report the percentage of students in their schools who identify themselves as Indigenous (i.e., First Nations, Métis, or Inuit). The proportion of students who identify as Indigenous in the PCAP sample of schools is relatively small, with principals in most provinces most likely to report that Indigenous students constitute 1 to 5 percent of their student body (Figure 4.12, Appendix A.4.7.1).



### **FIGURE 4.12** Percentage of students who identify as Indigenous in schools

Anglophone school systems in Saskatchewan and both school systems in Manitoba had the highest proportions of Indigenous students: in about one-fifth of these schools, principals reported that at least 25 percent of the student population identified as Indigenous (Table 4.8, Appendix A.4.7.2).

|     | 0%  | 1 to 5% | 6 to 10%       | 11 to 25% | 26 to 50% | More than<br>50% |
|-----|-----|---------|----------------|-----------|-----------|------------------|
|     |     | Ang     | lophone school | systems   |           |                  |
| BC  | 13‡ | 37      | 13‡            | 25‡       | 7‡        | 6‡               |
| AB  | 18‡ | 43      | 16‡            | 22‡       | 1‡        | 0                |
| SK  | 4‡  | 35      | 23             | 17‡       | 11‡       | 10‡              |
| MB  | 4‡  | 30      | 21‡            | 20        | 12‡       | 13‡              |
| ON  | 28  | 58      | 3‡             | 9‡        | 1‡        | 0‡               |
| QC  | 35‡ | 55‡     | 4‡             | 2‡        | 2‡        | 2‡               |
| NB  | 23‡ | 61      | 6‡             | 1‡        | 8‡        | 0                |
| NS  | 3‡  | 59      | 25‡            | 11‡       | 2‡        | 0                |
| PE  | 24‡ | 62‡     | 10‡            | 5‡        | 0         | 0                |
| NL  | 29‡ | 51      | 5‡             | 4‡        | 10‡       | 1‡               |
| CAN | 22  | 51      | 9              | 13        | 3         | 2‡               |
|     |     | Fran    | cophone school | systems   |           |                  |
| ВС  | 0   | 36‡     | 18‡            | 45‡       | 0         | 0                |
| AB  | 20‡ | 80‡     | 0              | 0         | 0         | 0                |
| SK  | 17‡ | 83‡     | 0              | 0         | 0         | 0                |
| MB  | 0   | 20‡     | 27‡            | 33‡       | 20‡       | 0                |
| ON  | 18‡ | 62      | 14‡            | 5‡        | 1‡        | 0                |
| QC  | 76  | 19‡     | 2‡             | 0         | 3‡        | 0                |
| NB  | 44‡ | 50‡     | 2‡             | 4‡        | 0         | 0                |
| NS  | 11‡ | 89‡     | 0              | 0         | 0         | 0                |
| CAN | 53  | 36      | 5‡             | 3‡        | 3‡        | 0                |

# TABLE 4.8Percentage of schools by proportion of students who identify as Indigenous and<br/>language of the school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

## Instructional time management

Within the school year, instructional time describes the time scheduled for instruction and other activities during which students and teachers interact and students are expected to be engaged in learning. Instructional time does not include days that teachers attend conferences or professional development activities.

Districts and schools organize instructional time and schedule subject-specific classes for students. Scheduling (i.e., by the semester or full year) affects classrooms by influencing the continuity of instruction and the types of pedagogy that teachers employ. Mathematics classes at the Grade 8/ Secondary II level are offered on a full-year or semestered basis, depending on the policy of the school, district, or province.

## Scheduling

As shown in Figure 4.13 and Table 4.9, the vast majority of schools across Canada offer Grade 8/ Secondary II mathematics courses on a full-year basis with the exception of British Columbia, where roughly one-quarter of schools in both anglophone and francophone school systems schedule their mathematics courses on a semestered basis (Appendices A.4.8.1, A.4.8.2).



#### FIGURE 4.13 Percentage of semestered and full-year mathematics classes

|     | Anglophone s | chool systems | Francophone s | chool systems |
|-----|--------------|---------------|---------------|---------------|
|     | Semester     | Full year     | Semester      | Full year     |
| BC  | 22‡          | 78            | 27‡           | 73‡           |
| AB  | 3‡           | 97            | 0             | 100‡          |
| SK  | 1‡           | 99            | 0             | 100‡          |
| MB  | 0            | 100           | 0             | 100‡          |
| ON  | 0‡           | 100           | 1‡            | 99            |
| QC  | 2‡           | 98            | 3‡            | 97            |
| NB  | 0            | 100           | 0             | 100           |
| NS  | 1‡           | 99            | 0             | 100‡          |
| PE  | 10‡          | 90‡           |               |               |
| NL  | 0            | 100           |               |               |
| CAN | 2            | 98            | 2‡            | 98            |

# TABLE 4.9Percentage of semestered and full-year mathematics classes, by language of the<br/>school system

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

## Weekly mathematics instruction time

School principals were asked to indicate the number of minutes of mathematics instruction offered each week to Grade 8/Secondary II students at their schools. In Canada overall, about half of schools offered 250 to 300 minutes of weekly instruction in mathematics; however, instructional time varied among the provinces (Figure 4.14, Appendix A.4.9.1).



#### FIGURE 4.14 Minutes of mathematics instruction offered each week

At the pan-Canadian level, 70 percent of schools in anglophone systems and 47 percent of schools in francophone systems provide over 250 minutes of weekly mathematics instruction. In anglophone school systems, the Atlantic provinces and Ontario had the highest proportion of schools offering more than 250 minutes of mathematics instruction per week, while British Columbia and Alberta had the lowest. In francophone school systems, New Brunswick had the highest proportion (91 percent) of schools offering more than 250 minutes of weekly instruction, while British Columbia had the lowest (9 percent). In the latter province, almost three-quarters of francophone schools offered 200 or fewer minutes of mathematics instruction each week (Table 4.10, Appendix A.4.9.2).

|     | 150 or fewer | 151 to 200 | 201 to 250       | 251 to 300 | More than 300 |
|-----|--------------|------------|------------------|------------|---------------|
|     |              | Anglophone | e school systems |            |               |
| BC  | 3‡           | 33         | 33               | 17‡        | 15‡           |
| AB  | 2‡           | 18‡        | 50               | 23‡        | 7‡            |
| SK  | 2‡           | 11‡        | 34               | 35         | 18‡           |
| MB  | 0            | 6‡         | 20‡              | 40         | 34            |
| ON  | 0            | 6‡         | 9‡               | 70         | 15‡           |
| QC  | 0            | 8‡         | 56‡              | 31‡        | 6‡            |
| NB  | 0            | 0          | 10‡              | 74         | 16‡           |
| NS  | 1‡           | 3‡         | 1‡               | 80         | 15‡           |
| PE  | 0            | 5‡         | 0                | 71‡        | 24‡           |
| NL  | 1‡           | 1‡         | 6‡               | 65         | 27‡           |
| CAN | 1‡           | 10         | 19               | 55         | 15            |
|     |              | Francophon | e school systems |            |               |
| BC  | 0            | 73‡        | 18‡              | 0          | 9‡            |
| AB  | 0            | 33‡        | 44‡              | 22‡        | 0             |
| SK  | 0            | 17‡        | 67‡              | 17‡        | 0             |
| MB  | 0            | 20‡        | 13‡              | 60‡        | 7‡            |
| ON  | 2‡           | 6‡         | 7‡               | 53         | 32            |
| QC  | 2‡           | 8‡         | 64               | 14‡        | 12‡           |
| NB  | 2‡           | 0          | 7‡               | 9‡         | 82            |
| NS  | 0            | 0          | 30‡              | 60‡        | 10‡           |
| CAN | 2‡           | 8          | 43               | 25         | 22            |

## TABLE 4.10 Percentage of schools by minutes of mathematics instruction offered each week and language of the school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

Comparisons with PCAP 2016, in which reading was the major domain (O'Grady, Fung, Brochu, et al., 2019), reveal that schools offered more instructional minutes per week in language arts than in mathematics. In Canada overall, 16 percent of schools that participated in PCAP 2019 offered more than 300 minutes of mathematics instruction per week, while over 40 percent of schools participating in PCAP 2016 offered more than 300 minutes of language arts instruction per week. In PCAP 2010, in which mathematics was first the major domain, around two-thirds of schools in Canada overall offered between 201 and 300 minutes of mathematics instruction per week (CMEC, 2012b, p. 114). This proportion is lower than in PCAP 2019, in which approximately three-quarter of schools allotted this much time to mathematics instruction per week.

As shown in Figure 4.15, the amount of time scheduled for weekly mathematics instruction is not significantly associated with achievement at the school level (Appendix A.4.9.3). This is consistent

with what was observed for science in PCAP 2013 (O'Grady & Houme, 2015, p. 106). For mathematics in PCAP 2010, schools that offered 201 to 250 minutes of mathematics instruction per week had higher scores than those schools that spent more time on mathematics instruction (CMEC, 2012b, p. 113). A similar trend was evident for reading in 2016, where students in schools offering less instructional time in language arts per week tended to perform better than those receiving more time (O'Grady, Fung, Brochu, et al., 2019, p. 127). As noted in past PCAP reports, caution is needed when interpreting findings on the association between instructional time and achievement, as it is possible that other factors may be confounded with instructional time, such as schools providing more instructional time to lower-achieving students.





*Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## **Daily mathematics instruction time**

Across the provinces, over three-quarters of schools provide daily mathematics instruction, with the exception of schools in British Columbia (41 percent) and Quebec (10 percent) (Figure 4.16, Appendix A.4.10.1). Almost all schools offer daily mathematics instruction in anglophone systems in Ontario, New Brunswick, and Nova Scotia, and in francophone systems in Manitoba and Nova Scotia. The lowest proportion of schools offering mathematics instruction on a daily basis was in francophone systems in Saskatchewan (0 percent) and Quebec (4 percent) (Table 4.11, Appendix A.4.10.2).



#### **FIGURE 4.16** Percentage of schools offering daily mathematics instruction

 TABLE 4.11 Percentage of schools offering daily mathematics instruction, by language of the school system

Anglophone school system Francophone school system BC 42 27‡ 79 30‡ AB 0 SK 87 MB 93 100‡ ON 99 95 4‡ QC 40‡ NB 99 95 NS 98 100‡ ΡE 90‡ ---NL 89 --CAN 89 38

<sup>‡</sup> There are fewer than 30 observations.

*Note*: Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

As shown in Figure 4.17, mathematics achievement of students in schools that offer daily mathematics instruction was lower than that of students in schools that do not offer daily mathematics instruction (Appendix A.4.10.3). As with instructional time, caution is needed in interpreting findings on the association between frequency of mathematics instruction and achievement, as it is possible that other factors may be confounded with the frequency of mathematics instruction, such as schools providing instruction more frequently to lower-achieving students.



# FIGURE 4.17 Relationship between daily mathematics instruction and mathematics achievement

\* Significance difference compared to the *mathematics instruction offered daily* category *Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## Enrichment and extracurricular activities

In addition to resources directed toward classroom instruction, school resources are also allocated to enrich students' learning through extracurricular experiences. Schools strive to engage students by offering a wide assortment of options that pique their interest and expand their skills. Abalde (2014) found that larger schools were able to provide a wider complement of these offerings, whereas smaller schools were likely to leverage equitable student participation, drawing on and reinforcing a deeper sense of community.

Figure 4.18 lists various activities offered by schools participating in PCAP 2019, categorizing them into enrichment activities, clubs, and other extracurricular activities. For enrichment activities, mathematics interventions were offered most often (78 percent), while enrichment mathematics activities were offered the least frequently (37 percent). With respect to the second category, clubs, participating schools offered a variety of choices, including robotics/coding clubs, chess clubs, mathematics and other academic clubs, and debating clubs or activities, of which debating clubs/ activities were offered the least frequently (13 percent). In addition, a high proportion of principals (84 percent) reported that their school offered a type of club other than those specified in the questionnaire. The third category, other extracurricular activities, included an array of activities, with volunteering or service activities most often offered by schools (70 percent) (Appendix A.4.11.1).

These activities were subjected to principal component analysis. Through this process, three items resolved into a component that had a correlation above .20 with mathematics achievement. The *enrichment and extracurricular activities index* comprises three items related to the availability of clubs: mathematics clubs, academic clubs (other than mathematics), and robotics or coding clubs. As shown in Figure 4.18, 45 percent of schools that participated in PCAP 2019 offered a robotics or coding club for Grade 8/Secondary II students outside of school hours. Twenty percent of schools offered a mathematics club, and 30 percent offered an academic club that focused on a subject other than mathematics.



#### FIGURE 4.18 Enrichment and extracurricular activities offered by schools

Schools were grouped into quartiles according to whether or not they offered the three types of activities that constitute the enrichment and extracurricular activities index. The top quarter of the index represents schools that were most likely to offer these three types of activities, while the bottom quarter represents schools that were least likely to offer them. The relationship between this index and mathematics achievement at the pan-Canadian level is shown in Figure 4.19. Mathematics scores are higher for students in schools that are most likely to provide access to these clubs, with a significant difference in achievement found between schools in the top quarter and third quarter of this index, as well as between the top quarter and the bottom quarter (Appendix A.4.11.2).

# FIGURE 4.19 Relationship between the enrichment and extracurricular activities index and mathematics achievement



\* Significant difference compared to the adjacent lower quarter; the bottom quarter is compared to top quarter *Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

## Challenges to teaching and learning

## Challenges to providing instruction

As with previous PCAP administrations, principals were asked to share their views on how shortages or an inadequate number of resources affected the capacity of their schools to provide instruction. Administrators' perspectives on the adequacy of school facilities, human resources, and instructional resources are shown Figure 4.20. The majority (69 percent) of school principals reported that the availability of qualified mathematics teachers was never or rarely a concern. By contrast, other types of shortages were reported as sometimes or often affecting the capacity of the school to provide instruction; these include shortages of qualified substitute teachers (70 percent), of qualified education assistants for mathematics classrooms (46 percent), and of mathematics specialists to support mathematics teachers (46 percent). Shortages of instructional resources for mathematics were not often cited as an issue; less than 30 percent of principals reported that shortages of resources such as computers, digital resources or software, instructional materials, and library materials sometimes or often affect mathematics instruction in their school. In general, principals who responded to the school questionnaire reported being less concerned about the adequacy of school facilities and instructional resources than about the availability of qualified educational personnel able to provide additional support to mathematics teachers in various capacities (Appendix A.4.12).

#### FIGURE 4.20 Challenges to providing instruction, as reported by school principals



Compared with principals' responses in PCAP 2016, in which reading was the major domain, a higher percentage of principals in 2019 indicated that shortages of qualified teachers and of various additional educational personnel were affecting their school's capacity to provide instruction. However, it is not possible to conclude based on the PCAP data alone if this change is due to a decrease in qualified educational personnel or to mathematics being more affected than reading by resource issues.

## Challenges to learning

School principals were asked to give their opinion on the extent to which a number of challenges hindered student learning in their schools. The 14 items included in the school questionnaire are shown in Table 4.12. Through the use of principal components analysis, five of the items were organized into one component, which was correlated with student achievement in mathematics. These five items form the *challenges to learning index*.

#### TABLE 4.12 Questionnaire items related to school-level challenges to learning

#### In your school, to what extent is student learning hindered by the following?

- Student absenteeism (all excused absences)
- Class composition
- Students intimidating or bullying other students
- Students skipping classes
- Student use of alcohol or illegal drugs
- Teachers' low expectations of students
- Poor student-teacher relations
- Disruption of classes by students
- Teachers not meeting individual students' needs
- Teacher absenteeism
- Students lacking respect for teachers
- Staff resisting change
- Teachers being too strict with students
- Lack of time to provide instructional leadership for staff

Note: The shaded items constitute the challenges to learning index.

Generally, schools reported that the challenges in this index did not greatly hinder student learning. As shown in Figure 4.21, absenteeism (all excused absences) and class composition were most frequently reported as hindering learning. Principals reported that skipped classes and the use of alcohol or drugs by students were less frequently challenges in their schools (Appendix A.4.13.1).

## FIGURE 4.21 Percentage of schools by principals' responses to questionnaire items constituting the challenges to learning index



Schools were grouped into quartiles according to the extent to which they reported that student learning was hindered by the five items that constitute the challenges to learning index. Schools in the top quarter of the index most frequently reported that student learning was hindered a lot, while those in the bottom quarter were least likely to indicate that the challenges hindered learning. As shown in Figure 4.22, significant differences in mathematics scores were found between the bottom half and the top half of this index, with mathematics achievement higher in schools that reported fewer challenges to learning (Appendix A.4.13.2).



## FIGURE 4.22 Relationship between the challenges to learning index and mathematics achievement

\* Significant difference compared to the adjacent lower quarter; the bottom quarter is compared to the top quarter *Note:* "Mean score in mathematics – school level" refers to the mean student score in a school. The scores shown are the Canadian averages of those means.

In the majority of schools participating in PCAP 2019, principals reported that student absences presented at least some challenges to learning (Appendix A.4.13.1). Figure 4.23 shows the responses to the question posed to principals about the percentage of students who were absent on a typical day from their school for reasons other than a school-sponsored activity. Overall in Canada, close to half of schools reported that 5 to 10 percent of students are absent on a typical day. The highest rates of absenteeism, with at least 1 in 10 schools reporting more than 10 percent of the students absent on a typical day, were found in British Columbia, Alberta, Saskatchewan, and Newfoundland and Labrador. Quebec schools reported the lowest rate of absenteeism, with 68 percent of schools reporting that less than 5 percent of students are absent on a typical day (Figure 4.23, Appendix A.4.14.1). Francophone schools reported lower absentee rates than anglophone schools in Canada overall; with only a couple of exceptions, absentee rates are also lower in francophone schools in the provinces (Table 4.13, Appendix A.4.14.2).



### FIGURE 4.23 Percentage of students absent on a typical day for reasons other than schoolsponsored activities

## TABLE 4.13Percentage of schools by proportion of students absent on a typical day for reasons<br/>other than school-sponsored activities and language of school system

|     | Anglophone school systems |          |                  | Francophone school systems |          |                  |  |  |
|-----|---------------------------|----------|------------------|----------------------------|----------|------------------|--|--|
|     | Less than 5%              | 5 to 10% | More than<br>10% | Less than 5%               | 5 to 10% | More than<br>10% |  |  |
| BC  | 44                        | 45       | 10‡              | 73‡                        | 27‡      | 0                |  |  |
| AB  | 57                        | 30       | 13‡              | 67‡                        | 33‡      | 0                |  |  |
| SK  | 48                        | 41       | 11‡              | 83‡                        | 17‡      | 0                |  |  |
| MB  | 47                        | 45       | 7‡               | 87‡                        | 13‡      | 0                |  |  |
| ON  | 39                        | 56       | 5‡               | 68                         | 29‡      | 3‡               |  |  |
| QC  | 51‡                       | 39‡      | 10‡              | 71                         | 27‡      | 2‡               |  |  |
| NB  | 32‡                       | 55       | 13‡              | 73                         | 23‡      | 4‡               |  |  |
| NS  | 34                        | 61       | 5‡               | 67‡                        | 22‡      | 11‡              |  |  |
| PE  | 35‡                       | 65‡      | 0                |                            |          |                  |  |  |
| NL  | 35                        | 52       | 13‡              |                            |          |                  |  |  |
| CAN | 43                        | 49       | 8                | 71                         | 27       | 2‡               |  |  |

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however, these results are included in the calculations for the overall Canadian and provincial totals and means.

## Summary

This chapter has presented information on the schools that participated in PCAP 2019, based on responses by principals to the school questionnaire. Four aspects of the schools were examined: demographic background and grade configuration; diversity of the school population; factors influencing learning; and challenges to learning and teaching.

Schools that participated in PCAP 2019 were located in a range of locales, from large cities to small rural settings. The schools, which were both public and private, ranged in size from more than 1,000 to fewer than 100 students, and were configured with fewer than four to nine or more grades. The number of Grade 8/Secondary II students, the target population for this study, ranged from more than 200 to fewer than 25 students per school. Mathematics achievement was found to be higher in schools in larger communities, in larger schools, and in schools with a higher enrolment of Grade 8/Secondary II students. Private schools outperformed public schools in PCAP 2019.

Canadian schools are diverse and welcome newcomers from around the world. Schools across Canada offer English- and French-second-language programs. This chapter has presented demographic information on these language programs and on Indigenous self-identification at the pan-Canadian and provincial levels, as well as by language of the school system.

Mathematics instruction was scheduled mostly in full-year courses at the Grade 8/Secondary II level. In Canada overall, schools were most likely to offer between 200 and 300 minutes per week of instruction in mathematics, and most schools scheduled daily mathematics classes. Although the amount of time spent on mathematics instruction was not related to achievement, schools that did not offer daily instruction had better results than those that did.

In schools, resources are directed both to the classroom and to extracurricular activities that enrich students' learning. Schools that offered mathematics clubs, academic clubs (other than mathematics), and robotics or coding clubs as extracurricular activities were found to have better outcomes in mathematics.

With respect to providing mathematics instruction, the challenge that school principals reported most frequently was the availability of qualified substitute and occasional teachers, and, to a lesser extent, qualified education assistants and mathematics specialists to support their teachers. The challenges to student learning reported most frequently by principals were student absenteeism and class composition.
This report is the second of two reports providing results from PCAP 2019. The initial report focused on achievement results in the three domains assessed by PCAP (mathematics, reading, and science). The current report, which is complementary to the first, looks at contextual variables associated with mathematics achievement. The variables analyzed in this report are drawn from information provided in the three questionnaires administered to students, teachers, and schools as part of PCAP 2019. The information presented here is provided to inform policy, research, and practice. This report includes variables that are correlated with achievement and others that provide a descriptive lens on practices in Grade 8/Secondary II mathematics classrooms in Canada. Over the upcoming months, analysis will continue, and results will be published by CMEC on specific topics of interest.

## A profile of students

No gender gap was found in mathematics in PCAP 2019. This finding is consistent with the results in PCAP 2010, when mathematics was first the major domain. However, these results differ from those in the most recent international large-scale assessments in which Canada participated: boys outperformed girls in mathematics at the Grade 4 level in TIMSS 2019 and at age 15 in PISA 2018.

Canada-wide, both English- and French-first-language students in francophone schools achieved higher scores in mathematics than their peers in anglophone schools; however, scores were similar among students who reported first languages other than French or English in both language systems.

In anglophone school systems, students who were enrolled in a second-language program, either currently or in the past, achieved significantly lower mean scores in mathematics compared to students who had never been enrolled in such a program. In francophone school systems, the difference in mathematics achievement between students who are currently enrolled and those who had never been enrolled in a second-language program was not statistically significant.

Two proxies for socioeconomic status are used in PCAP contextual reports: parents' educational levels and the number of books in students' homes. Student achievement in mathematics was highest for students whose parents had a university-level education and in homes with the greatest number of books.

In PCAP 2019, students who were not born in Canada had statistically lower scores in mathematics compared to those of their Canadian-born counterparts. Among Indigenous students, those who identified themselves as Inuit or Métis achieved higher scores in mathematics compared to those who self-identified as First Nations.

Better mathematics outcomes were associated with students who had more positive attitudes toward mathematics as well as higher mathematics self-efficacy. Higher achievement scores in mathematics were attained by students who scored in the top quarter of the student effort index and who often completed their homework and had regular attendance at school. Unsurprisingly, students who had a good grasp of mathematical terms had better assessment outcomes than their peers who were less familiar with these terms.

Motivating students to learn and to value mathematics is an important aspect of teaching this subject. Integrating mathematics into other subject areas is one way to help students see the relevance of mathematics. A majority of students reported that they have used what they learned in mathematics in science and technology classes, and almost 40 percent of students reported applying what they had learned in mathematics in their art classes.

Students reported that teachers most frequently used homework or teacher-developed classroom assessments to assess learning. The most common homework activities, according to students, were paper-and-pencil calculations, word problems, and studying for assessments. Achievement scores in mathematics were highest for students whose teachers often provided rubrics as well as feedback that included hints or strategies to help them solve problems. The majority of students participating in PCAP agreed with statements that indicated a sense of belonging or connectedness at school, and achievement was related to the degree to which they liked school.

## A profile of teaching

The majority of Canadian mathematics teachers in Grade 8/Secondary II classrooms hold a bachelor of education (BEd) or equivalent, have at least five years of teaching experience, consider themselves to be a specialist by either experience or both education and experience, and reported having a good understanding of the mathematics concepts taught in earlier and later grades. Most teachers reported that they engage in discussions with their colleagues at least two or three times a month on how to teach a particular topic in mathematics. More men than women teach mathematics at the Grade 8/Secondary II level. Nearly all teachers reported that they are somewhat or very confident in their ability to perform a number of mathematical tasks, although a majority of teachers were not confident about their ability to do coding or programming. At the pan-Canadian level, over 90 percent of teachers agreed or strongly agreed with the following statements: that practice is important for student learning in mathematics; that it is possible for all students to succeed in mathematics; that success in mathematics requires hard work; and that student success in mathematics requires good teaching.

About three-quarters of teachers reported that they had participated in three or more days of professional development (PD) related to the teaching of mathematics in the past five years. The most common form of PD was attending workshops or conferences, but around 80 percent of teachers had also pursued PD opportunities to learn more about teaching strategies, integrating technology into teaching, and strategies to adapt instruction/resources to students' interests and needs. Over half of the teachers who completed the teacher questionnaire had attended PD sessions focused on mathematics content knowledge, and 61 percent reported receiving support in mathematics teaching.

Higher achievement was found in classrooms where teachers had at least five years' teaching experience, where at least 40 percent of their teaching assignment was in mathematics, and where teachers considered themselves specialists by both education and experience.

Similar to earlier PCAP cycles, class size varies considerably across provinces; in PCAP 2019, the most common class sizes reported ranged from 20 to 29 students. Thirty percent of Grade 8/ Secondary II classes participating in PCAP are in multi-grade classrooms. Achievement was highest in classes with 25 or more students and with only one grade level.

Canadian teachers adapt their teaching to serve the needs of their students. These adaptations may include a vast array of instructional strategies and resources; in addition, various accommodations are provided for students who need them. Modifications that teachers were most likely to use often in classrooms to support student learning were allowing more time to accomplish a task and adapting teaching methods. At the pan-Canadian level, more than 40 percent of Grade 8/Secondary II mathematics teachers had additional adults besides themselves in the classroom at least some of the time to support students' learning. Instructional modifications or additional adults were more often required to support student learning in classrooms with lower mathematics scores.

Over half of teachers reported that instructional time was sometimes or often lost due to student misbehaviour or to disruptions outside the control of teachers. A higher frequency for both of these factors was associated with lower mathematics achievement.

In terms of classroom practices, a majority of teachers reported often using calculators and mathematics curriculum documents in their mathematics instruction. Around 40 percent of teachers often used activities that they designed, interactive white boards, worksheets, and textbooks. Most teachers reported using mathematical language to probe student understanding daily or almost daily, and the majority of teachers asked students daily or almost daily to justify their reasoning and to give oral and written explanations.

Over two-thirds of Grade 8/Secondary II teachers participating in PCAP 2019 reported assigning homework involving paper-and-pencil calculations and word problems at least once a week. More than half the teachers indicated never using homework assignments as part of students' grades. Students in classes taught by a teacher who assigned more than 30 minutes of homework per week had higher mathematics achievement than those with teachers who assigned less than 30 minutes of homework per week. Nearly three-quarters of teachers reported often having a class discussion on homework to gauge the level of effort student had put into it. A majority of teachers also reported that they often monitored completion of homework and had students correct their homework in class.

To assess their students' understanding, mathematics teachers at this level used written-response questions more frequently than selected-response questions. The most common type of writtenresponse question required multi-step solutions, and more frequent use of this type of question was associated with higher achievement in mathematics. Teachers reported using different types of questions in mathematics assessments to measure levels of thinking; questions to measure students' ability to apply knowledge and understanding were used most frequently.

In PCAP 2019, teachers were asked to report on challenges that they face in teaching mathematics. Over 40 percent of teachers reported that the range of student ability in their classes presents a lot of challenges to their mathematics teaching. About one-quarter of teachers reported that they are challenged to a large extent by large class sizes, disruptive students, and diverse classroom composition. The challenges that teachers reported least frequently were concerns about safety and their own limited background knowledge in mathematics.

## A profile of schools

School characteristics vary across the country. The PCAP 2019 school questionnaire, which was completed by the principals of the schools participating in the PCAP assessment, provided data that were used to examine characteristics of schools at the provincial and pan-Canadian levels.

The demographic information provided by principals shows the variability across schools in Canada. Schools that participated in PCAP 2019 were located in a range of locales, from large cities to small towns and rural areas. The schools, which were both public and private, ranged in size from more than 1,000 to fewer than 100 students; they had as few as one to four grades and as many as nine or more grades. The number of Grade 8/Secondary II students ranged from 25 or fewer to more than 200 per school. At the pan-Canadian level, the majority of schools offered Grade 8/Secondary II mathematics as a full-year class and provided daily mathematics instruction. Most schools offered between 200 and 300 minutes of mathematics instruction per week, with about half of schools offering 250 to 300 minutes of instruction.

Generally, higher achievement scores in mathematics were found in large schools that were located in larger centres, and in schools with more than 50 Grade 8/Secondary II students. Higher mathematics achievement was found in private, compared to public, schools and in schools that did not offer daily mathematics instruction.

In addition to the resources required for classroom learning, many schools provide resources for extracurricular experiences to enrich students' learning. Schools that offered mathematics clubs, academic clubs (other than mathematics), and robotics or coding clubs had better outcomes in mathematics.

School principals reported that the most frequent challenge to providing instruction was the lack of availability of qualified educational personnel to serve as substitute or occasional teachers and to assist mathematics teachers in various capacities. The challenges hindering student learning that were reported most frequently by principals were student absenteeism and class composition.

## **Final statement**

The results from the PCAP 2019 assessment provide a comprehensive picture of Grade 8/ Secondary II students' mathematics skills at the provincial and pan-Canadian levels. Data from the PCAP contextual questionnaires also highlight different factors in the students' homes, classrooms, and school environments that contribute to their performance in mathematics. This report helps to contextualize the learning and teaching of mathematics in Canadian schools. Over the coming months, CMEC, in collaboration with ministries and departments of education, will continue to analyze the results from PCAP in conjunction with other education indicators to better inform the teaching of mathematics and related educational policies.

- Aaron, S., Horn, A.S., Sung, T.J. (2017). *The impact of graduate education on teacher effectiveness: Does a master's degree matter?* Midwestern Higher Education Compact.
- Abalde, M.A. (2014). *School size policies: A literature review*. OECD Education Working Papers, No. 106. OECD Publishing. http://dx.doi.org/10.1787/5jxt472ddkjl-en
- Alberta Education. (2016). *Mathematics Kindergarten to Grade 9*. https://education.alberta.ca/ media/3115252/2016\_k\_to\_9\_math\_pos.pdf
- Andersen, I., & Reimer, D. (2019). Same-gender teacher assignment, instructional strategies, and student achievement: New evidence on the mechanisms generating same-gender teacher effects. *Research in Social Stratification and Mobility*, 62, 100406. https://doi.org/10.1016/j. rssm.2019.05.001
- Anderson, K., & Winthrop, R. (2016). Building global consensus on measuring learning. In S. McGrath & Q. Gu (Eds.), *Routledge handbook of international education and development* (pp. 93–107). Routledge.
- Anderson, R.H. (1992). *The nongraded elementary school: Lessons from history*. Paper presented at the American Educational Research Association in San Francisco, CA.
- Andon, A., Thompson, C.G., & Becker, B.J. (2014). A quantitative synthesis of the immigrant achievement gap across OECD countries. *Large-Scale Assessments in Education*, 2(7), 1–20. http://doi.org/10.1186/s40536-014-0007-2
- Antecol, H., Eren, O., & Ozbeklik, S. (2012). The effect of teacher gender on student achievement in primary school: Evidence from a randomized experiment. IZA Discussion Paper No. 6453. Institute for the Study of Labour. http://ftp.iza.org/dp6453.pdf
- Arora, A., & Pawlowski, E. (2017). Examining gender differences in the mathematical literacy of 15-yearolds and the numeracy skills of the age cohorts as adults. U.S. PIAAC. https://static1.squarespace. com/static/51bb74b8e4b0139570ddf020/t/588a18c13a04118a0c68116e/1485445313820/ Arora\_Pawlowski\_PIAAC\_2017.pdf
- Badgett, K. (2014). The influence of teacher graduate degrees on student reading achievement. AASA Journal of Scholarship & Practice, 11(1), 4–25.
- Ball, D.L., Thames, M.H., & Phelps, G. (2008). Content knowledge for teaching: What makes it special? *Journal of Teacher Education*, 59(5), 389–407. https://doi.org/10.1177/ 0022487108324554
- Bandura, A. (1977). Social learning theory. Prentice-Hall.
- Bandura, A. (1997). Self-efficacy: The exercise of control. Macmillan.
- Batchelor, S., Torbeyns, J., & Verschaffel, L. (2019). Affect and mathematics in young children: An introduction. *Educational Studies in Mathematics*, 100(3), 201–209. https://doi-org.qe2a-proxy. mun.ca/10.1007/s10649-018-9864-x

- Birchinall, L. (2013). Case study of trainee teachers' responses to the impact on engagement and motivation in learning through a model of cross-curricular context-based learning: "Keeping fit and healthy." *Curriculum Journal*, 24(1), 27–49. http://doi.org/10.1080/09585176.2012.731014
- Black, L., Mendick, H., & Solomon, Y. (Eds.). (2009). *Mathematical relationships in education: Identities and participation*. Routledge.
- Blum, R.W. (2005). A case for school connectedness. Educational Leadership, 62(7), 16-20.
- Boaler, J. (2002). *Experiencing school mathematics: Traditional and reform approaches to teaching and their impact on student learning* (rev. and expanded ed.). Lawrence Erlbaum.
- Bolyard, J.J., & Moyer-Packenham, P.S. (2008). A review of the literature on mathematics and science teacher quality. *Peabody Journal of Education*, 83(4), 509–535.
- Borgonovi, F., Choi, A., & Paccagnella, M. (2018). The evolution of gender gaps in numeracy and literacy between childhood and adulthood. OECD Education Working Paper No. 184. OECD Publishing. http://www.oecd.org/officialdocuments/publicdisplaydocumentpdf/?cote=EDU/ WKP(2018)20&docLanguage=En
- Borland, M.V., & Howsen, R.M. (2003). An examination of the effect of elementary school size on student academic achievement. *International Review of Education 49*, 463–474.
- Bruckauf, Z. (2016). Falling behind: Socio-demographic profiles of educationally disadvantaged youth. Evidence from PISA 2000–2012. UNICEF Office of Research, Innocenti Working Paper, WP-2016-11. https://www.unicef-irc.org/publications/pdf/IWP\_2016\_11.pdf
- Burns, R., & Mason, D. (2002). Class composition and student achievement in elementary schools. American Educational Research Journal, 39(1), 207–233. https://doi. org/10.3102/00028312039001207
- Causa, O., Dantan, S., & Johansson, Å. (2009). Intergenerational social mobility in European OECD countries. OECD Economics Department Working Papers, No. 709. OECD Publishing. http:// doi.org/10.1787/223043801483
- Cheryan, S., Plaut, V.C., Davies, P., & Steele, C.M. (2009). Ambient belonging: How stereotypical cues impact gender participation in computer science. *Journal of Personality and Social Psychology*, 97, 1045–1060. https://doi.org/10.1037/a0016239
- Cheung, A.C.K., & Slavin, R.E. (2013). The effectiveness of educational technology applications for enhancing mathematics achievement in K–12 classrooms: A meta-analysis. *Educational Research Review, 9*, 88–113.
- Chevalier, A., Harmon, C., O'Sullivan, V., & Walker, I. (2013). The impact of parental income and education on the schooling of their children. *IZA Journal of Labor Economics*, *2*(8), 1–22. http://doi.org/10.1186/2193-8997-2-8
- Chingos, M.M., & Whitehurst, G.J. 2011. *Class size: What research says and what it means for state policy*. Brown Center on Educational Policy, Brookings Institution. https://www.brookings.edu/research/class-size-what-research-says-and-what-it-means-for-state-policy/
- Ciampa, K., & Gallagher, T. (2018). A comparative examination of Canadian and American preservice teachers' self-efficacy beliefs for literacy instruction. *Reading & Writing*, 31(2), 457–481. https://doi.org/10.1007/s11145-017-9793-6

- Cobb, P. (1986). Contexts, goals, beliefs, and learning mathematics. *For the Learning of Mathematics*, 6(2), 2–9.
- Cobb, P., & Yackel, E. (2011). Foreword. In A. Sfard, K. Gravemeijer, & E. Yackel (Ed.), *A journey in mathematics education research insights from the work of Paul Cobb* (p. 48). Mathematics Education Library.
- Collin, C., & Jensen, H. (2009). *A statistical profile of poverty in Canada*. (PRB 09-17E). Parliamentary Information and Research Service of the Library of Parliament.
- Collins, J.L. (1985). *Self-efficacy and ability in achievement behaviour*. Stanford University Dissertation Publishing.
- Conference Board of Canada. (2013). Percentage of graduates in science, math, computer science, and engineering. https://www.conferenceboard.ca/hcp/Details/education/graduates-science-math-computer-science-engineerin.aspx
- Cooper, H., Lindsay, J.J., Nye, B., & Greathouse, S. (1998). Relationships among attitudes about homework, amount of homework assigned and completed, and student achievement. *Journal of Educational Psychology*, *90*, 70–83.
- Cooper, H., Robinson, J.C., & Patall, E.A. (2006). Does homework improve academic achievement? A synthesis of research. *Review of Educational Research*, *76*, 1–62. https://doi.org/10.3102/00346543076001001
- Council of Ministers of Education, Canada (CMEC). (2010). Teacher education and development study in mathematics 2008: Canadian report. Author. http://www.cmec.ca/Publications/Lists/ Publications/Attachments/277/WEB%20TEDS-M\_Report\_Eng.pdf
- Council of Ministers of Education, Canada (CMEC). (2011). PCAP 2010: Report on the pan-Canadian assessment of mathematics, science, and reading. Author. https://cmec.ca/Publications/ Lists/Publications/Attachments/274/pcap2010.pdf
- Council of Ministers of Education, Canada (CMEC). (2012a). *PCAP-13 2007: Jurisdictional* profiles and achievement equity. Author. http://www.cmec.ca/Publications/Lists/Publications/ Attachments/299/PCAP2007\_Jurisdictional\_Web\_EN.pdf
- Council of Ministers of Education, Canada (CMEC). (2012b). PCAP 2010: Contextual report on student achievement in mathematics. Author. http://www.cmec.ca/docs/pcap/pcap2010/ English/6\_ContextualAnalyse/Context%20ReportEn.pdf
- Council of Ministers of Education, Canada (CMEC). (2014). Homework alert: How much is enough? *Assessment Matters!*, 7, 1–6. Author. http://www.cmec.ca/Publications/Lists/ Publications/Attachments/338/AMatters\_No7\_Homework\_EN.pdf
- Council of Ministers of Education, Canada (CMEC). (2015). Immigrants in Canada: Does socioeconomic background matter? *Assessment Matters*, 9, 1–8. Author. https://cmec.ca/ Publications/Lists/Publications/Attachments/343/AMatters\_No9\_EN.pdf
- Council of Ministers of Education, Canada (CMEC). (2020). *PCAP 2019: Assessment framework*. Author. https://cmec.ca/docs/pcap/pcap2019/PCAP-2019-Assessment-Framework-EN.pdf
- Council of Ministers of Education, Canada (CMEC). (2021). Trends in STEM and BHASE graduates from public postsecondary institutions across Canadian provinces and territories, 2010 to 2018. Author. https://cmec.ca/Publications/Lists/Publications/Attachments/420/STEM\_BHASE\_ graduates\_report\_Final\_EN.pdf

- Crowe, C.C. (2013). A longitudinal investigation of parent educational involvement and student achievement: Disentangling parent socialization and child evocative effects across development. *Journal of Educational Research and Policy Studies, 13*(2), 1–33.
- Cvencek, D., Meltzoff, A.N., & Greenwald, A.G. (2011). Math-gender stereotypes in elementary school children. *Child Development*, 82(3), 766–779. https://doi.org/10.1111/j.1467-8624.2010.01529.x
- Daker, R.J., Gattas, S.U., Sokolowski, H.M., Green, A.E., & Lyons, I.M. (2021). First-year students' math anxiety predicts STEM avoidance and underperformance throughout university, independently of math ability. *Science of Learning 6*, 17. https://doi 10.1038/s41539-021-00095-7
- Danielson, C. (2007). *Enhancing professional practice*. Association for Supervision and Curriculum Development.
- Darling-Hammond, L., Herman, J., Pellegrino, J., Abedi, J., Aber, J.L., Baker, E., Bennett, R., Gordon, E., Haertel, E., Hakuta, K., Ho, A., Linn, R.L., Pearson, P.D., Popham, J., Resnick, L., Schoenfeld, A.H., Shavelson, R., Shepard, L.A., Shulman, L. & Steele, C.M. (2013). *Criteria for high-quality assessment*. Stanford Center for Opportunity Policy in Education.
- Davis, B., Sumara, D., & Kieren, T. (1996). Cognition, co-emergence, curriculum. Journal of Curriculum Studies, 28(2), 151–169.
- De Jong, R., Westerhof, K.J., & Creemers, B.P.M. (2000). Homework and student math achievement in junior high schools. *Educational Research and Evaluation*, *6*, 130–157.
- Devlin, K. (2019). How technology has changed what it means to think mathematically. In M. Danesi (Ed.), *Interdisciplinary perspectives on math cognition* (pp. 53–78). Springer International.
- Duff, P.A., & Becker-Zayas, A. (2017). Demographics and heritage languages in Canada. In O. Kagan, M. Carreira, & C. Hitchens (Eds.), *The Routledge handbook of heritage language education: From innovation to program building* (pp. 57–67). Routledge.
- Eccles, J.S., Wigfield, A., Midgley, C., Reuman, D., MacIver, D., & Feldlaufer, H. (1993). Negative effects of traditional middle schools on students' motivation. *Elementary School Journal*, 95(5), 553–574. https://doi.org/10.1086/461740
- Echazarra, A., & Radinger, T. (2019). Does attending a rural school make a difference in how and what you learn? *PISA in Focus*, No. 94. OECD Publishing. https://www.oecd-ilibrary.org/ education/does-attending-a-rural-school-make-a-difference-in-how-and-what-you-learn\_ d076ecc3-en
- Escueta, M., Quan, V., Nickow, A.J., & Oreopoulos, P. (2017). *Education technology: An evidencebased review*. National Bureau of Economic Research Working Papers, No. 23744. https://www. nber.org/system/files/working\_papers/w23744/w23744.pdf
- Evans, M.D.R., Kelley, J., & Sikora, J. (2014). Scholarly culture and academic performance in 42 nations. *Social Forces*, *92*(4), 1573–1605.
- Fan, H., Xu, J., Cai, Z., He, J., & Fan, X. (2017). Homework and students' achievement in math and science: A 30-year meta-analysis, 1986–2015. *Educational Research Review, 20*, 35–54. https://doi.org/10.1016/j.edurev.2016.11.003

- Farnia, F., & Geva, E. (2011). Cognitive correlates of vocabulary growth in English language learners. *Applied Psycholinguistics*, 32, 711–738. https://doi.org/10.1017/S0142716411000038
- Frenette, M., & Chan, P.C.W. (2015). Academic outcomes of public and private high schools students: What lies behind the difference. Analytical Studies Branch Research Papers Series. No. 367. Statistics Canada. https://publications.gc.ca/collections/collection\_2015/ statcan/11f0019m/11f0019m2015367-eng.pdf
- Frenzel, A.C., Pekrun, R., & Goetz, T. (2007). Girls and mathematics: A "hopeless" issue? A controlvalue approach to gender differences in emotions towards mathematics. *European Journal of Psychology of Education*, 22, 497–514. http://dx.doi.org/10.1007/BF03173468
- Galliott, N., Graham, L., & Sweller, N. (2015). Who struggles most in making a career choice and why? Findings from a cross-sectional survey of Australian high-school students. *Journal of Psychologists and Counsellors in Schools*, 25(2), 133–151. https://doi.org/10.1017/ jgc.2015.7
- Galloway, M., Conner, J., & Pope, D. (2013). Nonacademic effects of homework in privileged, high performing high schools. *Journal of Experimental Education*, *81*(4), 379–389.
- Ganley, C.M., & Lubienski, S.T. (2016). Mathematics confidence, interest, and performance: Examining gender patterns and reciprocal relations. *Learning and Individual Differences*, 47, 182–193.
- Gershenson, S., & Langbein, L. (2015). The effect of primary school size on academic achievement. *Educational Evaluation and Policy Analysis*, *37*, 135–155. https://doi.org/10.3102/0162373715576075
- Godfrey Anderson, J. (2016). *Be innovative but don't be wrong: Are 21<sup>st</sup> century students experiencing* 21<sup>st</sup> century mathematics education? Doctoral dissertation, University of New Brunswick. https:// unbscholar.lib.unb.ca/islandora/object/unbscholar%3A8121
- Goldhaber, D.D., & Brewer, D.J. (1996). Evaluating the effect of teacher degree level on educational performance. National Center for Educational Statistics, U.S. Department of Education. http:// nces.ed.gov/pubs97/975351.pdf
- Gottfried, M., Estrada, F., & Sublett, C. (2015). STEM education and sexual minority youth: Examining math and science coursetaking patterns among high school students. *High School Journal*, *99*(1), 66–87. https://www.jstor.org/stable/44075333
- Government of Canada. (2017). Official languages annual report, 2015–2016. http://open.canada.ca/ data/dataset/3eb6e5a0-1618-4ced-8bc5-49615ed5b43d/resource/82cdc735-7896-4d6a-828c-851219be02a7/download/rapport15-16finalen-finalqc-1.pdf
- Government of Canada. (2020). Annual report to Parliament 2020. https://www.sac-isc.gc.ca/ eng/1602010609492/1602010631711
- Graham, L., White, S., Cologon, K., & Pianta, R. (2020). Do teachers' years of experience make a difference in the quality of teaching? *Teaching and Teacher Education, 96*, 103190. https://doi.org/10.1016/j.tate.2020.103190
- Gronlund, N., & Waugh, C.K. (2009). Assessment of student achievement. Pearson Education.
- Han, W. (2012). Bilingualism and academic achievement. *Child Development*, 83(1), 300–321. https://doi.org/10.1111/j.1467-8624.2011.01686.x

- Hanushek, E.A. (2003). The failure of input-based schooling policies. *Economic Journal, 113* (485), 64–98.
- Hanushek, E.A., Kain, J.F., O'Brien, D.M., & Rivkin, S.G. (2005, February). The market for teacher quality. Working Paper 11154. National Bureau of Economic Research. http://www.nber.org/ papers/w11154
- Hattie, J. (2009). Visible learning: A synthesis of over 800 meta-analyses relating to achievement. Routledge.
- Heck, R. (2009). Teacher effectiveness and student achievement: Investigating a multilevel cross-classified model. *Journal of Educational Administration*, 47(2), 227–249. https://doi. org/10.1108/09578230910941066
- Herbel-Eisenmann, B., Choppin, J., Wagner, D., & Pimm, D. (2012). Inherent connections between discourse and equity in mathematics classrooms. In *Equity in discourse for mathematics education: Theories, practices, and policies* (pp. 1–13). Springer.
- Herrmann, M.A., & Rockoff, J.E. (2010). *Worker absence and productivity: Evidence from teaching*. National Bureau of Economic Research.
- Hill, H., & Ball, D.L. (2009). The curious and crucial case of mathematical knowledge for teaching. *Phi Delta Kappan*, *91*(2), 68–71. https://doi.org/10.1177/003172170909100215
- Hill, N., Castellino, D., Lansford, J., Nowlin, P., Dodge, K., Bates, J., & Pettit, G. (2004).
  Parent academic involvement as related to school behavior, achievement, and aspirations:
  Demographic variations across adolescence. *Child Development*, 75(5), 1491–1509. https://doi.org/10.1111/j.1467-8624.2004.00753.x
- Holmes, L.K. (2020). The impact of school attendance on student achievement in mathematics and reading. In *Dissertation Abstracts International Section A: Humanities and Social Sciences* (Vol. 81, Issue 6-A). https://www.proquest.com/openview/247d38942b009223fa49a0bff0e9b677/1?cbl= 18750&diss=y&pq-origsite=gscholar
- Holmlund, H., & Böhlmark, A. (2019). Does grade configuration matter? Effects of school reorganisation on pupils' educational experience. *Journal of Urban Economics*, 109, 14–26. https://doi.org/10.1016/j.jue.2018.11.004
- Hoover-Dempsey, K.V., Battiato, A.C., Walker, J.M.T., Reed, R.P., DeJong, J.M., & Jones, K.P. (2001). Parental involvement in homework. *Educational Psychologist*, *36*, 195–209.
- Horn, A.S., & Jang, S.T. (2017, March). The impact of graduate education on teacher effectiveness: Does a masters's degree matter? *Midwestern Higher Education Compact*, 1–6. https://files.eric. ed.gov/fulltext/ED587432.pdf
- Huang, C. (2013). Gender differences in academic self-efficacy: A meta-analysis. *European Journal Psychology of Education 28*, 1–35. https://doi.org/10.1007/s10212-011-0097-y
- Huang, H., & Liang, G. (2016). Parental cultural capital and student school performance in mathematics and science across nations. *Journal of Educational Research*, 109(3), 286–295. http://10.0.4.56/00220671.2014.946122
- Huba, A., & Freed, J.E. (2000). Learner-centered assessment on college campuses: Shifting the focus from teaching to learning. Allyn & Bacon.

- Jones, K.R., & Ezeife, A.N. (2011). School size as a factor in the academic achievement of elementary school students. *Psychology*, 2(8), 859–868. https://www.scirp.org/pdf/ PSYCH20110800007\_34224480.pdf
- Kahn, S., & Ginther, D.K. (2017). Women and STEM. In S.L. Averett, L.M. Argys, & S.D. Hoffman (Eds.), *The Oxford handbook on the economics of women*. Oxford University Press.
- Kilpatrick, J., Swafford, J., & Findell, B. (2001). *Adding it up: Helping children learn mathematics*. National Academy Press.
- Knifsend, C., & Graham, S. (2012). Too much of a good thing? How breadth of extracurricular participation relates to school-related affect and academic outcomes during adolescence. *Journal* of Youth and Adolescence, 41, 379–389.
- Koellner, K., & Jacobs, J. (2015). Distinguishing models of professional development: The case of an adaptive model's impact on teachers' knowledge, instruction, and student achievement. *Journal of Teacher Education*, 66(1), 51–67. https://doi.org/10.1177/0022487114549599
- Kurz, A., Elliott, S., Wehby, J., & Smithson, J. (2010). Alignment of the intended, planned, and enacted curriculum in general and special education and its relation to student achievement. *Journal of Special Education*, 44(3), 131–145. https://doi.org/10.1177/0022466909341196
- Ladd, H.F., & Sorensen, L.C. (2017). Returns to teacher experience: Student achievement and motivation in middle school. *Education Finance and Policy*, 12(2), 241–279. https://doi. org/10.1162/EDFP\_a\_00194
- Lam, T., & Ho, E. (2013). Effect of family social capital on students' literacy performance: Insights from PISA. *Chinese University Education Journal*, 41(1/2), 65–83.
- Lave, J. (1988). Cognition in practice: Mind, mathematics, and culture in everyday life. Cambridge University Press.
- Lave, J., Murtaugh, M., & de la Rosa, O. (1984). The dialectic of arithmetic in grocery shopping. In B. Rogoff & J. Lave (Eds.), *Everyday cognition: Its development in social context* (pp. 67–94). Harvard University Press.
- Lave, J., & Wenger, E. (1991). *Situated learning legitimate peripheral participation*. Cambridge University Press.
- Learn Alberta. (2008). Mathematics: Creating an effective mathematics environment. https://www.learnalberta.ca/content/mewa/html/teaching/index.html
- Lee, J., Rhee, D., & Rudolf, R. (2019). Teacher gender, student gender, and primary school achievement: Evidence from ten francophone African countries. *Journal of Development Studies*, 55(4), 661–679. https://doi.org/10.1080/00220388.2018.1453604
- Lerman, S. (2000). The social turn in mathematics education research. In J. Boaler (Ed.), *Multiple perspectives on mathematics teaching and learning*. Ablex.
- Li, Q., & Ma, X. (2010). A meta-analysis of the effects of computer technology on school students' mathematics learning. *Educational Psychology Review*, 22, 215–243.
- Li, Y. (2008). Mathematical preparation of elementary school teachers: Generalists versus content specialists. *School Science and Mathematics*, *108*(5), 169–172.

- Lindemann, D.J. (2015). Gender and numeracy skill use: Cross-national revelations from PIAAC. https://static1.squarespace.com/static/51bb74b8e4b0139570ddf020/t/54da75e1e4b02b70886 14c36/1423603169721/Lindemann\_PIAAC.pdf
- Liu, J., Lee, M., & Gershenson, S. (2021). The short- and long-run impacts of secondary school absences. *Journal of Public Economics, 199.* https://doi. 10.1016/j.jpubeco.2021.104441
- Mangu, D.M., Lee, A.R., Middleton, J.A., & Nelson, J.K. (2015). Motivational factors predicting STEM and engineering career intentions for high school students. In 2015 IEEE Frontiers in Education conference (FIE) (pp. 1–8). http://ieeexplore.ieee.org/xpls/abs\_all. jsp?arnumber=7344065
- Martin, D.P., & Rimm-Kaufman, S.E. (2015). Do student self-efficacy and teacher-student interaction quality contribute to emotional and social engagement in fifth grade math? *Journal* of School Psychology, 53(5), 359–373.
- Mason, J., Burton, L., & Stacey, K. (1982). Thinking mathematically. Addison-Wesley.
- Matthews, J.S. (2018). When am I ever going to use this in the real world? Cognitive flexibility and urban adolescents' negotiation of the value of mathematics. *Journal of Educational Psychology*, 110(5), 726–746. https://doi.org/10.1037/edu0000242
- McLeod, D.B. (1992). Research on affect in mathematics education: A reconceptualization. In D.A. Grouws (Ed.), Handbook of research on mathematics teaching and learning: A project of the National Council of Teachers of Mathematics (pp. 575–596). Macmillan.
- Merry, J.J. (2013). Tracing the U.S. deficit in PISA reading skills to early childhood. *Sociology of Education*, *86*(3), 234–252. http://doi.org/10.1177/0038040712472913
- Middleton, J.A., Jansen, A., & Goldin, G.A. (2016). Motivation. In M.S. Hannula, P. Di Martino,
  & M. Pantziara (Eds.), *Attitudes, beliefs, motivation and identify in mathematics education:* An overview of the field and future directions (pp. 17–22). Springer Open.
- Moslemi, N., & Mousavi, A. (2019). A psychometric re-examination of the science teaching efficacy and beliefs instrument (STEBI) in a Canadian context. *Education Sciences*, 9(1), 17. https://doi. org/10.3390/educsci9010017
- Muis, K.R., Ranellucci, G.M., Franco, G.M., & Crippen, K.J. (2013). The interactive effects of personal achievement goals and performance feedback in an undergraduate science class. *Journal of Experimental Education*, 81(4), 556–578, https://doi.org/10.1080/00220973.2012.7 38257
- Mullis, I.V.S., Martin, M.O., Foy, P., & Arora, A. (2012). TIMSS 2011 international results in mathematics. International Association for the Evaluation of Educational Achievement, Boston College. https://timssandpirls.bc.edu/timss2011/index.html
- Mullis, I.V.S., Martin, M.O., Goh, S., & Prendergast, C. (2017). *PIRLS 2016 encyclopedia: Education policy and curriculum in reading*. TIMMS & PIRLS International Study Center, Boston College. http://timssandpirls.bc.edu/pirls2016/ encyclopedia/
- Multon, K.D., Brown, S.D., & Lent, R.W. (1991). Relation of self-efficacy beliefs to academic outcomes: A meta-analytic investigation. *Journal of Counseling Psychology*, 38, 30–38.
- National Council of Teachers of Mathematics. (2000). *Principles and standards for school mathematics*. Author.

- Natural Science and Engineering Research Council of Canada. (2017). *Women in science and engineering in Canada*. https://www.nserc-crsng.gc.ca/\_doc/Reports-Rapports/WISE2017\_e. pdf
- Ng, C., Bartlett, B., & Elliott, S. (2018). Empowering engagement: Creating learning opportunities for students from challenging backgrounds. Springer.
- Nova Scotia. Department of Education and Early Childhood Education. (2019). *Mathematics 1: Guide*. https://curriculum.novascotia.ca/sites/default/files/documents/curriculum-files/ Mathematics%201%20Guide%20%282019%29.pdf
- Núñez, J.C., Rosário, P., Vallejo, G., & González-Pienda, J.A. (2013). A longitudinal assessment of the effectiveness of a school-based mentoring program in middle school. *Contemporary Educational Psychology*, *38*, 11–21. https://doi.org/10.1016/j.cedpsych.2012.10.002
- OECD. (2010). PISA 2009 results: Vol. 2. Overcoming social background Equity in learning opportunities and outcomes. OECD Publishing. http://dx.doi.org/10.1787/9789264091504-en
- OECD. (2012). Public and private schools: How management and funding relate to their socio-economic profile. OECD Publishing. http://www.oecd.org/pisa/50110750.pdf
- OECD (2013a). PISA 2012 results. Ready to learn: Vol. 3. Students' engagement, drive and self-beliefs. OECD Publishing. https://doi.org/10.1787/9789264201170-en
- OECD. (2013b). What makes urban schools different? *PISA in Focus*, No. 28, OECD Publishing. https://doi.org/10.1787/5k46l8w342jc-en
- OECD. (2015). Students, computers and learning: Making the connection. OECD Publishing. https:// doi.org/10.1787/9789264239555-en
- OECD. (2016a). How is learning time organised in primary and secondary education? Education indicators in focus. OECD Publishing. https://www.oecd-ilibrary.org/education/how-is-learningtime-organised-in-primary-and-secondary-education\_5jm3tqsm1kq5-en
- OECD. (2016b). Low-performing students: Why they fall behind and how to help them succeed. OECD Publishing. http://dx.doi.org/10.1787/9789264250246-en
- OECD. (2017). PISA 2015 results: Vol. 3. Students' well-being. OECD Publishing. http://dx.doi. org/10.1787/9789264273856-en
- OECD. (2018). A brave new world: Technology and education. OECD Publishing. https://doi. org/10.1787/9b181d3c-en
- OECD. (2019). PISA 2018 results: Vol. II. Where all students can succeed. OECD Publishing. https://doi.org/10.1787/b5fd1b8f-en
- Office of the Commissioner of Official Languages. (2019, February). Official Languages Commissioner calls on federal government to address national shortage of French-as-a-secondlanguage teachers. https://www.clo-ocol.gc.ca/en/news/releases/2019/2019-02-13
- O'Grady, K., Deussing, M-A., Scerbina, T., Tao, Y., Fung, K., Elez, V., & Monk, J. (2019). *Measuring up: Canadian results of the OECD PISA 2018 study. The performance of Canadian 15-year-olds in reading, mathematics, and science*. Council of Ministers of Education, Canada. https://cmec.ca/ Publications/Lists/Publications/Attachments/396/PISA2018\_PublicReport\_EN.pdf

- O'Grady, K., Fung, K., Brochu, P., Servage, L., & Tao, Y. (2019). *PCAP 2016: Contextual report on achievement in reading*. Council of Ministers of Education, Canada. https://cmec.ca/ Publications/Lists/Publications/Attachments/393/PCAP2016\_Contextual\_Report\_EN\_ FINAL.pdf
- O'Grady, K., Fung, K., Servage, L., & Khan, G. (2018). *PCAP 2016: Report on the pan-Canadian assessment of reading, mathematics, and science.* Council of Ministers of Education, Canada. https://cmec.ca/Publications/Lists/Publications/Attachments/381/PCAP-2016-Public-Report-EN.pdf
- O'Grady, K., & Houme, K. (2015). *PCAP 2013: Contextual report on student achievement in science*. Council of Ministers of Education, Canada. https://www.cmec.ca/Publications/Lists/ Publications/Attachments/350/PCAP2013\_ContextualReport\_Final\_Web\_EN.pdf
- O'Grady, K., Houme, K., Costa, E., Rostamian, A., & Tao, Y. (2021). *PCAP 2019: Report on the pan-Canadian assessment of mathematics, reading, and science.* Council of Ministers of Education, Canada. https://cmec.ca/Publications/Lists/Publications/Attachments/426/PCAP2019-Public-Report-EN.pdf
- O'Grady, K., Rostamian, A., Monk, J., Tao, Y., Scerbina, T., & Elez, V. (2021). *TIMSS 2019: Canadian results from the Trends in International Mathematics and Science Study.* Council of Ministers of Education, Canada. https://cmec.ca/Publications/Lists/Publications/ Attachments/417/TIMSS19\_Report\_EN.pdf
- Onion, A. (2004). What use is maths to me? A report on the outcomes from student focus groups. *Teaching Mathematics and Its Applications*, 23(4), 189–194. https://doi.org/10.1093/teamat/23.4.189
- Ontario Ministry of Education. (2020–21). Mathematics (2020): The mathematical processes. https://www.dcp.edu.gov.on.ca/en/curriculum/elementary-mathematics/context/the-mathematical-processes
- Onuzo, U., Garcia, A.F., Hernandez, A., Peng, Y., & Lecoq, T. (2013). Intergenerational equity: Understanding the linkages between parents and children — A systematic review. London School of Economics and Political Science.
- Pajares, F., & Kranzler, J. (1995). Self-efficacy beliefs and general mental ability in mathematical problem-solving. *Contemporary Educational Psychology*, *20*(4), 426–443.
- Pajares, F., & Schunk, D.H. (2001). Self-beliefs and school success: Self-efficacy, self-concept, and school achievement. In R.J. Riding & S.G. Rayner (Eds.), *Self-perception*, 239–265. Ablex.
- Pardini, P. (2005). The slowdown of the multiage classroom. School Administrator, 62(3), 23-30.
- Parkin, A. (2015). International report card on public education: Key facts on Canadian achievement and equity. Environics Institute. http://www.environicsinstitute.org/uploads/institute-projects/ environics institute - parkin - international report on education - final report.pdf
- Perry, L.B., & McConney, A. (2010). Does the SES of the school matter? An examination of socioeconomic status and student achievement using PISA 2003. *Teachers College Record*, 112(4), 1137–1162.
- Pimm, D. (1991). Communicating mathematically. In K. Durkin & B. Shire (Eds.), Language in mathematical education: Research and practice (pp. 17–23). Open University Press.
- Pólya, G. (1957). How to solve it: A new aspect of mathematical method (2<sup>nd</sup> ed.). Doubleday.

- Porter, A.C. (2002). Measuring the content of instruction: Uses in research and practice. *Educational Researcher*, *31*(7), 3–14. https://doi.org/10.3102/0013189X031007003
- Proehl, R.A., Douglas, S., Elias, D., Johnson, A.H., & Westsmith, W. (2003). A collaborative approach: Assessing the impact of multi-grade classrooms. *Catholic Education: A Journal of Inquiry & Practice*, 16(2): 417–440.
- Proulx, J. (2019). Mental mathematics under the lens: Strategies, oral mathematics, enactments of meanings. *Journal of Mathematical Behavior*, 56, 100725. https://doi.org/10.1016/j. jmathb.2019.100725
- Rachford, M.K. (2011). An investigation into the process of transference, through the integration of art with science and math curricula, in a California community college: A case study. In *ProQuest LLC*. https://www.proquest.com/docview/861936516
- Ramdass, D., & Zimmerman, B.J. (2011). Developing self-regulation skills: The important role of homework. *Journal of Advanced Academics*, 22(2), 194–218. https://doi. org/10.1177/1932202X1102200202
- Rice, J.K. (2010). The impact of teacher experience: Examining the evidence and policy implications. National Center for the Analysis of Longitudinal Data in Education Research, Urban Institute. https://www.urban.org/research/publication/impact-teacher-experience
- Richards, J. (2014). Are we making progress? New evidence on Aboriginal education outcomes in provincial and reserve schools. C.D. Howe Institute.
- Riggs, I.M., & Enochs, L.G. (1990). Toward the development of an elementary teacher's science teaching efficacy belief instrument. *Science Education*, 74(6), 625–637.
- Rivkin, S.G., Hanushek, E.A., & Kain, J.F. (2005). Teachers, schools, and academic achievement. *Econometrica*, 73(2), 417–458.
- Sadler, P.M., Sonnert, G., Hazari, Z., & Tai, R. (2012). Stability and volatility of STEM career interest in high school: A gender study. *Science Education*, *96*(3), 411–427.
- Sakellariou, C. (2020). The contribution of self-beliefs to the mathematics gender achievement gap and its link to gender equality. *Oxford Review of Education*, 46(6), 804–821. https://doi. 10.1080/03054985.2020.1807313
- Santagata, R., & Lee, J. (2021). Mathematical knowledge for teaching and the mathematical quality of instruction: A study of novice elementary school teachers. *Journal of Mathematics Teacher Education*, 24(1) 33–60. https://doi: 10.1007/s10857-019-09447-y
- Schnepf, S. (2008). *Inequality of learning amongst immigrant children in industrialised countries*. Institute for the Study of Labour, University of Bonn. http://ftp.iza.org/dp3337.pdf
- Schoenfeld, A.H. (1989). Explorations of students' mathematical beliefs and behavior. Journal for Research in Mathematics Education, 20(4), 338–355.
- Schwartz, A.E., Stiefel, L., Rubenstein, R., & Zabel, J. (2011). The path not taken: How does school organization affect eighth-grade achievement? *Educational Evaluation and Policy Analysis*, 33(3), 293–317.
- Shepard, L.A. (2000). The role of assessment in a learning culture. *Educational Researcher, 29*(7), 4–14.

- Shepard, L.A. (2005). Linking formative assessment to scaffolding. *Educational Leadership*, 63(3), 67–70.
- Shipley, L. (2011). A profile of minority-language students and schools in Canada: Results from the Programme for International Assessment (PISA), 2009. Statistics Canada. http://www.statcan. gc.ca/pub/81-595-m/81-595-m2011092-eng.pdf
- Shulman, L.S. (1986). Those who understand: Knowledge growth in teaching. *Educational Researcher*, *15*(2), 4–14. https://www.wcu.edu/webfiles/pdfs/shulman.pdf
- Simpkins, S.D., Davis-Kean, P.E., & Eccles, J.S. (2006). Math and science motivation. *Developmental Psychology*, *42*(1), 70–83.
- Slavin, R.E. (1987). Grouping for instruction in the elementary school. *Educational Psychologist*, 22(2), 109–127.
- Statistics Canada. (2009). Data accuracy and quality. https://www150.statcan.gc.ca/n1/ pub/13f0026m/2007001/ch5-eng.htm
- Statistics Canada. (2015). Linguistic characteristics of Canadians. http://www12.statcan.gc.ca/ census-recensement/2011/as-sa/98-314-x/98-314-x2011001-eng.cfm
- Statistics Canada. (2016a). Demand growing for second language immersion programs. https://www150.statcan.gc.ca/n1/pub/11-402-x/2012000/chap/lang/lang01-eng.htm
- Statistics Canada. (2016b). Language highlight tables, 2016 Census: Mother tongue by age (total), % distribution (2016) for the population excluding institutional residents of Canada, provinces and territories, 2016 Census. https://www12.statcan.gc.ca/census-recensement/2016/dp-pd/hltst/lang/Table.cfm?Lang=E&T=11&Geo=00&SP=1&view=2&age=1
- Statistics Canada. (2017a). Aboriginal peoples in Canada: Key results from the 2016 census. *The Daily*. https://www150.statcan.gc.ca/n1/daily-quotidien/171025/dq171025a-eng.htm
- Statistics Canada. (2017b). Aboriginal peoples highlight tables, 2016 Census: Aboriginal identity population by both sexes, total, age, 2016 counts, Canada, provinces and territories, 2016 Census. https://www12.statcan.gc.ca/census-recensement/2016/dp-pd/hlt-fst/abo-aut/Table. cfm?Lang=Eng&S=99&O=A&RPP=25
- Statistics Canada. (2017c). Census in brief: Linguistic integration of immigrants and official language populations in Canada. https://www12.statcan.gc.ca/census-recensement/2016/as-sa/98-200-x/2016017/98-200-x2016017-eng.cfm
- Statistics Canada. (2017d). Distribution (in percentage) of recent immigrants in Canada by provinces and territories, 1981 to 2016. https://www.statcan.gc.ca/eng/dai/btd/othervisuals/other008
- Statistics Canada. (2017e). An increasingly diverse linguistic profile: Corrected data from the 2016 Census. https://www150.statcan.gc.ca/n1/daily-quotidien/170817/dq170817a-eng. htm?indid=17282-2&indgeo=0
- Statistics Canada. (2019). Statistics on official languages in Canada. https://www.canada.ca/en/ canadian-heritage/services/official-languages-bilingualism/publications/statistics.html
- Statistics Canada (2020a). English, French and non-official languages spoken at home by geography, 2001 to 2016. Table 15-10-0009-01. https://doi.org/10.25318/1510000901-eng

- Statistics Canada. (2020b). Number of students in elementary and secondary schools, by school type and program type. Table 37-10-0109-01. https://www150.statcan.gc.ca/n1/daily-quotidien/201015/dq201015a-eng.htm
- Statistics Canada. (2021a). Number of students in official languages programs, public elementary and secondary schools, by program type, grade and sex. Table 37-10-0009-01. https://doi.org/10.25318/3710000901-eng
- Statistics Canada. (2021b). Proportion of educators in public elementary and secondary schools by sex. Table 37-10-0153-03. https://www150.statcan.gc.ca/t1/tbl1/en/tv.action?pid=3710015303
- Stein, M.K., Grover, B.W., & Henningsen, M.A. (1996). Building student capacity for mathematical thinking and reasoning: An analysis of mathematical tasks used in reform classrooms. *American Educational Research Journal*, 33(4), 455–488.
- Suh, J., & Seshaiyer, P. (2015). Examining teachers' understanding of the mathematical learning progression through vertical articulation during lesson study. *Journal of Mathematics Teacher Education*, 18(3), 207–229. https://doi.org/10.1007/s10857-014-9282-7
- Suppa, S., DiNapoli, J., Thanheiser, E., Tobias, J.M., & Yeo, S. (2020). Supporting novice mathematics teacher educators teaching elementary mathematics content courses for the first time. *Mathematics Enthusiast*, 17(2/3), 493–536. https://scholarworks.umt.edu/tme/vol17/ iss2/7/
- Swanson, H.L., Kong, J., & Petcu, S.D. (2019). Growth in math computation among monolingual and English language learners: Does the executive system have a role? *Developmental Neuropsychology*, 44(8), 566–593. https://doi.org/10.1080/87565641.2019.1688328
- Thanheiser, E. (2009). Preservice elementary school teachers' conceptions of multidigit whole numbers. *Journal for Research in Mathematics Education*, 40(3), 251–281.
- Tomlinson, C.A. (2017). What differentiated instruction is and isn't. In *How to differentiate instruction in academically diverse classrooms* (3<sup>rd</sup> ed.). ASCD. http://www1.ascd.org/publications/ books/117032/chapters/What-Differentiated-Instruction-Is%E2%80%94and-Isn't.aspx
- Toronto District School Board Research and Information Services. (2011). The effects of gradespan configurations on student achievement: Implications for the Toronto District School Board (TDSB) — A summary. https://www.tdsb.on.ca/portals/\_default/ARC\_helpful\_info\_docs/ GradeSpanConfigurationSUMMARY08Feb11.pdf
- Trautwein, U., & Köller, O. (2003). The relationship between homework and achievement: Still much of a mystery. *Educational Psychology Review*, *15*, 115–145.
- Trautwein, U., Lüdtke, O., Schnyder, I., & Niggli, A. (2006). Predicting homework effort: Support for a domain-specific, multilevel homework model. *Journal of Educational Psychology*, 98, 438– 456. https://doi.org/10.1037/0022-0663.98.2.438
- Truth and Reconciliation Commission of Canada. (2012). *Truth and Reconciliation Commission of Canada: Calls to Action*. Author.

- Uppal, S. (2017). Young men and women without a high school diploma. Catalogue no. 75-006-X. Statistics Canada. https://www150.statcan.gc.ca/n1/pub/75-006-x/2017001/article/14824-eng. htm
- Warton, P.M. (2001). The forgotten voices in homework: Views of students. *Educational Psychologist*, 36, 155–165.
- Watt, H.M., Shapka, J.D., Morris, Z.A., Durik, A.M., Keating, D.P., & Eccles, J.S. (2012). Gendered motivational processes affecting high school mathematics participation, educational aspirations, and career plans: A comparison of samples from Australia, Canada, and the United States. *Developmental Psychology*, 48, 1594–1611. http://dx.doi.org/ 10.1037/a0027838
- Webb, N.L. (2002). Depth of knowledge levels for four content areas. https://www.maine.gov/doe/ sites/maine.gov.doe/files/inline-files/dok.pdf
- Webel, C., Conner, K., Sheffel, C., Tarr, J., & Austin, C. (2017). Elementary mathematics specialists in "departmentalized" teaching assignments: Affordances and constraints. *Journal of Mathematical Behavior*, 46, 196–214. https://doi.org/10.1016/j.jmathb.2016.12.006
- Wech, D., & Weinkam, T. (2016, September). Determinants of the educational situation of young migrants. CESifo DICE Report, 14(3), 65–68. https://www.econstor.eu/ bitstream/10419/167276/1/ifo-dice-report-v14-y2016-i3-p65-68.pdf
- Weidemann, W. (1995). Problem solving in math class: "Word problems" were never like this. *Middle School Journal*, 27(1), 11–17.
- Western and Northern Canadian Protocol. (2006). *The common curriculum framework for K–9 mathematics*. https://open.alberta.ca/dataset/465efca1-3b52-4355-8fbd-88ce553b4384/ resource/6d788800-5b0b-4e7b-a501-b249ee03e306/download/ccfkto9.pdf
- Whitlock, J. (2003) Fostering school connectedness. ACT for Youth: Upstate Center of Excellence.
- Wilkelmann, T.J. (2016). *The impact of teacher gender on elementary students' academic performance*. Doctoral dissertation, Baker University.
- Willms, J.D., Friessen, S., & Milton, P. (2009). What did you do in school today? Transforming classrooms through social, academic and intellectual engagement. Canadian Education Association.
- Wilson, S.M., Floden, R.E., & Ferrini-Mundy, J. (2001). Teacher preparation research: Current knowledge, gaps, and recommendations. Center for the Study of Teaching and Policy, University of Washington. https://www.education.uw.edu/ctp/sites/default/files/ctpmail/PDFs/ TeacherPrep-WFFM-02-2001.pdf
- Xu, J. (2005). Purposes for doing homework reported by middle and high school students. *Journal of Educational Research, 99*, 46–55.
- Zimmerman, B.J., & Kitsantas, A. (2005). Homework practices and academic achievement: The mediating role of self-efficacy and perceived responsibility beliefs. *Contemporary Educational Psychology*, 30(4), 397–417. https://doi.org/10.1016/j.cedpsych.2005.05.003

### Rounding numbers

Because of rounding, some numbers in tables may not add up to exactly the totals shown. Percentages, mean scores, and differences are always calculated on the basis of the exact numbers and are rounded only after calculation.

#### Data accuracy

Data quality indicators presented in this report are adapted from data accuracy standards set forth by Statistics Canada (Statistics Canada, 2009); estimates based on fewer than 30 observations are flagged with the symbol ‡, and estimates with a coefficient of variation greater than 33.3 percent are considered to be too unreliable to be published and are therefore suppressed and represented by a "U."

|                           | Fen  | nale  | M    | ale   | l identify<br>myself in<br>another way |       | l prefe<br>to : | er not<br>say |
|---------------------------|------|-------|------|-------|--|-------|-----------------|---------------|
| Canada and provinces      | %    | SE    | %    | SE    | %                                      | SE    | %               | SE            |
| British Columbia          | 47.7 | (0.7) | 48.4 | (0.7) | 1.7                                    | (0.2) | 2.2             | (0.3)         |
| Alberta                   | 50.2 | (0.9) | 45.9 | (0.9) | 2.5                                    | (0.4) | 1.4             | (0.2)         |
| Saskatchewan              | 49.8 | (0.9) | 47.0 | (0.9) | 1.7                                    | (0.2) | 1.4             | (0.2)         |
| Manitoba                  | 49.3 | (0.9) | 47.1 | (0.9) | 1.5                                    | (0.2) | 2.1             | (0.3)         |
| Ontario                   | 48.6 | (1.0) | 48.6 | (1.0) | 1.2                                    | (0.3) | 1.6             | (0.3)         |
| Quebec                    | 46.5 | (1.3) | 49.5 | (1.4) | 1.8                                    | (0.3) | 2.2             | (0.3)         |
| New Brunswick             | 48.7 | (0.0) | 49.0 | (0.0) | 1.1‡                                   | (0.0) | 1.1‡            | (0.0)         |
| Nova Scotia               | 46.4 | (0.1) | 50.3 | (0.1) | 1.8                                    | (0.0) | 1.5‡            | (0.0)         |
| Prince Edward Island      | 44.2 | (0.0) | 51.5 | (0.0) | 1.5‡                                   | (0.0) | 2.9‡            | (0.0)         |
| Newfoundland and Labrador | 48.4 | (0.5) | 47.3 | (0.4) | 2.3‡                                   | (0.1) | 2.0‡            | (0.2)         |
| Canada                    | 48.3 | (0.5) | 48.3 | (0.5) | 1.6                                    | (0.1) | 1.8             | (0.1)         |

#### TABLE A.1.1 Percentage of students by gender self-identification

‡ There are fewer than 30 observations.

#### TABLE A.1.2 Percentage of students by gender self-identification and language of school system

|                              |      | Anglophone school systems |      |       |                           |                               |               |               | Francophone school systems |       |      |                           |                               |               |               |       |
|------------------------------|------|---------------------------|------|-------|---------------------------|-------------------------------|---------------|---------------|----------------------------|-------|------|---------------------------|-------------------------------|---------------|---------------|-------|
|                              | Fen  | nale                      | M    | ale   | l ide<br>myse<br>ano<br>w | ntify<br>elf in<br>ther<br>ay | l pr<br>not t | efer<br>o say | Female Male                |       | ale  | l ide<br>myse<br>ano<br>w | ntify<br>elf in<br>ther<br>ay | l pr<br>not t | efer<br>o say |       |
| Canada and<br>provinces      | %    | SE                        | %    | SE    | %                         | SE                            | %             | SE            | %                          | SE    | %    | SE                        | %                             | SE            | %             | SE    |
| British Columbia             | 47.6 | (0.7)                     | 48.5 | (0.7) | 1.7                       | (0.2)                         | 2.2           | (0.3)         | 51.7                       | (0.0) | 43.4 | (0.0)                     | 2.8‡                          | (0.0)         | 2.1‡          | (0.0) |
| Alberta                      | 50.2 | (0.9)                     | 45.9 | (0.9) | 2.5                       | (0.4)                         | 1.4           | (0.2)         | 47.3                       | (2.9) | 45.9 | (2.3)                     | 4.8‡                          | (1.0)         | 2.0‡          | (0.5) |
| Saskatchewan                 | 49.9 | (0.9)                     | 46.9 | (0.9) | 1.8                       | (0.2)                         | 1.4           | (0.2)         | 40.7                       | (0.0) | 59.3 | (0.0)                     | 0.0                           | (0.0)         | 0.0           | (0.0) |
| Manitoba                     | 49.4 | (0.9)                     | 47.0 | (0.9) | 1.6                       | (0.2)                         | 2.1           | (0.3)         | 45.3                       | (0.0) | 51.1 | (0.0)                     | 0.7‡                          | (0.0)         | 2.8‡          | (0.0) |
| Ontario                      | 48.9 | (1.1)                     | 48.3 | (1.0) | 1.2                       | (0.3)                         | 1.6           | (0.3)         | 43.4                       | (1.0) | 53.6 | (0.9)                     | 0.7‡                          | (0.1)         | 2.3           | (0.3) |
| Quebec                       | 48.1 | (1.9)                     | 47.8 | (1.9) | 2.5‡                      | (0.3)                         | 1.6‡          | (0.3)         | 46.3                       | (1.5) | 49.7 | (1.5)                     | 1.7                           | (0.3)         | 2.2           | (0.3) |
| New Brunswick                | 49.2 | (0.0)                     | 48.1 | (0.0) | 1.4‡                      | (0.0)                         | 1.4‡          | (0.0)         | 47.4                       | (0.0) | 51.5 | (0.0)                     | 0.5‡                          | (0.0)         | 0.6‡          | (0.0) |
| Nova Scotia                  | 46.4 | (0.1)                     | 50.1 | (0.1) | 1.9                       | (0.0)                         | 1.6‡          | (0.0)         | 45.2                       | (1.0) | 54.1 | (0.9)                     | 0.7‡                          | (0.2)         | 0.0           | (0.0) |
| Prince Edward<br>Island      | 44.0 | (0.0)                     | 51.5 | (0.0) | 1.5‡                      | (0.0)                         | 3.0‡          | (0.0)         |                            |       |      |                           |                               |               |               |       |
| Newfoundland and<br>Labrador | 48.4 | (0.5)                     | 47.3 | (0.4) | 2.3‡                      | (0.1)                         | 2.0‡          | (0.2)         |                            |       |      |                           |                               |               |               |       |
| Canada                       | 48.9 | (0.6)                     | 47.8 | (0.5) | 1.6                       | (0.2)                         | 1.7           | (0.1)         | 46.1                       | (1.2) | 50.1 | (1.3)                     | 1.6                           | (0.3)         | 2.2           | (0.3) |

**‡** There are fewer than 30 observations.

|                                  | Mean score | SE    |
|----------------------------------|------------|-------|
| Female                           | 512        | (2.1) |
| Male                             | 509        | (2.1) |
| I identify myself in another way | 495*       | (6.7) |
| l prefer not to say              | 493*       | (7.4) |

\* Significant difference compared to the *female* category

### TABLE A.1.4 Percentage of students by first language spoken

|                           | Eng  | lish  | Frer | nch   | Aı<br>Indige<br>Iangı | n<br>nous<br>iage | Oth<br>Iangu | ier<br>iage |
|---------------------------|------|-------|------|-------|-----------------------|-------------------|--------------|-------------|
| Canada and provinces      | %    | SE    | %    | SE    | %                     | SE                | %            | SE          |
| British Columbia          | 76.8 | (1.4) | 1.0  | (0.1) | 0.7‡                  | (0.1)             | 21.5         | (1.4)       |
| Alberta                   | 78.9 | (1.5) | 1.7  | (0.3) | 0.8‡                  | (0.2)             | 18.6         | (1.5)       |
| Saskatchewan              | 84.6 | (1.2) | 0.6  | (0.1) | 1.3‡                  | (0.3)             | 13.5         | (1.1)       |
| Manitoba                  | 78.5 | (1.1) | 2.1  | (0.2) | 1.4                   | (0.2)             | 18.0         | (1.1)       |
| Ontario                   | 77.6 | (1.6) | 2.4  | (0.2) | 0.4‡                  | (0.1)             | 19.6         | (1.6)       |
| Quebec                    | 13.2 | (0.9) | 77.6 | (1.5) | 1.4                   | (0.3)             | 7.8          | (0.8)       |
| New Brunswick             | 70.0 | (0.0) | 24.9 | (0.0) | 0.4‡                  | (0.0)             | 4.6          | (0.0)       |
| Nova Scotia               | 92.3 | (0.2) | 2.7  | (0.2) | 0.3‡                  | (0.0)             | 4.7          | (0.1)       |
| Prince Edward Island      | 93.8 | (0.0) | 2.1  | (0.0) | 0.4‡                  | (0.0)             | 3.6‡         | (0.0)       |
| Newfoundland and Labrador | 95.4 | (0.2) | 0.7  | (0.1) | 0.8‡                  | (0.1)             | 3.1          | (0.2)       |
| Canada                    | 65.6 | (0.8) | 17.5 | (0.5) | 0.7                   | (0.1)             | 16.2         | (0.7)       |

‡ There are fewer than 30 observations.

| Anglophone school systems    |      |       |      |       |               |               |             | Francophone school systems |      |       |      |       |               |               |              |             |
|------------------------------|------|-------|------|-------|---------------|---------------|-------------|----------------------------|------|-------|------|-------|---------------|---------------|--------------|-------------|
|                              | Eng  | lish  | Fre  | nch   | Indig<br>lang | enous<br>uage | Otl<br>lang | ner<br>uage                | Eng  | lish  | Fre  | nch   | Indig<br>lang | enous<br>uage | Otł<br>Iangu | ner<br>Jage |
| Canada and<br>provinces      | %    | SE    | %    | SE    | %             | SE            | %           | SE                         | %    | SE    | %    | SE    | %             | SE            | %            | SE          |
| British Columbia             | 76.9 | (1.4) | 0.8‡ | (0.1) | 0.7‡          | (0.1)         | 21.6        | (1.4)                      | 59.6 | (0.0) | 29.3 | (0.0) | 0.0           | (0.0)         | 11.1‡        | (0.0)       |
| Alberta                      | 79.4 | (1.6) | 1.3‡ | (0.3) | 0.8‡          | (0.2)         | 18.6        | (1.5)                      | 45.6 | (2.6) | 33.4 | (1.0) | 0.0           | (0.0)         | 21.0         | (2.7)       |
| Saskatchewan                 | 84.9 | (1.2) | U‡   | (0.1) | 1.3‡          | (0.3)         | 13.5        | (1.1)                      | 54.9 | (0.0) | 35.4 | (0.0) | 1.4‡          | 0.0           | 8.3‡         | 0.0         |
| Manitoba                     | 79.5 | (1.1) | 0.8‡ | (0.2) | 1.4           | (0.2)         | 18.3        | (1.1)                      | 40.8 | (0.0) | 53.2 | (0.0) | 0.4‡          | (0.0)         | 5.7‡         | (0.0)       |
| Ontario                      | 79.0 | (1.7) | 0.6‡ | (0.2) | 0.4‡          | (0.1)         | 20.0        | (1.7)                      | 49.9 | (1.7) | 37.6 | (1.6) | 0.3‡          | (0.1)         | 12.2         | (1.0)       |
| Quebec                       | 70.8 | (1.9) | 17.8 | (1.7) | 1.4‡          | (0.4)         | 10.0        | (1.0)                      | 6.6  | (0.8) | 84.5 | (1.5) | 1.4           | (0.4)         | 7.6          | (0.9)       |
| New Brunswick                | 92.3 | (0.0) | 1.6‡ | (0.0) | 0.6‡          | (0.0)         | 5.5         | (0.0)                      | 15.0 | (0.0) | 82.6 | (0.0) | 0.1‡          | (0.0)         | 2.3‡         | (0.0)       |
| Nova Scotia                  | 93.5 | (0.1) | 1.4‡ | (0.0) | 0.3‡          | (0.0)         | 4.8         | (0.1)                      | 66.3 | (1.7) | 31.0 | (1.9) | U‡            | (0.1)         | 2.3‡         | 0.4         |
| Prince Edward<br>Island      | 95.3 | (0.0) | 0.6‡ | (0.0) | 0.4‡          | (0.0)         | 3.7‡        | (0.0)                      |      |       |      |       |               |               |              |             |
| Newfoundland and<br>Labrador | 95.5 | (0.2) | 0.6‡ | (0.1) | 0.8‡          | (0.1)         | 3.1         | (0.2)                      |      |       |      |       |               |               |              |             |
| Canada                       | 79.8 | (0.9) | 1.2  | (0.1) | 0.6           | (0.1)         | 18.3        | (0.9)                      | 12.2 | (0.8) | 78.7 | (1.4) | 1.2           | (0.3)         | 8.0          | (0.8)       |

#### TABLE A.1.5 Percentage of students by first language spoken and language of school system

‡ There are fewer than 30 observations.

U Too unreliable to be published

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

#### TABLE A.1.6 Mathematics achievement by students' first language and language of the school system, Canada

|                            | First language | Mean score | SE    |
|----------------------------|----------------|------------|-------|
| Anglophone school systems  | English        | 503        | (2.1) |
|                            | French         | 486        | (8.4) |
|                            | Other          | 505        | (4.0) |
| Francophone school systems | English        | 522*       | (3.9) |
|                            | French         | 540*       | (3.5) |
|                            | Other          | 519        | (8.3) |

\* Significant difference compared to the anglophone school systems category

| Canada and provinces%SE%%SE%%SE%%SE%%SE%% </th <th></th> <th>Eng<br/>onl<sup>i</sup><br/>mo<br/>Eng</th> <th>lish<br/>y or<br/>stly<br/>lish</th> <th>Fre<br/>onl<br/>mo<br/>Fre</th> <th>nch<br/>y or<br/>stly<br/>nch</th> <th>Eng<br/>and F<br/>equ</th> <th>glish<br/>French<br/>Jally</th> <th>Eng<br/>an<br/>lang<br/>othei<br/>Fre</th> <th>glish<br/>d a<br/>uage<br/>r than<br/>nch</th> <th>Fre<br/>an<br/>lang<br/>othe<br/>Eng</th> <th>nch<br/>d a<br/>uage<br/>r than<br/>glish</th> <th>Most<br/>Indig<br/>lang</th> <th>tly an<br/>enous<br/>uage</th> <th>Mo<br/>oti<br/>langi</th> <th>stly<br/>her<br/>Jages</th>  |                              | Eng<br>onl <sup>i</sup><br>mo<br>Eng | lish<br>y or<br>stly<br>lish | Fre<br>onl<br>mo<br>Fre | nch<br>y or<br>stly<br>nch | Eng<br>and F<br>equ | glish<br>French<br>Jally | Eng<br>an<br>lang<br>othei<br>Fre | glish<br>d a<br>uage<br>r than<br>nch | Fre<br>an<br>lang<br>othe<br>Eng | nch<br>d a<br>uage<br>r than<br>glish | Most<br>Indig<br>lang | tly an<br>enous<br>uage | Mo<br>oti<br>langi | stly<br>her<br>Jages |
|--|------------------------------|--------------------------------------|------------------------------|-------------------------|----------------------------|---------------------|--------------------------|-----------------------------------|---------------------------------------|----------------------------------|---------------------------------------|-----------------------|-------------------------|--------------------|----------------------|
| British Columbia       65.3       (1.5)       0.6       (0.1)       2.1       (0.3)       19.7       (1.0)       0.3‡       (0.1)       0.5‡       (0.1)       11.6       (0.7)         Alberta       69.1       (1.8)       0.5‡       (0.1)       2.4       (0.3)       17.9       (1.2)       U‡       (0.2)       0.3‡       (0.1)       9.4       (0.8)         Saskatchewan       77.1       (1.3)       0.8‡       (0.2)       2.4       (0.3)       13.2       (1.0)       0.5‡       (0.1)       1.2‡       (0.3)       4.8       (0.5)         Manitoba       68.5       (1.1)       0.9       (0.1)       4.6       (0.4)       15.0       (0.8)       0.2‡       (0.0)       1.2‡       (0.2)       9.4       (0.6)         Ontario       61.5       (1.6)       1.3       (0.2)       4.1       (0.4)       23.0       (1.3)       0.6       (0.1)       0.5‡       (0.1)       0.5‡       (0.1)       9.0       (0.7)         Quebec       7.5       (0.6)       60.1       1.7       18.0       (0.7)       3.7       (0.4)       6.5       (0.8)       0.4‡       (0.0)       2.3       (0.1)         New Bru              | Canada and provinces         | %                                    | SE                           | %                       | SE                         | %                   | SE                       | %                                 | SE                                    | %                                | SE                                    | %                     | SE                      | %                  | SE                   |
| Alberta69.1(1.8)0.5‡(0.1)2.4(0.3)17.9(1.2)U‡(0.2)0.3‡(0.1)9.4(0.8)Saskatchewan77.1(1.3)0.8‡(0.2)2.4(0.3)13.2(1.0)0.5‡(0.1)1.2‡(0.3)4.8(0.5)Manitoba68.5(1.1)0.9(0.1)4.6(0.4)15.0(0.8)0.2‡(0.0)1.2‡(0.2)9.4(0.6)Ontario61.5(1.6)1.3(0.2)4.1(0.4)23.0(1.3)0.6(0.1)0.5‡(0.1)9.0(0.7)Quebec7.5(0.6)60.1(1.7)18.0(0.7)3.7(0.4)6.5(0.8)0.6‡(0.2)3.5(0.4)New Brunswick62.1(0.0)12.9(0.0)18.0(0.0)4.0(0.0)0.4‡(0.0)2.3(0.0)Nova Scotia83.4(0.2)1.2‡(0.0)6.4(0.1)4.7(0.1)0.5‡(0.0)0.4‡(0.0)2.1‡(0.0)Newfoundland and<br>Labrador90.2(0.4)0.2‡(0.0)2.5‡(0.2)4.2(0.2)0.1‡(0.0)0.5‡(0.1)2.4‡(0.2)Canada54.0(0.8)13.0(0.4)6.7(0.2)0.1‡(0.0)2.5‡(0.2)1.5‡(0.2)0.1‡(0.0)0.5‡(0.1)2.4‡(0.2)New Gruns Scotia85.9(0.4)0.2‡ <td< td=""><td>British Columbia</td><td>65.3</td><td>(1.5)</td><td>0.6</td><td>(0.1)</td><td>2.1</td><td>(0.3)</td><td>19.7</td><td>(1.0)</td><td>0.3‡</td><td>(0.1)</td><td>0.5‡</td><td>(0.1)</td><td>11.6</td><td>(0.7)</td></td<>   | British Columbia             | 65.3                                 | (1.5)                        | 0.6                     | (0.1)                      | 2.1                 | (0.3)                    | 19.7                              | (1.0)                                 | 0.3‡                             | (0.1)                                 | 0.5‡                  | (0.1)                   | 11.6               | (0.7)                |
| Saskatchewan77.1(1.3)0.8‡(0.2)2.4(0.3)13.2(1.0)0.5‡(0.1)1.2‡(0.3)4.8(0.5)Manitoba68.5(1.1)0.9(0.1)4.6(0.4)15.0(0.8)0.2‡(0.0)1.2‡(0.2)9.4(0.6)Ontario61.5(1.6)1.3(0.2)4.1(0.4)23.0(1.3)0.6(0.1)0.5‡(0.1)9.0(0.7)Quebec7.5(0.6)60.1(1.7)18.0(0.7)3.7(0.4)6.5(0.8)0.6‡(0.2)3.5(0.4)New Brunswick62.1(0.0)12.9(0.0)18.0(0.0)4.0(0.0)0.4‡(0.0)2.3(0.0)Nova Scotia83.4(0.2)1.7(0.0)6.4(0.1)4.7(0.1)0.5‡(0.0)0.4‡(0.0)2.9(0.1)Prince Edward Island85.9(0.0)1.2‡(0.0)2.5‡(0.2)3.9‡(0.0)0.3‡(0.0)2.4‡(0.2)Newfoundland and Labrador90.2(0.4)0.2‡(0.0)2.5‡(0.2)4.2(0.2)0.1‡(0.1)2.4‡(0.2)Canada54.0(0.8)13.0(0.4)6.7(0.2)16.3(0.6)1.7(0.2)0.5(0.1)7.8(0.2)   | Alberta                      | 69.1                                 | (1.8)                        | 0.5‡                    | (0.1)                      | 2.4                 | (0.3)                    | 17.9                              | (1.2)                                 | U‡                               | (0.2)                                 | 0.3‡                  | (0.1)                   | 9.4                | (0.8)                |
| Manitoba68.5(1.1)0.9(0.1)4.6(0.4)15.0(0.8)0.2‡(0.0)1.2‡(0.2)9.4(0.6)Ontario61.5(1.6)1.3(0.2)4.1(0.4)23.0(1.3)0.6(0.1)0.5‡(0.1)9.0(0.7)Quebec7.5(0.6)60.1(1.7)18.0(0.7)3.7(0.4)6.5(0.8)0.6‡(0.2)3.5(0.4)New Brunswick62.1(0.0)12.9(0.0)18.0(0.0)4.0(0.0)0.4‡(0.0)2.3(0.0)Nova Scotia83.4(0.2)1.7(0.0)6.4(0.1)4.7(0.1)0.5‡(0.0)0.3‡(0.0)2.9(0.1)Prince Edward Island85.9(0.0)1.2‡(0.0)6.3(0.0)3.9‡(0.0)0.3‡(0.0)2.1‡(0.0)Newfoundland and Labrador90.2(0.4)0.2‡(0.2)2.5‡(0.2)4.2(0.2)0.1‡(0.1)2.4‡(0.2)Canada54.0(0.8)13.0(0.4)6.7(0.2)16.3(0.6)1.7(0.2)0.5(0.1)7.8(0.3)   | Saskatchewan                 | 77.1                                 | (1.3)                        | 0.8‡                    | (0.2)                      | 2.4                 | (0.3)                    | 13.2                              | (1.0)                                 | 0.5‡                             | (0.1)                                 | 1.2‡                  | (0.3)                   | 4.8                | (0.5)                |
| Ontario       61.5       (1.6)       1.3       (0.2)       4.1       (0.4)       23.0       (1.3)       0.6       (0.1)       0.5‡       (0.1)       9.0       (0.7)         Quebec       7.5       (0.6)       60.1       (1.7)       18.0       (0.7)       3.7       (0.4)       6.5       (0.8)       0.6‡       (0.2)       3.5       (0.4)         New Brunswick       62.1       (0.0)       12.9       (0.0)       18.0       (0.0)       4.0       (0.0)       0.4‡       (0.0)       0.4‡       (0.0)       2.3       (0.0)         Nova Scotia       83.4       (0.2)       1.7       (0.0)       6.4       (0.1)       4.7       (0.1)       0.5‡       (0.0)       0.4‡       (0.0)       2.9       (0.1)         Prince Edward Island       85.9       (0.0)       1.2‡       (0.0)       6.3       (0.0)       3.9‡       (0.0)       0.3‡       (0.0)       2.4‡       (0.0)         Newfoundland and Labrador       90.2       (0.4)       0.2‡       (0.2)       4.2       (0.2)       0.1‡       0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       < | Manitoba                     | 68.5                                 | (1.1)                        | 0.9                     | (0.1)                      | 4.6                 | (0.4)                    | 15.0                              | (0.8)                                 | 0.2‡                             | (0.0)                                 | 1.2‡                  | (0.2)                   | 9.4                | (0.6)                |
| Quebec       7.5       (0.6)       60.1       (1.7)       18.0       (0.7)       3.7       (0.4)       6.5       (0.8)       0.6‡       (0.2)       3.5       (0.4)         New Brunswick       62.1       (0.0)       12.9       (0.0)       18.0       (0.0)       4.0       (0.0)       0.4‡       (0.0)       0.4‡       (0.0)       2.3       (0.0)         Nova Scotia       83.4       (0.2)       1.7       (0.0)       6.4       (0.1)       4.7       (0.1)       0.5‡       (0.0)       0.4‡       (0.0)       2.9       (0.1)         Prince Edward Island       85.9       (0.0)       1.2‡       (0.0)       6.3       (0.0)       3.9‡       (0.0)       0.3‡       (0.0)       2.1‡       (0.0)         Newfoundland and<br>Labrador       90.2       (0.4)       0.2‡       (0.0)       2.5‡       (0.2)       4.2       (0.2)       0.1‡       (0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       16.3       (0.6)       1.7       (0.2)       0.5       (0.1)       7.8       (0.3)  | Ontario                      | 61.5                                 | (1.6)                        | 1.3                     | (0.2)                      | 4.1                 | (0.4)                    | 23.0                              | (1.3)                                 | 0.6                              | (0.1)                                 | 0.5‡                  | (0.1)                   | 9.0                | (0.7)                |
| New Brunswick       62.1       (0.0)       12.9       (0.0)       18.0       (0.0)       4.0       (0.0)       0.4‡       (0.0)       0.4‡       (0.0)       2.3       (0.0)         Nova Scotia       83.4       (0.2)       1.7       (0.0)       6.4       (0.1)       4.7       (0.1)       0.5‡       (0.0)       0.3‡       (0.0)       2.9       (0.1)         Prince Edward Island       85.9       (0.0)       1.2‡       (0.0)       6.3       (0.0)       3.9‡       (0.0)       0.3‡       (0.0)       2.1‡       (0.0)         Newfoundland and Labrador       90.2       (0.4)       0.2‡       (0.0)       2.5‡       (0.2)       4.2       (0.2)       0.1‡       (0.0)       0.5‡       (0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       16.3       (0.6)       1.7       (0.2)       0.5       (0.1)       7.8       (0.3)  | Quebec                       | 7.5                                  | (0.6)                        | 60.1                    | (1.7)                      | 18.0                | (0.7)                    | 3.7                               | (0.4)                                 | 6.5                              | (0.8)                                 | 0.6‡                  | (0.2)                   | 3.5                | (0.4)                |
| Nova Scotia       83.4       (0.2)       1.7       (0.0)       6.4       (0.1)       4.7       (0.1)       0.5‡       (0.0)       0.3‡       (0.0)       2.9       (0.1)         Prince Edward Island       85.9       (0.0)       1.2‡       (0.0)       6.3       (0.0)       3.9‡       (0.0)       0.3‡       (0.0)       2.1‡       (0.0)         Newfoundland and<br>Labrador       90.2       (0.4)       0.2‡       (0.0)       2.5‡       (0.2)       4.2       (0.2)       0.1‡       (0.0)       0.5‡       (0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       16.3       (0.6)       1.7       (0.2)       0.5       (0.1)       7.8       (0.3)  | New Brunswick                | 62.1                                 | (0.0)                        | 12.9                    | (0.0)                      | 18.0                | (0.0)                    | 4.0                               | (0.0)                                 | 0.4‡                             | (0.0)                                 | 0.4‡                  | (0.0)                   | 2.3                | (0.0)                |
| Prince Edward Island       85.9       (0.0)       1.2‡       (0.0)       6.3       (0.0)       3.9‡       (0.0)       0.3‡       (0.0)       0.4‡       (0.0)       2.1‡       (0.0)         Newfoundland and<br>Labrador       90.2       (0.4)       0.2‡       (0.0)       2.5‡       (0.2)       4.2       (0.2)       0.1‡       (0.0)       0.5‡       (0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       16.3       (0.6)       1.7       (0.2)       0.5‡       (0.1)       7.8       (0.3)   | Nova Scotia                  | 83.4                                 | (0.2)                        | 1.7                     | (0.0)                      | 6.4                 | (0.1)                    | 4.7                               | (0.1)                                 | 0.5‡                             | (0.0)                                 | 0.3‡                  | (0.0)                   | 2.9                | (0.1)                |
| Newfoundland and<br>Labrador       90.2       (0.4)       0.2‡       (0.0)       2.5‡       (0.2)       4.2       (0.2)       0.1‡       (0.0)       0.5‡       (0.1)       2.4‡       (0.2)         Canada       54.0       (0.8)       13.0       (0.4)       6.7       (0.2)       16.3       (0.6)       1.7       (0.2)       0.5‡       (0.1)       2.4‡       (0.2)   | Prince Edward Island         | 85.9                                 | (0.0)                        | 1.2‡                    | (0.0)                      | 6.3                 | (0.0)                    | 3.9‡                              | (0.0)                                 | 0.3‡                             | (0.0)                                 | 0.4‡                  | (0.0)                   | 2.1‡               | (0.0)                |
| Canada         54.0         (0.8)         13.0         (0.4)         6.7         (0.2)         16.3         (0.6)         1.7         (0.2)         0.5         (0.1)         7.8         (0.3)  | Newfoundland and<br>Labrador | 90.2                                 | (0.4)                        | 0.2‡                    | (0.0)                      | 2.5‡                | (0.2)                    | 4.2                               | (0.2)                                 | 0.1‡                             | (0.0)                                 | 0.5‡                  | (0.1)                   | 2.4‡               | (0.2)                |
|  | Canada                       | 54.0                                 | (0.8)                        | 13.0                    | (0.4)                      | 6.7                 | (0.2)                    | 16.3                              | (0.6)                                 | 1.7                              | (0.2)                                 | 0.5                   | (0.1)                   | 7.8                | (0.3)                |

#### TABLE A.1.7 Percentage of students by language used in everyday life

**‡** There are fewer than 30 observations.

U Too unreliable to be published

|                              | Anglophone school systems |                                 |                                      |                            |                     |                       |                                  |                                 |                                 |                                   |                        |                         |                    |                      |
|------------------------------|---------------------------|---------------------------------|--------------------------------------|----------------------------|---------------------|-----------------------|----------------------------------|---------------------------------|---------------------------------|-----------------------------------|------------------------|-------------------------|--------------------|----------------------|
|                              | Eng<br>onl<br>mo<br>Eng   | glish<br>y or<br>ostly<br>glish | Fre<br>onl <sup>y</sup><br>mo<br>Fre | nch<br>y or<br>stly<br>nch | Eng<br>and F<br>equ | lish<br>rench<br>ally | Englis<br>a lang<br>other<br>Fre | h and<br>guage<br>• than<br>nch | Frenc<br>a lang<br>othei<br>Eng | h and<br>guage<br>' than<br>¦lish | Most<br>Indigo<br>Iang | tly an<br>enous<br>uage | Mo<br>otl<br>langu | stly<br>ner<br>Jages |
| Canada and provinces         | %                         | SE                              | %                                    | SE                         | %                   | SE                    | %                                | SE                              | %                               | SE                                | %                      | SE                      | %                  | SE                   |
| British Columbia             | 65.4                      | (1.5)                           | 0.5‡                                 | (0.1)                      | 1.9                 | (0.3)                 | 19.7                             | (1.0)                           | 0.3‡                            | (0.1)                             | 0.5‡                   | (0.1)                   | 11.6               | (0.7)                |
| Alberta                      | 69.7                      | (1.8)                           | 0.4‡                                 | (0.1)                      | 1.9                 | (0.3)                 | 17.9                             | (1.2)                           | U‡                              | (0.2)                             | 0.3‡                   | (0.1)                   | 9.4                | (0.8)                |
| Saskatchewan                 | 77.3                      | (1.3)                           | 0.7‡                                 | (0.2)                      | 2.2                 | (0.3)                 | 13.2                             | (1.0)                           | 0.5‡                            | (0.1)                             | 1.2‡                   | (0.3)                   | 4.9                | (0.5)                |
| Manitoba                     | 69.5                      | (1.1)                           | 0.7‡                                 | (0.1)                      | 3.6                 | (0.4)                 | 15.3                             | (0.9)                           | U‡                              | (0.0)                             | 1.2‡                   | (0.2)                   | 9.7                | (0.6)                |
| Ontario                      | 62.7                      | (1.7)                           | 0.8‡                                 | (0.2)                      | 2.6                 | (0.4)                 | 23.6                             | (1.4)                           | 0.5‡                            | (0.1)                             | 0.5‡                   | (0.1)                   | 9.3                | (0.8)                |
| Quebec                       | 45.1                      | (1.9)                           | 6.3                                  | (0.8)                      | 29.2                | (1.8)                 | 14.3                             | (0.7)                           | 1.2‡                            | (0.3)                             | 0.8‡                   | (0.2)                   | 3.3                | (0.3)                |
| New Brunswick                | 83.9                      | (0.0)                           | 1.1‡                                 | (0.0)                      | 6.3                 | (0.0)                 | 5.3                              | (0.0)                           | 0.2‡                            | (0.0)                             | 0.5‡                   | (0.0)                   | 2.7                | (0.0)                |
| Nova Scotia                  | 85.0                      | (0.1)                           | 1.3‡                                 | (0.0)                      | 5.2                 | (0.1)                 | 4.6                              | (0.1)                           | 0.5‡                            | (0.0)                             | 0.3‡                   | (0.0)                   | 3.0                | (0.1)                |
| Prince Edward Island         | 88.4                      | (0.0)                           | 0.9‡                                 | (0.0)                      | 3.9‡                | (0.0)                 | 4.0‡                             | (0.0)                           | 0.2‡                            | (0.0)                             | 0.4‡                   | (0.0)                   | 2.1‡               | (0.0)                |
| Newfoundland and<br>Labrador | 90.2                      | (0.4)                           | 0.1‡                                 | (0.0)                      | 2.4‡                | (0.2)                 | 4.2                              | (0.2)                           | 0.1‡                            | (0.0)                             | 0.6‡                   | (0.1)                   | 2.4‡               | (0.2)                |
| Canada                       | 66.3                      | (0.9)                           | 0.9                                  | (0.1)                      | 3.2                 | (0.2)                 | 19.7                             | (0.7)                           | 0.4                             | (0.1)                             | 0.5                    | (0.1)                   | 8.9                | (0.4)                |
|                              |                           |                                 |                                      |                            |                     | Franco                | phone                            | school                          | system                          | S                                 |                        |                         |                    |                      |
|                              | Eng<br>onl<br>mo<br>Eng   | glish<br>y or<br>ostly<br>glish | Fre<br>onl <sup>y</sup><br>mo<br>Fre | nch<br>y or<br>stly<br>nch | Eng<br>and F<br>equ | lish<br>rench<br>ally | Englis<br>a lang<br>other<br>Fre | h and<br>guage<br>than<br>nch   | Frenc<br>a lang<br>othei<br>Eng | h and<br>guage<br>' than<br>¦lish | Most<br>Indig<br>Iang  | tly an<br>enous<br>uage | Mo<br>otl<br>langu | stly<br>ner<br>Jages |
| Canada and provinces         | %                         | SE                              | %                                    | SE                         | %                   | SE                    | %                                | SE                              | %                               | SE                                | %                      | SE                      | %                  | SE                   |
| British Columbia             | 41.3                      | (0.0)                           | 7.9‡                                 | (0.0)                      | 33.3                | (0.0)                 | 9.7‡                             | (0.0)                           | 4.2‡                            | (0.0)                             | 0.0                    | (0.0)                   | 3.7‡               | (0.0)                |
| Alberta                      | 28.9                      | (3.4)                           | 6.0‡                                 | (1.1)                      | 38.1                | (2.8)                 | 14.4‡                            | (2.3)                           | 5.8‡                            | (1.9)                             | 0.0                    | (0.0)                   | 6.9‡               | (1.5)                |
| Saskatchewan                 | 48.9                      | (0.0)                           | 10.3‡                                | (0.0)                      | 28.8‡               | (0.0)                 | 4.5‡                             | (0.0)                           | 3.6‡                            | (0.0)                             | 1.2‡                   | (0.0)                   | 2.8‡               | (0.0)                |
| Manitoba                     | 30.6                      | (0.0)                           | 10.3‡                                | (0.0)                      | 47.8                | (0.0)                 | 5.5‡                             | (0.0)                           | 4.3‡                            | (0.0)                             | 0.4‡                   | (0.0)                   | 1.1‡               | (0.0)                |
| Ontario                      | 39.2                      | (1.4)                           | 9.2                                  | (1.0)                      | 33.3                | (0.9)                 | 11.3                             | (1.1)                           | 3.5                             | (0.4)                             | U‡                     | (0.1)                   | 3.4                | (0.3)                |
| Quebec                       | 3.2                       | (0.6)                           | 66.3                                 | (1.8)                      | 16.7                | (0.7)                 | 2.5                              | (0.4)                           | 7.2                             | (0.9)                             | 0.6‡                   | (0.2)                   | 3.5                | (0.4)                |
| New Brunswick                | 7.8                       | (0.0)                           | 42.3                                 | (0.0)                      | 47.0                | (0.0)                 | 0.7‡                             | (0.0)                           | 0.8‡                            | (0.0)                             | 0.1‡                   | (0.0)                   | 1.4‡               | (0.0)                |
| Nova Scotia                  | 49.4                      | (1.6)                           | 8.7‡                                 | (0.5)                      | 31.3                | (1.3)                 | 6.5‡                             | (0.6)                           | 2.2‡                            | (0.4)                             | 0.0                    | (0.0)                   | 2.0‡               | (0.5)                |
| Canada                       | 7.7                       | (0.6)                           | 58.6                                 | (1.7)                      | 19.7                | (0.6)                 | 3.5                              | (0.4)                           | 6.6                             | (0.8)                             | 0.5‡                   | (0.1)                   | 3.4                | (0.4)                |

#### TABLE A.1.8 Percentage of students by language used in everyday life and language of school system

**‡** There are fewer than 30 observations.

U Too unreliable to be published

|                           | English<br>immersion |       | Fre<br>imme | nch<br>ersion | Indigenous-<br>language<br>immersion |      |       | Otł<br>lang<br>imme | ner-<br>uage<br>ersion |       |
|---------------------------|----------------------|-------|-------------|---------------|--------------------------------------|------|-------|---------------------|------------------------|-------|
| Canada and provinces      | %                    | % SE  |             | %             | SE                                   | %    | SE    |                     | %                      | SE    |
| British Columbia          | 25.0                 | (1.1) |             | 24.4          | (1.8)                                | 2.6  | (0.3) |                     | 7.2                    | (0.5) |
| Alberta                   | 25.0                 | (1.3) |             | 22.4          | (2.1)                                | 0.6‡ | (0.2) |                     | 8.0                    | (0.9) |
| Saskatchewan              | 23.5                 | (1.2) |             | 19.4          | (1.9)                                | 4.4  | (0.7) |                     | 4.1                    | (0.5) |
| Manitoba                  | 33.4                 | (1.2) |             | 27.9          | (2.0)                                | 3.8  | (0.6) |                     | 6.7                    | (0.4) |
| Ontario                   | 20.6                 | (1.2) |             | 25.5          | (2.6)                                | 1.2  | (0.3) |                     | 6.5                    | (0.5) |
| Quebec                    | 32.6                 | (1.3) |             | 18.5          | (0.9)                                | 1.3  | (0.2) |                     | 9.7                    | (0.8) |
| New Brunswick             | 28.0                 | (0.0) |             | 47.9          | (0.0)                                | 2.0  | (0.0) |                     | 3.6                    | (0.0) |
| Nova Scotia               | 29.1                 | (0.3) |             | 44.3          | (0.5)                                | 1.2‡ | (0.0) |                     | 3.1                    | (0.1) |
| Prince Edward Island      | 24.4                 | (0.0) |             | 51.4          | (0.0)                                | 1.0‡ | (0.0) |                     | 2.6‡                   | (0.0) |
| Newfoundland and Labrador | 28.3                 | (0.6) |             | 33.2          | (1.2)                                | 1.2‡ | (0.1) |                     | 3.0                    | (0.2) |
| Canada                    | 25.2                 | (0.6) |             | 24.6          | (1.1)                                | 1.5  | (0.1) |                     | 7.2                    | (0.3) |

#### TABLE A.1.9 Percentage of students enrolled in language-immersion programs

‡ There are fewer than 30 observations.

## TABLE A.1.10 Percentage of students by enrolment in second-language programs and language of school system, Canada

|                                  | An    | glophoi        | ne school sy | sten         | ns            | Fra         | Francophone school systems |               |                 |  |  |  |
|----------------------------------|-------|----------------|--------------|--------------|---------------|-------------|----------------------------|---------------|-----------------|--|--|--|
|                                  | Curre | ently<br>olled | Pr<br>e      | evio<br>nrol | ously<br>lled | Curr<br>enr | rently<br>olled            | Previ<br>enro | iously<br>olled |  |  |  |
|                                  | %     | SE             | %            |              | SE            | %           | SE                         | %             | SE              |  |  |  |
| English-second-language program* | 17.8  | (0.5)          | 11.          | 1            | (0.5)         | 22.9        | (1.5)                      | 18.0          | (0.9)           |  |  |  |
| French-second-language program** | 25.3  | (1.0)          | 12.          | 2            | (0.4)         | 12.7        | (0.7)                      | 5.9           | (0.4)           |  |  |  |

\* These include English-language-learners (ELL), English-as-an-additional-language (EAL), and English-as-a-second-language (ESL) programs.

\*\* These include extended and intensive French.

*Note:* Although students were also asked about second-language programs in Indigenous and other languages, the response rate was too low for the results to be valid.

## TABLE A.1.11 Mathematics achievement by second-language learning status and language of school system, Canada

|                            | Enrolment in a<br>second-language<br>program | Mean score | SE    |
|----------------------------|--|------------|-------|
| Anglophone school systems  | Currently enrolled                           | 471*       | (3.3) |
|                            | Previously enrolled                          | 492*       | (3.8) |
|                            | Never enrolled                               | 514        | (2.2) |
| Francophone school systems | Currently enrolled                           | 536        | (5.6) |
|                            | Previously enrolled                          | 515*       | (6.0) |
|                            | Never enrolled                               | 537        | (3.4) |

\* Significant difference compared to the never enrolled category

### TABLE A.1.12 Percentage of students by their parents' education as reported by students

|                           | Univo<br>degr | ersity<br>ee(s) | Comp<br>colle<br>cé | oleted<br>ge or<br>gep | So<br>postse | ome<br>econdary | Comp<br>high s | oleted | Did<br>com<br>high s | not<br>plete<br>chool | l do<br>kn | on't<br>ow |
|---------------------------|---------------|-----------------|---------------------|------------------------|--------------|-----------------|----------------|--------|----------------------|-----------------------|------------|------------|
| Canada and provinces      | %             | SE              | %                   | SE                     | %            | SE              | %              | SE     | %                    | SE                    | %          | SE         |
| British Columbia          | 47.7          | (1.4)           | 13.0                | (0.7)                  | 3.7          | (0.3)           | 8.5            | (0.6)  | 3.1                  | (0.2)                 | 24.1       | (0.9)      |
| Alberta                   | 45.3          | (1.7)           | 13.4                | (0.8)                  | 3.2          | (0.3)           | 8.5            | (0.8)  | 3.9                  | (0.6)                 | 25.7       | (1.0)      |
| Saskatchewan              | 40.0          | (1.5)           | 13.7                | (0.7)                  | 5.2          | (0.4)           | 11.4           | (0.7)  | 5.3                  | (0.5)                 | 24.4       | (1.0)      |
| Manitoba                  | 38.5          | (1.3)           | 12.7                | (0.6)                  | 3.5          | (0.3)           | 13.6           | (0.7)  | 5.1                  | (0.4)                 | 26.6       | (0.8)      |
| Ontario                   | 47.3          | (1.7)           | 17.7                | (0.9)                  | 3.7          | (0.4)           | 6.3            | (0.6)  | 2.5                  | (0.3)                 | 22.4       | (1.0)      |
| Quebec                    | 44.5          | (1.4)           | 12.6                | (0.6)                  | 3.3          | (0.3)           | 8.6            | (0.6)  | 3.5                  | (0.4)                 | 27.5       | (0.8)      |
| New Brunswick             | 42.9          | (0.0)           | 16.0                | (0.0)                  | 4.0          | (0.0)           | 10.9           | (0.0)  | 3.4                  | (0.0)                 | 22.7       | (0.0)      |
| Nova Scotia               | 45.7          | (0.2)           | 17.4                | (0.1)                  | 3.1          | (0.1)           | 9.4            | (0.1)  | 4.2                  | (0.1)                 | 20.2       | (0.1)      |
| Prince Edward Island      | 45.4          | (0.0)           | 16.5                | (0.0)                  | 2.1‡         | (0.0)           | 14.9           | (0.0)  | 6.5‡                 | (0.0)                 | 14.6       | (0.0)      |
| Newfoundland and Labrador | 42.1          | (0.6)           | 16.9                | (0.3)                  | 2.7          | (0.2)           | 9.3            | (0.2)  | 3.4                  | (0.2)                 | 25.6       | (0.4)      |
| Canada                    | 45.7          | (0.8)           | 15.2                | (0.4)                  | 3.6          | (0.2)           | 8.0            | (0.3)  | 3.2                  | (0.2)                 | 24.2       | (0.5)      |

**‡** There are fewer than 30 observations.

Note: "Some postsecondary" refers to any kind of education after high school.

|                              |              |                 |                      |                                  | Anglo  | phone sch                    | nool syst | ems       |      |                          |         |       |
|------------------------------|--------------|-----------------|----------------------|----------------------------------|--------|------------------------------|-----------|-----------|------|--------------------------|---------|-------|
|                              | Univ<br>degr | ersity<br>ee(s) | Comp<br>colle<br>cég | Completed<br>college or<br>cégep |        | Some Com<br>stsecondary high |           | Completed |      | l not<br>plete<br>school | I don't | know  |
| Canada and provinces         | %            | SE              | %                    | SE                               | %      | SE                           | %         | SE        | %    | SE                       | %       | SE    |
| British Columbia             | 47.7         | (1.4)           | 13.1                 | (0.7)                            | 3.7    | (0.3)                        | 8.5       | (0.6)     | 3.1  | (0.2)                    | 24.0    | (0.9) |
| Alberta                      | 45.3         | (1.7)           | 13.4                 | (0.8)                            | 3.2    | (0.3)                        | 8.6       | (0.8)     | 3.9  | (0.6)                    | 25.5    | (1.0) |
| Saskatchewan                 | 39.9         | (1.5)           | 13.8                 | (0.7)                            | 5.2    | (0.4)                        | 11.5      | (0.7)     | 5.4  | (0.5)                    | 24.4    | (1.0) |
| Manitoba                     | 38.2         | (1.3)           | 12.9                 | (0.6)                            | 3.5    | (0.3)                        | 13.7      | (0.7)     | 5.2  | (0.4)                    | 26.5    | (0.8) |
| Ontario                      | 47.1         | (1.8)           | 18.0                 | (1.0)                            | 3.9    | (0.4)                        | 6.5       | (0.6)     | 2.6  | (0.4)                    | 21.9    | (1.0) |
| Quebec                       | 49.6         | (1.7)           | 12.9                 | (0.6)                            | 7.0    | (0.7)                        | 7.7       | (0.8)     | 3.1  | (0.4)                    | 19.8    | (1.1) |
| New Brunswick                | 44.5         | (0.0)           | 16.7                 | (0.0)                            | 4.8    | (0.0)                        | 12.8      | (0.0)     | 3.5  | (0.0)                    | 17.7    | (0.0) |
| Nova Scotia                  | 45.5         | (0.2)           | 17.7                 | (0.1)                            | 3.2    | (0.1)                        | 9.5       | (0.1)     | 4.3  | (0.1)                    | 19.7    | (0.1) |
| Prince Edward Island         | 45.5         | (0.0)           | 16.2                 | (0.0)                            | 2.2‡   | (0.0)                        | 15.2      | (0.0)     | 6.7‡ | (0.0)                    | 14.2    | (0.0) |
| Newfoundland and<br>Labrador | 42.1         | (0.6)           | 17.0                 | (0.3)                            | 2.7    | (0.2)                        | 9.3       | (0.2)     | 3.4  | (0.2)                    | 25.6    | (0.4) |
| Canada                       | 46.1         | (1.0)           | 15.9                 | (0.5)                            | 3.8    | (0.2)                        | 8.0       | (0.4)     | 3.2  | (0.2)                    | 23.0    | (0.6) |
|                              |              |                 |                      |                                  | Franco | phone scl                    | hool sys  | tems      |      |                          |         |       |
|                              | Univ         | orcity          | Comp                 | pleted                           | 50     | <b>m</b> 0                   | Com       | alatad    | Did  | not                      |         |       |

## TABLE A.1.13 Percentage of students by their parents' education as reported by students, and language of school system

|                      |              |                 |                      |                                  | Tranco | phone sei                 | 1001 393 | Completed<br>high school         Did not<br>complete<br>high school         I don't know           %         SE         %         SE         %         SE           5.5‡         (0.0)         1.2‡         (0.0)         37.5         (0.0)           U‡         (1.2)         2.5‡         (0.5)         34.6         (1.7)           5.9‡         (0.0)         2.1‡         (0.0)         33.0         (0.0) |      |                                    |       |       |  |  |  |  |  |  |
|----------------------|--------------|-----------------|----------------------|----------------------------------|--------|---------------------------|----------|--|------|------------------------------------|-------|-------|--|--|--|--|--|--|
|                      | Univ<br>degr | ersity<br>ee(s) | Comp<br>colle<br>cég | Completed<br>college or<br>cégep |        | Some C<br>oostsecondary h |          | Completed<br>high school   |      | Did not<br>complete<br>high school |       | know  |  |  |  |  |  |  |
| Canada and provinces | %            | SE              | %                    | SE                               | %      | SE                        | %        | SE   | %    | SE                                 | %     | SE    |  |  |  |  |  |  |
| British Columbia     | 49.5         | (0.0)           | 4.3‡                 | (0.0)                            | 2.0‡   | (0.0)                     | 5.5‡     | (0.0)  | 1.2‡ | (0.0)                              | 37.5  | (0.0) |  |  |  |  |  |  |
| Alberta              | 46.5         | (1.8)           | 11.7‡                | (0.9)                            | U‡     | (0.6)                     | U‡       | (1.2)  | 2.5‡ | (0.5)                              | 34.6  | (1.7) |  |  |  |  |  |  |
| Saskatchewan         | 52.9         | (0.0)           | 9.0‡                 | (0.0)                            | 1.1‡   | (0.0)                     | 5.9‡     | (0.0)  | 2.3‡ | (0.0)                              | 28.7‡ | (0.0) |  |  |  |  |  |  |
| Manitoba             | 48.9         | (0.0)           | 7.8‡                 | (0.0)                            | 1.7‡   | (0.0)                     | 6.6‡     | (0.0)  | 2.1‡ | (0.0)                              | 33.0  | (0.0) |  |  |  |  |  |  |
| Ontario              | 49.4         | (1.4)           | 12.7                 | (0.7)                            | 1.7    | (0.2)                     | 3.6      | (0.5)  | 1.4  | (0.2)                              | 31.2  | (0.9) |  |  |  |  |  |  |
| Quebec               | 44.0         | (1.6)           | 12.5                 | (0.6)                            | 2.9    | (0.3)                     | 8.7      | (0.6)  | 3.5  | (0.4)                              | 28.4  | (0.8) |  |  |  |  |  |  |
| New Brunswick        | 39.0         | (0.0)           | 14.3                 | (0.0)                            | 2.1‡   | (0.0)                     | 6.4      | (0.0)  | 3.2‡ | (0.0)                              | 35.0  | (0.0) |  |  |  |  |  |  |
| Nova Scotia          | 50.3         | (1.2)           | 9.1‡                 | (0.5)                            | U‡     | (0.1)                     | 8.5‡     | (1.1)  | 2.2‡ | (0.4)                              | 29.6  | (1.6) |  |  |  |  |  |  |
| Canada               | 44.5         | (1.4)           | 12.5                 | (0.6)                            | 2.7    | (0.3)                     | 8.1      | (0.5)  | 3.3  | (0.4)                              | 28.9  | (0.7) |  |  |  |  |  |  |

**‡** There are fewer than 30 observations.

U Too unreliable to be published

Note: "Some postsecondary" refers to any kind of education after high school.

#### TABLE A.1.14 Mathematics achievement by parents' education as reported by students, Canada

|  | Mean<br>score | SE    |
|--|---------------|-------|
| Did not complete high school                       | 462           | (6.2) |
| Completed high school                              | 465           | (2.8) |
| Had some postsecondary education after high school | 496*          | (4.6) |
| Completed education at a college or cégep          | 508*          | (4.0) |
| Completed one or more university degrees           | 536*          | (1.9) |
| I don't know                                       | 486*          | (2.3) |

\* Significant difference compared to the completed high school category

#### TABLE A.1.15 Percentage of students by the number of books in their home

|                           | 0–10 | books | 11–25<br>books |       | 26-<br>bo | 100<br>oks | 101–200<br>books |       | More th<br>200 boo |       |
|---------------------------|------|-------|----------------|-------|-----------|------------|------------------|-------|--------------------|-------|
| Canada and provinces      | %    | SE    | %              | SE    | %         | SE         | %                | SE    | %                  | SE    |
| British Columbia          | 8.8  | (0.7) | 12.9           | (0.6) | 33.9      | (0.9)      | 20.8             | (0.7) | 23.6               | (1.0) |
| Alberta                   | 8.5  | (0.8) | 13.4           | (0.9) | 32.4      | (1.0)      | 23.5             | (0.9) | 22.2               | (1.1) |
| Saskatchewan              | 10.9 | (0.8) | 16.1           | (0.9) | 30.2      | (0.9)      | 22.0             | (0.9) | 20.8               | (0.9) |
| Manitoba                  | 12.2 | (0.7) | 15.2           | (0.8) | 30.7      | (0.6)      | 19.0             | (0.7) | 22.9               | (0.9) |
| Ontario                   | 8.9  | (0.7) | 15.9           | (0.9) | 34.5      | (1.0)      | 20.5             | (0.8) | 20.2               | (1.1) |
| Quebec                    | 13.8 | (0.8) | 18.6           | (0.7) | 35.1      | (0.9)      | 17.5             | (0.6) | 14.9               | (0.7) |
| New Brunswick             | 10.7 | (0.0) | 14.6           | (0.0) | 30.9      | (0.0)      | 20.4             | (0.0) | 23.3               | (0.0) |
| Nova Scotia               | 9.2  | (0.1) | 13.7           | (0.1) | 30.5      | (0.1)      | 22.5             | (0.1) | 24.1               | (0.2) |
| Prince Edward Island      | 7.8  | (0.0) | 12.5           | (0.0) | 34.5      | (0.0)      | 18.3             | (0.0) | 26.9               | (0.0) |
| Newfoundland and Labrador | 8.9  | (0.3) | 13.8           | (0.2) | 37.2      | (0.5)      | 19.8             | (0.3) | 20.3               | (0.4) |
| Canada                    | 10.1 | (0.4) | 15.6           | (0.4) | 33.9      | (0.5)      | 20.3             | (0.4) | 20.1               | (0.5) |

|                           | 0–10  | books | 11–25 | books | 26 –100 | ) books | 101–200 | 0 books | More<br>200 b | than<br>books |
|---------------------------|-------|-------|-------|-------|---------|---------|---------|---------|---------------|---------------|
|                           |       |       |       |       | Angle   | ophone  |         |         |               |               |
| Canada and provinces      | %     | SE    | %     | SE    | %       | SE      | %       | SE      | %             | SE            |
| British Columbia          | 8.8   | (0.7) | 12.9  | (0.6) | 33.9    | (0.9)   | 20.8    | (0.7)   | 23.6          | (1.0)         |
| Alberta                   | 8.4   | (0.8) | 13.4  | (0.9) | 32.5    | (1.0)   | 23.4    | (0.9)   | 22.2          | (1.1)         |
| Saskatchewan              | 10.9  | (0.8) | 16.1  | (0.9) | 30.2    | (0.9)   | 22.0    | (0.9)   | 20.8          | (0.9)         |
| Manitoba                  | 12.5  | (0.7) | 15.3  | (0.8) | 30.7    | (0.7)   | 18.7    | (0.7)   | 22.8          | (0.9)         |
| Ontario                   | 8.9   | (0.7) | 15.8  | (1.0) | 34.6    | (1.0)   | 20.5    | (0.9)   | 20.3          | (1.1)         |
| Quebec                    | 8.4   | (0.8) | 13.0  | (1.0) | 30.4    | (1.0)   | 25.7    | (1.1)   | 22.4          | (0.8)         |
| New Brunswick             | 7.8   | (0.0) | 12.3  | (0.0) | 30.2    | (0.0)   | 22.5    | (0.0)   | 27.3          | (0.0)         |
| Nova Scotia               | 9.1   | (0.1) | 13.5  | (0.1) | 30.6    | (0.1)   | 22.4    | (0.1)   | 24.3          | (0.1)         |
| Prince Edward Island      | 7.7‡  | (0.0) | 12.7  | (0.0) | 34.7    | (0.0)   | 17.7    | (0.0)   | 27.1          | (0.0)         |
| Newfoundland and Labrador | 8.9   | (0.3) | 13.8  | (0.2) | 37.2    | (0.5)   | 19.8    | (0.3)   | 20.4          | (0.4)         |
| Canada                    | 9.0   | (0.4) | 14.7  | (0.5) | 33.5    | (0.6)   | 21.2    | (0.5)   | 21.6          | (0.6)         |
|                           |       |       |       |       | Franc   | ophone  |         |         |               |               |
| Canada and provinces      | %     | SE    | %     | SE    | %       | SE      | %       | SE      | %             | SE            |
| British Columbia          | 10.8‡ | (0.0) | 12.5  | (0.0) | 28.3    | (0.0)   | 22.9    | (0.0)   | 25.5          | (0.0)         |
| Alberta                   | 11.8‡ | (1.8) | 12.1‡ | (1.8) | 25.4    | (2.8)   | 29.1    | (2.7)   | 21.7          | (2.3)         |
| Saskatchewan              | 5.7‡  | (0.0) | 16.4‡ | (0.0) | 28.0‡   | (0.0)   | 23.3‡   | (0.0)   | 26.5‡         | (0.0)         |
| Manitoba                  | 4.1‡  | (0.0) | 10.7  | (0.0) | 31.0    | (0.0)   | 27.5    | (0.0)   | 26.7          | (0.0)         |
| Ontario                   | 10.4  | (0.6) | 17.5  | (1.0) | 33.1    | (0.9)   | 20.1    | (0.7)   | 18.9          | (1.1)         |
| Quebec                    | 14.4  | (0.9) | 19.3  | (0.8) | 35.6    | (1.0)   | 16.6    | (0.7)   | 14.1          | (0.8)         |
| New Brunswick             | 18.1  | (0.0) | 20.3  | (0.0) | 32.9    | (0.0)   | 15.2    | (0.0)   | 13.6          | (0.0)         |
| Nova Scotia               | 10.5‡ | (1.0) | 18.1‡ | (0.8) | 28.9    | (1.3)   | 22.9    | (1.4)   | 19.6‡         | (1.1)         |
| Prince Edward Island      |       |       |       |       |         |         |         |         |               |               |
| Newfoundland and Labrador |       |       |       |       |         |         |         |         |               |               |
| Canada                    | 14.0  | (0.7) | 19.0  | (0.7) | 35.1    | (0.8)   | 17.1    | (0.6)   | 14.7          | (0.7)         |

## TABLE A.1.16 Percentage of students by the number of books in their home and language of school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

#### TABLE A.1.17 Mathematics achievement by the number of books in the home, Canada

|                     | Mean score | SE    |
|---------------------|------------|-------|
| 0–10 books          | 463        | (2.9) |
| 11–25 books         | 485*       | (3.2) |
| 26–100 books        | 510*       | (2.1) |
| 101–200 books       | 527*       | (2.5) |
| More than 200 books | 535*       | (2.5) |

\* Significant difference compared to the 0–10 books category

|                           | Born in | Canada | Not born in | Canada |
|---------------------------|---------|--------|-------------|--------|
| Canada and provinces      | %       | SE     | %           | SE     |
| British Columbia          | 82.2    | (1.0)  | 17.8        | (1.0)  |
| Alberta                   | 82.0    | (1.4)  | 18.0        | (1.4)  |
| Saskatchewan              | 84.3    | (1.3)  | 15.7        | (1.3)  |
| Manitoba                  | 79.8    | (1.3)  | 20.2        | (1.3)  |
| Ontario                   | 84.7    | (1.3)  | 15.3        | (1.3)  |
| Quebec                    | 90.8    | (0.8)  | 9.2         | (0.8)  |
| New Brunswick             | 91.9    | (0.0)  | 8.1         | (0.0)  |
| Nova Scotia               | 91.8    | (0.1)  | 8.2         | (0.1)  |
| Prince Edward Island      | 93.4    | (0.0)  | 6.6‡        | (0.0)  |
| Newfoundland and Labrador | 94.8    | (0.3)  | 5.2         | (0.3)  |
| Canada                    | 85.5    | (0.6)  | 14.5        | (0.6)  |

#### TABLE A.1.18 Percentage of students by immigrant status

**‡** There are fewer than 30 observations.

#### TABLE A.1.19 Percentage of students by immigrant status and language of school system

|                           | A                 | nglopho | ne school syster | ns             | Fr         | Francophone school systems |              |               |  |  |  |
|---------------------------|-------------------|---------|------------------|----------------|------------|----------------------------|--------------|---------------|--|--|--|
|                           | Born in<br>Canada |         | Not b<br>Car     | orn in<br>Iada | Bor<br>Can | n in<br>ada                | Not b<br>Can | orn in<br>ada |  |  |  |
| Canada and provinces      | %                 | SE      | %                | SE             | %          | SE                         | %            | SE            |  |  |  |
| British Columbia          | 82.2              | (1.0)   | 17.8             | (1.0)          | 82.5       | (0.0)                      | 17.5         | (0.0)         |  |  |  |
| Alberta                   | 82.0              | (1.4)   | 18.0             | (1.4)          | 76.4       | (3.1)                      | 23.6         | (3.1)         |  |  |  |
| Saskatchewan              | 84.3              | (1.3)   | 15.7             | (1.3)          | 82.3       | (0.0)                      | 17.7‡        | (0.0)         |  |  |  |
| Manitoba                  | 79.6              | (1.3)   | 20.4             | (1.3)          | 87.5       | (0.0)                      | 12.5         | (0.0)         |  |  |  |
| Ontario                   | 84.6              | (1.3)   | 15.4             | (1.3)          | 86.0       | (1.2)                      | 14.0         | (1.2)         |  |  |  |
| Quebec                    | 91.9              | (0.9)   | 8.1              | (0.9)          | 90.7       | (0.9)                      | 9.3          | (0.9)         |  |  |  |
| New Brunswick             | 90.8              | (0.0)   | 9.2              | (0.0)          | 94.7       | (0.0)                      | 5.3          | (0.0)         |  |  |  |
| Nova Scotia               | 91.7              | (0.1)   | 8.3              | (0.1)          | 93.9       | (0.5)                      | 6.1‡         | (0.5)         |  |  |  |
| Prince Edward Island      | 93.4              | (0.0)   | 6.6‡             | (0.0)          |            |                            |              |               |  |  |  |
| Newfoundland and Labrador | 94.8              | (0.3)   | 5.2              | (0.3)          |            |                            |              |               |  |  |  |
| Canada                    | 84.3              | (0.7)   | 15.7             | (0.7)          | 90.2       | (0.8)                      | 9.8          | (0.8)         |  |  |  |

**‡** There are fewer than 30 observations.

|                    | Mean<br>score | SE    |
|--------------------|---------------|-------|
| Born in Canada     | 511           | (1.9) |
| Not born in Canada | 503*          | (3.3) |

### TABLE A.1.20 Mathematics achievement by immigrant status, Canada

\* Significant difference

|                              | 0            |             |         |        | 0    |       | -    | •     |                         |                                       |      |                               |
|------------------------------|--------------|-------------|---------|--------|------|-------|------|-------|-------------------------|---------------------------------------|------|-------------------------------|
|                              | Ne<br>Indige | ot<br>enous | First N | ations | In   | uit   | Mé   | étis  | Indige<br>mult<br>ident | Indigenous,<br>multiple<br>identities |      | oth<br>enous<br>non-<br>enous |
| Canada and provinces         | %            | SE          | %       | SE     | %    | SE    | %    | SE    | %                       | SE                                    | %    | SE                            |
| British Columbia             | 90.6         | (0.8)       | 5.7     | (0.6)  | U‡   | (0.1) | 2.4  | (0.3) | 0.3‡                    | (0.1)                                 | 0.8‡ | (0.1)                         |
| Alberta                      | 90.1         | (1.1)       | 4.2     | (0.9)  | U‡   | (0.1) | 3.8  | (0.4) | 1.0‡                    | (0.2)                                 | 0.7‡ | (0.2)                         |
| Saskatchewan                 | 79.4         | (1.6)       | 10.9    | (1.3)  | U‡   | (0.1) | 7.5  | (0.6) | 1.2‡                    | (0.2)                                 | 0.9‡ | (0.2)                         |
| Manitoba                     | 79.7         | (1.4)       | 9.4     | (0.9)  | U‡   | (0.1) | 9.4  | (0.8) | 0.9‡                    | (0.2)                                 | 0.4‡ | (0.1)                         |
| Ontario                      | 94.8         | (0.7)       | 3.6     | (0.6)  | U‡   | (0.1) | 0.9  | (0.2) | U‡                      | (0.1)                                 | 0.4‡ | (0.1)                         |
| Quebec                       | 93.1         | (0.5)       | 3.0     | (0.4)  | 0.6‡ | (0.1) | 1.9  | (0.2) | U‡                      | (0.1)                                 | 1.1  | (0.2)                         |
| New Brunswick                | 91.9         | (0.0)       | 5.6     | (0.0)  | 0.1‡ | (0.0) | 1.2‡ | (0.0) | 0.4‡                    | (0.0)                                 | 0.9‡ | (0.0)                         |
| Nova Scotia                  | 89.1         | (0.1)       | 6.7     | (0.1)  | 0.3‡ | (0.0) | 2.4  | (0.0) | 0.7‡                    | (0.0)                                 | 0.8‡ | (0.0)                         |
| Prince Edward Island         | 95.4         | (0.0)       | 3.6‡    | (0.0)  | 0.2‡ | (0.0) | 0.3‡ | (0.0) | 0.0                     | (0.0)                                 | 0.4‡ | (0.0)                         |
| Newfoundland and<br>Labrador | 87.5         | (0.5)       | 6.5     | (0.3)  | 2.4‡ | (0.1) | 1.6‡ | (0.1) | 0.3‡                    | (0.0)                                 | 1.6‡ | (0.1)                         |
| Canada                       | 92.0         | (0.4)       | 4.4     | (0.3)  | 0.3  | (0.0) | 2.3  | (0.1) | 0.4                     | (0.0)                                 | 0.7  | (0.1)                         |
| ‡ There are fewer than 30 ob | servations.  |             |         |        |      |       |      |       |                         |                                       |      |                               |

#### TABLE A.1.21 Percentage of students by their Indigenous identity as reported by students

U Too unreliable to be published

|                           | Anglophone school systems |             |               |       |        |         |          |       |      |                                       |      |                               |
|---------------------------|---------------------------|-------------|---------------|-------|--------|---------|----------|-------|------|---------------------------------------|------|-------------------------------|
|                           | N<br>Indige               | ot<br>enous | First Nations |       | Inuit  |         | Mé       | Métis |      | Indigenous,<br>multiple<br>identities |      | oth<br>enous<br>non-<br>enous |
| Canada and provinces      | %                         | SE          | %             | SE    | %      | SE      | %        | SE    | %    | SE                                    | %    | SE                            |
| British Columbia          | 90.6                      | (0.8)       | 5.7           | (0.6) | U‡     | (0.1)   | 2.4      | (0.3) | 0.3‡ | (0.1)                                 | 0.8‡ | (0.1)                         |
| Alberta                   | 90.1                      | (1.2)       | 4.3           | (0.9) | U‡     | (0.1)   | 3.8      | (0.4) | 1.0‡ | (0.2)                                 | 0.7‡ | (0.2)                         |
| Saskatchewan              | 79.4                      | (1.7)       | 11.0          | (1.3) | U‡     | (0.1)   | 7.4      | (0.6) | 1.2‡ | (0.2)                                 | 0.9‡ | (0.2)                         |
| Manitoba                  | 79.9                      | (1.4)       | 9.6           | (1.0) | U‡     | (0.1)   | 9.1      | (0.8) | 0.9‡ | (0.2)                                 | 0.4‡ | (0.1)                         |
| Ontario                   | 94.9                      | (0.7)       | 3.6           | (0.6) | U‡     | (0.1)   | 0.8‡     | (0.2) | U‡   | (0.1)                                 | 0.4‡ | (0.1)                         |
| Quebec                    | 92.1                      | (0.7)       | 4.8           | (0.5) | 0.8‡   | (0.2)   | 1.2‡     | (0.2) | U‡   | (0.0)                                 | 1.0‡ | (0.2)                         |
| New Brunswick             | 92.1                      | (0.0)       | 6.1           | (0.0) | 0.0‡   | (0.0)   | 0.6‡     | (0.0) | 0.2‡ | (0.0)                                 | 0.9‡ | (0.0)                         |
| Nova Scotia               | 89.1                      | (0.1)       | 6.8           | (0.1) | 0.3‡   | (0.0)   | 2.3      | (0.0) | 0.7‡ | (0.0)                                 | 0.8‡ | (0.0)                         |
| Prince Edward Island      | 95.6                      | (0.0)       | 3.5‡          | (0.0) | 0.2‡   | (0.0)   | 0.2‡     | (0.0) | 0.0  | (0.0)                                 | 0.4‡ | (0.0)                         |
| Newfoundland and Labrador | 87.5                      | (0.5)       | 6.5           | (0.3) | 2.4‡   | (0.1)   | 1.6‡     | (0.1) | 0.3‡ | (0.0)                                 | 1.6‡ | (0.1)                         |
| Canada                    | 91.7                      | (0.4)       | 4.9           | (0.4) | 0.2    | (0.0)   | 2.2      | (0.1) | 0.4  | (0.1)                                 | 0.6  | (0.1)                         |
|                           |                           |             |               |       | Franco | phone s | chool sy | stems |      |                                       |      |                               |

#### TABLE A.1.22 Percentage of students by their Indigenous identity as reported by students and language of school system

|                      | N<br>Indige | ot<br>enous | First Nations |       | Inuit |       | Métis |       | Indigenous,<br>multiple<br>identities |       | Both<br>Indigenous<br>and non-<br>Indigenous |       |
|----------------------|-------------|-------------|---------------|-------|-------|-------|-------|-------|---------------------------------------|-------|--|-------|
| Canada and provinces | %           | SE          | %             | SE    | %     | SE    | %     | SE    | %                                     | SE    | %  | SE    |
| British Columbia     | 92.0        | (0.0)       | 2.4‡          | (0.0) | 0.0   | (0.0) | 4.8‡  | (0.0) | 0.4‡                                  | (0.0) | 0.4‡   | (0.0) |
| Alberta              | 96.4        | (0.9)       | U‡            | (0.3) | U‡    | (0.3) | U‡    | (0.3) | 1.2‡                                  | (0.4) | U‡   | (0.7) |
| Saskatchewan         | 86.1        | (0.0)       | 0.0           | (0.0) | 0.0   | (0.0) | 12.8‡ | (0.0) | 0.0                                   | (0.0) | 1.1‡   | (0.0) |
| Manitoba             | 73.1        | (0.0)       | 0.3‡          | (0.0) | 0.6‡  | (0.0) | 23.7  | (0.0) | 1.1‡                                  | (0.0) | 1.1‡   | (0.0) |
| Ontario              | 92.3        | (0.8)       | 2.5           | (0.4) | U‡    | (0.1) | 4.1   | (0.4) | 0.4‡                                  | (0.1) | 0.6‡   | (0.1) |
| Quebec               | 93.2        | (0.6)       | 2.7           | (0.4) | 0.6‡  | (0.2) | 2.0   | (0.2) | U‡                                    | (0.1) | 1.1  | (0.2) |
| New Brunswick        | 91.3        | (0.0)       | 4.3           | (0.0) | 0.3‡  | (0.0) | 2.5‡  | (0.0) | 0.7‡                                  | (0.0) | 1.0‡   | (0.0) |
| Nova Scotia          | 87.9        | (0.9)       | 5.8‡          | (0.4) | 1.6‡  | (0.4) | 2.8‡  | (0.3) | 1.6‡                                  | (0.4) | 0.4‡   | (0.1) |
| Canada               | 93.0        | (0.5)       | 2.7           | (0.4) | 0.6‡  | (0.1) | 2.3   | (0.2) | 0.3                                   | (0.1) | 1.1  | (0.2) |

**‡** There are fewer than 30 observations.

U Too unreliable to be published

|                                    | Mean score | SE    |
|------------------------------------|------------|-------|
| Not Indigenous                     | 515*       | (1.9) |
| First Nations                      | 446        | (5.3) |
| Inuit                              | 473*       | (8.8) |
| Métis                              | 484*       | (4.2) |
| Indigenous, multiple identities    | 457        | (7.4) |
| Both Indigenous and non-Indigenous | 487*       | (9.1) |

### TABLE A.1.23 Mathematics achievement by Indigenous identity as reported by students, Canada

\* Significant difference compared to the First Nations category

## TABLE A.2.1.1 Distribution of students by questionnaire items constituting the attitude toward mathematics index, Canada

|  | Strongly disagree <sup>1</sup> |       |               | Disagree <sup>2</sup> |                             |       |               |       |
|--|--------------------------------|-------|---------------|-----------------------|-----------------------------|-------|---------------|-------|
|  | %                              | SE    | Mean<br>score | SE                    | %                           | SE    | Mean<br>score | SE    |
| Strong correlation with mathematics achievement (> 0.2)                              |                                |       |               |                       |                             |       |               |       |
| I understand most of the mathematics I am taught                                     | 5.3                            | (0.2) | 439*          | (3.1)                 | 11.5                        | (0.3) | 455*          | (2.3) |
| I like problem solving   | 12.5                           | (0.3) | 467*          | (2.3)                 | 25.9                        | (0.4) | 483*          | (2.6) |
| I like doing mental mathematics  | 17.6                           | (0.4) | 482*          | (2.3)                 | 31.9                        | (0.5) | 495*          | (2.5) |
| Weak or no correlation with mathematics achievement (< 0.2)                          |                                |       |               |                       |                             |       |               |       |
| I like doing paper-pencil calculations   | 13.8                           | (0.4) | 482*          | (2.4)                 | 24.0                        | (0.4) | 496*          | (2.5) |
| I need to keep taking mathematics for the kind of job I want after<br>I leave school | 8.4                            | (0.3) | 477*          | (3.2)                 | 19.6                        | (0.4) | 498*          | (2.2) |
| The mathematics I learn now will be useful in my everyday life                       | 11.7                           | (0.3) | 486*          | (2.7)                 | 26.4                        | (0.4) | 508*          | (2.0) |
| I like hands-on mathematics activities   | 11.2                           | (0.3) | 487*          | (2.9)                 | 23.2                        | (0.4) | 508*          | (2.7) |
| I like mathematics questions that involve a lot of reading                           | 34.0                           | (0.5) | 500*          | (1.9)                 | 43.0                        | (0.5) | 515*          | (2.3) |
| I like to write or explain the methods or procedures I use in mathematics            | 20.9                           | (0.3) | 504*          | (2.5)                 | 36.6                        | (0.5) | 509*          | (2.1) |
| I like estimating  | 18.1                           | (0.4) | 503*          | (2.7)                 | 37.7                        | (0.5) | 513*          | (2.4) |
|  |                                | Ag    | ree³          |                       | Strongly agree <sup>₄</sup> |       |               |       |
|  | %                              | SE    | Mean<br>score | SE                    | %                           | SE    | Mean<br>score | SE    |
| Strong correlation with mathematics achievement (> 0.2)                              |                                |       |               |                       |                             |       |               |       |
| I understand most of the mathematics I am taught                                     | 48.7                           | (0.5) | 494*          | (2.0)                 | 34.4                        | (0.6) | 562*          | (2.1) |
| I like problem solving   | 45.8                           | (0.5) | 520*          | (1.9)                 | 15.9                        | (0.4) | 559*          | (2.9) |
| I like doing mental mathematics  | 34.5                           | (0.4) | 520*          | (2.2)                 | 16.0                        | (0.4) | 550*          | (2.4) |
| Weak or no correlation with mathematics achievement (< 0.2)                          |                                |       |               |                       |                             |       |               |       |
| I like doing paper-pencil calculations   | 43.3                           | (0.5) | 514*          | (2.0)                 | 18.8                        | (0.4) | 542*          | (2.9) |
| I need to keep taking mathematics for the kind of job I want after<br>I leave school | 43.0                           | (0.4) | 509*          | (2.2)                 | 29.0                        | (0.5) | 530*          | (2.4) |
| The mathematics I learn now will be useful in my everyday life                       | 46.1                           | (0.5) | 514*          | (2.4)                 | 15.8                        | (0.4) | 522*          | (2.7) |
| I like hands-on mathematics activities   | 46.1                           | (0.5) | 513           | (1.9)                 | 19.6                        | (0.4) | 520*          | (3.0) |
| I like mathematics questions that involve a lot of reading                           | 18.5                           | (0.4) | 519           | (3.1)                 | 4.6                         | (0.2) | 513           | (5.2) |
| I like to write or explain the methods or procedures I use in mathematics            | 32.2                           | (0.5) | 511           | (2.6)                 | 10.3                        | (0.3) | 528*          | (3.6) |
| I like estimating  | 35.9                           | (0.5) | 510           | (2.0)                 | 8.3                         | (0.3) | 516           | (3.4) |
| * Significant difference between:  |                                |       |               |                       |                             |       |               |       |

<sup>1</sup> "Strongly disagree" and "Strongly agree"

<sup>2</sup> "Disagree" and "Strongly disagree"

<sup>3</sup> "Agree" and "Disagree"

<sup>4</sup> "Strongly agree" and "Agree"

*Note:* Items have been sorted in descending order by correlation with mathematics achievement.

#### TABLE A.2.1.2 Attitude toward mathematics index scores

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 49.0*       | (0.2) |
| Alberta                   | 49.8        | (0.3) |
| Saskatchewan              | 50.7        | (0.2) |
| Manitoba                  | 50.4        | (0.2) |
| Ontario                   | 51.9*       | (0.2) |
| Quebec                    | 47.2*       | (0.3) |
| New Brunswick             | 50.0        | (0.0) |
| Nova Scotia               | 48.8*       | (0.0) |
| Prince Edward Island      | 48.2*       | (0.0) |
| Newfoundland and Labrador | 47.6*       | (0.1) |
| Canada                    | 50.0        | (0.1) |

\* Significant difference compared to Canada

# TABLE A.2.1.3 Attitude toward mathematics index scores by language of the school system Anglophone school systems Francophone school systems Difference

| Canada and provinces      | Anglophone sch | nool systems | Francophone scl | Difference** |         |
|---------------------------|----------------|--------------|-----------------|--------------|---------|
|                           | Index score    | SE           | Index score     | SE           | (A – F) |
| British Columbia          | 49.0*          | (0.2)        | 51.0*           | (0.0)        | -2**    |
| Alberta                   | 49.8*          | (0.3)        | 52.4*           | (0.7)        | -3**    |
| Saskatchewan              | 50.7           | (0.2)        | 53.9*           | (0.0)        | -3**    |
| Manitoba                  | 50.3           | (0.2)        | 52.2*           | (0.0)        | -2**    |
| Ontario                   | 51.8*          | (0.2)        | 52.8*           | (0.2)        | -1**    |
| Quebec                    | 47.4*          | (0.3)        | 47.2*           | (0.3)        | 0       |
| New Brunswick             | 50.1*          | (0.0)        | 49.8*           | (0.0)        | 0**     |
| Nova Scotia               | 48.6*          | (0.0)        | 51.9*           | (0.4)        | -3**    |
| Prince Edward Island      | 48.2*          | (0.0)        |                 |              |         |
| Newfoundland and Labrador | 47.6*          | (0.1)        |                 |              |         |
| Canada                    | 50.6           | (0.1)        | 47.9            | (0.2)        | 3**     |

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

| Canada and provinces      | Femal       | les   | Males       | Males |         |  |  |
|---------------------------|-------------|-------|-------------|-------|---------|--|--|
|                           | Index score | SE    | Index score | SE    | (F – M) |  |  |
| British Columbia          | 48.4*       | (0.2) | 49.5*       | (0.3) | -1**    |  |  |
| Alberta                   | 49.5        | (0.3) | 50.1        | (0.4) | -1      |  |  |
| Saskatchewan              | 50.6*       | (0.3) | 50.8        | (0.3) | 0       |  |  |
| Manitoba                  | 49.9        | (0.2) | 50.9        | (0.3) | -1**    |  |  |
| Ontario                   | 51.2*       | (0.3) | 52.5*       | (0.3) | -1**    |  |  |
| Quebec                    | 47.2*       | (0.3) | 47.2*       | (0.3) | 0       |  |  |
| New Brunswick             | 50.5*       | (0.0) | 49.5*       | (0.0) | 1**     |  |  |
| Nova Scotia               | 48.8*       | (0.0) | 48.8*       | (0.1) | 0       |  |  |
| Prince Edward Island      | 48.1*       | (0.0) | 48.3*       | (0.0) | 0**     |  |  |
| Newfoundland and Labrador | 48.3*       | (0.2) | 46.9*       | (0.2) | 1**     |  |  |
| Canada                    | 49.6        | (0.2) | 50.4        | (0.2) | -1**    |  |  |

#### TABLE A.2.1.4 Attitude toward mathematics index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

#### TABLE A.2.1.5 Mathematics achievement by the attitude toward mathematics index, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second quarter <sup>2</sup> |       | Third quarter <sup>3</sup> |       | Top quarter <sup>₄</sup> |       |
|-------------------------------|-----------------------------|-------|-----------------------------|-------|----------------------------|-------|--------------------------|-------|
|                               | Mean<br>score               | SE    | Mean<br>score               | SE    | Mean<br>score              | SE    | Mean<br>score            | SE    |
| Canada overall                | 477*                        | (2.0) | 502*                        | (2.6) | 524*                       | (2.5) | 541*                     | (2.6) |
| Language of the school system |                             |       |                             |       |                            |       |                          |       |
| Anglophone                    | 464*                        | (2.4) | 494*                        | (3.0) | 518*                       | (2.9) | 534*                     | (3.0) |
| Francophone                   | 508*                        | (3.8) | 534*                        | (4.2) | 551*                       | (4.0) | 570*                     | (4.0) |
| Gender                        |                             |       |                             |       |                            |       |                          |       |
| Female                        | 477*                        | (2.5) | 501*                        | (3.8) | 526*                       | (2.8) | 539*                     | (3.3) |
| Male                          | 477*                        | (2.5) | 504*                        | (2.8) | 522*                       | (3.7) | 542*                     | (3.2) |
|                               |                             |       |                             |       |                            |       |                          |       |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

 $^{\rm 2}$  Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter
|  | Not at all confident <sup>1</sup>                       |       |               | t1    | Sor  | newhat | confider      | nt²   |      | Very co | nfident <sup>3</sup> |       |
|--|---|-------|---------------|-------|------|--------|---------------|-------|------|---------|----------------------|-------|
|  | %   | SE    | Mean<br>score | SE    | %    | SE     | Mean<br>score | SE    | %    | SE      | Mean<br>score        | SE    |
| Strong correlation with                              | Strong correlation with mathematics achievement (> 0.2) |       |               |       |      |        |               |       |      |         |                      |       |
| Paper-pencil<br>calculations                         | 7.3   | (0.3) | 441*          | (3.2) | 49.4 | (0.5)  | 488*          | (1.9) | 43.3 | (0.5)   | 549*                 | (2.0) |
| Problem solving                                      | 11.3  | (0.3) | 448*          | (3.1) | 49.0 | (0.5)  | 493*          | (1.9) | 39.7 | (0.6)   | 550*                 | (2.0) |
| Mental math  | 19.6  | (0.4) | 471*          | (2.8) | 52.1 | (0.5)  | 507*          | (1.9) | 28.3 | (0.5)   | 545*                 | (2.2) |
| Reading to understand<br>the problem to be<br>solved | 11.1  | (0.3) | 459*          | (3.9) | 53.7 | (0.6)  | 504*          | (1.8) | 35.2 | (0.5)   | 538*                 | (2.2) |
| Explaining how I solved a problem                    | 20.2  | (0.4) | 477*          | (2.7) | 51.1 | (0.5)  | 509*          | (1.9) | 28.7 | (0.5)   | 537*                 | (2.3) |
| Estimation   | 16.9  | (0.4) | 478*          | (3.0) | 55.3 | (0.5)  | 508*          | (2.0) | 27.8 | (0.5)   | 536*                 | (2.3) |

#### TABLE A.2.2.1 Distribution of students by questionnaire items constituting the confidence with mathematical processes index, Canada

\* Significant difference between:

<sup>1</sup> "Not at all confident" and "Very confident"

<sup>2</sup> "Somewhat confident" and "Not at all confident"

<sup>3</sup> "Very confident" and "Somewhat confident"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 49.4        | (0.2) |
| Alberta                   | 50.0        | (0.3) |
| Saskatchewan              | 49.9        | (0.2) |
| Manitoba                  | 50.2        | (0.3) |
| Ontario                   | 51.4*       | (0.3) |
| Quebec                    | 47.9*       | (0.2) |
| New Brunswick             | 49.5*       | (0.0) |
| Nova Scotia               | 49.7        | (0.0) |
| Prince Edward Island      | 50.3        | (0.0) |
| Newfoundland and Labrador | 48.9*       | (0.1) |
| Canada                    | 50.1        | (0.1) |

### TABLE A.2.2.2 Confidence with mathematical processes index scores

\* Significant difference compared to Canada

| Consider and more in a set | Anglophone scho | ol systems | Francophone scho | Francophone school systems |         |  |  |
|----------------------------|-----------------|------------|------------------|----------------------------|---------|--|--|
| Canada and provinces       | Index score     | SE         | Index score      | SE                         | (A – F) |  |  |
| British Columbia           | 49.4*           | (0.2)      | 49.4*            | (0.0)                      | 0       |  |  |
| Alberta                    | 50.0            | (0.3)      | 51.1*            | (0.5)                      | -1      |  |  |
| Saskatchewan               | 49.9            | (0.2)      | 50.3*            | (0.0)                      | 0       |  |  |
| Manitoba                   | 50.2            | (0.3)      | 50.7*            | (0.0)                      | -1      |  |  |
| Ontario                    | 51.4*           | (0.3)      | 50.8*            | (0.2)                      | 1       |  |  |
| Quebec                     | 49.2*           | (0.3)      | 47.8*            | (0.2)                      | 1**     |  |  |
| New Brunswick              | 50.1*           | (0.0)      | 48.0             | (0.0)                      | 2**     |  |  |
| Nova Scotia                | 49.6*           | (0.0)      | 51.4*            | (0.2)                      | -2**    |  |  |
| Prince Edward Island       | 50.4            | (0.0)      |                  |                            |         |  |  |
| Newfoundland and Labrador  | 48.9*           | (0.1)      |                  |                            |         |  |  |
| Canada                     | 50.6            | (0.1)      | 48.1             | (0.2)                      | 2**     |  |  |

TABLE A.2.2.3 Confidence with mathematical processes index scores by language of the school system

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

| Canada and provinces      | Female      | es    | Males       | ;     | Difference** |
|---------------------------|-------------|-------|-------------|-------|--------------|
| -                         | Index score | SE    | Index score | SE    | (F – M)      |
| British Columbia          | 48.1        | (0.3) | 50.8        | (0.3) | -3**         |
| Alberta                   | 48.6        | (0.4) | 51.3        | (0.3) | -3**         |
| Saskatchewan              | 48.5        | (0.3) | 51.2        | (0.3) | -3**         |
| Manitoba                  | 48.8        | (0.3) | 51.5        | (0.3) | -3**         |
| Ontario                   | 50.0*       | (0.4) | 52.8*       | (0.3) | -3**         |
| Quebec                    | 46.3*       | (0.3) | 49.5*       | (0.3) | -3**         |
| New Brunswick             | 48.3        | (0.0) | 50.7*       | (0.0) | -2**         |
| Nova Scotia               | 48.6        | (0.0) | 51.0*       | (0.0) | -2**         |
| Prince Edward Island      | 49.2*       | (0.0) | 51.7        | (0.0) | -3**         |
| Newfoundland and Labrador | 48.4        | (0.2) | 49.3*       | (0.1) | -1**         |
| Canada                    | 48.6        | (0.2) | 51.5        | (0.1) | -3**         |

#### TABLE A.2.2.4 Confidence with mathematical processes index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

#### TABLE A.2.2.5 Mathematics achievement by the confidence with mathematical processes index, Canada

|                               | Bottom<br>quarter <sup>1</sup> |       | Second<br>quarter <sup>2</sup> |       | Third q       | Third quarter <sup>3</sup> |   | Top quarter <sup>₄</sup> |       |
|-------------------------------|--------------------------------|-------|--------------------------------|-------|---------------|----------------------------|---|--------------------------|-------|
|                               | Mean<br>score                  | SE    | Mean<br>score                  | SE    | Mean<br>score | SE                         | 1 | Mean<br>score            | SE    |
| Canada overall                | 467*                           | (2.3) | 491*                           | (2.3) | 529*          | (2.4)                      |   | 559*                     | (2.2) |
| Language of the school system |                                |       |                                |       |               |                            |   |                          |       |
| Anglophone                    | 453*                           | (2.7) | 482*                           | (2.7) | 520*          | (2.8)                      |   | 556*                     | (2.5) |
| Francophone                   | 505*                           | (3.5) | 524*                           | (3.7) | 563*          | (3.8)                      |   | 583*                     | (4.5) |
| Gender                        |                                |       |                                |       |               |                            |   |                          |       |
| Female                        | 469*                           | (2.8) | 498*                           | (2.4) | 529*          | (2.9)                      |   | 560*                     | (3.3) |
| Male                          | 463*                           | (2.7) | 484*                           | (3.2) | 529*          | (3.0)                      |   | 559*                     | (2.6) |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

### TABLE A.2.3.1 Distribution of students by questionnaire items constituting the confidence using technology index, Canada

|   | N    | Not at all confident <sup>1</sup> |               |       | S   | Somewhat confident <sup>2</sup> |               |       |      | Very confident <sup>3</sup> |               |       |  |
|---|------|-----------------------------------|---------------|-------|-----|---------------------------------|---------------|-------|------|-----------------------------|---------------|-------|--|
|   | %    | SE                                | Mean<br>score | SE    | %   | SE                              | Mean<br>score | SE    | %    | SE                          | Mean<br>score | SE    |  |
| Strong correlation with mathematics achievement (> 0.2)     |      |                                   |               |       |     |                                 |               |       |      |                             |               |       |  |
| Using calculators   | 2.5  | (0.2)                             | 434*          | (6.0) | 19. | 2 (0.4)                         | 470*          | (2.5) | 78.3 | (0.5)                       | 523*          | (1.8) |  |
| Weak or no correlation with mathematics achievement (< 0.2) |      |                                   |               |       |     |                                 |               |       |      |                             |               |       |  |
| Using computers   | 5.9  | (0.2)                             | 471*          | (4.2) | 31. | 2 (0.5)                         | 498*          | (2.1) | 62.9 | (0.5)                       | 520*          | (2.0) |  |
| Coding/programming  | 40.7 | (0.5)                             | 507*          | (2.3) | 41. | 0.6)                            | 511*          | (2.2) | 18.4 | (0.4)                       | 518*          | (2.4) |  |

\* Significant difference between:

<sup>1</sup> "Not at all confident" and "Very confident"

<sup>2</sup> "Somewhat confident" and "Not at all confident"

<sup>3</sup> "Very confident" and "Somewhat confident"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

### TABLE A.2.3.2 Confidence using technology index scores

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 49.3*       | (0.2) |
| Alberta                   | 50.0        | (0.3) |
| Saskatchewan              | 49.8        | (0.2) |
| Manitoba                  | 49.3        | (0.2) |
| Ontario                   | 50.7*       | (0.2) |
| Quebec                    | 49.3*       | (0.2) |
| New Brunswick             | 49.6*       | (0.0) |
| Nova Scotia               | 49.6*       | (0.0) |
| Prince Edward Island      | 50.3        | (0.0) |
| Newfoundland and Labrador | 49.1*       | (0.1) |
| Canada                    | 50.0        | (0.1) |

\* Significant difference compared to Canada

| Council and an a finance  | Anglophone scho | ool systems | Francophone sch | ool systems | Difference** |  |
|---------------------------|-----------------|-------------|-----------------|-------------|--------------|--|
| Canada and provinces      | Index score     | SE          | Index score     | SE          | (A – F)      |  |
| British Columbia          | 49.3*           | (0.2)       | 50.2*           | (0.0)       | -1**         |  |
| Alberta                   | 50.0            | (0.3)       | 50.0            | (0.9)       | 0            |  |
| Saskatchewan              | 49.9            | (0.2)       | 47.1*           | (0.0)       | 3**          |  |
| Manitoba                  | 49.3*           | (0.2)       | 49.4            | (0.0)       | 0            |  |
| Ontario                   | 50.8*           | (0.3)       | 50.0            | (0.2)       | 1**          |  |
| Quebec                    | 49.4            | (0.3)       | 49.2            | (0.2)       | 0            |  |
| New Brunswick             | 49.9            | (0.0)       | 49.0            | (0.0)       | 1**          |  |
| Nova Scotia               | 49.6*           | (0.0)       | 48.8            | (0.2)       | 1**          |  |
| Prince Edward Island      | 50.3            | (0.0)       |                 |             |              |  |
| Newfoundland and Labrador | 49.1*           | (0.1)       |                 |             |              |  |
| Canada                    | 50.2            | (0.1)       | 49.3            | (0.2)       | 1**          |  |

### TABLE A.2.3.3 Confidence using technology index scores by language of the school system

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

| Canada and                   | Female      | es    | Males       |       | Difforonco** (E - M) |
|------------------------------|-------------|-------|-------------|-------|----------------------|
| provinces                    | Index score | SE    | Index score | SE    |                      |
| British Columbia             | 47.6*       | (0.3) | 51.0        | (0.3) | -3**                 |
| Alberta                      | 48.4        | (0.4) | 51.5        | (0.3) | -3**                 |
| Saskatchewan                 | 48.6        | (0.3) | 51.0        | (0.3) | -2**                 |
| Manitoba                     | 48.0        | (0.3) | 50.6*       | (0.3) | -3**                 |
| Ontario                      | 48.9        | (0.4) | 52.6*       | (0.3) | -4**                 |
| Quebec                       | 48.2        | (0.2) | 50.4*       | (0.3) | -2**                 |
| New Brunswick                | 48.5        | (0.0) | 50.8*       | (0.0) | -2**                 |
| Nova Scotia                  | 48.4        | (0.0) | 50.9*       | (0.0) | -3**                 |
| Prince Edward Island         | 50.5*       | (0.0) | 50.1*       | (0.0) | 0**                  |
| Newfoundland and<br>Labrador | 48.3        | (0.1) | 49.8*       | (0.1) | -2**                 |
| Canada                       | 48.5        | (0.2) | 51.5        | (0.1) | -3**                 |

TABLE A.2.3.4 Confidence using technology index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

|                       |                             |       | -             |                     | <b>e e</b> , ·             |               |        |
|-----------------------|-----------------------------|-------|---------------|---------------------|----------------------------|---------------|--------|
|                       | Bottom quarter <sup>1</sup> |       | Second qu     | uarter <sup>2</sup> | Third quarter <sup>3</sup> | Top qu        | arter⁴ |
|                       | Mean<br>score               | SE    | Mean<br>score | SE                  | Mean SE<br>score           | Mean<br>score | SE     |
| Canada overall        | 481*                        | (2.5) | 515*          | (2.4)               | 528* (2.3)                 | 523*          | (2.4)  |
| Language of the schoo | l system                    |       |               |                     |                            |               |        |
| Anglophone            | 471*                        | (2.9) | 507*          | (2.8)               | 522* (2.7)                 | 517           | (2.7)  |
| Francophone           | 517*                        | (4.0) | 539*          | (4.2)               | 549* (3.8)                 | 547           | (4.2)  |
| Gender                |                             |       |               |                     |                            |               |        |
| Female                | 485*                        | (2.8) | 513*          | (3.1)               | 527* (2.5)                 | 519*          | (3.5)  |
| Male                  | 476*                        | (3.2) | 517*          | (3.0)               | 529* (3.3)                 | 525           | (2.7)  |

TABLE A.2.3.5 Mathematics achievement by the confidence using technology index, Canada

\* Significant difference between:

 $^{\rm 1}$  Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

#### TABLE A.2.4.1 Distribution of students by questionnaire items constituting the student effort index, Canada

|   | :                  | Strongly | disagree <sup>1</sup> | L     |      | Disa   | gree <sup>2</sup> |       |
|---|--------------------|----------|-----------------------|-------|------|--------|-------------------|-------|
|   | %                  | SE       | Mean<br>score         | SE    | %    | SE     | Mean<br>score     | SE    |
| Strong correlation with mathematics achievement (> 0.2)     |                    |          |                       |       |      |        |                   |       |
| I am prepared for my assessments                            | 4.5                | (0.2)    | 462*                  | (4.8) | 12.7 | (0.3)  | 478*              | (2.7) |
| I keep studying until I understand the material             | 5.8                | (0.3)    | 475*                  | (4.4) | 21.4 | (0.4)  | 487*              | (2.3) |
| Weak or no correlation with mathematics achievement (< 0.2) |                    |          |                       |       |      |        |                   |       |
| I pay attention in class                                    | 3.2                | (0.2)    | 477*                  | (6.2) | 10.8 | (0.3)  | 496*              | (3.3) |
| I practise concepts that have not been taught in class      | 18.5               | (0.4)    | 498*                  | (2.5) | 38.0 | (0.5)  | 507*              | (1.8) |
| I keep my work well-organized                               | 8.0                | (0.3)    | 491*                  | (3.9) | 22.6 | (0.4)  | 502*              | (2.9) |
| I avoid distractions when I am studying                     | 8.9                | (0.3)    | 495*                  | (4.4) | 35.7 | (0.5)  | 512*              | (2.2) |
|   | Agree <sup>3</sup> |          |                       |       |      | Strong | y agree⁴          |       |
|   | %                  | SE       | Mean<br>score         | SE    | %    | SE     | Mean<br>score     | SE    |
| Strong correlation with mathematics achievement (> 0.2)     |                    |          |                       |       |      |        |                   |       |
| I am prepared for my assessments                            | 59.1               | (0.5)    | 507*                  | (1.9) | 23.8 | (0.5)  | 547*              | (2.2) |
| I keep studying until I understand the material             | 48.5               | (0.5)    | 511*                  | (2.0) | 24.3 | (0.6)  | 540*              | (2.3) |
| Weak or no correlation with mathematics achievement (< 0.2) |                    |          |                       |       |      |        |                   |       |
| I pay attention in class                                    | 59.8               | (0.5)    | 506*                  | (2.0) | 26.2 | (0.5)  | 532*              | (2.2) |
| I practise concepts that have not been taught in class      | 31.2               | (0.4)    | 510                   | (2.6) | 12.3 | (0.4)  | 545*              | (3.3) |
| I keep my work well-organized                               | 44.6               | (0.5)    | 511*                  | (1.9) | 24.7 | (0.5)  | 526*              | (2.4) |
| I avoid distractions when I am studying                     | 41.9               | (0.5)    | 510                   | (1.9) | 13.4 | (0.4)  | 521*              | (3.0) |

\* Significant difference between:

<sup>1</sup> "Strongly disagree" and "Strongly agree"

<sup>2</sup> "Disagree" and "Strongly disagree"

<sup>3</sup> "Agree" and "Disagree"

<sup>4</sup> "Strongly agree" and "Agree"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

| TABLE A.2.4.2 Student effort index sco | ores |
|--|------|
|--|------|

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 49.8        | (0.2) |
| Alberta                   | 49.7        | (0.3) |
| Saskatchewan              | 50.2        | (0.2) |
| Manitoba                  | 50.2        | (0.2) |
| Ontario                   | 51.0*       | (0.3) |
| Quebec                    | 48.6*       | (0.3) |
| New Brunswick             | 50.1        | (0.0) |
| Nova Scotia               | 48.5*       | (0.0) |
| Prince Edward Island      | 47.7*       | (0.0) |
| Newfoundland and Labrador | 48.3*       | (0.1) |
| Canada                    | 50.0        | (0.1) |

\* Significant difference compared to Canada

| Consider and successive as | Anglophone sch | ool systems | Francophone scho | Francophone school systems |         |  |  |
|----------------------------|----------------|-------------|------------------|----------------------------|---------|--|--|
| Canada and provinces       | Index score    | SE          | Index score      | SE                         | (A – F) |  |  |
| British Columbia           | 49.8           | (0.2)       | 50.1*            | (0.0)                      | 0       |  |  |
| Alberta                    | 49.7           | (0.3)       | 50.3             | (0.6)                      | -1      |  |  |
| Saskatchewan               | 50.2           | (0.2)       | 50.9*            | (0.0)                      | -1**    |  |  |
| Manitoba                   | 50.1           | (0.2)       | 52.2*            | (0.0)                      | -2**    |  |  |
| Ontario                    | 51.0*          | (0.3)       | 52.0*            | (0.3)                      | -1**    |  |  |
| Quebec                     | 48.8*          | (0.2)       | 48.6*            | (0.3)                      | 0       |  |  |
| New Brunswick              | 49.7*          | (0.0)       | 51.1*            | (0.0)                      | -1**    |  |  |
| Nova Scotia                | 48.4*          | (0.0)       | 50.9*            | (0.4)                      | -3**    |  |  |
| Prince Edward Island       | 47.6*          | (0.0)       |                  |                            |         |  |  |
| Newfoundland and Labrador  | 48.3*          | (0.1)       |                  |                            |         |  |  |
| Canada                     | 50.3           | (0.2)       | 49.0             | (0.3)                      | 1**     |  |  |

### TABLE A.2.4.3 Student effort index scores by language of the school system

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

| Conside and provinces     | Female      | es    | Males       | Males |         |  |  |  |
|---------------------------|-------------|-------|-------------|-------|---------|--|--|--|
| Canada and provinces —    | Index score | SE    | Index score | SE    | (F – M) |  |  |  |
| British Columbia          | 50.3        | (0.3) | 49.3        | (0.3) | 1**     |  |  |  |
| Alberta                   | 50.4        | (0.4) | 49.0        | (0.3) | 1**     |  |  |  |
| Saskatchewan              | 51.2        | (0.3) | 49.3        | (0.3) | 2**     |  |  |  |
| Manitoba                  | 50.9        | (0.3) | 49.5        | (0.3) | 1**     |  |  |  |
| Ontario                   | 52.0*       | (0.3) | 50.1*       | (0.3) | 2**     |  |  |  |
| Quebec                    | 49.7*       | (0.3) | 47.5*       | (0.3) | 2**     |  |  |  |
| New Brunswick             | 51.5*       | (0.0) | 48.7        | (0.0) | 3**     |  |  |  |
| Nova Scotia               | 49.5*       | (0.0) | 47.3*       | (0.0) | 2**     |  |  |  |
| Prince Edward Island      | 48.3*       | (0.0) | 46.9*       | (0.0) | 1**     |  |  |  |
| Newfoundland and Labrador | 49.0*       | (0.1) | 47.6*       | (0.1) | 1**     |  |  |  |
| Canada                    | 50.9        | (0.2) | 49.2        | (0.2) | 2**     |  |  |  |

### TABLE A.2.4.4 Student effort index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

### TABLE A.2.4.5 Mathematics achievement by the student effort index, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second qu     | uarter² | Third qua     | rter <sup>3</sup> | Top qua       | rter⁴ |
|-------------------------------|-----------------------------|-------|---------------|---------|---------------|-------------------|---------------|-------|
|                               | Mean<br>score               | SE    | Mean<br>score | SE      | Mean<br>score | SE                | Mean<br>score | SE    |
| Canada overall                | 487*                        | (2.7) | 508*          | (2.4)   | 513           | (2.3)             | 538*          | (2.3) |
| Language of the school system |                             |       |               |         |               |                   |               |       |
| Anglophone                    | 476*                        | (3.1) | 500*          | (2.8)   | 507*          | (2.7)             | 533*          | (2.7) |
| Francophone                   | 520*                        | (4.2) | 540*          | (3.9)   | 534           | (4.2)             | 557*          | (4.1) |
| Gender                        |                             |       |               |         |               |                   |               |       |
| Female                        | 481*                        | (3.4) | 507*          | (3.0)   | 509           | (2.8)             | 533*          | (2.8) |
| Male                          | 491*                        | (2.9) | 509*          | (2.9)   | 516*          | (3.0)             | 543*          | (3.5) |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

|   | No hon<br>is ass | nework<br>igned | Less than 30<br>minutes<br>1 hour |       | Betwee<br>2 ho | Between 1 and<br>2 hours |      | Between 2 and<br>3 hours |     | More than<br>3 hours |     |       |
|---|------------------|-----------------|-----------------------------------|-------|----------------|--------------------------|------|--------------------------|-----|----------------------|-----|-------|
|   | %                | SE              | %                                 | SE    | %              | SE                       | %    | SE                       | %   | SE                   | %   | SE    |
| In mathematics                              | 15.1             | (0.6)           | 25.9                              | (0.5) | 33.5           | (0.5)                    | 15.0 | (0.4)                    | 5.6 | (0.3)                | 4.9 | (0.2) |
| In each of your<br>other school<br>subjects | 10.7             | (0.5)           | 20.6                              | (0.6) | 31.8           | (0.5)                    | 20.0 | (0.5)                    | 8.4 | (0.3)                | 8.6 | (0.4) |

### TABLE A.2.4.6 Distribution of students by hours per week spent on homework, Canada

### TABLE A.2.4.7 Distribution and scores of students by frequency of homework completion, Canada

|           | %    | SE    | Mean score | SE    |
|-----------|------|-------|------------|-------|
| Never     | 1.8  | (0.1) | 482*       | (6.8) |
| Rarely    | 7.6  | (0.3) | 478*       | (5.6) |
| Sometimes | 23.4 | (0.5) | 483*       | (2.5) |
| Often     | 67.2 | (0.6) | 530        | (1.9) |

\* Significant difference compared to the *often* category

### TABLE A.2.4.8 Distribution of students by the number of days absent or late for school, Canada

|   | 0–2 days |       |        |       |      | 3–5 days   |           |       |  |
|---|----------|-------|--------|-------|------|------------|-----------|-------|--|
|   | %        | SE    | Mean   | SE    | %    | SE         | Mean      | SE    |  |
| For reasons that are not school-related | 22.9     | (0.5) | 517    | (2.7) | 25.4 | (0.4)      | 518       | (2.2) |  |
| For school-related activities           | 47.5     | (0.7) | 505    | (2.2) | 31.3 | (0.6)      | 518       | (2.3) |  |
|   |          | 6–9   | days   |       |      | 10–14 days |           |       |  |
|   | %        | SE    | Mean   | SE    | %    | SE         | Mean      | SE    |  |
| For reasons that are not school-related | 19.8     | (0.3) | 514    | (2.5) | 14.0 | (0.4)      | 504       | (2.9) |  |
| For school-related activities           | 12.8     | (0.4) | 525    | (3.3) | 5.1  | (0.3)      | 513       | (5.3) |  |
|   |          | 15-2  | 0 days |       |      | More tha   | an 20 day | S     |  |
|   | %        | SE    | Mean   | SE    | %    | SE         | Mean      | SE    |  |
| For reasons that are not school-related | 8.7      | (0.3) | 498    | (3.3) | 9.3  | (0.3)      | 493       | (3.5) |  |
| For school-related activities           | 1.8      | (0.1) | 501    | (6.1) | 1.5  | (0.1)      | 473       | (9.8) |  |

|                              |      | Never  |       |       |  |      | 1 or 2 times |          |       |  |
|------------------------------|------|--------|-------|-------|--|------|--------------|----------|-------|--|
|                              | %    | SE     | Mean  | SE    |  | %    | SE           | Mean     | SE    |  |
| I skipped a whole school day | 77.4 | (0.6)  | 519   | (1.8) |  | 17.5 | (0.5)        | 490      | (2.9) |  |
| I skipped some classes       | 84.4 | (0.4)  | 517   | (1.8) |  | 12.0 | (0.4)        | 488      | (3.3) |  |
| I arrived late for school    | 58.8 | (0.7)  | 526   | (2.0) |  | 26.3 | (0.5)        | 499      | (2.5) |  |
|                              |      | 3 or 4 | times |       |  |      | 5 or mo      | re times |       |  |
|                              | %    | SE     | Mean  | SE    |  | %    | SE           | Mean     | SE    |  |
| I skipped a whole school day | 2.9  | (0.2)  | 469   | (6.1) |  | 2.2  | (0.2)        | 446      | (5.3) |  |
| I skipped some classes       | 2.0  | (0.1)  | 462   | (7.1) |  | 1.6  | (0.1)        | 449      | (8.1) |  |
| I arrived late for school    | 7.9  | (0.3)  | 484   | (3.1) |  | 7.0  | (0.3)        | 463      | (3.5) |  |

### TABLE A.2.4.9 Distribution and scores of students by the number of days absent or late for school in the past two weeks, Canada

## TABLE A.2.4.10 Distribution of students by hours per week spent on activities outside of school hours, Canada

|   | No t      | time  | Less t    | han 1 hour | 1-2                  | 2 hours |
|---|-----------|-------|-----------|------------|----------------------|---------|
|   | %         | SE    | %         | SE         | %                    | SE      |
| Playing mathematics-related games or puzzles          | 54.3      | (0.5) | 31.2      | (0.5)      | 10.2                 | (0.3)   |
| Doing extracurricular activities (e.g., clubs, music) | 31.1      | (0.5) | 13.1      | (0.3)      | 21.1                 | (0.4)   |
| Doing community service                               | 61.5      | (0.6) | 21.1      | (0.4)      | 11.4                 | (0.4)   |
| Doing physical activity (e.g., walking, sports)       | 4.3       | (0.2) | 11.2      | (0.3)      | 22.1                 | (0.4)   |
| Using a computer or device for personal reasons       | 1.5       | (0.1) | 4.4       | (0.2)      | 13.1                 | (0.4)   |
| Spending time with friends                            | 6.2       | (0.3) | 8.0       | (0.3)      | 15.4                 | (0.3)   |
|   | 3–4 hours |       | 5–6 hours |            | <br>More than 6 hour |         |
|   | %         | SE    | %         | SE         | %                    | SE      |
| Playing mathematics-related games or puzzles          | 2.6       | (0.2) | 0.7       | (0.1)      | 0.9                  | (0.1)   |
| Doing extracurricular activities (e.g., clubs, music) | 14.7      | (0.3) | 7.3       | (0.3)      | 12.6                 | (0.4)   |
| Doing community service                               | 3.8       | (0.2) | 1.0       | (0.1)      | 1.1                  | (0.1)   |
| Doing physical activity (e.g., walking, sports)       | 20.2      | (0.4) | 12.6      | (0.3)      | 29.6                 | (0.6)   |
| Using a computer or device for personal reasons       | 19.8      | (0.4) | 16.5      | (0.4)      | 44.7                 | (0.5)   |
| Spending time with friends                            | 22.3      | (0.4) | <br>17.7  | (0.4)      | 30.4                 | (0.5)   |

| Strong correlat | ion with ma | thematics a | achievement (> 0         | ).2)  |                                     |            |                             |       |  |
|-----------------|-------------|-------------|--------------------------|-------|-------------------------------------|------------|-----------------------------|-------|--|
|                 |             | Never h     | neard of it <sup>1</sup> |       |                                     | Heard of i | it a few times <sup>2</sup> |       |  |
|                 | %           | SE          | Mean score               | SE    | <br>%                               | SE         | Mean score                  | SE    |  |
| Percentage      | 1.3         | (0.1)       | 409*                     | (6.3) | 5.4                                 | (0.4)      | 434*                        | (6.4) |  |
| Perimeter       | 2.3         | (0.2)       | 428*                     | (5.7) | 5.6                                 | (0.2)      | 440                         | (4.7) |  |
| Equation        | 2.6         | (0.2)       | 443*                     | (5.6) | 4.4                                 | (0.2)      | 433                         | (5.2) |  |
| Square root     | 1.6         | (0.1)       | 421*                     | (5.9) | 7.2                                 | (0.3)      | 461*                        | (4.1) |  |
| Ratio           | 2.5         | (0.2)       | 429*                     | (7.3) | 8.3                                 | (0.4)      | 471*                        | (3.7) |  |
| Factors         | 3.2         | (0.2)       | 468*                     | (4.7) | 11.7                                | (0.4)      | 482*                        | (3.0) |  |
|                 |             | Heard       | of it often <sup>3</sup> |       | Know this concept well <sup>4</sup> |            |                             |       |  |
|                 | %           | SE          | Mean score               | SE    | <br>%                               | SE         | Mean score                  | SE    |  |
| Percentage      | 27.4        | (0.6)       | 472*                     | (1.9) | 65.9                                | (0.8)      | 535*                        | (1.7) |  |
| Perimeter       | 19.8        | (0.5)       | 462*                     | (1.7) | 72.2                                | (0.6)      | 533*                        | (1.8) |  |
| Equation        | 18.0        | (0.4)       | 461*                     | (2.5) | 74.9                                | (0.6)      | 530*                        | (1.6) |  |
| Square root     | 25.9        | (0.5)       | 476*                     | (2.3) | 65.4                                | (0.7)      | 533*                        | (1.8) |  |
| Ratio           | 25.5        | (0.5)       | 480*                     | (2.1) | 63.6                                | (0.7)      | 532*                        | (1.8) |  |
| Factors         | 32.4        | (0.5)       | 491*                     | (2.4) | 52.7                                | (0.7)      | 533*                        | (1.9) |  |

## TABLE A.2.5.1 Distribution of students by questionnaire items constituting the knowledge of general terms in mathematics index, Canada

\* Significant difference between:

<sup>1</sup> "Never heard of it" and "Know this concept well"

<sup>2</sup> "Heard of it a few times" and "Never heard of it"

<sup>3</sup> "Heard of it often" and "Heard of it a few times"

<sup>4</sup> "Know this concept well" and "Heard of it often"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 48.1*       | (0.3) |
| Alberta                   | 49.7        | (0.4) |
| Saskatchewan              | 49.3        | (0.3) |
| Manitoba                  | 49.1*       | (0.3) |
| Ontario                   | 51.0*       | (0.3) |
| Quebec                    | 50.2        | (0.2) |
| New Brunswick             | 49.3*       | (0.0) |
| Nova Scotia               | 50.1        | (0.0) |
| Prince Edward Island      | 48.8*       | (0.0) |
| Newfoundland and Labrador | 49.0*       | (0.2) |
| Canada                    | 50.1        | (0.2) |

#### TABLE A.2.5.2 Knowledge of general terms in mathematics index scores

\* Significant difference compared to Canada

### TABLE A.2.5.3 Knowledge of general terms in mathematics index scores by language of the school system

| Canada and provinces      | Anglophone s<br>systems | school | Francophone s<br>systems | Difference** (A – F) |      |
|---------------------------|-------------------------|--------|--------------------------|----------------------|------|
|                           | Index score             | SE     | Index score              | SE                   |      |
| British Columbia          | 48.1*                   | (0.3)  | 50.6                     | (0.0)                | -2** |
| Alberta                   | 49.7                    | (0.5)  | 50.6                     | (0.3)                | -1   |
| Saskatchewan              | 49.3                    | (0.3)  | 49.3*                    | (0.0)                | 0    |
| Manitoba                  | 49.0*                   | (0.3)  | 51.4*                    | (0.0)                | -2** |
| Ontario                   | 51.0*                   | (0.4)  | 51.7*                    | (0.3)                | -1   |
| Quebec                    | 50.4                    | (0.4)  | 50.2*                    | (0.2)                | 0    |
| New Brunswick             | 48.2*                   | (0.0)  | 52.0*                    | (0.0)                | -4** |
| Nova Scotia               | 50.0                    | (0.0)  | 50.6                     | (0.4)                | -1   |
| Prince Edward Island      | 48.8*                   | (0.0)  |                          |                      |      |
| Newfoundland and Labrador | 49.0*                   | (0.2)  |                          |                      |      |
| Canada                    | 50.1                    | (0.2)  | 50.4                     | (0.2)                | 0    |

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

| Canada and provinces      | Females     |       | Males       |       | Difforonco** (E _ M) |
|---------------------------|-------------|-------|-------------|-------|----------------------|
|                           | Index score | SE    | Index score | SE    | Difference (F - M)   |
| British Columbia          | 48.5*       | (0.4) | 47.8*       | (0.4) | 1**                  |
| Alberta                   | 50.4        | (0.5) | 49.1        | (0.5) | 1**                  |
| Saskatchewan              | 50.0        | (0.3) | 48.6        | (0.4) | 1**                  |
| Manitoba                  | 49.9        | (0.3) | 48.3*       | (0.4) | 2**                  |
| Ontario                   | 51.2        | (0.4) | 50.8*       | (0.4) | 0                    |
| Quebec                    | 51.1        | (0.3) | 49.3        | (0.3) | 2**                  |
| New Brunswick             | 50.0*       | (0.0) | 48.6*       | (0.0) | 1**                  |
| Nova Scotia               | 51.1        | (0.0) | 48.9*       | (0.1) | 2**                  |
| Prince Edward Island      | 49.9*       | (0.0) | 47.4*       | (0.0) | 2**                  |
| Newfoundland and Labrador | 50.6        | (0.1) | 47.4*       | (0.2) | 3**                  |
| Canada                    | 50.6        | (0.2) | 49.6        | (0.2) | 1**                  |

### TABLE A.2.5.4 Knowledge of general terms in mathematics index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

### TABLE A.2.5.5 Mathematics achievement by the knowledge of general terms in mathematics index, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second qu     | Second quarter <sup>2</sup> |               | Third quarter <sup>3</sup> |               | Top quarter <sup>₄</sup> |  |
|-------------------------------|-----------------------------|-------|---------------|-----------------------------|---------------|----------------------------|---------------|--------------------------|--|
|                               | Mean<br>score               | SE    | Mean<br>score | SE                          | Mean<br>score | SE                         | Mean<br>score | SE                       |  |
| Canada overall                | 454*                        | (2.5) | 506*          | (1.8)                       | 539*          | (2.6)                      | 553*          | (2.1)                    |  |
| Language of the school system |                             |       |               |                             |               |                            |               |                          |  |
| Anglophone                    | 446*                        | (2.8) | 496*          | (2.1)                       | 534*          | (3.2)                      | 549*          | (2.5)                    |  |
| Francophone                   | 490*                        | (4.0) | 531*          | (3.7)                       | 562*          | (4.0)                      | 570           | (4.2)                    |  |
| Gender                        |                             |       |               |                             |               |                            |               |                          |  |
| Female                        | 453*                        | (2.9) | 504*          | (2.4)                       | 537*          | (3.2)                      | 547*          | (2.5)                    |  |
| Male                          | 455*                        | (2.9) | 509*          | (2.5)                       | 541*          | (3.2)                      | 559*          | (2.8)                    |  |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

|  |           | Neve    | er heard of it <sup>1</sup> |       |                                     | Heard of | it a few times <sup>2</sup> |       |
|--|-----------|---------|-----------------------------|-------|-------------------------------------|----------|-----------------------------|-------|
|  | %         | SE      | Mean score                  | SE    | %                                   | SE       | Mean score                  | SE    |
| Strong correlation with mathem                   | atics acl | nieveme | ent (> 0.2)                 |       |                                     |          |                             |       |
| Familiar with mathematics terms: quadrant        | 17.1      | (0.5)   | 476*                        | (2.4) | 21.1                                | (0.5)    | 493*                        | (2.7) |
| Familiar with mathematics terms: Cartesian plane | 30.2      | (1.0)   | 484*                        | (2.4) | 15.3                                | (0.4)    | 496*                        | (3.0) |
|  |           | Hea     | rd of it often <sup>3</sup> |       | Know this concept well <sup>4</sup> |          |                             |       |
|  | %         | SE      | Mean score                  | SE    | %                                   | SE       | Mean score                  | SE    |
| Familiar with mathematics terms: quadrant        | 23.4      | (0.4)   | 498                         | (2.3) | 38.4                                | (0.8)    | 544*                        | (2.1) |
| Familiar with mathematics terms: Cartesian plane | 18.1      | (0.4)   | 496                         | (2.7) | 36.5                                | (1.0)    | 547*                        | (2.2) |

#### TABLE A.2.6.1 Distribution of students by questionnaire items constituting the knowledge of geometry and measurement terms index, Canada

\* Significant difference between:

<sup>1</sup> "Never heard of it" and "Know this concept well"

<sup>2</sup> "Heard of it a few times" and "Never heard of it"

<sup>3</sup> "Heard of it often" and "Heard of it a few times"

<sup>4</sup> "Know this concept well" and "Heard of it often"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 44.2*       | (0.3) |
| Alberta                   | 50.2        | (0.4) |
| Saskatchewan              | 47.0*       | (0.4) |
| Manitoba                  | 46.4*       | (0.3) |
| Ontario                   | 49.1*       | (0.4) |
| Quebec                    | 57.8*       | (0.2) |
| New Brunswick             | 48.1*       | (0.0) |
| Nova Scotia               | 46.0*       | (0.1) |
| Prince Edward Island      | 54.2*       | (0.0) |
| Newfoundland and Labrador | 42.8*       | (0.1) |
| Canada                    | 50.1        | (0.2) |

### TABLE A.2.6.2 Knowledge of geometry and measurement terms index scores

\* Significant difference compared to Canada

| Canada and provinces      | Anglophone school systems |       | Francophone sch | ool systems | Difference** |  |
|---------------------------|---------------------------|-------|-----------------|-------------|--------------|--|
|                           | Index score               | SE    | Index score     | SE          | (A – F)      |  |
| British Columbia          | 44.1*                     | (0.3) | 52.4*           | (0.0)       | -8**         |  |
| Alberta                   | 50.1*                     | (0.4) | 52.5*           | (0.4)       | -2**         |  |
| Saskatchewan              | 46.9*                     | (0.4) | 52.0*           | (0.0)       | -5**         |  |
| Manitoba                  | 46.4*                     | (0.3) | 47.2*           | (0.0)       | -1**         |  |
| Ontario                   | 48.9                      | (0.5) | 52.7*           | (0.3)       | -4**         |  |
| Quebec                    | 56.6*                     | (0.2) | 58.0*           | (0.2)       | -1**         |  |
| New Brunswick             | 46.2*                     | (0.0) | 52.8*           | (0.0)       | -7**         |  |
| Nova Scotia               | 45.7*                     | (0.0) | 52.3*           | (0.4)       | -7**         |  |
| Prince Edward Island      | 54.4*                     | (0.0) |                 |             |              |  |
| Newfoundland and Labrador | 42.7*                     | (0.1) |                 |             |              |  |
| Canada                    | 48.2                      | (0.3) | 57.2            | (0.2)       | -9**         |  |

### TABLE A.2.6.3 Knowledge of geometry and measurement terms index scores by language of the school system

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

Note: Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

### TABLE A.2.6.4 Knowledge of geometry and measurement terms index scores by gender

| Canada and provinces      | Females     |       | Males       |       | Difference** |  |
|---------------------------|-------------|-------|-------------|-------|--------------|--|
|                           | Index score | SE    | Index score | SE    | (F – M)      |  |
| British Columbia          | 43.7*       | (0.3) | 44.6*       | (0.3) | -1**         |  |
| Alberta                   | 49.9        | (0.5) | 50.4        | (0.4) | 0            |  |
| Saskatchewan              | 46.1*       | (0.4) | 47.7*       | (0.3) | -2**         |  |
| Manitoba                  | 46.0*       | (0.4) | 46.9*       | (0.3) | -1**         |  |
| Ontario                   | 48.6*       | (0.5) | 49.5        | (0.5) | -1**         |  |
| Quebec                    | 58.3*       | (0.3) | 57.4*       | (0.2) | 1**          |  |
| New Brunswick             | 47.7*       | (0.0) | 48.5*       | (0.0) | -1**         |  |
| Nova Scotia               | 45.8*       | (0.1) | 46.1*       | (0.1) | 0**          |  |
| Prince Edward Island      | 54.4*       | (0.0) | 53.9*       | (0.0) | 1**          |  |
| Newfoundland and Labrador | 41.8*       | (0.1) | 43.7*       | (0.1) | -2**         |  |
| Canada                    | 49.9        | (0.3) | 50.3        | (0.2) | 0            |  |

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

| index, Canada |                             |    |          |                             |       |                            |       |              |  |  |  |  |
|---------------|-----------------------------|----|----------|-----------------------------|-------|----------------------------|-------|--------------|--|--|--|--|
|               | Bottom quarter <sup>1</sup> |    | Second q | Second quarter <sup>2</sup> |       | Third quarter <sup>3</sup> |       | Top quarter⁴ |  |  |  |  |
|               | Mean                        | SE | Mean     | SE                          | Mean  | SE                         | Mean  | SE           |  |  |  |  |
|               | score                       |    | score    |                             | score |                            | score |              |  |  |  |  |

497\*

497\*

503\*

498\*

497\*

(2.2)

(2.5)

(4.0)

(3.2)

(2.6)

516\*

512\*

527\*

516\*

517\*

(2.4)

(2.8)

(4.0)

(3.0)

(3.0)

553\*

546\*

562\*

552\*

554\*

(2.5)

(3.7)

(3.5)

(2.9)

(3.2)

### TABLE A 2.6.5 Mathematics achievement by the knowledge of geometry and measurement terms

486\*

486\*

475\*

482\*

489\*

(2.6)

(2.7)

(6.3)

(3.1)

(3.1)

\* Significant difference between:

Language of the school system

Canada overall

Anglophone

Francophone

Gender Female

Male

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

### TABLE A.2.7.1 Distribution of students by questionnaire items constituting the teacher-directed mathematics activities index, Canada

|  |   | Ne  | ever1   |   |  | Rarely <sup>2</sup>   |  |  |
|--|---|---|---|---|--|---|--|--|
|  | %   | SE  | Mean  | SE  | %  | SE  | Mean   | SE   |
|  |   | I   | score   |   |  |   | score  |  |
| Strong correlation with mathematics achievement (> 0.2)  |   |   |   |   |  |   |  |  |
| Solve problems   | 1.7   | (0.1)   | 434*  | (7.1)   | 5.7  | (0.2)   | 460*   | (3.2)  |
| Work individually  | 1.7   | (0.1)   | 441*  | (6.9)   | 5.0  | (0.3)   | 468*   | (4.1)  |
| Practise skills  | 3.5   | (0.2)   | 470*  | (6.3)   | 11.3   | (0.3)   | 484*   | (2.6)  |
| Weak or no correlation with mathematics achievement (< 0.2)  |   |   |   |   |  |   |  |  |
| Pay attention to the teacher doing examples and giving explanations  | 1.8   | (0.1)   | 457*  | (7.3)   | 5.1  | (0.2)   | 483*   | (4.8)  |
| Observe teacher-guided problem solving and investigations  | 2.4   | (0.2)   | 466*  | (6.8)   | 8.2  | (0.3)   | 493*   | (4.2)  |
| Use calculators  | 1.8   | (0.2)   | 468*  | (8.2)   | 4.3  | (0.2)   | 490*   | (4.8)  |
| Copy notes   | 6.0   | (0.3)   | 513   | (4.8)   | 15.4   | (0.5)   | 518  | (3.1)  |
|  |   | Some  | etimes <sup>3</sup>   |   | Often⁴   |   |  |  |
|  | ~   | SE  | Mean  | SE  | %  | SE  | Mean   | SE   |
|  | %   | JL  |   |   |  | JL  | wicun  |  |
|  | %   | JL  | score   | 02  | ,,,  | JL  | score  |  |
| Strong correlation with mathematics achievement (> 0.2)  | %   | JL  | score   |   | ,,,  | JL  | score  |  |
| Strong correlation with mathematics achievement (> 0.2)<br>Solve problems  | %<br>29.3   | (0.5)   | <b>score</b><br>483*  | (2.1)   | 63.3   | (0.6)   | score  | (1.9)  |
| Strong correlation with mathematics achievement (> 0.2)<br>Solve problems<br>Work individually   | %<br>29.3<br>26.1   | (0.5)<br>(0.5)  | score<br>483*<br>488*                                       | (2.1)<br>(2.9)  | 63.3<br>67.3   | (0.6)<br>(0.6)  | score<br>532*<br>526*  | (1.9)<br>(1.7)   |
| Strong correlation with mathematics achievement (> 0.2)<br>Solve problems<br>Work individually<br>Practise skills  | %<br>29.3<br>26.1<br>36.3                                 | (0.5)<br>(0.5)<br>(0.5)                                     | score<br>483*<br>488*<br>497*                               | (2.1)<br>(2.9)<br>(2.2)                                     | 63.3<br>67.3<br>48.9                                 | (0.6)<br>(0.6)<br>(0.7)                                     | score<br>532*<br>526*<br>532*                                | (1.9)<br>(1.7)<br>(1.8)  |
| Strong correlation with mathematics achievement (> 0.2)<br>Solve problems<br>Work individually<br>Practise skills<br>Weak or no correlation with mathematics achievement (< 0.2)                     | %<br>29.3<br>26.1<br>36.3                                 | (0.5)<br>(0.5)<br>(0.5)                                     | score<br>483*<br>488*<br>497*                               | (2.1)<br>(2.9)<br>(2.2)                                     | 63.3<br>67.3<br>48.9                                 | (0.6)<br>(0.6)<br>(0.7)                                     | score<br>532*<br>526*<br>532*                                | (1.9)<br>(1.7)<br>(1.8)  |
| Strong correlation with mathematics achievement (> 0.2)         Solve problems         Work individually         Practise skills         Weak or no correlation with mathematics achievement (< 0.2) | %<br>29.3<br>26.1<br>36.3<br>25.6                         | (0.5)<br>(0.5)<br>(0.5)                                     | score<br>483*<br>488*<br>497*<br>492*                       | (2.1)<br>(2.9)<br>(2.2)<br>(2.6)                            | 63.3<br>67.3<br>48.9<br>67.5                         | (0.6)<br>(0.6)<br>(0.7)<br>(0.6)                            | 532*<br>526*<br>532*<br>532*                                 | (1.9)<br>(1.7)<br>(1.8)<br>(1.6)                                     |
| Strong correlation with mathematics achievement (> 0.2)         Solve problems         Work individually         Practise skills         Weak or no correlation with mathematics achievement (< 0.2) | %<br>29.3<br>26.1<br>36.3<br>25.6<br>36.0                 | (0.5)<br>(0.5)<br>(0.5)<br>(0.5)<br>(0.5)                   | score<br>483*<br>488*<br>497*<br>492*<br>500                | (2.1)<br>(2.9)<br>(2.2)<br>(2.6)<br>(2.3)                   | 63.3<br>67.3<br>48.9<br>67.5<br>53.4                 | (0.6)<br>(0.7)<br>(0.6)<br>(0.6)                            | 532*<br>526*<br>532*<br>532*<br>523*                         | (1.9)<br>(1.7)<br>(1.8)<br>(1.6)<br>(1.7)                            |
| Strong correlation with mathematics achievement (> 0.2)         Solve problems         Work individually         Practise skills         Weak or no correlation with mathematics achievement (< 0.2) | %<br>29.3<br>26.1<br>36.3<br>25.6<br>36.0<br>24.7         | (0.5)<br>(0.5)<br>(0.5)<br>(0.5)<br>(0.5)<br>(0.7)          | score<br>483*<br>488*<br>497*<br>492*<br>500<br>493         | (2.1)<br>(2.9)<br>(2.2)<br>(2.6)<br>(2.3)<br>(3.0)          | 63.3<br>67.3<br>48.9<br>67.5<br>53.4<br>69.1         | (0.6)<br>(0.6)<br>(0.7)<br>(0.6)<br>(0.6)<br>(0.8)          | 532*<br>526*<br>532*<br>523*<br>523*<br>524*<br>521*         | (1.9)<br>(1.7)<br>(1.8)<br>(1.6)<br>(1.7)<br>(1.7)                   |
| Strong correlation with mathematics achievement (> 0.2)         Solve problems         Work individually         Practise skills         Weak or no correlation with mathematics achievement (< 0.2) | %<br>29.3<br>26.1<br>36.3<br>25.6<br>36.0<br>24.7<br>31.1 | (0.5)<br>(0.5)<br>(0.5)<br>(0.5)<br>(0.5)<br>(0.7)<br>(0.6) | score<br>483*<br>488*<br>497*<br>492*<br>500<br>493<br>508* | (2.1)<br>(2.9)<br>(2.2)<br>(2.6)<br>(2.3)<br>(3.0)<br>(2.3) | 63.3<br>67.3<br>48.9<br>67.5<br>53.4<br>69.1<br>47.6 | (0.6)<br>(0.6)<br>(0.7)<br>(0.6)<br>(0.6)<br>(0.8)<br>(0.9) | 532*<br>526*<br>532*<br>523*<br>523*<br>524*<br>521*<br>521* | (1.9)<br>(1.7)<br>(1.8)<br>(1.6)<br>(1.7)<br>(1.7)<br>(1.7)<br>(2.0) |

\* Significant difference between:

<sup>1</sup> "Never" and "Often"

<sup>2</sup> "Rarely" and "Never"

<sup>3</sup> "Sometimes" and "Rarely"

<sup>4</sup> "Often" and "Sometimes"

Note: Items have been sorted in descending order by correlation with mathematics achievement.

| Canada and provinces      | Index score | SE    |
|---------------------------|-------------|-------|
| British Columbia          | 49.4        | (0.2) |
| Alberta                   | 49.7        | (0.3) |
| Saskatchewan              | 49.6        | (0.2) |
| Manitoba                  | 48.9*       | (0.3) |
| Ontario                   | 49.8        | (0.3) |
| Quebec                    | 51.6*       | (0.3) |
| New Brunswick             | 49.2*       | (0.0) |
| Nova Scotia               | 49.7        | (0.0) |
| Prince Edward Island      | 48.7*       | (0.0) |
| Newfoundland and Labrador | 50.0        | (0.1) |
| Canada                    | 50.1        | (0.1) |

### TABLE A.2.7.2 Teacher-directed mathematics activities index scores

\* Significant difference compared to Canada

| Canada and provinces      | Anglophone school systems |       | Francophone sch | ool systems | Difference** |  |
|---------------------------|---------------------------|-------|-----------------|-------------|--------------|--|
| -                         | Index score               | SE    | Index score     | SE          | (A – F)      |  |
| British Columbia          | 49.4                      | (0.2) | 51.6            | (0.0)       | -2**         |  |
| Alberta                   | 49.7                      | (0.3) | 51.1            | (0.9)       | -1           |  |
| Saskatchewan              | 49.6                      | (0.2) | 48.7*           | (0.0)       | 1**          |  |
| Manitoba                  | 48.9                      | (0.3) | 50.8            | (0.0)       | -2**         |  |
| Ontario                   | 49.8                      | (0.3) | 49.6*           | (0.3)       | 0            |  |
| Quebec                    | 50.8*                     | (0.3) | 51.7*           | (0.3)       | -1           |  |
| New Brunswick             | 48.7*                     | (0.0) | 50.6*           | (0.0)       | -2**         |  |
| Nova Scotia               | 49.7                      | (0.0) | 50.4            | (0.4)       | -1           |  |
| Prince Edward Island      | 48.7*                     | (0.0) |                 |             |              |  |
| Newfoundland and Labrador | 49.9                      | (0.1) |                 |             |              |  |
| Canada                    | 49.7                      | (0.2) | 51.5            | (0.3)       | -2**         |  |

### TABLE A.2.7.3 Teacher-directed mathematics activities index scores by language of the school system

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

*Note:* Due to small sample sizes, results for students in the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

| Canada and provinces      | Female      | 25    | Males       | Males |         |  |
|---------------------------|-------------|-------|-------------|-------|---------|--|
|                           | Index score | SE    | Index score | SE    | (F – M) |  |
| British Columbia          | 50.3*       | (0.3) | 48.5        | (0.3) | 2**     |  |
| Alberta                   | 50.9        | (0.4) | 48.7        | (0.4) | 2**     |  |
| Saskatchewan              | 51.2        | (0.3) | 48.2        | (0.3) | 3**     |  |
| Manitoba                  | 49.8*       | (0.3) | 48.1        | (0.4) | 2**     |  |
| Ontario                   | 51.1        | (0.3) | 48.6        | (0.3) | 2**     |  |
| Quebec                    | 53.5*       | (0.3) | 49.7        | (0.4) | 4**     |  |
| New Brunswick             | 50.8*       | (0.0) | 47.5*       | (0.0) | 3**     |  |
| Nova Scotia               | 51.3        | (0.0) | 48.0*       | (0.1) | 3**     |  |
| Prince Edward Island      | 50.1*       | (0.0) | 47.1*       | (0.0) | 3**     |  |
| Newfoundland and Labrador | 51.0        | (0.1) | 48.9        | (0.2) | 2**     |  |
| Canada                    | 51.4        | (0.2) | 48.8        | (0.2) | 3**     |  |

### TABLE A.2.7.4 Teacher-directed mathematics activities index scores by gender

\* Significant difference compared to Canada

\*\* Significant difference within Canada or within the province

### TABLE A.2.7.5 Mathematics achievement by the teacher-directed mathematics activities index, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second q      | Second quarter <sup>2</sup> |               | Third quarter <sup>3</sup> |               | arter <sup>4</sup> |
|-------------------------------|-----------------------------|-------|---------------|-----------------------------|---------------|----------------------------|---------------|--------------------|
|                               | Mean<br>score               | SE    | Mean<br>score | SE                          | Mean<br>score | SE                         | Mean<br>score | SE                 |
| Canada overall                | 474*                        | (2.6) | 515*          | (2.7)                       | 529*          | (1.9)                      | 536*          | (2.2)              |
| Language of the school system |                             |       |               |                             |               |                            |               |                    |
| Anglophone                    | 470*                        | (2.9) | 510*          | (3.3)                       | 522*          | (2.2)                      | 528*          | (2.6)              |
| Francophone                   | 497*                        | (4.7) | 533*          | (3.5)                       | 554*          | (3.5)                      | 558           | (3.9)              |
| Gender                        |                             |       |               |                             |               |                            |               |                    |
| Female                        | 464*                        | (3.3) | 507*          | (2.8)                       | 525*          | (2.4)                      | 536*          | (2.6)              |
| Male                          | 481*                        | (3.1) | 522*          | (3.9)                       | 534*          | (2.8)                      | 536           | (3.1)              |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

|                               | Ye   | 25    |     | No      |
|-------------------------------|------|-------|-----|---------|
|                               | %    | SE    | %   | SE      |
| Science                       | 86.1 | (0.4) | 13. | 9 (0.4) |
| Technology                    | 72.7 | (0.5) | 27. | 3 (0.5) |
| Art                           | 39.0 | (0.7) | 61. | 0 (0.7) |
| Social studies                | 27.0 | (0.5) | 73. | 0 (0.5) |
| Health and physical education | 24.4 | (0.5) | 75. | 6 (0.5) |
| Music/band                    | 23.9 | (0.6) | 76. | 1 (0.6) |
| Language arts                 | 20.7 | (0.5) | 79. | 3 (0.5) |

### TABLE A.2.8 Distribution of students who reported using what they learned in mathematics in other subjects, Canada

### TABLE A.2.9 Distribution of students by type of mathematics homework assigned, Canada

|   | Never or almost<br>never |       | 2 or 3<br>per n | times<br>nonth | 1 to 3<br>per v | times<br>week | Daily or<br>almost daily |       |
|---|--------------------------|-------|-----------------|----------------|-----------------|---------------|--------------------------|-------|
|   | %                        | SE    | %               | SE             | %               | SE            | %                        | SE    |
| Paper-pencil calculations   | 13.0                     | (0.4) | 21.6            | (0.6)          | 31.5            | (0.6)         | 33.8                     | (0.8) |
| Word problems   | 15.4                     | (0.5) | 28.3            | (0.6)          | 34.0            | (0.6)         | 22.3                     | (0.7) |
| Studying for assessments  | 9.8                      | (0.4) | 39.7            | (0.6)          | 35.8            | (0.5)         | 14.7                     | (0.4) |
| New concepts not taught in class                                      | 47.4                     | (0.6) | 28.4            | (0.5)          | 16.4            | (0.4)         | 7.9                      | (0.3) |
| Collaborative problem-solving activities                              | 34.0                     | (0.6) | 35.4            | (0.6)          | 23.0            | (0.6)         | 7.6                      | (0.3) |
| Viewing videos online (e.g.,<br>YouTube, Vimeo) to review<br>concepts | 50.4                     | (0.9) | 29.2            | (0.6)          | 13.3            | (0.5)         | 7.1                      | (0.3) |
| Creating problems   | 50.1                     | (0.7) | 30.5            | (0.6)          | 14.5            | (0.5)         | 4.9                      | (0.2) |
| Participation in online<br>discussion forums or blogs                 | 69.9                     | (0.7) | 17.1            | (0.5)          | 9.0             | (0.4)         | 4.0                      | (0.2) |
| Projects  | 40.0                     | (0.7) | 44.3            | (0.7)          | 11.7            | (0.4)         | 3.9                      | (0.3) |
| Activities using concrete or virtual manipulatives                    | 56.4                     | (0.7) | 28.3            | (0.6)          | 11.3            | (0.4)         | 3.9                      | (0.2) |

|  | Never |       | Rar  | Rarely |      | Sometimes |      | ten   |
|--|-------|-------|------|--------|------|-----------|------|-------|
|  | %     | SE    | %    | SE     | %    | SE        | %    | SE    |
| Homework   | 10.7  | (0.4) | 18.7 | (0.7)  | 30.3 | (0.6)     | 40.3 | (1.0) |
| Teacher-developed classroom assessments                                | 7.7   | (0.3) | 19.5 | (0.4)  | 45.3 | (0.4)     | 27.5 | (0.5) |
| Individual student assignments/projects                                | 14.7  | (0.4) | 21.8 | (0.4)  | 38.5 | (0.6)     | 25.0 | (0.6) |
| Common assessments (i.e., any assessment used by two or more teachers) | 14.2  | (0.4) | 24.5 | (0.5)  | 38.0 | (0.5)     | 23.3 | (0.6) |
| Self-assessment  | 21.7  | (0.5) | 24.1 | (0.4)  | 33.8 | (0.5)     | 20.5 | (0.5) |
| Group assignments/projects   | 17.5  | (0.5) | 26.0 | (0.5)  | 40.4 | (0.6)     | 16.1 | (0.5) |
| Performance assessments (include tasks with a real-world application)  | 22.5  | (0.4) | 29.6 | (0.5)  | 35.1 | (0.5)     | 12.8 | (0.4) |
| Peer assessment  | 28.9  | (0.6) | 30.8 | (0.5)  | 31.1 | (0.6)     | 9.2  | (0.4) |
| Student portfolios and/or journals                                     | 41.0  | (0.7) | 27.7 | (0.5)  | 22.5 | (0.5)     | 8.9  | (0.5) |

### TABLE A.2.10.1 Distribution of students by type of mathematics assessment method, Canada

### TABLE A.2.10.2 Distribution of students by use of scoring rubrics in mathematics, Canada

|  |      | Ye    | es            |       | No   |       |               |       |  |
|--|------|-------|---------------|-------|------|-------|---------------|-------|--|
|  | %    | SE    | Mean<br>score | SE    | %    | SE    | Mean<br>score | SE    |  |
| Do you know what a scoring rubric is for marking assessments or assignments?             | 76.7 | (0.6) | 520           | (1.9) | 23.3 | (0.6) | 481*          | (2.3) |  |
| Do you sometimes use a rubric when you start<br>an assignment in your mathematics class? | 48.9 | (0.8) | 517           | (2.8) | 51.1 | (0.8) | 523*          | (1.8) |  |

\* Significant difference

## TABLE A.2.10.3 Mathematics achievement of students by the frequency that teachers use rubrics, Canada

|           | %    | SE    | Mean  | SE    |
|-----------|------|-------|-------|-------|
|           |      |       | score |       |
| Never     | 13.8 | (0.5) | 522*  | (3.0) |
| Rarely    | 24.8 | (0.6) | 511*  | (2.2) |
| Sometimes | 32.2 | (0.6) | 512*  | (2.3) |
| Often     | 29.1 | (1.0) | 535   | (2.9) |

\* Significant difference compared to the often category

|   | Never |       | Rarely |       | Sometimes |       | Often |       |
|---|-------|-------|--------|-------|-----------|-------|-------|-------|
|   | %     | SE    | %      | SE    | %         | SE    | %     | SE    |
| Useful hints or strategies to solve a problem                       | 5.0   | (0.2) | 10.8   | (0.4) | 38.4      | (0.5) | 45.8  | (0.6) |
| Details about how your assessments or<br>assignments will be marked | 6.5   | (0.2) | 13.4   | (0.4) | 36.3      | (0.5) | 43.8  | (0.7) |
| Regular feedback to help you improve your<br>learning               | 8.0   | (0.3) | 15.2   | (0.4) | 36.5      | (0.5) | 40.2  | (0.7) |
| An opportunity to redo or resubmit work                             | 14.9  | (0.6) | 23.7   | (0.5) | 37.1      | (0.6) | 24.3  | (0.7) |

### TABLE A.2.11.1 Distribution of students by teacher feedback and support in mathematics class, Canada

### TABLE A.2.11.2 Mathematics achievement of students by teacher feedback and support in mathematics class, Canada

|  | Never or      | rarely <sup>1</sup> | Sometir       | nes²  | Often         | 3     |
|--|---------------|---------------------|---------------|-------|---------------|-------|
|  | Mean<br>score | SE                  | Mean<br>score | SE    | Mean<br>score | SE    |
| Useful hints or strategies to solve a problem                    | 498*          | (3.0)               | 507*          | (2.4) | 520*          | (2.0) |
| Details about how your assessments or assignments will be marked | 508*          | (2.6)               | 507           | (2.2) | 517*          | (2.2) |
| Regular feedback to help you improve your learning               | 515           | (2.6)               | 508*          | (2.0) | 512           | (2.6) |
| An opportunity to redo or resubmit work                          | 524*          | (2.4)               | 505*          | (2.3) | 501           | (2.6) |

\* Significant difference between:

<sup>1</sup> "Never or rarely" and "Often"

<sup>2</sup> "Sometimes" and "Never or rarely"

<sup>3</sup> "Often" and "Sometimes"

|  |      | Strongly | disagree <sup>1</sup> |       |      | Disa                        | gree <sup>2</sup> |       |  |  |
|--|------|----------|-----------------------|-------|------|-----------------------------|-------------------|-------|--|--|
|  | %    | SE       | Mean                  | SE    | %    | SE                          | Mean              | SE    |  |  |
| l like school  | 9.6  | (0.3)    | 478*                  | (2.9) | 21.1 | (0.4)                       | 501*              | (2.7) |  |  |
| At school, I am treated fairly                         | 4.6  | (0.2)    | 466*                  | (4.3) | 14.8 | (0.4)                       | 492*              | (2.6) |  |  |
| School staff care about me                             | 5.3  | (0.2)    | 479*                  | (4.2) | 14.2 | (0.4)                       | 503*              | (2.9) |  |  |
| At school, I feel that I belong                        | 6.8  | (0.3)    | 475*                  | (3.6) | 16.8 | (0.4)                       | 495*              | (2.6) |  |  |
| At school, I make friends easily                       | 6.0  | (0.2)    | 491*                  | (4.7) | 14.9 | (0.4)                       | 507*              | (4.3) |  |  |
| At school, I have a friend or friends whom I can trust | 3.2  | (0.2)    | 473*                  | (6.2) | 4.9  | (0.2)                       | 501*              | (5.8) |  |  |
|  |      | Ag       | ree³                  |       |      | Strongly agree <sup>4</sup> |                   |       |  |  |
|  | %    | SE       | Mean                  | SE    | %    | SE                          | Mean              | SE    |  |  |
| l like school  | 57.0 | (0.5)    | 515*                  | (1.9) | 12.3 | (0.4)                       | 527*              | (3.4) |  |  |
| At school, I am treated fairly                         | 58.8 | (0.5)    | 512*                  | (1.9) | 21.7 | (0.5)                       | 526*              | (2.6) |  |  |
| School staff care about me                             | 59.1 | (0.5)    | 513*                  | (1.9) | 21.4 | (0.5)                       | 515               | (2.5) |  |  |
| At school, I feel that I belong                        | 56.7 | (0.5)    | 514*                  | (2.0) | 19.7 | (0.4)                       | 522*              | (2.6) |  |  |
| At school, I make friends easily                       | 46.6 | (0.5)    | 513                   | (1.9) | 32.5 | (0.5)                       | 511               | (2.2) |  |  |
| At school, I have a friend or friends whom I can trust | 35.8 | (0.5)    | 508                   | (2.3) | 56.1 | (0.5)                       | 514*              | (1.9) |  |  |

### TABLE A.2.12 Distribution and scores of students by students' sense of belonging in school, Canada

\* Significant difference between:

<sup>1</sup> "Strongly disagree" and "Strongly agree"

<sup>2</sup> "Disagree" and "Strongly disagree"

<sup>3</sup> "Agree" and "Disagree"

<sup>4</sup> "Strongly agree" and "Agree"

## TABLE A.2.13.1 Distribution of students by type of problems encountered in mathematics class or assignment, Canada

|  | Ne  | ver   | Rar  | ely   | Some | times | Of   | ten   |  |
|--|-----|-------|------|-------|------|-------|------|-------|--|
|  | %   | SE    | %    | SE    | %    | SE    | %    | SE    |  |
| Level 1 problems: How often<br>encountered in mathematics lessons? | 3.6 | (0.2) | 10.4 | (0.3) | 43.2 | (0.6) | 42.7 | (0.7) |  |
| Level 1 problems: How often encountered on assessments?            | 4.0 | (0.2) | 12.6 | (0.4) | 44.9 | (0.5) | 38.5 | (0.7) |  |
| Level 3 problems: How often<br>encountered in mathematics lessons? | 5.5 | (0.2) | 16.6 | (0.3) | 45.5 | (0.5) | 32.3 | (0.6) |  |
| Level 3 problems: How often encountered on assessments?            | 6.7 | (0.3) | 17.3 | (0.4) | 46.2 | (0.5) | 29.8 | (0.5) |  |

*Note:* Level 1 problems describe scenarios and provide all the information required to solve them. Level 3 problems do not describe scenarios but require the use of mathematical knowledge to form conclusions.

| TABLE A.2.13.2 | Mathematics achievement of students by type of problems encountered in |
|----------------|--|
|                | mathematics class or assignment, Canada                                |

|  | Never or rarely <sup>1</sup> |       | Sometir       | nes²  | Often <sup>3</sup> |       |  |
|--|------------------------------|-------|---------------|-------|--------------------|-------|--|
|  | Mean<br>score                | SE    | Mean<br>score | SE    | Mean<br>score      | SE    |  |
| Level 1 problems: How often encountered in<br>mathematics lessons? | 464*                         | (3.3) | 501*          | (1.8) | 537*               | (2.0) |  |
| Level 1 problems: How often encountered on assessments?            | 471*                         | (3.1) | 505*          | (1.8) | 536*               | (2.1) |  |
| Level 3 problems: How often encountered in mathematics lessons?    | 503*                         | (3.2) | 508           | (2.0) | 523*               | (2.2) |  |
| Level 3 problems: How often encountered on assessments?            | 509*                         | (2.9) | 506           | (2.0) | 523*               | (2.2) |  |

\* Significant difference between:

<sup>1</sup> "Never or rarely" and "Often"

<sup>2</sup> "Sometimes" and "Never or rarely"

<sup>3</sup> "Often" and "Sometimes"

*Note:* Level 1 problems describe scenarios and provide all the information required to solve them. Level 3 problems do not describe scenarios but require the use of mathematical knowledge to form conclusions.

### TABLE A.3.1.1 Class size in Grade 8/Secondary II mathematics classes

|                           | Fewer<br>15 | than<br>i | 15–   | 19    | 20-   | -24    | 25-   | -29    | 30 or ı | more  |
|---------------------------|-------------|-----------|-------|-------|-------|--------|-------|--------|---------|-------|
| Canada and provinces      | %           | SE        | %     | SE    | %     | SE     | %     | SE     | %       | SE    |
| British Columbia          | 3.3‡        | (2.2)     | 10.1‡ | (3.1) | 36.0  | (4.5)  | 39.9  | (4.4)  | 10.6‡   | (2.8) |
| Alberta                   | 4.4‡        | (3.7)     | 9.5‡  | (3.3) | 19.3‡ | (4.1)  | 39.1  | (4.8)  | 27.7    | (4.2) |
| Saskatchewan              | 14.4‡       | (3.7)     | 22.3  | (3.9) | 24.4  | (4.0)  | 27.3  | (3.9)  | 11.5‡   | (2.7) |
| Manitoba                  | 9.3‡        | (2.6)     | 21.1  | (3.7) | 39.0  | (4.3)  | 22.5  | (3.6)  | 8.0‡    | (2.3) |
| Ontario                   | 6.8‡        | (2.3)     | 5.5‡  | (1.8) | 30.3  | (4.5)  | 43.5  | (4.5)  | 13.9‡   | (3.0) |
| Quebec                    | 5.4‡        | (2.3)     | 9.1‡  | (2.9) | 14.7‡ | (3.2)  | 53.9  | (4.6)  | 16.9‡   | (3.2) |
| New Brunswick             | 7.9‡        | (2.2)     | 16.5  | (3.4) | 29.3  | (5.2)  | 34.9  | (5.6)  | 11.4‡   | (3.8) |
| Nova Scotia               | 6.3‡        | (2.0)     | 17.9‡ | (4.2) | 36.4  | (5.4)  | 37.1  | (5.6)  | 2.3‡    | (1.0) |
| Prince Edward Island      | 8.4‡        | (4.2)     | 7.6‡  | (4.3) | 56.5‡ | (13.2) | 27.5‡ | (12.4) | 0.0     | (0.0) |
| Newfoundland and Labrador | 20.2        | (3.8)     | 8.3‡  | (2.5) | 17.7‡ | (4.7)  | 46.9‡ | (7.3)  | 6.8‡    | (4.4) |
| Canada                    | 6.5         | (1.2)     | 9.2   | (1.1) | 26.9  | (2.2)  | 42.4  | (2.3)  | 15.1    | (1.6) |

<sup>‡</sup> There are fewer than 30 observations.

|               | Mean score | SE     |
|---------------|------------|--------|
| Fewer than 15 | 469        | (18.5) |
| 15–19         | 501        | (9.2)  |
| 20–24         | 500        | (3.8)  |
| 25–29         | 510*       | (3.1)  |
| 30 or more    | 525*       | (4.6)  |

#### TABLE A.3.1.2 Achievement in mathematics by class size, Canada

\* Significant difference compared to the *fewer than 15* category

### TABLE A.3.1.3 Class size in Grade 8/Secondary II mathematics classes by language of the school system

|                           | Anglophone school systems  |        |       |       |       |        |       |        |            |       |
|---------------------------|----------------------------|--------|-------|-------|-------|--------|-------|--------|------------|-------|
|                           | Fewer t                    | han 15 | 15-   | ·19   | 20-   | -24    | 25–29 |        | 30 or more |       |
| Canada and provinces      | %                          | SE     | %     | SE    | %     | SE     | %     | SE     | %          | SE    |
| British Columbia          | 3.2‡                       | (2.2)  | 10.0‡ | (3.1) | 36.1  | (4.5)  | 40.0  | (4.5)  | 10.7‡      | (2.8) |
| Alberta                   | 3.9‡                       | (3.8)  | 9.6‡  | (3.4) | 19.1‡ | (4.2)  | 39.2  | (4.9)  | 28.1       | (4.3) |
| Saskatchewan              | 14.0‡                      | (3.8)  | 22.3‡ | (4.0) | 24.5  | (4.0)  | 27.6  | (3.9)  | 11.6‡      | (2.8) |
| Manitoba                  | 8.5‡                       | (2.6)  | 21.4‡ | (3.8) | 39.3  | (4.4)  | 22.8  | (3.7)  | 8.0‡       | (2.4) |
| Ontario                   | 6.7‡                       | (2.4)  | 5.4‡  | (1.9) | 29.9  | (4.7)  | 44.1  | (4.7)  | 13.9‡      | (3.2) |
| Quebec                    | 18.7‡                      | (6.2)  | 13.6‡ | (5.2) | 26.8‡ | (7.5)  | 35.4‡ | (9.0)  | 5.4‡       | (4.2) |
| New Brunswick             | 6.8‡                       | (2.6)  | 15.7‡ | (4.2) | 28.3‡ | (6.4)  | 41.4‡ | (7.0)  | 7.8‡       | (3.7) |
| Nova Scotia               | 5.9‡                       | (2.0)  | 18.4‡ | (4.4) | 37.1  | (5.6)  | 36.5  | (5.7)  | 2.1‡       | (1.0) |
| Prince Edward Island      | 5.0‡                       | (3.1)  | 6.7‡  | (4.3) | 60.0‡ | (14.0) | 28.3‡ | (13.4) | 0.0        | (0.0) |
| Newfoundland and Labrador | 19.9                       | (3.8)  | 8.3‡  | (2.5) | 17.8‡ | (4.7)  | 47.1‡ | (7.4)  | 6.9‡       | (4.4) |
| Canada                    | 6.8                        | (1.4)  | 9.3   | (1.3) | 29.2  | (2.6)  | 40.3  | (2.6)  | 14.4       | (1.8) |
|                           | Francophone school systems |        |       |       |       |        |       |        |            |       |

|                           |       |         |       |        | •     | •      |       |        |            |       |
|---------------------------|-------|---------|-------|--------|-------|--------|-------|--------|------------|-------|
|                           | Fewer | than 15 | 15-   | -19    | 20-   | -24    | 25–29 |        | 30 or more |       |
| Canada and provinces      | %     | SE      | %     | SE     | %     | SE     | %     | SE     | %          | SE    |
| British Columbia          | 22.2‡ | (13.9)  | 33.3‡ | (15.7) | 22.2‡ | (13.9) | 22.2‡ | (13.9) | 0.0        | (0.0) |
| Alberta                   | 36.0‡ | (15.2)  | 0.0   | (0.0)  | 36.0‡ | (17.6) | 28.0‡ | (19.1) | 0.0        | (0.0) |
| Saskatchewan              | 57.1‡ | (18.7)  | 28.6‡ | (17.1) | 14.3‡ | (13.2) | 0.0   | (0.0)  | 0.0        | (0.0) |
| Manitoba                  | 38.9‡ | (11.5)  | 11.1‡ | (7.4)  | 27.8‡ | (10.6) | 11.1‡ | (7.4)  | 11.1‡      | (7.4) |
| Ontario                   | 9.7‡  | (2.6)   | 6.5‡  | (2.0)  | 37.4  | (7.0)  | 31.6‡ | (6.3)  | 14.8‡      | (6.3) |
| Quebec                    | 3.5‡  | (2.4)   | 8.5‡  | (3.2)  | 12.9‡ | (3.4)  | 56.5  | (5.1)  | 18.6‡      | (3.6) |
| New Brunswick             | 11.0‡ | (3.9)   | 18.7‡ | (5.4)  | 31.9‡ | (8.0)  | 17.6‡ | (6.4)  | 20.9‡      | (9.1) |
| Nova Scotia               | 14.3‡ | (10.4)  | 7.1‡  | (7.3)  | 21.4‡ | (15.0) | 50.0‡ | (20.8) | 7.1‡       | (7.3) |
| Prince Edward Island      |       |         |       |        |       |        |       |        |            |       |
| Newfoundland and Labrador |       |         |       |        |       |        |       |        |            |       |
| Canada                    | 5.4   | (2.0)   | 8.6   | (2.7)  | 16.9  | (3.0)  | 51.4  | (4.3)  | 17.8       | (3.1) |

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

|                           | Ne    | ver    | Up to 1/4 of<br>the time |        | Up to 1/2 of<br>the time |       | Most<br>the t | or all<br>ime |
|---------------------------|-------|--------|--------------------------|--------|--------------------------|-------|---------------|---------------|
| Canada and provinces      | %     | SE     | %                        | SE     | %                        | SE    | %             | SE            |
| British Columbia          | 40.7  | (4.6)  | 23.6                     | (4.0)  | 12.7‡                    | (3.1) | 23.1          | (3.9)         |
| Alberta                   | 42.2  | (5.0)  | 24.1                     | (4.1)  | 9.0‡                     | (2.9) | 24.6‡         | (5.0)         |
| Saskatchewan              | 46.5  | (4.7)  | 15.7‡                    | (3.2)  | 9.7‡                     | (2.9) | 28.1          | (4.1)         |
| Manitoba                  | 36.5  | (4.3)  | 32.7                     | (4.1)  | 9.9‡                     | (2.5) | 20.8          | (3.7)         |
| Ontario                   | 66.0  | (4.4)  | 20.3                     | (3.3)  | 6.5‡                     | (2.3) | 7.2‡          | (3.2)         |
| Quebec                    | 62.3  | (4.6)  | 27.3                     | (4.2)  | 7.3‡                     | (2.7) | 3.1‡          | (1.3)         |
| New Brunswick             | 42.7  | (5.6)  | 23.9                     | (4.7)  | 10.3‡                    | (3.5) | 23.1          | (4.7)         |
| Nova Scotia               | 33.3  | (5.3)  | 51.0                     | (5.6)  | 9.8‡                     | (2.7) | 6.0‡          | (2.7)         |
| Prince Edward Island      | 39.7‡ | (13.6) | 51.9‡                    | (13.7) | 4.6‡                     | (2.8) | 3.8‡          | (2.8)         |
| Newfoundland and Labrador | 43.6  | (7.0)  | 49.8                     | (7.2)  | 4.6‡                     | (1.9) | 2.0‡          | (1.0)         |
| Canada                    | 55.7  | (2.3)  | 24.2                     | (1.8)  | 8.0                      | (1.2) | 12.1          | (1.6)         |

### TABLE A.3.2.1 Percentage of teachers indicating that other adults were present in their mathematics class

**‡** There are fewer than 30 observations.

## TABLE A.3.2.2 Achievement in mathematics by presence of other adults in the mathematics classroom, Canada

|                       | Mean score | SE    |
|-----------------------|------------|-------|
| Never                 | 515        | (3.5) |
| Up to 1/4 of the time | 501*       | (3.2) |
| Up to 1/2 of the time | 487*       | (7.6) |
| Most or all the time  | 484*       | (9.2) |

\* Significant difference compared to the *never* category

### TABLE A.3.3 Percentage of teachers and mathematics achievement by accomodations and modifications used in mathematics classrooms, Canada

|  | Never |       |        |        |       | Rarely |      |        |  |
|--|-------|-------|--------|--------|-------|--------|------|--------|--|
|  | %     | SE    | Mean   | SE     | %     | SE     | Mean | SE     |  |
| Program modifications (e.g., altering course expectations)                       | 13.4  | (1.5) | 540    | (5.8)  | 8.3   | (1.1)  | 520* | (5.7)  |  |
| More time in which to accomplish a task  | 1.2‡  | (0.5) | 561    | (13.8) | 1.0‡  | (0.3)  | 543  | (21.7) |  |
| Adapted teaching methods   | 1.7‡  | (0.5) | 547    | (12.0) | 3.2   | (0.7)  | 543  | (9.9)  |  |
| Withdrawal of student from class (e.g., for a short-term, targeted intervention) | 25.6  | (2.2) | 518    | (4.5)  | 26.1  | (2.0)  | 512  | (4.3)  |  |
| Assistive technologies   | 21.5  | (1.7) | 516    | (6.7)  | 24.3  | (1.9)  | 515  | (4.9)  |  |
| Help of an education assistant (e.g., teaching aide, interpreter)                | 38.8  | (2.3) | 519    | (4.5)  | 13.7  | (1.7)  | 501* | (5.7)  |  |
|  |       | Some  | etimes |        | Often |        |      |        |  |

|  | Sometimes |       |      |       | Often |       |      |       |
|--|-----------|-------|------|-------|-------|-------|------|-------|
|  | %         | SE    | Mean | SE    | %     | SE    | Mean | SE    |
| Program modifications (e.g., altering course expectations)                       | 30.8      | (2.1) | 509* | (3.7) | 47.6  | (2.4) | 492* | (3.9) |
| More time in which to accomplish a task  | 22.3      | (1.9) | 523* | (4.8) | 75.5  | (1.9) | 499* | (2.9) |
| Adapted teaching methods   | 36.1      | (2.2) | 517* | (3.5) | 59.1  | (2.3) | 496* | (3.4) |
| Withdrawal of student from class (e.g., for a short-term, targeted intervention) | 33.5      | (2.2) | 495* | (4.8) | 14.8  | (1.5) | 498* | (6.3) |
| Assistive technologies   | 35.4      | (2.3) | 501* | (3.1) | 18.9  | (1.9) | 493* | (6.3) |
| Help of an education assistant (e.g., teaching aide, interpreter)                | 22.4      | (1.8) | 508  | (3.7) | 25.0  | (2.0) | 488* | (5.3) |

\* Significant difference compared to the *never* category

‡ There are fewer than 30 observations.

|                           | One grade only |        | Two grade levels |        | Three o<br>grade | r more<br>levels |
|---------------------------|----------------|--------|------------------|--------|------------------|------------------|
| Canada and provinces      | %              | SE     | %                | SE     | %                | SE               |
| British Columbia          | 73.7           | (4.1)  | 12.8‡            | (3.1)  | 13.5‡            | (3.3)            |
| Alberta                   | 70.0           | (4.6)  | 14.9‡            | (3.5)  | 15.1‡            | (3.7)            |
| Saskatchewan              | 55.8           | (4.6)  | 32.0             | (4.5)  | 12.2‡            | (3.0)            |
| Manitoba                  | 55.9           | (4.4)  | 29.3             | (4.2)  | 14.7‡            | (3.1)            |
| Ontario                   | 60.4           | (4.6)  | 30.6             | (4.5)  | 9.0‡             | (2.6)            |
| Quebec                    | 94.2           | (1.8)  | 4.9‡             | (1.7)  | 0.9‡             | (0.5)            |
| New Brunswick             | 60.6           | (5.4)  | 24.4             | (4.8)  | 15.0‡            | (3.8)            |
| Nova Scotia               | 85.8           | (3.3)  | 11.1‡            | (3.0)  | 3.1‡             | (1.5)            |
| Prince Edward Island      | 77.1‡          | (10.9) | 13.7‡            | (10.0) | 9.2‡             | (5.4)            |
| Newfoundland and Labrador | 80.7           | (5.7)  | 12.0‡            | (4.5)  | 7.3‡             | (3.9)            |
| Canada                    | 69.6           | (2.3)  | 21.2             | (2.2)  | 9.2              | (1.3)            |

### TABLE A.3.4.1 Percentage of teachers by number of grade levels in their mathematics classrooms

<sup>‡</sup> There are fewer than 30 observations.

### TABLE A.3.4.2 Achievement in mathematics by number of grade levels, Canada

|                            | Mean score | SE    |
|----------------------------|------------|-------|
| One grade only             | 512        | (2.8) |
| Two grade levels           | 494*       | (5.8) |
| Three or more grade levels | 490*       | (9.6) |

\* Significant difference compared to the one grade only category

### TABLE A.3.5.1 Percentage of teachers by number of days taught by a substitute teacher

|                           | 5 or few | ver days | 6–9 days |        | 10–19 days |       | 20 or more da |        |
|---------------------------|----------|----------|----------|--------|------------|-------|---------------|--------|
| Canada and provinces      | %        | SE       | %        | SE     | %          | SE    | %             | SE     |
| British Columbia          | 61.8     | (4.4)    | 20.7     | (3.6)  | 10.5‡      | (2.8) | 7.0‡          | (2.4)  |
| Alberta                   | 54.6     | (5.0)    | 23.2     | (4.0)  | 12.7‡      | (3.2) | 9.6‡          | (2.8)  |
| Saskatchewan              | 51.5     | (4.6)    | 26.9     | (4.1)  | 16.3‡      | (3.7) | 5.3‡          | (2.2)  |
| Manitoba                  | 44.1     | (4.4)    | 38.3     | (4.3)  | 11.2‡      | (2.8) | 6.5‡          | (2.1)  |
| Ontario                   | 44.2     | (4.5)    | 34.1     | (4.5)  | 16.4       | (3.3) | 5.2‡          | (1.8)  |
| Quebec                    | 74.7     | (4.0)    | 16.6‡    | (3.4)  | 6.0‡       | (2.2) | 2.7‡          | (1.1)  |
| New Brunswick             | 47.8     | (5.5)    | 33.4     | (5.5)  | 5.7‡       | (1.9) | 13.2‡         | (4.0)  |
| Nova Scotia               | 40.3     | (5.5)    | 39.3     | (5.5)  | 13.3‡      | (3.8) | 7.1‡          | (2.4)  |
| Prince Edward Island      | 30.5‡    | (12.2)   | 35.1‡    | (13.5) | 17.6‡      | (9.3) | 16.8‡         | (11.3) |
| Newfoundland and Labrador | 36.5     | (6.4)    | 38.6     | (7.0)  | 23.9‡      | (6.8) | 1.0‡          | (0.7)  |
| Canada                    | 52.9     | (2.3)    | 28.2     | (2.2)  | 13.0       | (1.6) | 5.8           | (1.0)  |

**‡** There are fewer than 30 observations.

|                 | Mean score | SE    |
|-----------------|------------|-------|
| 5 or fewer days | 511*       | (3.9) |
| 6–9 days        | 505*       | (4.0) |
| 10–19 days      | 489        | (5.3) |
| 20 or more days | 511*       | (7.0) |

### TABLE A.3.5.2 Achievement in mathematics by number of days taught by a substitute teacher, Canada

\* Significant difference compared to the 10 to 19 days category

### TABLE A.3.6.1 Percentage of teachers by reasons for lost instructional time, Canada

|   | Never |       | Rarely |       | Sometimes |       | O    | ten   |
|---|-------|-------|--------|-------|-----------|-------|------|-------|
|   | %     | SE    | %      | SE    | %         | SE    | %    | SE    |
| Student misbehaviour  | 13.9  | (1.6) | 34.3   | (2.2) | 33.7      | (2.3) | 18.1 | (1.7) |
| Other disruptions (e.g., announcements, assemblies, visits) | 2.0   | (0.5) | 44.6   | (2.3) | 46.8      | (2.4) | 6.6  | (1.2) |
| Discussions unrelated to the mathematics lesson             | 9.2   | (1.3) | 60.6   | (2.3) | 27.1      | (2.0) | 3.2  | (0.8) |

### TABLE A.3.6.2 Achievement in mathematics by reasons for lost instructional time, Canada

|   | Never or rarely |       |      |       | Sometimes or often |       |      |       |
|---|-----------------|-------|------|-------|--------------------|-------|------|-------|
|   | %               | SE    | Mean | SE    | %                  | SE    | Mean | SE    |
| Student misbehaviour  | 48.2            | (2.3) | 518  | (3.7) | 51.8               | (2.3) | 495* | (3.1) |
| Other disruptions (e.g., announcements, assemblies, visits) | 46.5            | (2.3) | 513  | (2.9) | 53.5               | (2.3) | 500* | (3.8) |
| Discussions unrelated to the mathematics lesson             | 69.7            | (2.1) | 509  | (3.1) | 30.3               | (2.1) | 499  | (4.1) |

\* Significant difference compared to the *never or rarely* category

|                           | Female |        | Ма    | ale    | Anothe | er way | l prefer n | I prefer not to say |  |  |
|---------------------------|--------|--------|-------|--------|--------|--------|------------|---------------------|--|--|
| Canada and provinces      | %      | SE     | %     | SE     | %      | SE     | %          | SE                  |  |  |
| British Columbia          | 47.4   | (4.6)  | 51.7  | (4.6)  | 0.0    | (0.0)  | 0.9‡       | (0.8)               |  |  |
| Alberta                   | 37.5   | (4.8)  | 57.6  | (5.0)  | 0.0    | (0.0)  | 4.9‡       | (2.4)               |  |  |
| Saskatchewan              | 42.3   | (4.5)  | 56.5  | (4.5)  | 0.0    | (0.0)  | 1.2‡       | (0.9)               |  |  |
| Manitoba                  | 32.8   | (4.1)  | 65.3  | (4.1)  | 0.0    | (0.0)  | 1.9‡       | (1.3)               |  |  |
| Ontario                   | 37.8   | (4.3)  | 60.8  | (4.3)  | 0.0    | (0.0)  | 1.4‡       | (0.9)               |  |  |
| Quebec                    | 33.3   | (4.4)  | 66.0  | (4.4)  | 0.0    | (0.0)  | 0.7‡       | (0.7)               |  |  |
| New Brunswick             | 35.1   | (5.2)  | 63.7  | (5.3)  | 1.2‡   | (1.2)  | 0.0        | (0.0)               |  |  |
| Nova Scotia               | 37.1   | (5.3)  | 61.8  | (5.4)  | 0.0    | (0.0)  | 1.1‡       | (1.1)               |  |  |
| Prince Edward Island      | 44.3‡  | (13.6) | 55.7‡ | (13.6) | 0.0    | (0.0)  | 0.0        | (0.0)               |  |  |
| Newfoundland and Labrador | 49.1   | (7.2)  | 50.9  | (7.2)  | 0.0    | (0.0)  | 0.0        | (0.0)               |  |  |
| Canada                    | 38.0   | (2.2)  | 60.3  | (2.2)  | 0.0‡   | (0.0)  | 1.6‡       | (0.5)               |  |  |

### TABLE A.3.7.1 Percentage of teachers by gender self-identification

**‡** There are fewer than 30 observations.

### TABLE A.3.7.2 Achievement in mathematics by gender self-identification of teacher, Canada

|                     | Mean score | SE     |
|---------------------|------------|--------|
| Female              | 509        | (3.1)  |
| Male                | 503        | (3.6)  |
| Another way         |            |        |
| l prefer not to say | 540*       | (13.8) |

\* Significant difference compared to the *female* category

*Note:* Due to sample size and confidentiality issues, mathematics achievement of the one class whose teacher chose to identify themselves in another way is not reported.

|   | Yes  |       |      |        |      | No    |      |       |  |
|---|------|-------|------|--------|------|-------|------|-------|--|
|   | %    | SE    | Mean | SE     | %    | SE    | Mean | SE    |  |
| ВА  | 35.0 | (2.4) | 501  | (4.3)  | 65.0 | (2.4) | 509  | (3.0) |  |
| BEd or equivalent (e.g., at least one year of teacher training) | 88.3 | (1.4) | 505  | (2.7)  | 11.7 | (1.4) | 514  | (6.0) |  |
| BSc   | 24.1 | (1.9) | 507  | (6.0)  | 75.9 | (1.9) | 506  | (2.7) |  |
| Other bachelor's degree   | 12.8 | (1.5) | 513  | (6.2)  | 87.2 | (1.5) | 505  | (2.7) |  |
| MEd   | 9.3  | (1.1) | 506  | (6.7)  | 90.7 | (1.1) | 506  | (2.6) |  |
| Other master's degree   | 8.9  | (1.7) | 511  | (6.6)  | 91.1 | (1.7) | 506  | (2.6) |  |
| PhD or equivalent   | 0.7‡ | (0.3) | 544  | (20.5) | 99.3 | (0.3) | 506  | (2.5) |  |
| Other degree or diploma   | 8.0  | (1.2) | 497  | (6.5)  | 92.0 | (1.2) | 507  | (2.6) |  |

### TABLE A.3.8.1 Percentage of teachers and mathematics achievement by academic credentials, Canada

**‡** There are fewer than 30 observations.

### TABLE A.3.8.2 Percentage of teachers by courses taken during postsecondary studies, Canada

|  | 0 courses |       | 1–2 courses |       | 3–5 courses |       | 6–9 co | 6–9 courses |      | more<br>rses |
|--|-----------|-------|-------------|-------|-------------|-------|--------|-------------|------|--------------|
|  | %         | SE    | %           | SE    | %           | SE    | %      | SE          | %    | SE           |
| Courses in mathematics or mathematics-<br>related subjects (mathematics content) | 12.4      | (1.9) | 30.9        | (2.2) | 23.9        | (1.9) | 12.7   | (1.3)       | 20.0 | (1.6)        |
| Mathematics teaching-methodology courses (mathematics pedagogy)                  | 21.8      | (2.0) | 51.7        | (2.3) | 19.5        | (1.6) | 4.7    | (0.8)       | 2.3  | (0.5)        |

### TABLE A.3.8.3 Percentage of teachers by areas of study during formal pre-service education or training, Canada

|                               | Not at all |       | Overview or<br>introduction to<br>the topic |       | Are<br>empl | a of<br>hasis |
|-------------------------------|------------|-------|---|-------|-------------|---------------|
|                               | %          | SE    | %   | SE    | %           | SE            |
| Mathematics                   | 10.9       | (1.7) | 47.4  | (2.3) | 41.7        | (2.2)         |
| Pedagogy/teaching mathematics | 13.5       | (1.8) | 52.6  | (2.3) | 34.0        | (2.1)         |
| Educational psychology        | 5.9        | (1.0) | 71.1  | (2.1) | 23.0        | (2.0)         |
| Remedial mathematics          | 46.1       | (2.4) | 47.6  | (2.4) | 6.4         | (1.3)         |
| Special education             | 32.4       | (2.1) | 47.7  | (2.4) | 19.9        | (2.1)         |
| Assessment methods            | 4.9        | (0.8) | 68.8  | (2.1) | 26.3        | (2.0)         |

|                           | Fewer<br>5 ye | than<br>ars | 5–10 years 1 |       | 11–15 | 11–15 years |       | 16–20 years |       | More than<br>20 years |  |
|---------------------------|---------------|-------------|--------------|-------|-------|-------------|-------|-------------|-------|-----------------------|--|
| Canada and provinces      | %             | SE          | %            | SE    | %     | SE          | %     | SE          | %     | SE                    |  |
| British Columbia          | 17.2‡         | (3.5)       | 19.7‡        | (3.5) | 17.8  | (3.4)       | 21.3‡ | (3.8)       | 24.1  | (4.1)                 |  |
| Alberta                   | 19.8‡         | (4.7)       | 26.0         | (4.5) | 18.2‡ | (3.7)       | 15.9‡ | (3.4)       | 20.1‡ | (3.8)                 |  |
| Saskatchewan              | 20.3‡         | (3.7)       | 21.8         | (3.7) | 18.2‡ | (3.5)       | 14.0‡ | (3.1)       | 25.8  | (4.2)                 |  |
| Manitoba                  | 21.5          | (3.6)       | 16.2‡        | (3.3) | 26.1  | (3.9)       | 14.3‡ | (3.2)       | 21.9  | (3.6)                 |  |
| Ontario                   | 15.3          | (3.1)       | 24.7         | (4.3) | 21.6  | (3.6)       | 17.8  | (3.2)       | 20.5  | (3.7)                 |  |
| Quebec                    | 9.3‡          | (2.5)       | 12.2‡        | (3.0) | 25.5  | (4.1)       | 22.5  | (3.9)       | 30.5  | (4.3)                 |  |
| New Brunswick             | 7.9‡          | (2.8)       | 19.1         | (4.0) | 35.2  | (5.6)       | 16.2‡ | (4.0)       | 21.7‡ | (4.5)                 |  |
| Nova Scotia               | 11.4‡         | (3.7)       | 30.6         | (5.1) | 30.6  | (5.1)       | 13.0‡ | (3.9)       | 14.4‡ | (3.7)                 |  |
| Prince Edward Island      | 1.5‡          | (1.6)       | 6.1‡         | (4.8) | 19.1‡ | (10.4)      | 35.1‡ | (13.6)      | 38.2‡ | (13.4)                |  |
| Newfoundland and Labrador | 10.4‡         | (4.5)       | 20.8‡        | (5.4) | 19.6‡ | (5.2)       | 17.8‡ | (5.6)       | 31.4‡ | (7.1)                 |  |
| Canada                    | 15.1          | (1.6)       | 21.5         | (2.1) | 22.0  | (1.8)       | 18.4  | (1.7)       | 23.0  | (1.9)                 |  |

### TABLE A.3.8.4 Percentage of teachers by years of teaching experience

‡ There are fewer than 30 observations.

### TABLE A.3.8.5 Achievement in mathematics by teacher's years of teaching experience, Canada

|                    | Mean score | SE    |
|--------------------|------------|-------|
| Fewer than 5 years | 490        | (8.3) |
| 5–10 years         | 503        | (4.0) |
| 11–15 years        | 510*       | (4.7) |
| 16–20 years        | 515*       | (5.8) |
| More than 20 years | 509*       | (5.2) |

\* Significant difference compared to the *fewer than 5 years* category

|                           | Less than 20% |       | 20-   | 39%    | 40-   | 69%    | 70% o | r more |
|---------------------------|---------------|-------|-------|--------|-------|--------|-------|--------|
| Canada and provinces      | %             | SE    | %     | SE     | %     | SE     | %     | SE     |
| British Columbia          | 8.5‡          | (2.6) | 29.7  | (4.4)  | 26.8  | (4.1)  | 35.0  | (4.3)  |
| Alberta                   | 8.5‡          | (2.6) | 22.6‡ | (5.0)  | 34.0  | (4.7)  | 34.9  | (4.7)  |
| Saskatchewan              | 27.0          | (4.2) | 49.3  | (4.6)  | 16.4‡ | (3.4)  | 7.2‡  | (2.5)  |
| Manitoba                  | 16.9‡         | (3.4) | 31.9  | (4.1)  | 32.1  | (4.1)  | 19.1‡ | (3.6)  |
| Ontario                   | 9.5‡          | (2.6) | 53.2  | (4.5)  | 25.6  | (3.8)  | 11.7  | (2.6)  |
| Quebec                    | 0.6‡          | (0.4) | 1.4‡  | (0.9)  | 13.2‡ | (3.0)  | 84.8  | (3.2)  |
| New Brunswick             | 7.8‡          | (3.5) | 14.7  | (3.0)  | 33.8  | (5.0)  | 43.8  | (5.7)  |
| Nova Scotia               | 1.1‡          | (1.1) | 6.6‡  | (2.6)  | 37.6  | (5.6)  | 54.7  | (5.6)  |
| Prince Edward Island      | 0.0           | (0.0) | 43.5‡ | (13.4) | 39.7‡ | (13.5) | 16.8‡ | (11.2) |
| Newfoundland and Labrador | 5.9‡          | (3.9) | 8.3‡  | (2.7)  | 18.5‡ | (4.2)  | 67.3  | (6.0)  |
| Canada                    | 8.4           | (1.2) | 33.6  | (2.4)  | 25.0  | (1.9)  | 33.0  | (2.0)  |

### TABLE A.3.8.6 Percentage of teachers by proportion of schedule assigned to mathematics

**‡** There are fewer than 30 observations.

### TABLE A.3.8.7 Achievement in mathematics by proportion of teacher's schedule assigned to mathematics, Canada

|               | Mean score | SE    |
|---------------|------------|-------|
| Less than 20% | 496        | (8.0) |
| 20–39%        | 494        | (5.4) |
| 40–69%        | 509*       | (3.8) |
| 70% or more   | 520*       | (3.4) |

\* Significant difference compared to the 20 to 39% category

|                           | Yes, ba<br>educati | sed on<br>on only | Yes, based on<br>experience only |       | Yes, based on<br>education and<br>experience |        | No, I do not<br>consider myse<br>a mathematic<br>specialist |        |
|---------------------------|--------------------|-------------------|----------------------------------|-------|--|--------|---|--------|
| Canada and provinces      | %                  | SE                | %                                | SE    | %  | SE     | %   | SE     |
| British Columbia          | 2.9‡               | (1.5)             | 30.1                             | (4.2) | 39.9   | (4.5)  | 27.1  | (4.2)  |
| Alberta                   | 6.9‡               | (4.0)             | 26.5                             | (4.4) | 43.2   | (4.9)  | 23.4  | (4.2)  |
| Saskatchewan              | 1.0‡               | (0.9)             | 27.8                             | (4.0) | 32.6   | (4.5)  | 38.6  | (4.4)  |
| Manitoba                  | 2.2‡               | (1.2)             | 25.2                             | (3.8) | 44.3   | (4.4)  | 28.4  | (4.0)  |
| Ontario                   | 1.6‡               | (0.8)             | 23.2                             | (3.6) | 34.9   | (4.2)  | 40.3  | (4.7)  |
| Quebec                    | 2.5‡               | (1.4)             | 19.8                             | (3.7) | 72.4   | (4.1)  | 5.3‡  | (1.8)  |
| New Brunswick             | 5.4‡               | (2.7)             | 30.5                             | (5.1) | 38.1   | (5.5)  | 26.1  | (4.7)  |
| Nova Scotia               | 7.7‡               | (3.0)             | 13.9‡                            | (4.0) | 69.3   | (5.2)  | 9.1‡  | (3.3)  |
| Prince Edward Island      | 0.0                | (0.0)             | 6.1‡                             | (3.9) | 42.0‡  | (13.4) | 51.9‡   | (13.7) |
| Newfoundland and Labrador | 2.0‡               | (1.0)             | 7.0‡                             | (2.8) | 81.9   | (4.2)  | 9.1‡  | (2.9)  |
| Canada                    | 2.8                | (0.7)             | 23.7                             | (1.9) | 45.3   | (2.3)  | 28.2  | (2.3)  |

### TABLE A.3.8.8 Percentage of teachers by self-assessed specialization in mathematics

<sup>‡</sup> There are fewer than 30 observations.

### TABLE A.3.8.9 Achievement in mathematics by teacher's self-assessed specialization in mathematics, Canada

|   | Mean score | SE     |
|---|------------|--------|
| Yes, based on education only                          | 470        | (30.8) |
| Yes, based on experience only                         | 511        | (4.8)  |
| Yes, based on education and experience                | 510*       | (3.0)  |
| No, I do not consider myself a mathematics specialist | 499        | (4.9)  |

\* Significant difference compared to the no, I do not consider myself a mathematics specialist category

|                           |       | U     |          |        | •        | •     |          |        |                |        |
|---------------------------|-------|-------|----------|--------|----------|-------|----------|--------|----------------|--------|
|                           | 0 da  | ays   | 1–2 days |        | 3–4 days |       | 5–8 days |        | 9 or more days |        |
| Canada and provinces      | %     | SE    | %        | SE     | %        | SE    | %        | SE     | %              | SE     |
| British Columbia          | 6.2‡  | (2.1) | 20.9‡    | (4.0)  | 22.3     | (3.8) | 21.8     | (3.7)  | 28.8           | (4.2)  |
| Alberta                   | 10.6‡ | (2.9) | 15.5‡    | (3.4)  | 22.6‡    | (4.8) | 25.9     | (4.3)  | 25.5           | (4.4)  |
| Saskatchewan              | 12.0‡ | (3.3) | 33.7     | (4.3)  | 28.4     | (4.1) | 14.3‡    | (3.2)  | 11.7‡          | (2.8)  |
| Manitoba                  | 6.4‡  | (2.0) | 14.7‡    | (3.1)  | 29.6     | (4.0) | 21.7     | (3.7)  | 27.6           | (4.0)  |
| Ontario                   | 4.6‡  | (1.7) | 8.9‡     | (2.5)  | 21.4     | (3.7) | 27.3     | (4.4)  | 37.8           | (4.3)  |
| Quebec                    | 17.9‡ | (3.7) | 20.9     | (3.7)  | 25.5     | (4.1) | 14.2‡    | (3.2)  | 21.6           | (3.8)  |
| New Brunswick             | 23.8‡ | (5.0) | 27.6     | (5.2)  | 22.6     | (4.4) | 10.0‡    | (2.7)  | 16.0‡          | (4.0)  |
| Nova Scotia               | 4.3‡  | (2.6) | 12.3‡    | (3.8)  | 23.0‡    | (4.9) | 26.5‡    | (5.0)  | 33.9           | (5.1)  |
| Prince Edward Island      | 0.0   | (0.0) | 12.2‡    | (10.0) | 3.1‡     | (2.3) | 22.9‡    | (11.4) | 61.8‡          | (13.3) |
| Newfoundland and Labrador | 7.8‡  | (3.5) | 11.4‡    | (4.4)  | 28.1‡    | (6.5) | 37.8     | (7.2)  | 14.9‡          | (4.3)  |
| Canada                    | 8.8   | (1.1) | 15.0     | (1.5)  | 23.1     | (1.9) | 23.1     | (2.1)  | 30.0           | (2.1)  |

 TABLE A.3.9.1
 Percentage of teachers by number of days of professional development activities related to the teaching of mathematics in the past five years

**‡** There are fewer than 30 observations.

|   | Yes  |       | No   |       | Impact on student learning |       |          |       |      |       |       |       |  |
|---|------|-------|------|-------|----------------------------|-------|----------|-------|------|-------|-------|-------|--|
|   |      |       |      |       | None                       |       | A little |       | Some |       | A lot |       |  |
|   | %    | SE    | %    | SE    | %                          | SE    | %        | SE    | %    | SE    | %     | SE    |  |
| Academic courses (e.g., university)   | 28.7 | (2.1) | 71.3 | (2.1) | 2.4‡                       | (1.0) | 18.0     | (3.7) | 47.9 | (4.3) | 31.7  | (3.8) |  |
| Workshops or conferences  |      | (1.2) | 7.9  | (1.2) | 2.0‡                       | (0.6) | 14.1     | (1.6) | 57.1 | (2.5) | 26.8  | (2.4) |  |
| Professional learning communities   |      | (2.0) | 24.6 | (2.0) | 2.3‡                       | (0.8) | 20.1     | (2.1) | 47.1 | (2.8) | 30.6  | (2.7) |  |
| Development of common assessment items  |      | (2.1) | 31.4 | (2.1) | 2.2‡                       | (0.9) | 19.4     | (2.4) | 52.6 | (3.0) | 25.7  | (2.3) |  |
| Online training (e.g., webinars, videos)  | 36.6 | (2.3) | 63.4 | (2.3) | 6.2‡                       | (1.9) | 33.1     | (3.7) | 48.9 | (3.9) | 11.8  | (2.4) |  |
| Integration of technology into teaching   | 78.5 | (1.8) | 21.5 | (1.8) | 2.6‡                       | (0.8) | 19.7     | (1.9) | 47.8 | (2.8) | 29.9  | (2.7) |  |
| Formative assessment (assessment for learning, assessment as learning)  |      | (2.2) | 26.9 | (2.2) | 1.2‡                       | (0.5) | 17.6     | (2.1) | 46.5 | (2.7) | 34.8  | (2.5) |  |
| Differentiated instruction/resources to<br>adapt to students' interests and needs                             | 77.5 | (1.8) | 22.5 | (1.8) | 1.4‡                       | (0.6) | 21.2     | (2.1) | 44.4 | (2.8) | 33.0  | (2.6) |  |
| Implementation of new resources   | 65.2 | (2.1) | 34.8 | (2.1) | 1.4‡                       | (0.7) | 19.6     | (2.3) | 51.2 | (3.1) | 27.8  | (3.0) |  |
| Teaching strategies   | 81.2 | (1.6) | 18.8 | (1.6) | 0.8‡                       | (0.5) | 13.8     | (1.7) | 55.8 | (2.6) | 29.6  | (2.3) |  |
| Mathematics content knowledge   | 55.9 | (2.3) | 44.1 | (2.3) | 1.1‡                       | (0.7) | 18.3     | (2.5) | 48.7 | (3.5) | 32.0  | (3.4) |  |
| Responding to assessment data (school, provincial, national, international)                                   | 58.7 | (2.2) | 41.3 | (2.2) | 11.5                       | (2.2) | 35.3     | (3.3) | 41.0 | (3.2) | 12.2  | (1.9) |  |
| Receiving support in mathematics<br>teaching (e.g., lead teachers, coaches,<br>mentors, numeracy specialists) | 61.5 | (2.2) | 38.5 | (2.2) | 4.5‡                       | (1.6) | 20.5     | (2.5) | 50.1 | (3.3) | 24.9  | (2.6) |  |
| Receiving instructional feedback from an administrator  |      | (2.4) | 46.9 | (2.4) | 7.0                        | (1.7) | 30.2     | (3.2) | 45.7 | (3.6) | 17.2  | (2.5) |  |
| Mental health literacy/well-being   | 58.3 | (2.3) | 41.7 | (2.3) | 3.2                        | (0.9) | 29.0     | (2.8) | 48.5 | (3.2) | 19.3  | (2.6) |  |
| Social-emotional learning/self-regulation   |      | (2.4) | 53.0 | (2.4) | 3.4‡                       | (1.1) | 29.2     | (3.2) | 48.7 | (3.6) | 18.8  | (2.9) |  |

# TABLE A.3.9.2 Percentage of teachers by types of professional development and their perceived impact on student learning, Canada

‡ There are fewer than 30 observations.
#### TABLE A.3.9.3 Percentage of teachers by types of collaboration with other mathematics teachers, Canada

|   | Nev<br>alm<br>ne | Never or<br>almost<br>never |      | Several<br>times per<br>year |      | 2 or 3 times<br>per month |      | 1 to 3 times<br>per week |      | ly or<br>t daily |
|---|------------------|-----------------------------|------|------------------------------|------|---------------------------|------|--------------------------|------|------------------|
|   | %                | SE                          | %    | SE                           | %    | SE                        | %    | SE                       | %    | SE               |
| Discussion of how to teach a particular topic                   | 11.9             | (1.5)                       | 27.8 | (2.1)                        | 24.0 | (1.9)                     | 23.4 | (1.9)                    | 12.9 | (1.9)            |
| Collaboration on planning and preparing instructional materials | 21.1             | (1.8)                       | 29.2 | (2.1)                        | 23.8 | (2.0)                     | 18.0 | (2.0)                    | 7.9  | (1.1)            |
| Sharing what I have learned about my<br>teaching experiences    | 13.7             | (1.5)                       | 35.2 | (2.2)                        | 25.0 | (2.0)                     | 18.2 | (2.0)                    | 7.9  | (1.1)            |
| Visiting another classroom to learn more about teaching         | 72.6             | (2.2)                       | 17.7 | (2.0)                        | 4.7  | (0.9)                     | 3.8  | (0.9)                    | 1.1‡ | (0.3)            |
| Analyzing assessment data and using it to<br>inform instruction | 26.5             | (2.0)                       | 43.6 | (2.3)                        | 17.2 | (2.0)                     | 9.3  | (1.3)                    | 3.5  | (0.8)            |
| Working together on scoring student work                        | 59.4             | (2.3)                       | 28.4 | (2.3)                        | 8.2  | (1.2)                     | 3.0  | (0.7)                    | 0.9‡ | (0.3)            |
| Developing common assessments                                   | 34.2             | (2.2)                       | 40.1 | (2.3)                        | 16.5 | (1.6)                     | 6.0  | (1.1)                    | 3.3  | (0.8)            |

‡ There are fewer than 30 observations.

#### TABLE A.3.10 Percentage of teachers by confidence in their ability to do mathematics and to help students understand mathematics, Canada

|   | Not a<br>confi | at all<br>ident | Some<br>confi | ewhat<br>ident | Ve<br>confi | ry<br>dent |
|---|----------------|-----------------|---------------|----------------|-------------|------------|
|   | %              | SE              | %             | SE             | %           | SE         |
| Confident in their ability to do mathematics                          |                |                 |               |                |             |            |
| Paper-pencil calculations   | 0.2‡           | (0.2)           | 7.9           | (1.4)          | 91.8        | (1.5)      |
| Mental math   | 1.7‡           | (0.6)           | 24.7          | (2.0)          | 73.6        | (2.1)      |
| Estimation  | 0.3‡           | (0.2)           | 16.5          | (1.8)          | 83.2        | (1.8)      |
| Solve complex problems  | 1.6‡           | (0.8)           | 30.1          | (2.1)          | 68.3        | (2.2)      |
| Use technology  | 4.9            | (0.9)           | 41.0          | (2.3)          | 54.1        | (2.3)      |
| Coding/programming  | 54.8           | (2.3)           | 36.4          | (2.2)          | 8.9         | (1.2)      |
| Use online platforms for instruction and/or assessment                | 15.0           | (1.6)           | 51.4          | (2.3)          | 33.6        | (2.2)      |
| Encourage students to use technology or online resources              | 8.4            | (1.1)           | 50.0          | (2.3)          | 41.6        | (2.3)      |
| Confidence in their ability to help students develop understanding in | n mather       | matics          |               |                |             |            |
| Numbers and operations  | 0.0            | (0.0)           | 7.5           | (1.1)          | 92.5        | (1.1)      |
| Geometry and measurement  | 0.2‡           | (0.2)           | 14.8          | (1.7)          | 85.0        | (1.7)      |
| Patterns and relationships (algebra)                                  | 0.2‡           | (0.2)           | 12.2          | (1.6)          | 87.6        | (1.6)      |
| Data management and probability (statistics)                          | 0.7‡           | (0.3)           | 15.6          | (1.5)          | 83.7        | (1.5)      |

|  | Stro<br>disa | ngly<br>gree | gly Disag<br>ree |       | Agree |       | Stro<br>ag | ongly<br>ree |
|--|--------------|--------------|------------------|-------|-------|-------|------------|--------------|
|  | %            | SE           | %                | SE    | %     | SE    | %          | SE           |
| Natural ability                            | 6.1          | (1.1)        | 39.1             | (2.3) | 49.3  | (2.3) | 5.6        | (1.0)        |
| Work ethic                                 | 0.5‡         | (0.3)        | 1.4‡             | (0.5) | 46.2  | (2.3) | 51.9       | (2.3)        |
| Teaching                                   | 0.0‡         | (0.0)        | 3.2              | (0.7) | 74.2  | (2.0) | 22.6       | (1.9)        |
| Parents/guardians                          | 1.4‡         | (0.5)        | 15.2             | (1.6) | 64.5  | (2.2) | 18.8       | (1.8)        |
| Peer influence                             | 0.8‡         | (0.4)        | 17.5             | (1.7) | 66.3  | (2.2) | 15.5       | (1.7)        |
| Prior knowledge                            | 0.0‡         | (0.0)        | 2.7              | (0.8) | 70.7  | (2.1) | 26.6       | (2.0)        |
| Availability of help outside the classroom | 3.4          | (0.8)        | 25.8             | (2.0) | 61.9  | (2.2) | 9.0        | (1.3)        |
| Learning from the errors they make         | 0.8‡         | (0.4)        | 3.2              | (0.8) | 60.0  | (2.2) | 36.0       | (2.2)        |
| My high expectations for all students      | 0.7‡         | (0.3)        | 10.6             | (1.3) | 57.6  | (2.3) | 31.1       | (2.2)        |

# TABLE A.3.11.1 Percentage of teachers by level of agreement with statements about factors influencing student performance in mathematics, Canada

<sup>‡</sup> There are fewer than 30 observations.

# TABLE A.3.11.2 Percentage of teachers by level of agreement with statements about attitudes related to teaching and learning mathematics, Canada

|   | Strongly<br>disagree |       | Disa | gree  | Ag   | ree   | Stro<br>ag | ngly<br>ree |
|---|----------------------|-------|------|-------|------|-------|------------|-------------|
|   | %                    | SE    | %    | SE    | %    | SE    | %          | SE          |
| It is possible for all students to succeed in mathematics   | 0.8‡                 | (0.3) | 6.6  | (1.1) | 40.5 | (2.2) | 52.1       | (2.3)       |
| Success in mathematics requires hard work   | 0.0‡                 | (0.0) | 4.9  | (0.9) | 49.2 | (2.3) | 45.9       | (2.4)       |
| Success in mathematics requires natural ability   | 14.1                 | (1.5) | 65.0 | (2.2) | 18.7 | (1.8) | 2.2‡       | (0.7)       |
| Practice is important for student learning in mathematics   | 0.1‡                 | (0.1) | 0.3‡ | (0.2) | 28.6 | (2.1) | 71.0       | (2.1)       |
| Because calculators are easily available, there is less need to emphasize basic computational skills in teaching mathematics                    | 38.2                 | (2.2) | 49.7 | (2.3) | 10.2 | (1.3) | 1.8‡       | (0.5)       |
| There is not enough emphasis on basic computational skills in the early grades  | 3.0                  | (0.8) | 23.1 | (1.9) | 41.3 | (2.3) | 32.5       | (2.3)       |
| Student success in mathematics requires good teaching   | 0.0‡                 | (0.0) | 2.3  | (0.5) | 55.6 | (2.3) | 42.0       | (2.3)       |
| By the time students reach Grade 8/Secondary II, the emphasis in mathematics teaching should be more on problem solving                         | 1.0‡                 | (0.5) | 27.5 | (2.1) | 55.6 | (2.3) | 16.0       | (1.7)       |
| Students should not be allowed to use calculators until they have mastered basic computational skills   | 8.9                  | (1.4) | 43.9 | (2.3) | 35.6 | (2.3) | 11.5       | (1.4)       |
| Students should be given the opportunity to engage in computational thinking (e.g., programming, coding, robotics) in the mathematics classroom | 2.0‡                 | (0.5) | 17.1 | (1.6) | 58.1 | (2.3) | 22.8       | (2.0)       |

|  | Ne   | ver   | Rai  | rely  | Some | times | Of   | en    |
|--|------|-------|------|-------|------|-------|------|-------|
|  | %    | SE    | %    | SE    | %    | SE    | %    | SE    |
| Mathematics curriculum documents   | 5.1  | (0.9) | 10.5 | (1.2) | 28.2 | (2.1) | 56.2 | (2.3) |
| Textbooks  | 11.2 | (1.4) | 17.4 | (1.7) | 33.8 | (2.4) | 37.6 | (2.2) |
| Teachers' guides   | 13.9 | (1.5) | 28.5 | (2.1) | 36.2 | (2.3) | 21.3 | (1.8) |
| Worksheets   | 1.8‡ | (0.8) | 13.6 | (1.7) | 45.8 | (2.4) | 38.8 | (2.2) |
| Other print resources  | 3.5  | (0.8) | 12.7 | (1.5) | 48.9 | (2.4) | 34.9 | (2.2) |
| Calculators  | 0.7‡ | (0.3) | 6.0  | (1.0) | 33.9 | (2.2) | 59.4 | (2.3) |
| Computer software  | 18.3 | (1.5) | 32.9 | (2.1) | 36.0 | (2.3) | 12.8 | (1.8) |
| Web-based resources (other than worksheets)  | 12.4 | (1.3) | 27.2 | (2.0) | 43.1 | (2.3) | 17.2 | (2.2) |
| Measuring devices (e.g., protractors, balances)  | 3.5  | (0.7) | 21.5 | (1.6) | 59.4 | (2.3) | 15.5 | (2.1) |
| Interactive white boards   | 26.7 | (2.1) | 12.9 | (1.6) | 21.4 | (1.9) | 39.1 | (2.3) |
| Packaged instructional programs (e.g., Fountas and Pinnell, Leaps and Bounds, Jump Math) | 51.7 | (2.4) | 24.8 | (2.2) | 17.0 | (1.9) | 6.6  | (1.3) |
| Activities you have designed   | 2.1  | (0.5) | 7.2  | (0.9) | 45.5 | (2.3) | 45.2 | (2.4) |
| Online platforms (e.g., Google Classroom)  | 26.0 | (1.8) | 20.3 | (2.1) | 26.7 | (2.1) | 27.0 | (2.1) |

### TABLE A.3.12.1 Percentage of teachers by resources used in mathematics instruction, Canada

**‡** There are fewer than 30 observations.

# TABLE A.3.12.2Percentage of teachers by use of variety of opportunities for students to show<br/>understanding in mathematics, Canada

|   | Never or<br>almost<br>never |       | r or 2 or 3 t<br>ost per m<br>er |       | 1 to 3 times<br>per week |       | Dai<br>aln<br>da | ly or<br>nost<br>nily |
|---|-----------------------------|-------|----------------------------------|-------|--------------------------|-------|------------------|-----------------------|
|   | %                           | SE    | %                                | SE    | %                        | SE    | %                | SE                    |
| Give oral explanations  | 1.9‡                        | (0.5) | 8.3                              | (1.2) | 27.7                     | (2.0) | 62.1             | (2.2)                 |
| Give written explanations   | 1.6                         | (0.5) | 9.9                              | (1.2) | 33.8                     | (2.3) | 54.7             | (2.3)                 |
| Use mathematical language   | 0.3‡                        | (0.2) | 1.6                              | (0.4) | 16.5                     | (1.6) | 81.6             | (1.6)                 |
| Justify their reasoning   | 0.3‡                        | (0.1) | 5.9                              | (0.9) | 28.6                     | (2.0) | 65.2             | (2.2)                 |
| Make generalizations and conjectures  | 8.0                         | (1.2) | 23.2                             | (1.9) | 37.0                     | (2.3) | 31.7             | (2.2)                 |
| Make connections among multiple representations (e.g., concrete, pictorial, symbolic, abstract, text) | 3.2                         | (0.7) | 20.0                             | (1.9) | 41.4                     | (2.3) | 35.5             | (2.3)                 |
| Integrate technology in their learning  | 20.5                        | (1.7) | 36.0                             | (2.2) | 29.0                     | (2.2) | 14.5             | (1.8)                 |
| Analyze sources of errors and identify ways to overcome them  | 5.6                         | (1.1) | 18.4                             | (1.6) | 43.8                     | (2.3) | 32.2             | (2.2)                 |

# TABLE A.3.12.3 Percentage of teachers by amount of time in mathematics class spent on mathematics subdomains, Canada

|  | Less | than a | verage t | ime   | More than average tim |       |      |       |  |
|--|------|--------|----------|-------|-----------------------|-------|------|-------|--|
|  | %    | SE     | Mean     | SE    | %                     | SE    | Mean | SE    |  |
| Numbers and operations                       | 53.8 | (2.3)  | 515      | (3.7) | 46.2                  | (2.3) | 496* | (3.2) |  |
| Geometry and measurement                     | 51.6 | (2.3)  | 502      | (3.5) | 48.4                  | (2.3) | 510  | (3.6) |  |
| Patterns and relationships (algebra)         | 53.2 | (2.3)  | 501      | (3.3) | 46.8                  | (2.3) | 512* | (3.8) |  |
| Data management and probability (statistics) | 51.4 | (2.3)  | 504      | (3.0) | 48.6                  | (2.3) | 508  | (4.0) |  |

\* Significant difference

### TABLE A.3.12.4 Percentage of teachers by understanding of vertical articulation, Canada

|   | Strongly<br>disagree |       | ngly Disagree<br>gree |       | Agree |       | Strongl<br>agree |       |
|---|----------------------|-------|-----------------------|-------|-------|-------|------------------|-------|
|   | %                    | SE    | %                     | SE    | %     | SE    | %                | SE    |
| I have a deep understanding of the mathematics concepts taught<br>in earlier grades and how they connect to the Grade 8/Secondary II<br>mathematics curriculum. | 1.3‡                 | (0.4) | 7.1                   | (1.0) | 48.4  | (2.3) | 43.1             | (2.3) |
| I have a deep understanding of the mathematics concepts taught<br>in later grades and how they connect to the Grade 8/Secondary II<br>mathematics curriculum.   | 2.2                  | (0.6) | 20.2                  | (2.1) | 45.6  | (2.3) | 32.0             | (2.1) |

<sup>‡</sup> There are fewer than 30 observations.

|                           | l do not<br>mather<br>home | lo not assign<br>1athematics<br>homework |       | Less than<br>30 minutes |       | Between<br>30 minutes<br>and<br>1 hour |       | Between Betw<br>30 minutes and<br>and<br>1 hour |       | een 1<br>hours | More<br>2 ho | than<br>ours |
|---------------------------|----------------------------|--|-------|-------------------------|-------|--|-------|---|-------|----------------|--------------|--------------|
| Canada and provinces      | %                          | SE                                       | %     | SE                      | %     | SE                                     | %     | SE  | %     | SE             |              |              |
| British Columbia          | 13.9‡                      | (3.2)                                    | 10.5‡ | (2.7)                   | 30.5  | (4.3)                                  | 30.8  | (4.3)   | 14.4‡ | (3.3)          |              |              |
| Alberta                   | 14.7‡                      | (3.8)                                    | 26.0‡ | (4.8)                   | 25.3  | (4.2)                                  | 28.0  | (4.3)   | 6.0‡  | (2.3)          |              |              |
| Saskatchewan              | 24.7                       | (3.9)                                    | 29.6  | (4.2)                   | 30.3  | (4.4)                                  | 14.0‡ | (3.2)   | 1.6‡  | (1.0)          |              |              |
| Manitoba                  | 23.6                       | (3.9)                                    | 22.1  | (3.6)                   | 30.0  | (4.1)                                  | 19.7  | (3.5)   | 4.7‡  | (1.9)          |              |              |
| Ontario                   | 7.8‡                       | (2.2)                                    | 19.5  | (4.2)                   | 25.0  | (3.8)                                  | 31.3  | (4.1)   | 16.4  | (3.3)          |              |              |
| Quebec                    | 5.6‡                       | (2.0)                                    | 8.6‡  | (2.8)                   | 34.2  | (4.5)                                  | 37.2  | (4.5)   | 14.3  | (3.0)          |              |              |
| New Brunswick             | 34.7                       | (5.2)                                    | 18.6  | (4.0)                   | 31.2  | (5.2)                                  | 14.4‡ | (4.6)   | 1.2‡  | (0.8)          |              |              |
| Nova Scotia               | 11.3‡                      | (3.4)                                    | 20.0‡ | (4.8)                   | 27.6  | (5.0)                                  | 29.5  | (5.1)   | 11.6‡ | (3.5)          |              |              |
| Prince Edward Island      | 3.1‡                       | (2.3)                                    | 35.1‡ | (13.0)                  | 32.1‡ | (13.5)                                 | 29.8‡ | (12.3)  | 0.0   | (0.0)          |              |              |
| Newfoundland and Labrador | 14.6‡                      | (5.5)                                    | 5.9‡  | (2.1)                   | 10.5‡ | (3.8)                                  | 44.4  | (7.1)   | 24.6‡ | (6.4)          |              |              |
| Canada                    | 11.1                       | (1.2)                                    | 17.9  | (2.1)                   | 27.7  | (2.0)                                  | 30.4  | (2.1)   | 12.9  | (1.6)          |              |              |

# TABLE A.3.13.1 Percentage of teachers by time they expect students to spend on mathematics homework each week

‡ There are fewer than 30 observations.

### TABLE A.3.13.2 Achievement in mathematics by time teachers expect students to spend on mathematics homework each week, Canada

|                                      | Mean score | SE    |
|--------------------------------------|------------|-------|
| I do not assign mathematics homework | 477        | (7.2) |
| Less than 30 minutes                 | 491        | (7.5) |
| Between 30 minutes and 1 hour        | 510*       | (3.6) |
| Between 1 and 2 hours                | 518*       | (4.3) |
| More than 2 hours                    | 514*       | (5.1) |

\* Significant difference compared to the less than 30 minutes category

|  | Never or<br>almost<br>never |       | 2 or<br>3 times per<br>month |       | or 1 to<br>s per 3 times<br>hth wee |       | to Dail<br>es per alm<br>eek da |       |
|--|-----------------------------|-------|------------------------------|-------|-------------------------------------|-------|---------------------------------|-------|
|  | % SE                        |       | %                            | % SE  |                                     | SE    | %                               | SE    |
| Paper-pencil calculations  | 10.5                        | (1.5) | 22.4                         | (2.3) | 42.1                                | (2.5) | 25.1                            | (2.1) |
| Word problems  | 4.2                         | (1.5) | 29.8                         | (2.3) | 45.0                                | (2.5) | 21.0                            | (2.0) |
| Projects   | 55.1                        | (2.6) | 40.6                         | (2.5) | 4.0‡                                | (1.2) | 0.3‡                            | (0.3) |
| Creating problems  | 58.1                        | (2.5) | 35.2                         | (2.4) | 6.4                                 | (1.4) | 0.3‡                            | (0.1) |
| Studying for assessments   | 8.2                         | (1.2) | 70.9                         | (2.1) | 17.0                                | (1.8) | 3.9                             | (0.7) |
| Activities using concrete or virtual manipulatives (e.g., base-<br>ten blocks, colour tiles, geometric solids) | 49.6                        | (2.5) | 31.7                         | (2.3) | 14.3                                | (1.8) | 4.4                             | (1.1) |
| New concepts not taught in class   | 75.8                        | (2.1) | 16.3                         | (1.7) | 6.6                                 | (1.3) | 1.3‡                            | (0.5) |
| Collaborative problem-solving activities   | 46.5                        | (2.5) | 31.7                         | (2.3) | 17.5                                | (2.0) | 4.4                             | (1.0) |
| Participation in online discussion forums or blogs   | 89.3                        | (1.6) | 8.5                          | (1.5) | 1.9‡                                | (0.7) | 0.3‡                            | (0.2) |
| Viewing videos online (e.g., YouTube, Vimeo) to review concepts  | 54.0                        | (2.5) | 32.8                         | (2.4) | 11.0                                | (2.0) | 2.2‡                            | (0.6) |

#### TABLE A.3.13.3 Percentage of teachers by types of homework assigned, Canada

<sup>‡</sup> There are fewer than 30 observations.

### TABLE A.3.13.4 Percentage of teachers by frequency of monitoring student homework, Canada

|   | Never |       | Ra   | rely  | Som  | Sometimes |      | ften  |
|---|-------|-------|------|-------|------|-----------|------|-------|
|   | %     | SE    | %    | SE    | %    | SE        | %    | SE    |
| Monitor whether or not the homework has been completed  | 2.3‡  | (0.8) | 6.3  | (1.0) | 29.0 | (2.2)     | 62.5 | (2.4) |
| Collect and correct the homework                        | 14.3  | (1.7) | 25.3 | (2.3) | 32.9 | (2.3)     | 27.4 | (2.2) |
| Have a class discussion on the homework                 | 0.5‡  | (0.2) | 2.4  | (0.4) | 23.4 | (2.2)     | 73.7 | (2.2) |
| Provide individual student feedback on the homework     | 3.3   | (0.8) | 18.9 | (1.9) | 47.3 | (2.5)     | 30.4 | (2.3) |
| Have students correct their homework in class           | 5.1   | (1.0) | 9.2  | (1.3) | 31.7 | (2.4)     | 54.0 | (2.5) |
| Use homework to contribute to students' marks or grades | 50.3  | (2.5) | 22.3 | (2.0) | 17.5 | (1.8)     | 10.0 | (1.4) |

#### TABLE A.3.14.1 Percentage of teachers by types of questions used on mathematics assessment, Canada

|   | Never |       | Rarely |       | Sometimes |       | Often |       |
|---|-------|-------|--------|-------|-----------|-------|-------|-------|
|   | %     | SE    | %      | SE    | %         | SE    | %     | SE    |
| Selected-response items (e.g., true/false, multiple choice)                       | 9.5   | (1.3) | 27.9   | (2.1) | 35.7      | (2.3) | 26.8  | (2.1) |
| Short-response items (e.g., one or two words, one-step problems, short sentences) | 0.0‡  | (0.0) | 5.6    | (1.1) | 40.2      | (2.3) | 54.1  | (2.3) |
| Extended-response items requiring multi-step solutions                            | 0.2‡  | (0.1) | 2.2‡   | (0.7) | 26.6      | (2.0) | 71.0  | (2.0) |
| Extended-response items requiring an explanation or justification                 | 1.1‡  | (0.5) | 6.7    | (0.9) | 37.8      | (2.2) | 54.4  | (2.3) |
| Extended-response items requiring students to generate problems                   | 22.0  | (1.8) | 31.3   | (2.1) | 35.8      | (2.3) | 10.9  | (1.5) |
|   |       |       |        |       |           |       |       |       |

**‡** There are fewer than 30 observations.

# TABLE A.3.14.2 Achievement in mathematics by frequency of teachers' use of extended-response items requiring multi-step solutions, Canada

| Mean score | SE                               |
|------------|----------------------------------|
| 444        | (32.0)                           |
| 493        | (4.9)                            |
| 513*       | (2.6)                            |
|            | Mean score<br>444<br>493<br>513* |

\* Significant difference compared to the sometimes category

# TABLE A.3.14.3 Percentage of teachers by mathematics assessment used to measure different levels of thinking, Canada

|  | Never |       | Rarely |       | Sometimes |       | Often |       |
|--|-------|-------|--------|-------|-----------|-------|-------|-------|
|  | %     | SE    | %      | SE    | %         | SE    | %     | SE    |
| Knowledge of facts and concepts (e.g., recall, identify, label)  | 0.5‡  | (0.2) | 7.8    | (1.2) | 33.0      | (2.1) | 58.8  | (2.2) |
| Ability to apply knowledge and understanding (e.g., solve a problem, apply information to a new context) | 0.0   | (0.0) | 0.9‡   | (0.6) | 16.0      | (1.6) | 83.1  | (1.7) |
| Ability to explain, justify, evaluate, and generalize  | 0.8‡  | (0.4) | 6.7    | (1.0) | 32.1      | (2.0) | 60.4  | (2.2) |
|  |       |       |        |       |           |       |       |       |

|  | Not at all |       | A little |       | More than<br>a little |       | Α    | lot   |
|--|------------|-------|----------|-------|-----------------------|-------|------|-------|
|  | %          | SE    | %        | SE    | %                     | SE    | %    | SE    |
| The range of student abilities in the class  | 3.1        | (0.7) | 23.0     | (1.9) | 33.2                  | (2.3) | 40.7 | (2.3) |
| Students coming from a wide variety of backgrounds (e.g., socioeconomic, linguistic, cultural, etc.) | 26.4       | (2.2) | 28.8     | (2.0) | 23.0                  | (1.9) | 21.9 | (1.9) |
| Disruptive students  | 15.0       | (1.6) | 36.9     | (2.2) | 24.2                  | (2.2) | 23.8 | (1.9) |
| Time of day for instruction  | 36.6       | (2.3) | 36.5     | (2.3) | 15.8                  | (1.5) | 11.1 | (1.4) |
| Pressure from parents/guardians  | 55.6       | (2.3) | 32.1     | (2.2) | 9.1                   | (1.2) | 3.2  | (0.7) |
| Curriculum inappropriate for the grade level   | 66.7       | (2.1) | 23.0     | (1.9) | 7.9                   | (1.2) | 2.4  | (0.6) |
| Shortage of computer hardware or software  | 51.9       | (2.3) | 27.4     | (2.2) | 11.6                  | (1.5) | 9.1  | (1.4) |
| Inadequate physical facilities   | 72.8       | (2.0) | 16.7     | (1.6) | 7.7                   | (1.3) | 2.8  | (0.7) |
| Too much content in the curriculum   | 34.6       | (2.2) | 33.4     | (2.1) | 22.1                  | (2.0) | 9.9  | (1.5) |
| Large class sizes  | 27.9       | (2.0) | 24.3     | (2.1) | 23.7                  | (2.1) | 24.1 | (1.9) |
| Low morale in the school   | 54.4       | (2.3) | 27.7     | (2.2) | 11.7                  | (1.5) | 6.2  | (1.0) |
| Concerns for personal safety or the safety of students   | 82.0       | (1.8) | 13.0     | (1.5) | 3.1                   | (0.8) | 1.8‡ | (0.7) |
| Inadequate resources for lesson planning   | 55.2       | (2.3) | 32.7     | (2.2) | 7.9                   | (1.3) | 4.2  | (1.0) |
| Insufficient time for planning   | 26.9       | (2.0) | 36.0     | (2.3) | 22.9                  | (2.0) | 14.2 | (1.5) |
| Limitations in my background in the subject  | 76.1       | (2.1) | 17.8     | (2.0) | 4.8                   | (0.9) | 1.2  | (0.3) |
| External assessments or standardized tests   | 58.1       | (2.3) | 27.4     | (2.1) | 10.8                  | (1.5) | 3.7  | (0.9) |
| Insufficient professional development  | 49.3       | (2.3) | 34.3     | (2.3) | 12.2                  | (1.5) | 4.2  | (1.0) |
| Inadequate collegial support (e.g., mentoring)   | 66.2       | (2.2) | 22.3     | (1.8) | 7.6                   | (1.3) | 3.9  | (0.9) |
| Inadequate support from school administrators  | 72.9       | (2.0) | 18.2     | (1.7) | 6.9                   | (1.1) | 2.1‡ | (0.6) |
| Shortage of resources for mathematics instruction  | 43.9       | (2.3) | 36.4     | (2.3) | 12.3                  | (1.5) | 7.5  | (1.2) |

### TABLE A.3.15.1 Percentage of teachers by challenges to mathematics teaching, Canada

# TABLE A.3.15.2 Relationship between the challenges to teaching index and mathematics achievement, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second quarter <sup>2</sup> |       | Third quarter <sup>3</sup> |        | Top quarter <sup>₄</sup> |       |
|-------------------------------|-----------------------------|-------|-----------------------------|-------|----------------------------|--------|--------------------------|-------|
|                               | Mean<br>score               | SE    | Mean<br>score               | SE    | Mean<br>score              | SE     | Mean<br>score            | SE    |
| Canada overall                | 530*                        | (4.4) | 512*                        | (4.1) | 495*                       | (5.5)  | 486                      | (4.9) |
| Language of the school system |                             |       |                             |       |                            |        |                          |       |
| Anglophone                    | 523*                        | (4.9) | 509*                        | (4.7) | 486*                       | (6.2)  | 483                      | (5.4) |
| Francophone                   | 562*                        | (8.0) | 524*                        | (7.9) | 523                        | (10.2) | 504                      | (8.6) |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

#### TABLE A.4.1.1 Percentage of schools by size of communities in which they are located

|                           | Rural/<br>tov | 'small<br>wn | Medium<br>town |       | Medium Small ci<br>town |       | Medium<br>city |       | Medium Lar<br>city |       | Large | arge city |  |
|---------------------------|---------------|--------------|----------------|-------|-------------------------|-------|----------------|-------|--------------------|-------|-------|-----------|--|
| Canada and provinces      | %             | SE           | %              | SE    | %                       | SE    | %              | SE    | %                  | SE    |       |           |  |
| British Columbia          | 17.3‡         | (7.4)        | 10.2‡          | (2.7) | 26.5                    | (6.9) | 24.4           | (5.4) | 21.6               | (7.0) |       |           |  |
| Alberta                   | 24.8‡         | (7.9)        | 23.7‡          | (6.4) | 10.6‡                   | (3.3) | 1.7‡           | (0.8) | 39.3               | (6.4) |       |           |  |
| Saskatchewan              | 47.6          | (5.1)        | 8.2‡           | (2.0) | 6.7‡                    | (2.1) | 37.2           | (4.6) | 0.4‡               | (0.4) |       |           |  |
| Manitoba                  | 47.6          | (5.3)        | 12.6‡          | (3.2) | 3.7‡                    | (1.4) | 0.7‡           | (0.7) | 35.5               | (4.7) |       |           |  |
| Ontario                   | 13.9          | (4.2)        | 26.5           | (5.6) | 12.3                    | (2.9) | 27.4           | (4.4) | 19.9               | (3.4) |       |           |  |
| Quebec                    | 17.8‡         | (5.9)        | 25.3           | (4.5) | 22.5                    | (4.1) | 13.9‡          | (3.1) | 20.5               | (5.1) |       |           |  |
| New Brunswick             | 53.9          | (4.3)        | 21.7‡          | (3.6) | 20.6‡                   | (3.5) | 3.7‡           | (1.6) | 0.0                | (0.0) |       |           |  |
| Nova Scotia               | 45.1          | (4.7)        | 32.7           | (4.4) | 6.2‡                    | (2.3) | 15.1‡          | (3.4) | 0.9‡               | (0.9) |       |           |  |
| Prince Edward Island      | 73.5‡         | (9.3)        | 18.4‡          | (8.3) | 8.2‡                    | (5.5) | 0.0            | (0.0) | 0.0                | (0.0) |       |           |  |
| Newfoundland and Labrador | 74.8          | (4.5)        | 13.7‡          | (3.5) | 2.1‡                    | (1.5) | 9.4‡           | (3.0) | 0.0                | (0.0) |       |           |  |
| Canada                    | 23.2          | (2.6)        | 22.5           | (3.2) | 12.9                    | (1.7) | 21.1           | (2.4) | 20.2               | (2.1) |       |           |  |

**‡** There are fewer than 30 observations.

Note: Population of communties:

Rural/small town: fewer than 5,000

Medium town: 5,000–24,999

Small city: 25,000–99,999

Medium city: 100,000–499,999

Large city: over 500,000

|                           |                     |       |           | Angl           | ophone s | chool sy | stems       |       |            |       |
|---------------------------|---------------------|-------|-----------|----------------|----------|----------|-------------|-------|------------|-------|
|                           | Rural/small<br>town |       | Mec<br>to | Medium<br>town |          | l city   | Medium city |       | Large city |       |
| Canada and provinces      | %                   | SE    | %         | SE             | %        | SE       | %           | SE    | %          | SE    |
| British Columbia          | 17.9‡               | (7.6) | 10.2‡     | (2.8)          | 25.4     | (7.1)    | 24.6        | (5.5) | 21.9‡      | (7.2) |
| Alberta                   | 24.9‡               | (8.0) | 24.0‡     | (6.5)          | 10.6‡    | (3.4)    | 1.2‡        | (0.7) | 39.3       | (6.6) |
| Saskatchewan              | 47.8                | (5.2) | 8.3‡      | (2.0)          | 6.3‡     | (2.1)    | 37.3        | (4.7) | 0.4‡       | (0.4) |
| Manitoba                  | 46.5                | (5.5) | 12.9‡     | (3.3)          | 3.5‡     | (1.5)    | 0.8‡        | (0.8) | 36.3       | (4.9) |
| Ontario                   | 13.0‡               | (4.4) | 27.0‡     | (6.0)          | 11.9‡    | (3.1)    | 28.1        | (4.7) | 20.0       | (3.6) |
| Quebec                    | 17.0‡               | (5.2) | 24.5‡     | (5.9)          | 17.0‡    | (5.2)    | 17.0‡       | (5.2) | 24.6‡      | (5.9) |
| New Brunswick             | 50.6                | (5.7) | 18.2‡     | (4.4)          | 26.0‡    | (5.0)    | 5.2‡        | (2.5) | 0.0        | (0.0) |
| Nova Scotia               | 45.6                | (4.9) | 33.0      | (4.6)          | 5.8‡     | (2.3)    | 14.6‡       | (3.5) | 1.0‡       | (1.0) |
| Prince Edward Island      | 81.0‡               | (8.6) | 9.5‡      | (6.4)          | 9.5‡     | (6.4)    | 0.0         | (0.0) | 0.0        | (0.0) |
| Newfoundland and Labrador | 75.5                | (4.4) | 13.8‡     | (3.6)          | 2.1‡     | (1.5)    | 8.5‡        | (2.9) | 0.0        | (0.0) |
| Canada                    | 22.9                | (2.9) | 22.5      | (3.6)          | 11.7     | (1.9)    | 22.3        | (2.7) | 20.6       | (2.3) |

# TABLE A.4.1.2 Percentage of schools by size of communities in which they are located and language of school system

|                           | Francophone school systems |                                      |       |            |       |             |       |        |        |        |  |
|---------------------------|----------------------------|--------------------------------------|-------|------------|-------|-------------|-------|--------|--------|--------|--|
|                           | Rural,<br>to               | <sup>/</sup> small Medium<br>vn town |       | Small city |       | Medium city |       | Large  | e city |        |  |
| Canada and provinces      | %                          | SE                                   | %     | SE         | %     | SE          | %     | SE     | %      | SE     |  |
| British Columbia          | 0.0                        | (0.0)                                | 9.1‡  | (8.7)      | 63.6‡ | (14.5)      | 18.2‡ | (11.6) | 9.1‡   | (8.7)  |  |
| Alberta                   | 20.0‡                      | (12.7)                               | 10.0‡ | (9.5)      | 10.0‡ | (9.5)       | 20.0‡ | (12.7) | 40.0‡  | (15.5) |  |
| Saskatchewan              | 33.3‡                      | (19.3)                               | 0.0   | (0.0)      | 33.3‡ | (19.3)      | 33.3‡ | (19.3) | 0.0    | (0.0)  |  |
| Manitoba                  | 66.7‡                      | (12.2)                               | 6.7‡  | (6.4)      | 6.7‡  | (6.4)       | 0.0   | (0.0)  | 20.0‡  | (10.3) |  |
| Ontario                   | 30.0                       | (4.6)                                | 19.0‡ | (3.9)      | 19.0‡ | (3.9)       | 14.0‡ | (3.5)  | 18.0‡  | (3.8)  |  |
| Quebec                    | 18.0‡                      | (7.0)                                | 25.5‡ | (5.2)      | 23.6  | (4.9)       | 13.3‡ | (3.6)  | 19.6‡  | (6.0)  |  |
| New Brunswick             | 58.4                       | (6.6)                                | 26.5‡ | (5.9)      | 13.3‡ | (4.7)       | 1.8‡  | (1.8)  | 0.0    | (0.0)  |  |
| Nova Scotia               | 40.0‡                      | (15.5)                               | 30.0‡ | (14.5)     | 10.0‡ | (9.5)       | 20.0‡ | (12.7) | 0.0    | (0.0)  |  |
| Prince Edward Island      |                            |                                      |       |            |       |             |       |        |        |        |  |
| Newfoundland and Labrador |                            |                                      |       |            |       |             |       |        |        |        |  |
| Canada                    | 25.6                       | (4.2)                                | 23.0  | (3.2)      | 21.3  | (3.0)       | 12.8  | (2.3)  | 17.3   | (3.7)  |  |

**‡** There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

Note: Population of communties:

Rural/small town: fewer than 5,000

Medium town: 5,000-24,999

Small city: 25,000–99,999

Medium city: 100,000–499,999

Large city: over 500,000

#### TABLE A.4.1.3 Achievement in mathematics by size of communities in which schools are located, Canada

|                  | Mean score | SE    |
|------------------|------------|-------|
| Rural/small town | 481        | (6.8) |
| Medium town      | 497        | (5.9) |
| Small city       | 507*       | (6.5) |
| Medium city      | 510*       | (6.6) |
| Large city       | 516*       | (5.2) |
|                  |            |       |

\* Significant difference compared to the *rural/small town* category

Note: Population of communties:

Rural/small town: fewer than 5,000

Medium town: 5,000-24,999

Small city: 25,000-99,999

Medium city: 100,000-499,999

Large city: over 500,000

| TABLE A.4.2.1         Percentage of public and private | schools |       |         |   |  |
|--|---------|-------|---------|---|--|
|  | Put     | olic  | Private |   |  |
| Canada and provinces                                   | %       | SE    | %       |   |  |
| British Columbia                                       | 71.1    | (8.7) | 28.9‡   | ( |  |
| Alberta  | 88.5    | (5.9) | 11.5‡   | ( |  |
| Saskatchewan   | 97.6    | (1.8) | 2.4‡    | ( |  |
| Manitoba   | 87.2    | (3.8) | 12.8‡   | ( |  |
| Ontario  | 94.2    | (3.5) | 5.8‡    | ( |  |
| Quebec   | 70.7    | (5.5) | 29.3    | ( |  |
| New Brunswick  | 100.0   | (0.0) | 0.0     | ( |  |
| Nova Scotia  | 100.0   | (0.0) | 0.0     | ( |  |
| Prince Edward Island                                   | 100.0‡  | (0.0) | 0.0     | ( |  |
| Newfoundland and Labrador                              | 96.8    | (1.8) | 3.2‡    | ( |  |

90.2

(2.2)

<sup>‡</sup> There are fewer than 30 observations.

Canada

SE

(8.7)

(5.9)

(1.8)

(3.8)

(3.5)

(5.5) (0.0)

(0.0)

(0.0)

(1.8)

(2.2)

9.8

|                           | Anglophone school systems |       |       |       |        | cophone | school syst | ems   |
|---------------------------|---------------------------|-------|-------|-------|--------|---------|-------------|-------|
|                           | Public                    |       | Priv  | ate   | Pub    | olic    | Priv        | ate   |
| Canada and provinces      | %                         | SE    | %     | SE    | %      | SE      | %           | SE    |
| British Columbia          | 70.2                      | (8.9) | 29.8‡ | (8.9) | 100.0‡ | (0.0)   | 0.0         | (0.0) |
| Alberta                   | 88.2                      | (6.1) | 11.8‡ | (6.1) | 100.0‡ | (0.0)   | 0.0         | (0.0) |
| Saskatchewan              | 97.5                      | (1.8) | 2.5‡  | (1.8) | 100.0‡ | (0.0)   | 0.0         | (0.0) |
| Manitoba                  | 86.5                      | (4.0) | 13.5‡ | (4.0) | 100.0‡ | (0.0)   | 0.0         | (0.0) |
| Ontario                   | 93.9                      | (3.7) | 6.1‡  | (3.7) | 100.0  | (0.0)   | 0.0         | (0.0) |
| Quebec                    | 71.5                      | (6.2) | 28.5‡ | (6.2) | 70.5   | (6.5)   | 29.5‡       | (6.5) |
| New Brunswick             | 100.0                     | (0.0) | 0.0   | (0.0) | 100.0  | (0.0)   | 0.0         | (0.0) |
| Nova Scotia               | 100.0                     | (0.0) | 0.0   | (0.0) | 100.0‡ | (0.0)   | 0.0         | (0.0) |
| Prince Edward Island      | 100.0‡                    | (0.0) | 0.0   | (0.0) |        |         |             |       |
| Newfoundland and Labrador | 96.8                      | (1.8) | 3.2‡  | (1.8) |        |         |             |       |
| Canada                    | 91.2                      | (2.4) | 8.8   | (2.4) | 82.7   | (4.1)   | 17.3‡       | (4.1) |

#### TABLE A.4.2.2 Percentage of public and private schools by language of school system

<sup>‡</sup> There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

#### TABLE A.4.2.3 Achievement in mathematics by school governance, Canada

|         | Mean score | SE    |
|---------|------------|-------|
| Public  | 497        | (2.9) |
| Private | 540*       | (9.7) |
|         |            |       |

\* Significant difference compared to the *public* category

#### TABLE A.4.3.1 Percentage of schools by total enrolment

|                           | 100 or fewer |       | 101–500 |        | 501–1,000 |       | More than<br>1,000 |       |
|---------------------------|--------------|-------|---------|--------|-----------|-------|--------------------|-------|
| Canada and provinces      | %            | SE    | %       | SE     | %         | SE    | %                  | SE    |
| British Columbia          | 2.3‡         | (1.6) | 52.8    | (7.4)  | 28.2      | (5.2) | 16.7               | (3.3) |
| Alberta                   | 15.2‡        | (7.7) | 49.0    | (7.2)  | 34.4      | (6.0) | 1.4‡               | (0.7) |
| Saskatchewan              | 9.5‡         | (3.5) | 83.0    | (3.8)  | 7.3‡      | (1.7) | 0.2‡               | (0.2) |
| Manitoba                  | 5.0‡         | (3.2) | 80.5    | (3.8)  | 13.7      | (2.5) | 0.8‡               | (0.5) |
| Ontario                   | 4.4‡         | (3.3) | 66.8    | (4.7)  | 26.0      | (3.9) | 2.8‡               | (1.0) |
| Quebec                    | 2.3‡         | (1.2) | 43.2    | (6.1)  | 27.2      | (4.4) | 27.4               | (4.0) |
| New Brunswick             | 8.2‡         | (2.4) | 72.7    | (3.9)  | 19.1‡     | (3.4) | 0.0                | (0.0) |
| Nova Scotia               | 2.6‡         | (1.5) | 79.1    | (3.8)  | 18.3‡     | (3.6) | 0.0                | (0.0) |
| Prince Edward Island      | 18.4‡        | (8.3) | 61.2‡   | (10.1) | 20.4‡     | (8.2) | 0.0                | (0.0) |
| Newfoundland and Labrador | 23.1‡        | (4.3) | 66.4    | (4.8)  | 10.5‡     | (3.2) | 0.0                | (0.0) |
| Canada                    | 6.2          | (2.0) | 64.1    | (2.8)  | 24.5      | (2.2) | 5.2                | (0.7) |

|                           | Anglophone school systems |       |       |        |               |       |             |            |  |  |  |  |
|---------------------------|---------------------------|-------|-------|--------|---------------|-------|-------------|------------|--|--|--|--|
|                           | 100 or                    | fewer | 101-  | -500   | <b>501</b> –1 | L,000 | More<br>1,0 | than<br>00 |  |  |  |  |
| Canada and provinces      | %                         | SE    | %     | SE     | %             | SE    | %           | SE         |  |  |  |  |
| British Columbia          | 1.5‡                      | (1.5) | 52.8  | (7.6)  | 28.5          | (5.4) | 17.2        | (3.5)      |  |  |  |  |
| Alberta                   | 15.4‡                     | (7.9) | 47.9  | (7.3)  | 35.3          | (6.2) | 1.4‡        | (0.7)      |  |  |  |  |
| Saskatchewan              | 9.1‡                      | (3.6) | 83.2  | (3.8)  | 7.4‡          | (1.7) | 0.3‡        | (0.3)      |  |  |  |  |
| Manitoba                  | 4.9‡                      | (3.3) | 80.6  | (4.0)  | 13.7          | (2.6) | 0.8‡        | (0.6)      |  |  |  |  |
| Ontario                   | 3.6‡                      | (3.5) | 66.9  | (4.9)  | 26.7          | (4.1) | 2.8‡        | (1.1)      |  |  |  |  |
| Quebec                    | 7.5‡                      | (3.6) | 52.9‡ | (6.9)  | 24.5‡         | (5.9) | 15.1‡       | (4.9)      |  |  |  |  |
| New Brunswick             | 6.5‡                      | (2.8) | 70.1  | (5.2)  | 23.4‡         | (4.8) | 0.0         | (0.0)      |  |  |  |  |
| Nova Scotia               | 2.9‡                      | (1.6) | 79.0  | (4.0)  | 18.1‡         | (3.8) | 0.0         | (0.0)      |  |  |  |  |
| Prince Edward Island      | 9.5‡                      | (6.4) | 66.7‡ | (10.3) | 23.8‡         | (9.3) | 0.0         | (0.0)      |  |  |  |  |
| Newfoundland and Labrador | 22.3‡                     | (4.3) | 67.0  | (4.9)  | 10.6‡         | (3.2) | 0.0         | (0.0)      |  |  |  |  |
| Canada                    | 6.1                       | (2.3) | 65.7  | (3.1)  | 24.9          | (2.5) | 3.4         | (0.7)      |  |  |  |  |

### TABLE A.4.3.2 Percentage of schools by total enrolment, by language of the school system

|                           | Francophone school systems |        |             |        |       |        |             |             |  |  |  |  |
|---------------------------|----------------------------|--------|-------------|--------|-------|--------|-------------|-------------|--|--|--|--|
|                           | 100 or fewe                |        | wer 101–500 |        |       | 1,000  | More<br>1,0 | than<br>100 |  |  |  |  |
| Canada and provinces      | %                          | SE     | %           | SE     | %     | SE     | %           | SE          |  |  |  |  |
| British Columbia          | 27.3‡                      | (13.4) | 54.5‡       | (15.0) | 18.2‡ | (11.6) | 0.0         | (0.0)       |  |  |  |  |
| Alberta                   | 10.0‡                      | (9.5)  | 90.0‡       | (9.5)  | 0.0   | (0.0)  | 0.0         | (0.0)       |  |  |  |  |
| Saskatchewan              | 33.3‡                      | (19.3) | 66.7‡       | (19.3) | 0.0   | (0.0)  | 0.0         | (0.0)       |  |  |  |  |
| Manitoba                  | 6.7‡                       | (6.4)  | 80.0‡       | (10.3) | 13.3‡ | (8.8)  | 0.0         | (0.0)       |  |  |  |  |
| Ontario                   | 17.8‡                      | (3.8)  | 65.3        | (4.7)  | 13.9‡ | (3.4)  | 3.0‡        | (1.7)       |  |  |  |  |
| Quebec                    | 1.2‡                       | (1.2)  | 41.2‡       | (7.4)  | 27.7  | (5.2)  | 29.8        | (4.8)       |  |  |  |  |
| New Brunswick             | 10.6‡                      | (4.1)  | 76.1        | (5.8)  | 13.3‡ | (4.7)  | 0.0         | (0.0)       |  |  |  |  |
| Nova Scotia               | 0.0                        | (0.0)  | 80.0‡       | (12.7) | 20.0‡ | (12.7) | 0.0         | (0.0)       |  |  |  |  |
| Prince Edward Island      |                            |        |             |        |       |        |             |             |  |  |  |  |
| Newfoundland and Labrador |                            |        |             |        |       |        |             |             |  |  |  |  |
| Canada                    | 7.4                        | (1.4)  | 52.8        | (4.1)  | 21.5  | (3.1)  | 18.2        | (2.5)       |  |  |  |  |

**‡** There are fewer than 30 observations.

|                 | Mean score | SE     |
|-----------------|------------|--------|
| 100 or fewer    | 502        | (13.4) |
| 101–500         | 494        | (3.8)  |
| 501–1,000       | 516*       | (4.7)  |
| More than 1,000 | 521*       | (5.6)  |

### TABLE A.4.3.3 Achievement in mathematics by total enrolment, Canada

\* Significant difference compared to the 101 to 500 category

### TABLE A.4.4.1 Percentage of schools by number of grade levels

|                           | 1–4 grades |       | 5–8 g | rades | 9 or ı<br>grad | more<br>des |
|---------------------------|------------|-------|-------|-------|----------------|-------------|
| Canada and provinces      | %          | SE    | %     | SE    | %              | SE          |
| British Columbia          | 23.4       | (4.7) | 38.9  | (6.5) | 37.7‡          | (9.0)       |
| Alberta                   | 20.8       | (4.0) | 19.9‡ | (4.6) | 59.4           | (6.4)       |
| Saskatchewan              | 1.0‡       | (0.5) | 10.2‡ | (2.3) | 88.8           | (2.4)       |
| Manitoba                  | 14.9       | (2.6) | 13.1‡ | (3.6) | 72.0           | (4.2)       |
| Ontario                   | 6.1        | (1.4) | 8.1   | (2.2) | 85.7           | (2.6)       |
| Quebec                    | 11.5‡      | (3.3) | 71.5  | (5.7) | 17.0‡          | (5.4)       |
| New Brunswick             | 21.7‡      | (3.6) | 13.5‡ | (3.0) | 64.8           | (4.1)       |
| Nova Scotia               | 35.4       | (4.5) | 20.2‡ | (3.8) | 44.5           | (4.6)       |
| Prince Edward Island      | 28.6‡      | (9.2) | 4.1‡  | (4.0) | 67.3‡          | (9.5)       |
| Newfoundland and Labrador | 19.0‡      | (4.0) | 20.0‡ | (4.1) | 61.1           | (5.0)       |
| Canada                    | 10.9       | (1.1) | 18.2  | (1.7) | 71.0           | (2.1)       |

|                           |       |        | Anglophone s  | chool systems |          |           |
|---------------------------|-------|--------|---------------|---------------|----------|-----------|
|                           | 1–4 g | grades | 5–8 g         | rades         | 9 or moi | re grades |
| Canada and provinces      | %     | SE     | %             | SE            | %        | SE        |
| British Columbia          | 24.1  | (4.9)  | 39.5          | (6.8)         | 36.3‡    | (9.4)     |
| Alberta                   | 21.1  | (4.1)  | 19.3‡         | (4.7)         | 59.6     | (6.5)     |
| Saskatchewan              | 1.0‡  | (0.5)  | 9.8‡          | (2.3)         | 89.1     | (2.4)     |
| Manitoba                  | 15.7  | (2.7)  | 13.5‡         | (3.8)         | 70.8     | (4.4)     |
| Ontario                   | 6.3‡  | (1.4)  | 6.6‡          | (2.3)         | 87.2     | (2.7)     |
| Quebec                    | 3.8‡  | (2.6)  | 58.5          | (6.8)         | 37.8‡    | (6.7)     |
| New Brunswick             | 32.5‡ | (5.3)  | 16.9‡         | (4.3)         | 50.6     | (5.7)     |
| Nova Scotia               | 39.0  | (4.8)  | 18.1‡         | (3.8)         | 42.9     | (4.8)     |
| Prince Edward Island      | 33.3‡ | (10.3) | 4.8‡          | (4.6)         | 61.9‡    | (10.6)    |
| Newfoundland and Labrador | 19.1‡ | (4.1)  | 19.1‡         | (4.1)         | 61.7     | (5.0)     |
| Canada                    | 11.1  | (1.2)  | 12.9          | (1.6)         | 76.0     | (2.1)     |
|                           |       |        | Francophone s | chool systems |          |           |

### TABLE A.4.4.2 Percentage of schools by number of grade levels and language of school system

|                           | 1–4 g | rades | 5–8 g | rades  | 9 or mor | e grades |
|---------------------------|-------|-------|-------|--------|----------|----------|
| Canada and provinces      | %     | SE    | %     | SE     | %        | SE       |
| British Columbia          | 0.0   | (0.0) | 18.2‡ | (11.6) | 81.8‡    | (11.6)   |
| Alberta                   | 10.0‡ | (9.5) | 40.0‡ | (15.5) | 50.0‡    | (15.8)   |
| Saskatchewan              | 0.0   | (0.0) | 33.3‡ | (19.3) | 66.7‡    | (19.3)   |
| Manitoba                  | 0.0   | (0.0) | 6.7‡  | (6.4)  | 93.3‡    | (6.4)    |
| Ontario                   | 4.0‡  | (1.9) | 34.7  | (4.7)  | 61.4     | (4.8)    |
| Quebec                    | 13.0‡ | (3.9) | 74.1  | (6.7)  | 12.9‡    | (6.5)    |
| New Brunswick             | 7.1‡  | (3.4) | 8.8‡  | (3.8)  | 84.1     | (4.9)    |
| Nova Scotia               | 0.0   | (0.0) | 40.0‡ | (15.5) | 60.0‡    | (15.5)   |
| Prince Edward Island      |       |       |       |        |          |          |
| Newfoundland and Labrador |       |       |       |        |          |          |
| Canada                    | 9.4‡  | (2.4) | 54.9  | (4.2)  | 35.7     | (4.0)    |

‡ There are fewer than 30 observations.

|                  | Moon cooro | CE.   |
|------------------|------------|-------|
|                  | wear score | JE    |
| 1–4 grades       | 500        | (3.9) |
| 5–8 grades       | 512        | (4.8) |
| 9 or more grades | 499*       | (3.8) |

### TABLE A.4.4.3 Achievement in mathematics by number of grade levels, Canada

\* Significant difference compared to the 5–8 grades category

| TABLE A. F.J. T electricage of schools by total enrollient of Grade by Secondary in students |       |        |       |       |       |       |        |       |         |       |            |            |
|--|-------|--------|-------|-------|-------|-------|--------|-------|---------|-------|------------|------------|
|  | 25 or | fewer  | 26–50 |       | 51–75 |       | 76–100 |       | 101–200 |       | More<br>20 | than<br>00 |
| Canada and provinces   | %     | SE     | %     | SE    | %     | SE    | %      | SE    | %       | SE    | %          | SE         |
| British Columbia   | 27.5‡ | (9.8)  | 7.2‡  | (3.8) | 10.6‡ | (4.0) | 11.3‡  | (3.4) | 24.9    | (4.7) | 18.6       | (3.5)      |
| Alberta  | 30.4‡ | (8.5)  | 16.4‡ | (5.0) | 18.7‡ | (4.5) | 14.5‡  | (3.4) | 15.8    | (3.3) | 4.2‡       | (1.2)      |
| Saskatchewan   | 55.0  | (4.8)  | 34.1  | (4.3) | 8.7‡  | (1.8) | 0.2‡   | (0.2) | 1.7‡    | (0.7) | 0.2‡       | (0.2)      |
| Manitoba   | 44.0  | (5.5)  | 21.3  | (3.7) | 14.0  | (2.7) | 9.4‡   | (2.1) | 10.6‡   | (2.1) | 0.7‡       | (0.5)      |
| Ontario  | 20.8  | (5.7)  | 38.3  | (5.3) | 23.2  | (3.8) | 9.0‡   | (2.1) | 7.0     | (1.5) | 1.8‡       | (0.5)      |
| Quebec   | 13.8‡ | (5.9)  | 10.2‡ | (3.8) | 5.5‡  | (2.7) | 10.6‡  | (4.3) | 32.8    | (4.8) | 27.1       | (3.9)      |
| New Brunswick  | 39.7  | (4.2)  | 25.5  | (3.8) | 11.2‡ | (2.7) | 10.1‡  | (2.7) | 12.0‡   | (2.8) | 1.5‡       | (1.1)      |
| Nova Scotia  | 22.0‡ | (3.9)  | 25.1‡ | (4.0) | 13.9‡ | (3.2) | 19.1‡  | (3.7) | 17.3‡   | (3.5) | 2.6‡       | (1.5)      |
| Prince Edward Island   | 55.1‡ | (10.2) | 8.2‡  | (5.5) | 16.3‡ | (7.5) | 0.0    | (0.0) | 20.4‡   | (8.2) | 0.0        | (0.0)      |
| Newfoundland and<br>Labrador   | 52.6  | (5.1)  | 17.9‡ | (3.9) | 6.3‡  | (2.5) | 6.3‡   | (2.5) | 9.5‡    | (3.0) | 7.4‡       | (2.7)      |
| Canada   | 26.9  | (3.2)  | 28.8  | (3.0) | 17.8  | (2.1) | 9.4    | (1.2) | 11.7    | (1.1) | 5.4        | (0.6)      |

### TABLE A.4.5.1 Percentage of schools by total enrolment of Grade 8/Secondary II students

|                           | Anglophone school systems |        |       |       |       |       |       |       |       |       |            |            |
|---------------------------|---------------------------|--------|-------|-------|-------|-------|-------|-------|-------|-------|------------|------------|
|                           | 25 or                     | fewer  | 26-   | -50   | 51-   | -75   | 76-   | 100   | 101-  | -200  | More<br>20 | than<br>00 |
| Canada and provinces      | %                         | SE     | %     | SE    | %     | SE    | %     | SE    | %     | SE    | %          | SE         |
| British Columbia          | 26.1‡                     | (10.2) | 6.9‡  | (3.9) | 10.6‡ | (4.1) | 11.6‡ | (3.5) | 25.6  | (4.9) | 19.2       | (3.7)      |
| Alberta                   | 29.9‡                     | (8.8)  | 16.0‡ | (5.1) | 18.9‡ | (4.7) | 14.7‡ | (3.5) | 16.2  | (3.4) | 4.3‡       | (1.3)      |
| Saskatchewan              | 55.0                      | (4.8)  | 33.9  | (4.3) | 8.8‡  | (1.9) | 0.3‡  | (0.3) | 1.8‡  | (0.7) | 0.3‡       | (0.3)      |
| Manitoba                  | 42.7‡                     | (5.8)  | 21.0‡ | (3.9) | 14.8  | (2.8) | 9.9‡  | (2.2) | 11.3‡ | (2.2) | 0.4‡       | (0.4)      |
| Ontario                   | 19.0‡                     | (6.1)  | 39.2  | (5.6) | 24.0  | (4.1) | 9.3‡  | (2.2) | 6.8‡  | (1.5) | 1.8‡       | (0.5)      |
| Quebec                    | 26.4‡                     | (6.1)  | 20.8‡ | (5.6) | 11.3‡ | (4.4) | 9.4‡  | (4.0) | 20.7‡ | (5.6) | 11.3‡      | (4.3)      |
| New Brunswick             | 31.2‡                     | (5.3)  | 24.7‡ | (4.9) | 13.0‡ | (3.8) | 11.7‡ | (3.7) | 18.2‡ | (4.4) | 1.3‡       | (1.3)      |
| Nova Scotia               | 18.1‡                     | (3.8)  | 26.7‡ | (4.3) | 14.3‡ | (3.4) | 19.0‡ | (3.8) | 19.0‡ | (3.8) | 2.9‡       | (1.6)      |
| Prince Edward Island      | 47.6‡                     | (10.9) | 9.5‡  | (6.4) | 19.0‡ | (8.6) | 0.0   | (0.0) | 23.8‡ | (9.3) | 0.0        | (0.0)      |
| Newfoundland and Labrador | 52.1                      | (5.2)  | 18.1‡ | (4.0) | 6.4‡  | (2.5) | 6.4‡  | (2.5) | 9.6‡  | (3.0) | 7.4‡       | (2.7)      |
| Canada                    | 26.6                      | (3.7)  | 30.8  | (3.4) | 19.5  | (2.4) | 9.5   | (1.4) | 10.1  | (1.1) | 3.5        | (0.5)      |

#### TABLE A.4.5.2 Percentage of schools by enrolment of Grade 8/Secondary II students and language of the school system

|                           | Francophone school systems |        |       |        |       |       |       |        |       |       |            |            |
|---------------------------|----------------------------|--------|-------|--------|-------|-------|-------|--------|-------|-------|------------|------------|
|                           | 25 or                      | fewer  | 26-   | -50    | 51-   | -75   | 76-   | 100    | 101-  | -200  | More<br>20 | than<br>00 |
| Canada and provinces      | %                          | SE     | %     | SE     | %     | SE    | %     | SE     | %     | SE    | %          | SE         |
| British Columbia          | 72.7‡                      | (13.4) | 18.2‡ | (11.6) | 9.1‡  | (8.7) | 0.0   | (0.0)  | 0.0   | (0.0) | 0.0        | (0.0)      |
| Alberta                   | 50.0‡                      | (15.8) | 30.0‡ | (14.5) | 10.0‡ | (9.5) | 10.0‡ | (9.5)  | 0.0   | (0.0) | 0.0        | (0.0)      |
| Saskatchewan              | 50.0‡                      | (20.4) | 50.0‡ | (20.4) | 0.0   | (0.0) | 0.0   | (0.0)  | 0.0   | (0.0) | 0.0        | (0.0)      |
| Manitoba                  | 66.7‡                      | (12.2) | 26.7‡ | (11.4) | 0.0   | (0.0) | 0.0   | (0.0)  | 0.0   | (0.0) | 6.7‡       | (6.4)      |
| Ontario                   | 51.5                       | (5.0)  | 22.8‡ | (4.2)  | 8.9‡  | (2.8) | 4.0‡  | (1.9)  | 10.9‡ | (3.1) | 2.0‡       | (1.4)      |
| Quebec                    | 11.2‡                      | (7.2)  | 8.1‡  | (4.5)  | 4.4‡  | (3.1) | 10.8‡ | (5.1)  | 35.2  | (5.8) | 30.2       | (4.8)      |
| New Brunswick             | 51.3‡                      | (6.7)  | 26.5‡ | (5.9)  | 8.8‡  | (3.8) | 8.0‡  | (3.9)  | 3.5‡  | (2.5) | 1.8‡       | (1.8)      |
| Nova Scotia               | 60.0‡                      | (15.5) | 10.0‡ | (9.5)  | 10.0‡ | (9.5) | 20.0‡ | (12.7) | 0.0   | (0.0) | 0.0        | (0.0)      |
| Prince Edward Island      |                            |        |       |        |       |       |       |        |       |       |            |            |
| Newfoundland and Labrador |                            |        |       |        |       |       |       |        |       |       |            |            |
| Canada                    | 28.9                       | (4.1)  | 14.6  | (2.9)  | 5.9‡  | (2.0) | 8.5‡  | (3.1)  | 23.5  | (3.3) | 18.5       | (2.5)      |

**‡** There are fewer than 30 observations.

|               | Mean score | SE    |
|---------------|------------|-------|
| 25 or fewer   | 488        | (7.7) |
| 26–50         | 492        | (4.6) |
| 51–75         | 522*       | (5.6) |
| 76–100        | 505        | (5.5) |
| 101–200       | 511*       | (4.7) |
| More than 200 | 518*       | (4.3) |

#### TABLE A.4.5.3 Achievement in mathematics by total enrolment of Grade 8/Secondary II students, Canada

\* Significant difference compared to the 25 or fewer category

|                           | 09    | %     | 1–5%  |        | 6–10% |       | 11–25% |       | 26–50% |       | More<br>50 | than<br>% |
|---------------------------|-------|-------|-------|--------|-------|-------|--------|-------|--------|-------|------------|-----------|
| Canada and provinces      | %     | SE    | %     | SE     | %     | SE    | %      | SE    | %      | SE    | %          | SE        |
| British Columbia          | 8.5‡  | (3.8) | 56.1  | (7.3)  | 17.9‡ | (4.4) | 11.7‡  | (2.9) | 1.5‡   | (0.8) | 4.3‡       | (2.8)     |
| Alberta                   | 14.9‡ | (6.4) | 26.0‡ | (6.7)  | 24.5‡ | (6.6) | 16.7‡  | (4.8) | 9.3‡   | (2.7) | 8.6‡       | (3.7)     |
| Saskatchewan              | 20.3‡ | (4.5) | 42.2  | (5.1)  | 9.1‡  | (2.5) | 16.5‡  | (3.3) | 7.2‡   | (2.4) | 4.7‡       | (1.6)     |
| Manitoba                  | 16.3‡ | (4.4) | 39.8  | (5.3)  | 9.8‡  | (2.3) | 11.2‡  | (2.5) | 12.2‡  | (3.0) | 10.7‡      | (3.7)     |
| Ontario                   | 16.1  | (4.3) | 44.7  | (5.3)  | 12.0‡ | (3.1) | 12.8   | (2.8) | 5.4‡   | (2.0) | 9.0        | (3.7)     |
| Quebec                    | 52.1  | (5.7) | 20.8  | (3.6)  | 3.7‡  | (1.6) | 9.2‡   | (4.6) | 3.6‡   | (1.1) | 10.7‡      | (2.9)     |
| New Brunswick             | 39.2  | (4.3) | 31.2  | (4.0)  | 6.8‡  | (2.2) | 5.3‡   | (2.0) | 3.8‡   | (1.7) | 13.7‡      | (3.0)     |
| Nova Scotia               | 30.4  | (4.3) | 39.2  | (4.6)  | 7.1‡  | (2.4) | 0.0    | (0.0) | 1.8‡   | (1.2) | 21.5‡      | (3.9)     |
| Prince Edward Island      | 22.4‡ | (8.9) | 53.1‡ | (10.3) | 8.2‡  | (5.5) | 4.1‡   | (4.0) | 0.0    | (0.0) | 12.2‡      | (6.6)     |
| Newfoundland and Labrador | 54.2  | (5.1) | 14.9‡ | (3.7)  | 1.1‡  | (1.1) | 2.1‡   | (1.5) | 0.0    | (0.0) | 27.7‡      | (4.6)     |
| Canada                    | 20.7  | (2.5) | 39.7  | (3.0)  | 12.3  | (1.8) | 12.4   | (1.7) | 5.7    | (1.1) | 9.3        | (2.0)     |

### TABLE A.4.6.1 Percentage of schools by proportion of second-language learners in schools

<sup>‡</sup> There are fewer than 30 observations.

| Anglophone school systems |  |   |  |   |  |  |   |  |  |  |   |
|---------------------------|--|---|--|---|--|--|---|--|--|--|---|
| 09                        | %  | 1-  | 5%   | 6–1   | 0%   | 11-2   | 25%   | 26-!   | 50%  | More<br>50   | than<br>%   |
| %                         | SE   | %   | SE   | %   | SE   | %  | SE  | %  | SE   | %  | SE  |
| 7.9‡                      | (3.9)  | 57.6  | (7.4)  | 18.1‡   | (4.6)  | 11.2‡  | (2.9)   | 1.2‡   | (0.7)  | 3.9‡   | (2.8)   |
| 13.9‡                     | (6.6)  | 26.4‡   | (6.9)  | 24.6‡   | (6.8)  | 17.2‡  | (4.9)   | 9.5‡   | (2.7)  | 8.3‡   | (3.8)   |
| 20.0‡                     | (4.5)  | 42.0  | (5.1)  | 9.2‡  | (2.6)  | 16.7‡  | (3.3)   | 7.3‡   | (2.4)  | 4.8‡   | (1.6)   |
| 15.6‡                     | (4.6)  | 40.4  | (5.5)  | 9.2‡  | (2.3)  | 10.7‡  | (2.5)   | 12.8‡  | (3.1)  | 11.3‡  | (3.9)   |
| 15.2‡                     | (4.6)  | 46.2  | (5.6)  | 12.2‡   | (3.2)  | 12.9‡  | (3.0)   | 5.0‡   | (2.2)  | 8.4‡   | (3.9)   |
| 20.7‡                     | (5.6)  | 22.6‡   | (5.7)  | 5.7‡  | (3.2)  | 13.2‡  | (4.7)   | 15.1‡  | (4.9)  | 22.6‡  | (5.7)   |
| 28.0‡                     | (5.2)  | 40.0  | (5.7)  | 9.3‡  | (3.4)  | 5.3‡   | (2.6)   | 5.3‡   | (2.6)  | 12.0‡  | (3.8)   |
| 30.1                      | (4.5)  | 39.8  | (4.8)  | 7.8‡  | (2.6)  | 0.0  | (0.0)   | 1.9‡   | (1.4)  | 20.4‡  | (4.0)   |
| 19.0‡                     | (8.6)  | 57.1‡   | (10.8)   | 4.8‡  | (4.6)  | 4.8‡   | (4.6)   | 0.0  | (0.0)  | 14.3‡  | (7.6)   |
| 53.8                      | (5.2)  | 15.1‡   | (3.7)  | 1.1‡  | (1.1)  | 2.2‡   | (1.5)   | 0.0  | (0.0)  | 28.0‡  | (4.7)   |
| 16.5                      | (2.8)  | 42.5  | (3.4)  | 13.3  | (2.1)  | 12.9   | (1.8)   | 6.0  | (1.3)  | 8.9  | (2.3)   |
|                           | 09<br>7.9‡<br>13.9‡<br>20.0‡<br>15.6‡<br>15.2‡<br>20.7‡<br>28.0‡<br>30.1<br>19.0‡<br>53.8<br><b>16.5</b> | 0%           %         SE           7.9‡         (3.9)           13.9‡         (6.6)           20.0‡         (4.5)           15.6‡         (4.6)           20.7‡         (5.6)           20.7‡         (5.6)           20.7‡         (5.6)           20.7‡         (5.6)           20.7‡         (5.2)           30.1         (4.5)           19.0‡         (8.6)           53.8         (5.2)           16.5         (2.8) | 0%         1           %         SE         %           7.9‡         (3.9)         57.6           13.9‡         (6.6)         26.4‡           20.0‡         (4.5)         42.0           15.6‡         (4.6)         40.4           15.2‡         (4.6)         46.2           20.7‡         (5.6)         22.6‡           28.0‡         (5.2)         40.0           30.1         (4.5)         39.8           19.0‡         (8.6)         57.1‡           53.8         (5.2)         15.1‡           16.5         (2.8)         42.5 | 0%         1−5%           %         SE         %         SE           7.9‡         (3.9)         57.6         (7.4)           13.9‡         (6.6)         26.4‡         (6.9)           20.0‡         (4.5)         42.0         (5.1)           15.6‡         (4.6)         40.4         (5.5)           15.2‡         (4.6)         46.2         (5.7)           20.7‡         (5.6)         22.6‡         (5.7)           28.0‡         (5.2)         40.0         (5.7)           30.1         (4.5)         39.8         (4.8)           19.0‡         (8.6)         57.1‡         (10.8)           53.8         (5.2)         15.1‡         (3.7)           16.5         (2.8)         42.5         (3.4) | O%         1-5%         6-1           %         SE         %         SE         %           7.9‡         (3.9)         57.6         (7.4)         18.1‡           13.9‡         (6.6)         26.4‡         (6.9)         24.6‡           20.0‡         (4.5)         42.0         (5.1)         9.2‡           15.6‡         (4.6)         40.4         (5.5)         9.2‡           15.2‡         (4.6)         46.2         (5.6)         12.2‡           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡           28.0‡         (5.2)         40.0         (5.7)         9.3‡           30.1         (4.5)         39.8         (4.8)         7.8‡           19.0‡         (8.6)         57.1‡         (10.8)         4.8‡           53.8         (5.2)         15.1‡         (3.7)         1.1‡      16.5         (2.8)         42.5         (3.4)         13.3 | Maglophone s           0%         1-5%         6-10%           %         SE         %         SE           7.9‡         (3.9)         57.6         (7.4)         18.1‡         (4.6)           13.9‡         (6.6)         26.4‡         (6.9)         24.6‡         (6.8)           20.0‡         (4.5)         42.0         (5.1)         9.2‡         (2.6)           15.6‡         (4.6)         40.4         (5.5)         9.2‡         (2.3)           15.2‡         (4.6)         46.2         (5.6)         12.2‡         (3.2)           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)           20.7‡         (5.6)         39.8         (4.8)         7.8‡         (2.6)           30.1         (4.5)         39.8         (4.8)         7.8‡         (2.6)           19.0‡         (8.6)         57.1‡         (10.8)         4.8‡         (4.6)           53.8         (5.2)         15.1‡         (3.7) | O%         I -5%         G-10%         I1-2           %         SE         %         SE         %         SE         %           7.9‡         (3.9)         57.6         (7.4)         18.1‡         (4.6)         11.2‡           13.9‡         (6.6)         26.4‡         (6.9)         24.6‡         (6.8)         17.2‡           20.0‡         (4.5)         42.0         (5.1)         9.2‡         (2.6)         16.7‡           15.6‡         (4.6)         40.4         (5.5)         9.2‡         (2.3)         10.7‡           15.6‡         (4.6)         40.4         (5.5)         9.2‡         (3.2)         12.9‡           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)         13.2‡           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)         13.2‡           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)         13.2‡           30.1         (4.5)         39.8         (4.8)         7.8‡         (2.6)         0.0           19.0‡         (8.6)         57.1‡         (10.8)         4.8‡         (4.6)         4.8‡ | O%         I − 5%         G−1 ⋅ K         I1 − 25%           %         SE         %         SE         %         SE         %         SE           7.9‡         (3.9)         57.6         (7.4)         18.1‡         (4.6)         11.2‡         (2.9)           13.9‡         (6.6)         26.4‡         (6.9)         24.6‡         (6.8)         17.2‡         (4.9)           20.0‡         (4.5)         42.0         (5.1)         9.2‡         (2.6)         16.7‡         (3.3)           15.6‡         (4.6)         40.4         (5.5)         9.2‡         (2.3)         10.7‡         (2.5)           15.6‡         (4.6)         40.4         (5.5)         9.2‡         (3.2)         12.9‡         (3.0)           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)         13.2‡         (4.7)           20.7‡         (5.6)         22.6‡         (5.7)         5.7‡         (3.2)         13.2‡         (4.7)           28.0‡         (5.2)         40.0         (5.7)         5.7‡         (3.2)         13.2‡         (4.6)           30.1         (4.5)         39.8         (4.8)         7.8‡         (2.6) <td>Anglophone school systems           <math>0\%</math> <math>1-5\%</math> <math>6-10\%</math> <math>11-25\%</math> <math>26-5\%</math> <math>\%</math>         SE         <math>\%</math> <math>\%</math></td> <td>Anglo&gt;hore s-bool systems<math>0\%</math><math>1-5\%</math><math>6-10\%</math><math>11-25\%</math><math>26-50\%</math><math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>7.9‡</math><math>(3.9)</math><math>57.6</math><math>(7.4)</math><math>18.1\ddagger</math><math>(4.6)</math><math>11.2\ddagger</math><math>(2.9)</math><math>1.2\ddagger</math><math>(0.7)</math><math>13.9\ddagger</math><math>(6.6)</math><math>26.4\ddagger</math><math>(6.9)</math><math>24.6\ddagger</math><math>(6.8)</math><math>17.2\ddagger</math><math>(4.9)</math><math>9.5\ddagger</math><math>(2.7)</math><math>20.0\ddagger</math><math>(4.5)</math><math>42.0</math><math>(5.1)</math><math>9.2\ddagger</math><math>(2.6)</math><math>16.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>15.6\ddagger</math><math>(4.6)</math><math>40.4</math><math>(5.5)</math><math>9.2\ddagger</math><math>(2.3)</math><math>10.7\ddagger</math><math>(2.5)</math><math>12.8\ddagger</math><math>(3.1)</math><math>15.2\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>12.2\ddagger</math><math>(3.2)</math><math>12.9\ddagger</math><math>(3.0)</math><math>5.0\ddagger</math><math>(2.2)</math><math>20.7\ddagger</math><math>(5.6)</math><math>22.6\ddagger</math><math>(5.7)</math><math>5.7\ddagger</math><math>(3.2)</math><math>13.2\ddagger</math><math>(4.7)</math><math>15.1\ddagger</math><math>(4.9)</math><math>28.0\ddagger</math><math>(5.2)</math><math>40.0</math><math>(5.7)</math><math>9.3\ddagger</math><math>(3.4)</math><math>5.3\ddagger</math><math>(2.6)</math><math>5.3\ddagger</math><math>(2.6)</math><math>30.1</math><math>(4.5)</math><math>39.8</math><math>(4.8)</math><math>7.8\ddagger</math><math>(2.6)</math><math>0.0</math><math>1.9\ddagger</math><math>(1.4)</math><math>19.0\ddagger</math><math>(8.6)</math><math>57.1\ddagger</math><math>(10.8)</math><math>4.8\ddagger</math><math>(4.6)</math><math>4.8\ddagger</math><math>(4.6)</math><math>0.0</math><math>(0.0)</math><math>53.8</math><math>(5.2)</math><math>15.1\ddagger</math><math>(3.7)</math><math>1.1\ddagger</math><math>(1.1)</math><math>2.2\ddagger</math><math>(1.8)</math><math>6.0</math><math>(1.3)</math><math>19.0\ddagger</math><math>42.5</math><math>(3.4)</math><math>1.33</math><math>(2.1)</math><math>1.29</math><math>(1.8)</math><math>6.0</math><math>(1.3)</math><!--</td--><td>Anglophone skylow skylow<math>0\%</math><math>1-5\%</math><math>6-10\%</math><math>11-25\%</math><math>26-50\%</math>More stop<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math><math>7.9\ddagger</math><math>(3.9)</math><math>57.6</math><math>(7.4)</math><math>18.1\ddagger</math><math>(4.6)</math><math>11.2\ddagger</math><math>(2.9)</math><math>1.2\ddagger</math><math>(0.7)</math><math>3.9\ddagger</math><math>13.9\ddagger</math><math>(6.6)</math><math>26.4\ddagger</math><math>(6.9)</math><math>24.6\ddagger</math><math>(6.8)</math><math>17.2\ddagger</math><math>(4.9)</math><math>9.5\ddagger</math><math>(2.7)</math><math>8.3\ddagger</math><math>20.0\ddagger</math><math>(4.5)</math><math>42.0</math><math>(5.1)</math><math>9.2\ddagger</math><math>(2.6)</math><math>16.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.6\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.1)</math><math>9.2\ddagger</math><math>(2.6)</math><math>16.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.6\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>9.2\ddagger</math><math>(2.3)</math><math>10.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.2\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>9.2\ddagger</math><math>(2.3)</math><math>10.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>20.7\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>12.2\ddagger</math><math>(3.2)</math><math>12.9\ddagger</math><math>(3.0)</math><math>5.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>20.7\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.7)</math><math>5.7\ddagger</math><math>(3.2)</math><math>12.9\ddagger</math><math>(3.0)</math><math>5.3\ddagger</math><math>(2.6)</math><math>5.3\ddagger</math><math>(2.6)</math><math>2.4\%</math><math>20.7\ddagger</math><math>(4.6)</math><math>39.8</math><math>(4.8)</math><math>7.8\ddagger</math><math>(2.6)</math><math>10.4</math><math>(2.6)</math><math>12.4</math><math>(3.7)</math><math>(2.4)</math><math>(3.7)</math><math>(3.4)</math><math>20.7\pm</math><math>(4.6)</math><math>39.8</math>&lt;</td></td> | Anglophone school systems $0\%$ $1-5\%$ $6-10\%$ $11-25\%$ $26-5\%$ $\%$ SE $\%$ | Anglo>hore s-bool systems $0\%$ $1-5\%$ $6-10\%$ $11-25\%$ $26-50\%$ $\%$ SE $\%$ SE $\%$ SE $\%$ SE $7.9‡$ $(3.9)$ $57.6$ $(7.4)$ $18.1\ddagger$ $(4.6)$ $11.2\ddagger$ $(2.9)$ $1.2\ddagger$ $(0.7)$ $13.9\ddagger$ $(6.6)$ $26.4\ddagger$ $(6.9)$ $24.6\ddagger$ $(6.8)$ $17.2\ddagger$ $(4.9)$ $9.5\ddagger$ $(2.7)$ $20.0\ddagger$ $(4.5)$ $42.0$ $(5.1)$ $9.2\ddagger$ $(2.6)$ $16.7\ddagger$ $(3.3)$ $7.3\ddagger$ $(2.4)$ $15.6\ddagger$ $(4.6)$ $40.4$ $(5.5)$ $9.2\ddagger$ $(2.3)$ $10.7\ddagger$ $(2.5)$ $12.8\ddagger$ $(3.1)$ $15.2\ddagger$ $(4.6)$ $46.2$ $(5.6)$ $12.2\ddagger$ $(3.2)$ $12.9\ddagger$ $(3.0)$ $5.0\ddagger$ $(2.2)$ $20.7\ddagger$ $(5.6)$ $22.6\ddagger$ $(5.7)$ $5.7\ddagger$ $(3.2)$ $13.2\ddagger$ $(4.7)$ $15.1\ddagger$ $(4.9)$ $28.0\ddagger$ $(5.2)$ $40.0$ $(5.7)$ $9.3\ddagger$ $(3.4)$ $5.3\ddagger$ $(2.6)$ $5.3\ddagger$ $(2.6)$ $30.1$ $(4.5)$ $39.8$ $(4.8)$ $7.8\ddagger$ $(2.6)$ $0.0$ $1.9\ddagger$ $(1.4)$ $19.0\ddagger$ $(8.6)$ $57.1\ddagger$ $(10.8)$ $4.8\ddagger$ $(4.6)$ $4.8\ddagger$ $(4.6)$ $0.0$ $(0.0)$ $53.8$ $(5.2)$ $15.1\ddagger$ $(3.7)$ $1.1\ddagger$ $(1.1)$ $2.2\ddagger$ $(1.8)$ $6.0$ $(1.3)$ $19.0\ddagger$ $42.5$ $(3.4)$ $1.33$ $(2.1)$ $1.29$ $(1.8)$ $6.0$ $(1.3)$ </td <td>Anglophone skylow skylow<math>0\%</math><math>1-5\%</math><math>6-10\%</math><math>11-25\%</math><math>26-50\%</math>More stop<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math>SE<math>\%</math><math>7.9\ddagger</math><math>(3.9)</math><math>57.6</math><math>(7.4)</math><math>18.1\ddagger</math><math>(4.6)</math><math>11.2\ddagger</math><math>(2.9)</math><math>1.2\ddagger</math><math>(0.7)</math><math>3.9\ddagger</math><math>13.9\ddagger</math><math>(6.6)</math><math>26.4\ddagger</math><math>(6.9)</math><math>24.6\ddagger</math><math>(6.8)</math><math>17.2\ddagger</math><math>(4.9)</math><math>9.5\ddagger</math><math>(2.7)</math><math>8.3\ddagger</math><math>20.0\ddagger</math><math>(4.5)</math><math>42.0</math><math>(5.1)</math><math>9.2\ddagger</math><math>(2.6)</math><math>16.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.6\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.1)</math><math>9.2\ddagger</math><math>(2.6)</math><math>16.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.6\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>9.2\ddagger</math><math>(2.3)</math><math>10.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>15.2\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>9.2\ddagger</math><math>(2.3)</math><math>10.7\ddagger</math><math>(3.3)</math><math>7.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>20.7\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.6)</math><math>12.2\ddagger</math><math>(3.2)</math><math>12.9\ddagger</math><math>(3.0)</math><math>5.3\ddagger</math><math>(2.4)</math><math>4.8\ddagger</math><math>20.7\ddagger</math><math>(4.6)</math><math>46.2</math><math>(5.7)</math><math>5.7\ddagger</math><math>(3.2)</math><math>12.9\ddagger</math><math>(3.0)</math><math>5.3\ddagger</math><math>(2.6)</math><math>5.3\ddagger</math><math>(2.6)</math><math>2.4\%</math><math>20.7\ddagger</math><math>(4.6)</math><math>39.8</math><math>(4.8)</math><math>7.8\ddagger</math><math>(2.6)</math><math>10.4</math><math>(2.6)</math><math>12.4</math><math>(3.7)</math><math>(2.4)</math><math>(3.7)</math><math>(3.4)</math><math>20.7\pm</math><math>(4.6)</math><math>39.8</math>&lt;</td> | Anglophone skylow skylow $0\%$ $1-5\%$ $6-10\%$ $11-25\%$ $26-50\%$ More stop $\%$ SE $\%$ SE $\%$ SE $\%$ SE $\%$ SE $\%$ $7.9\ddagger$ $(3.9)$ $57.6$ $(7.4)$ $18.1\ddagger$ $(4.6)$ $11.2\ddagger$ $(2.9)$ $1.2\ddagger$ $(0.7)$ $3.9\ddagger$ $13.9\ddagger$ $(6.6)$ $26.4\ddagger$ $(6.9)$ $24.6\ddagger$ $(6.8)$ $17.2\ddagger$ $(4.9)$ $9.5\ddagger$ $(2.7)$ $8.3\ddagger$ $20.0\ddagger$ $(4.5)$ $42.0$ $(5.1)$ $9.2\ddagger$ $(2.6)$ $16.7\ddagger$ $(3.3)$ $7.3\ddagger$ $(2.4)$ $4.8\ddagger$ $15.6\ddagger$ $(4.6)$ $46.2$ $(5.1)$ $9.2\ddagger$ $(2.6)$ $16.7\ddagger$ $(3.3)$ $7.3\ddagger$ $(2.4)$ $4.8\ddagger$ $15.6\ddagger$ $(4.6)$ $46.2$ $(5.6)$ $9.2\ddagger$ $(2.3)$ $10.7\ddagger$ $(3.3)$ $7.3\ddagger$ $(2.4)$ $4.8\ddagger$ $15.2\ddagger$ $(4.6)$ $46.2$ $(5.6)$ $9.2\ddagger$ $(2.3)$ $10.7\ddagger$ $(3.3)$ $7.3\ddagger$ $(2.4)$ $4.8\ddagger$ $20.7\ddagger$ $(4.6)$ $46.2$ $(5.6)$ $12.2\ddagger$ $(3.2)$ $12.9\ddagger$ $(3.0)$ $5.3\ddagger$ $(2.4)$ $4.8\ddagger$ $20.7\ddagger$ $(4.6)$ $46.2$ $(5.7)$ $5.7\ddagger$ $(3.2)$ $12.9\ddagger$ $(3.0)$ $5.3\ddagger$ $(2.6)$ $5.3\ddagger$ $(2.6)$ $2.4\%$ $20.7\ddagger$ $(4.6)$ $39.8$ $(4.8)$ $7.8\ddagger$ $(2.6)$ $10.4$ $(2.6)$ $12.4$ $(3.7)$ $(2.4)$ $(3.7)$ $(3.4)$ $20.7\pm$ $(4.6)$ $39.8$ < |

# TABLE A.4.6.2 Percentage of schools by proportion of second-language learners and language of the school system

|                           |       |        |       |        | Franco | ophone s | school s | ystems |       |       |            |              |
|---------------------------|-------|--------|-------|--------|--------|----------|----------|--------|-------|-------|------------|--------------|
|                           | 0     | %      | 1-    | 5%     | 6–1    | 10%      | 11–      | 25%    | 26-   | 50%   | More<br>50 | e than<br>0% |
| Canada and provinces      | %     | SE     | %     | SE     | %      | SE       | %        | SE     | %     | SE    | %          | SE           |
| British Columbia          | 27.3‡ | (13.4) | 9.1‡  | (8.7)  | 9.1‡   | (8.7)    | 27.3‡    | (13.4) | 9.1‡  | (8.7) | 18.2‡      | (11.6)       |
| Alberta                   | 50.0‡ | (15.8) | 10.0‡ | (9.5)  | 20.0‡  | (12.7)   | 0.0      | (0.0)  | 0.0   | (0.0) | 20.0‡      | (12.7)       |
| Saskatchewan              | 40.0‡ | (21.9) | 60.0‡ | (21.9) | 0.0    | (0.0)    | 0.0      | (0.0)  | 0.0   | (0.0) | 0.0        | (0.0)        |
| Manitoba                  | 28.6‡ | (12.1) | 28.6‡ | (12.1) | 21.4‡  | (11.0)   | 21.4‡    | (11.0) | 0.0   | (0.0) | 0.0        | (0.0)        |
| Ontario                   | 30.0  | (4.6)  | 20.0‡ | (4.0)  | 9.0‡   | (2.9)    | 12.0‡    | (3.3)  | 11.0‡ | (3.1) | 18.0‡      | (3.8)        |
| Quebec                    | 58.3  | (6.5)  | 20.4  | (4.2)  | 3.3‡   | (1.8)    | 8.4‡     | (5.4)  | 1.3‡  | (0.7) | 8.3‡       | (3.2)        |
| New Brunswick             | 54.0  | (6.7)  | 19.5‡ | (5.3)  | 3.5‡   | (2.5)    | 5.3‡     | (3.0)  | 1.8‡  | (1.8) | 15.9‡      | (4.9)        |
| Nova Scotia               | 33.3‡ | (15.7) | 33.3‡ | (15.7) | 0.0    | (0.0)    | 0.0      | (0.0)  | 0.0   | (0.0) | 33.3‡      | (15.7)       |
| Prince Edward Island      |       |        |       |        |        |          |          |        |       |       |            |              |
| Newfoundland and Labrador |       |        |       |        |        |          |          |        |       |       |            |              |
| Canada                    | 49.5  | (4.2)  | 20.5  | (2.7)  | 5.6‡   | (1.4)    | 9.1‡     | (3.3)  | 3.6‡  | (0.9) | 11.7       | (2.3)        |

‡ There are fewer than 30 observations.

|                           | 09    | %     | 1-5   | 1–5% 6–10% 11–25% 26–50%<br> |       | % More t<br>50% |       |       |       |       |       |       |
|---------------------------|-------|-------|-------|------------------------------|-------|-----------------|-------|-------|-------|-------|-------|-------|
| Canada and provinces      | %     | SE    | %     | SE                           | %     | SE              | %     | SE    | %     | SE    | %     | SE    |
| British Columbia          | 12.5‡ | (7.1) | 36.9  | (7.4)                        | 12.9‡ | (3.2)           | 25.9  | (7.0) | 6.4‡  | (2.4) | 5.4‡  | (3.4) |
| Alberta                   | 18.4‡ | (6.9) | 44.2  | (7.0)                        | 15.2‡ | (4.1)           | 21.0‡ | (6.3) | 1.3‡  | (1.3) | 0.0   | (0.0) |
| Saskatchewan              | 4.1‡  | (2.0) | 35.8  | (4.8)                        | 22.9  | (4.3)           | 16.5‡ | (3.5) | 10.8‡ | (2.8) | 10.0‡ | (3.3) |
| Manitoba                  | 3.5‡  | (2.6) | 29.5  | (4.7)                        | 21.6  | (4.6)           | 20.9  | (3.6) | 12.2‡ | (2.9) | 12.3‡ | (4.5) |
| Ontario                   | 27.9  | (5.1) | 57.9  | (5.6)                        | 3.8‡  | (1.6)           | 8.9‡  | (4.6) | 1.2‡  | (1.1) | 0.4‡  | (0.4) |
| Quebec                    | 69.3  | (4.7) | 24.9  | (4.0)                        | 2.1‡  | (1.2)           | 0.3‡  | (0.3) | 3.0‡  | (2.7) | 0.3‡  | (0.3) |
| New Brunswick             | 32.2  | (4.1) | 56.6  | (4.3)                        | 4.5‡  | (1.8)           | 2.2‡  | (1.3) | 4.5‡  | (1.8) | 0.0   | (0.0) |
| Nova Scotia               | 3.7‡  | (1.8) | 62.0  | (4.6)                        | 22.6‡ | (4.0)           | 9.9‡  | (2.8) | 1.8‡  | (1.3) | 0.0   | (0.0) |
| Prince Edward Island      | 24.5‡ | (8.7) | 63.3‡ | (9.8)                        | 8.2‡  | (5.5)           | 4.1‡  | (4.0) | 0.0   | (0.0) | 0.0   | (0.0) |
| Newfoundland and Labrador | 28.7‡ | (4.7) | 51.0  | (5.2)                        | 5.3‡  | (2.3)           | 4.3‡  | (2.1) | 9.6‡  | (3.0) | 1.1‡  | (1.1) |
| Canada                    | 25.7  | (3.0) | 48.7  | (3.1)                        | 8.5   | (1.2)           | 11.7  | (2.6) | 3.3   | (0.7) | 2.0‡  | (0.5) |

 TABLE A.4.7.1 Percentage of schools by proportion of students who identify as Indigenous in schools

|                           |       |       |       |        | Anglo | phone s | chool sy | stems |       |       |            |           |
|---------------------------|-------|-------|-------|--------|-------|---------|----------|-------|-------|-------|------------|-----------|
|                           | 0     | %     | 1–    | 5%     | 6–1   | .0%     | 11-2     | 25%   | 26-   | 50%   | More<br>50 | than<br>% |
| Canada and provinces      | %     | SE    | %     | SE     | %     | SE      | %        | SE    | %     | SE    | %          | SE        |
| British Columbia          | 12.9‡ | (7.4) | 36.9  | (7.6)  | 12.7‡ | (3.3)   | 25.3‡    | (7.2) | 6.6‡  | (2.5) | 5.5‡       | (3.5)     |
| Alberta                   | 18.3‡ | (7.0) | 43.3  | (7.1)  | 15.6‡ | (4.2)   | 21.5‡    | (6.5) | 1.3‡  | (1.3) | 0.0        | (0.0)     |
| Saskatchewan              | 3.9‡  | (2.0) | 35.1  | (4.9)  | 23.2  | (4.3)   | 16.7‡    | (3.6) | 10.9‡ | (2.9) | 10.1‡      | (3.4)     |
| Manitoba                  | 3.7‡  | (2.7) | 30.0  | (4.9)  | 21.3‡ | (4.9)   | 20.2     | (3.7) | 11.7‡ | (3.0) | 13.0‡      | (4.7)     |
| Ontario                   | 28.4  | (5.4) | 57.7  | (5.9)  | 3.2‡  | (1.7)   | 9.1‡     | (4.9) | 1.2‡  | (1.2) | 0.4‡       | (0.4)     |
| Quebec                    | 35.3‡ | (6.7) | 54.9‡ | (7.0)  | 3.9‡  | (2.7)   | 2.0‡     | (1.9) | 2.0‡  | (1.9) | 2.0‡       | (1.9)     |
| New Brunswick             | 23.4‡ | (4.8) | 61.0  | (5.6)  | 6.5‡  | (2.8)   | 1.3‡     | (1.3) | 7.8‡  | (3.1) | 0.0        | (0.0)     |
| Nova Scotia               | 3.0‡  | (1.7) | 59.4  | (4.9)  | 24.8‡ | (4.3)   | 10.9‡    | (3.1) | 2.0‡  | (1.4) | 0.0        | (0.0)     |
| Prince Edward Island      | 23.8‡ | (9.3) | 61.9‡ | (10.6) | 9.5‡  | (6.4)   | 4.8‡     | (4.6) | 0.0   | (0.0) | 0.0        | (0.0)     |
| Newfoundland and Labrador | 29.0‡ | (4.7) | 50.5  | (5.2)  | 5.4‡  | (2.3)   | 4.3‡     | (2.1) | 9.7‡  | (3.1) | 1.1‡       | (1.1)     |
| Canada                    | 21.8  | (3.4) | 50.6  | (3.5)  | 9.0   | (1.3)   | 12.9     | (2.9) | 3.4   | (0.8) | 2.3‡       | (0.6)     |

#### TABLE A.4.7.2 Percentage of schools by proportion of students who identify as Indigenous and language of the school system

|                           |       |        |       |        | Franco | ophone s | school s | ystems |       |        |            |              |
|---------------------------|-------|--------|-------|--------|--------|----------|----------|--------|-------|--------|------------|--------------|
|                           | 0     | %      | 1–    | 5%     | 6–1    | L0%      | 11–      | 25%    | 26–   | 50%    | More<br>50 | e than<br>0% |
| Canada and provinces      | %     | SE     | %     | SE     | %      | SE       | %        | SE     | %     | SE     | %          | SE           |
| British Columbia          | 0.0   | (0.0)  | 36.4‡ | (14.5) | 18.2‡  | (11.6)   | 45.5‡    | (15.0) | 0.0   | (0.0)  | 0.0        | (0.0)        |
| Alberta                   | 20.0‡ | (12.7) | 80.0‡ | (12.7) | 0.0    | (0.0)    | 0.0      | (0.0)  | 0.0   | (0.0)  | 0.0        | (0.0)        |
| Saskatchewan              | 16.7‡ | (15.2) | 83.3‡ | (15.2) | 0.0    | (0.0)    | 0.0      | (0.0)  | 0.0   | (0.0)  | 0.0        | (0.0)        |
| Manitoba                  | 0.0   | (0.0)  | 20.0‡ | (10.3) | 26.7‡  | (11.4)   | 33.3‡    | (12.2) | 20.0‡ | (10.3) | 0.0        | (0.0)        |
| Ontario                   | 18.0‡ | (3.8)  | 62.0  | (4.9)  | 14.0‡  | (3.5)    | 5.0‡     | (2.2)  | 1.0‡  | (1.0)  | 0.0        | (0.0)        |
| Quebec                    | 76.0  | (5.1)  | 19.0‡ | (4.2)  | 1.8‡   | (1.3)    | 0.0      | (0.0)  | 3.2‡  | (3.2)  | 0.0        | (0.0)        |
| New Brunswick             | 44.2‡ | (6.6)  | 50.4‡ | (6.7)  | 1.8‡   | (1.8)    | 3.5‡     | (2.5)  | 0.0   | (0.0)  | 0.0        | (0.0)        |
| Nova Scotia               | 11.1‡ | (10.5) | 88.9‡ | (10.5) | 0.0    | (0.0)    | 0.0      | (0.0)  | 0.0   | (0.0)  | 0.0        | (0.0)        |
| Prince Edward Island      |       |        |       |        |        |          |          |        |       |        |            |              |
| Newfoundland and Labrador |       |        |       |        |        |          |          |        |       |        |            |              |
| Canada                    | 53.4  | (4.1)  | 35.6  | (3.5)  | 5.4‡   | (1.2)    | 3.0‡     | (0.8)  | 2.6‡  | (1.9)  | 0.0        | (0.0)        |

‡ There are fewer than 30 observations.

|                           | Seme | ester | Full y | /ear  |
|---------------------------|------|-------|--------|-------|
| Canada and provinces      | %    | SE    | %      | SE    |
| British Columbia          | 22.6 | (5.2) | 77.4   | (5.2) |
| Alberta                   | 2.7‡ | (1.5) | 97.3   | (1.5) |
| Saskatchewan              | 0.6‡ | (0.6) | 99.4   | (0.6) |
| Manitoba                  | 0.0  | (0.0) | 100.0  | (0.0) |
| Ontario                   | 0.4‡ | (0.4) | 99.6   | (0.4) |
| Quebec                    | 2.8‡ | (2.2) | 97.2   | (2.2) |
| New Brunswick             | 0.0  | (0.0) | 100.0  | (0.0) |
| Nova Scotia               | 0.9‡ | (0.9) | 99.1   | (0.9) |
| Prince Edward Island      | 8.2‡ | (5.5) | 91.8‡  | (5.5) |
| Newfoundland and Labrador | 0.0  | (0.0) | 100.0  | (0.0) |
| Canada                    | 2.4  | (0.5) | 97.6   | (0.5) |

#### TABLE A.4.8.1 Percentage of schools by semester and full-year classes

<sup>‡</sup> There are fewer than 30 observations.

#### TABLE A.4.8.2 Percentage of schools by semester and full-year classes and language of school system

|                           | An    | Anglophone school systems Fra |        |       |       |        | Francophone school systems |        |  |  |  |
|---------------------------|-------|-------------------------------|--------|-------|-------|--------|----------------------------|--------|--|--|--|
|                           | Seme  | ester                         | Full y | /ear  | Sem   | ester  | Full                       | year   |  |  |  |
| Canada and provinces      | %     | SE                            | %      | SE    | %     | SE     | %                          | SE     |  |  |  |
| British Columbia          | 22.4‡ | (5.4)                         | 77.6   | (5.4) | 27.3‡ | (13.4) | 72.7‡                      | (13.4) |  |  |  |
| Alberta                   | 2.8‡  | (1.6)                         | 97.2   | (1.6) | 0.0   | (0.0)  | 100.0‡                     | (0.0)  |  |  |  |
| Saskatchewan              | 0.7‡  | (0.7)                         | 99.3   | (0.7) | 0.0   | (0.0)  | 100.0‡                     | (0.0)  |  |  |  |
| Manitoba                  | 0.0   | (0.0)                         | 100.0  | (0.0) | 0.0   | (0.0)  | 100.0‡                     | (0.0)  |  |  |  |
| Ontario                   | 0.4‡  | (0.4)                         | 99.6   | (0.4) | 1.0‡  | (1.0)  | 99.0                       | (1.0)  |  |  |  |
| Quebec                    | 1.9‡  | (1.9)                         | 98.1   | (1.9) | 3.0‡  | (2.6)  | 97.0                       | (2.6)  |  |  |  |
| New Brunswick             | 0.0   | (0.0)                         | 100.0  | (0.0) | 0.0   | (0.0)  | 100.0                      | (0.0)  |  |  |  |
| Nova Scotia               | 1.0‡  | (1.0)                         | 99.0   | (1.0) | 0.0   | (0.0)  | 100.0‡                     | (0.0)  |  |  |  |
| Prince Edward Island      | 9.5‡  | (6.4)                         | 90.5‡  | (6.4) |       |        |                            |        |  |  |  |
| Newfoundland and Labrador | 0.0   | (0.0)                         | 100.0  | (0.0) |       |        |                            |        |  |  |  |
| Canada                    | 2.4   | (0.5)                         | 97.6   | (0.5) | 2.5‡  | (1.6)  | 97.5                       | (1.6)  |  |  |  |

**‡** There are fewer than 30 observations.

|                           | 150 or | fewer | 151–200 |       | 201–250 |       | 251–300 |       | More<br>30 | than<br>0 |
|---------------------------|--------|-------|---------|-------|---------|-------|---------|-------|------------|-----------|
| Canada and provinces      | %      | SE    | %       | SE    | %       | SE    | %       | SE    | %          | SE        |
| British Columbia          | 2.5‡   | (1.5) | 33.9    | (7.1) | 32.1    | (7.5) | 16.2‡   | (6.9) | 15.3‡      | (4.4)     |
| Alberta                   | 1.5‡   | (1.5) | 18.4‡   | (6.4) | 50.0    | (7.2) | 23.3‡   | (6.2) | 6.7‡       | (3.1)     |
| Saskatchewan              | 2.2‡   | (1.7) | 10.9‡   | (2.9) | 34.7    | (4.7) | 34.5    | (4.8) | 17.8‡      | (4.1)     |
| Manitoba                  | 0.0    | (0.0) | 6.5‡    | (2.0) | 19.9    | (4.0) | 41.4    | (5.0) | 32.3       | (5.1)     |
| Ontario                   | 0.1‡   | (0.1) | 6.3‡    | (3.5) | 8.6‡    | (3.7) | 69.1    | (5.5) | 15.8       | (4.3)     |
| Quebec                    | 1.9‡   | (1.5) | 7.6‡    | (2.3) | 62.9    | (5.4) | 17.0    | (3.2) | 10.6‡      | (4.8)     |
| New Brunswick             | 0.8‡   | (0.8) | 0.0     | (0.0) | 9.1‡    | (2.5) | 46.8    | (4.3) | 43.4       | (4.3)     |
| Nova Scotia               | 0.9‡   | (0.9) | 2.7‡    | (1.5) | 3.8‡    | (1.9) | 78.4    | (3.9) | 14.3‡      | (3.3)     |
| Prince Edward Island      | 0.0    | (0.0) | 4.1‡    | (4.0) | 0.0     | (0.0) | 75.5‡   | (8.7) | 20.4‡      | (8.2)     |
| Newfoundland and Labrador | 1.1‡   | (1.1) | 1.1‡    | (1.1) | 7.4‡    | (2.7) | 63.9    | (5.0) | 26.6‡      | (4.6)     |
| Canada                    | 0.8‡   | (0.3) | 9.8     | (2.1) | 22.4    | (2.4) | 50.9    | (3.1) | 16.1       | (2.4)     |

TABLE A.4.9.1 Percentage of schools by minutes of mathematics instruction per week

<sup>‡</sup> There are fewer than 30 observations.

|                           | Anglophone school systems |       |       |       |       |       |       |       |            |            |  |
|---------------------------|---------------------------|-------|-------|-------|-------|-------|-------|-------|------------|------------|--|
|                           | 150 or                    | fewer | 151-  | -200  | 201-  | -250  | 251-  | -300  | More<br>3( | than<br>00 |  |
| Canada and provinces      | %                         | SE    | %     | SE    | %     | SE    | %     | SE    | %          | SE         |  |
| British Columbia          | 2.6‡                      | (1.5) | 32.7  | (7.3) | 32.5  | (7.7) | 16.7‡ | (7.1) | 15.5‡      | (4.5)      |  |
| Alberta                   | 1.6‡                      | (1.6) | 18.0‡ | (6.6) | 50.2  | (7.4) | 23.3‡ | (6.3) | 6.9‡       | (3.1)      |  |
| Saskatchewan              | 2.2‡                      | (1.7) | 10.8‡ | (3.0) | 34.3  | (4.7) | 34.7  | (4.9) | 18.0‡      | (4.2)      |  |
| Manitoba                  | 0.0                       | (0.0) | 5.7‡  | (2.1) | 20.2‡ | (4.2) | 40.3  | (5.3) | 33.8       | (5.3)      |  |
| Ontario                   | 0.0                       | (0.0) | 6.4‡  | (3.6) | 8.7‡  | (3.9) | 70.1  | (5.9) | 14.8‡      | (4.5)      |  |
| Quebec                    | 0.0                       | (0.0) | 7.7‡  | (3.7) | 55.8‡ | (6.9) | 30.8‡ | (6.4) | 5.8‡       | (3.2)      |  |
| New Brunswick             | 0.0                       | (0.0) | 0.0   | (0.0) | 10.4‡ | (3.5) | 74.0  | (5.0) | 15.6‡      | (4.1)      |  |
| Nova Scotia               | 1.0‡                      | (1.0) | 2.9‡  | (1.7) | 1.0‡  | (1.0) | 80.4  | (3.9) | 14.7‡      | (3.5)      |  |
| Prince Edward Island      | 0.0                       | (0.0) | 4.8‡  | (4.6) | 0.0   | (0.0) | 71.4‡ | (9.9) | 23.8‡      | (9.3)      |  |
| Newfoundland and Labrador | 1.1‡                      | (1.1) | 1.1‡  | (1.1) | 6.5‡  | (2.5) | 64.5  | (5.0) | 26.9‡      | (4.6)      |  |
| Canada                    | 0.6‡                      | (0.3) | 10.0  | (2.3) | 19.5  | (2.6) | 54.6  | (3.5) | 15.3       | (2.7)      |  |

#### TABLE A.4.9.2 Percentage of schools by minutes of mathematics instruction offered each week and language of the school system

|                           | Francophone school systems           |       |       |        |       |        |       |        |                  |       |  |
|---------------------------|--------------------------------------|-------|-------|--------|-------|--------|-------|--------|------------------|-------|--|
|                           | 150 or fewer 151–200 201–250 251–300 |       |       |        |       |        |       |        | More than<br>300 |       |  |
| Canada and provinces      | %                                    | SE    | %     | SE     | %     | SE     | %     | SE     | %                | SE    |  |
| British Columbia          | 0.0                                  | (0.0) | 72.7‡ | (13.4) | 18.2‡ | (11.6) | 0.0   | (0.0)  | 9.1‡             | (8.7) |  |
| Alberta                   | 0.0                                  | (0.0) | 33.3‡ | (15.7) | 44.4‡ | (16.6) | 22.2‡ | (13.9) | 0.0              | (0.0) |  |
| Saskatchewan              | 0.0                                  | (0.0) | 16.7‡ | (15.2) | 66.7‡ | (19.3) | 16.7‡ | (15.2) | 0.0              | (0.0) |  |
| Manitoba                  | 0.0                                  | (0.0) | 20.0‡ | (10.3) | 13.3‡ | (8.8)  | 60.0‡ | (12.7) | 6.7‡             | (6.4) |  |
| Ontario                   | 2.0‡                                 | (1.4) | 6.1‡  | (2.4)  | 7.1‡  | (2.6)  | 52.5  | (5.0)  | 32.3             | (4.7) |  |
| Quebec                    | 2.2‡                                 | (1.8) | 7.6‡  | (2.7)  | 64.3  | (6.4)  | 14.3‡ | (3.4)  | 11.6‡            | (5.7) |  |
| New Brunswick             | 1.8‡                                 | (1.8) | 0.0   | (0.0)  | 7.2‡  | (3.5)  | 9.0‡  | (3.8)  | 82.0             | (5.2) |  |
| Nova Scotia               | 0.0                                  | (0.0) | 0.0   | (0.0)  | 30.0‡ | (14.5) | 60.0‡ | (15.5) | 10.0‡            | (9.5) |  |
| Prince Edward Island      |                                      |       |       |        |       |        |       |        |                  |       |  |
| Newfoundland and Labrador |                                      |       |       |        |       |        |       |        |                  |       |  |
| Canada                    | 1.9‡                                 | (1.1) | 8.5   | (1.8)  | 43.1  | (4.4)  | 24.9  | (2.9)  | 21.6             | (3.6) |  |

‡ There are fewer than 30 observations.

|                | Mean score | SE     |
|----------------|------------|--------|
| Fewer than 200 | 484        | (14.5) |
| 201–250        | 508        | (3.8)  |
| 251–300        | 504        | (4.0)  |
| More than 300  | 499        | (3.5)  |

### TABLE A.4.9.3 Achievement in mathematics by minutes of mathematics instruction per week, Canada

#### TABLE A.4.10.1 Percentage of schools by daily mathematics instruction

|                           | Mathematics<br>instruction<br>offered daily |       | Mather<br>instruct<br>offered | matics<br>ion not<br>I daily |
|---------------------------|---|-------|-------------------------------|------------------------------|
| Canada and provinces      | %   | SE    | %                             | SE                           |
| British Columbia          | 41.3  | (7.5) | 58.7                          | (7.5)                        |
| Alberta                   | 77.5  | (5.0) | 22.5‡                         | (5.0)                        |
| Saskatchewan              | 85.5  | (3.1) | 14.5‡                         | (3.1)                        |
| Manitoba                  | 93.7  | (2.0) | 6.3‡                          | (2.0)                        |
| Ontario                   | 98.4  | (1.0) | 1.6‡                          | (1.0)                        |
| Quebec                    | 9.7‡  | (2.2) | 90.3                          | (2.2)                        |
| New Brunswick             | 97.0  | (1.5) | 3.0‡                          | (1.5)                        |
| Nova Scotia               | 98.2  | (1.2) | 1.8‡                          | (1.2)                        |
| Prince Edward Island      | 91.8‡                                       | (5.5) | 8.2‡                          | (5.5)                        |
| Newfoundland and Labrador | 89.5  | (3.2) | 10.5‡                         | (3.2)                        |
| Canada                    | 82.7  | (1.6) | 17.3                          | (1.6)                        |

|                           | Ang                        | glophone                    | school syste                  | ms                           | Francophone school system                   |        |                              | ems   |  |
|---------------------------|----------------------------|-----------------------------|-------------------------------|------------------------------|---|--------|------------------------------|---|--|
|                           | Mathe<br>instru<br>offeree | matics<br>Iction<br>d daily | Mather<br>instruct<br>offered | matics<br>ion not<br>I daily | Mathematics<br>instruction<br>offered daily |        | Mathe<br>instruct<br>offered | Mathematics<br>instruction not<br>offered daily |  |
| Canada and provinces      | %                          | SE                          | %                             | SE                           | %   | SE     | %                            | SE  |  |
| British Columbia          | 41.7                       | (7.7)                       | 58.3                          | (7.7)                        | 27.3‡                                       | (13.4) | 72.7‡                        | (13.4)  |  |
| Alberta                   | 78.7                       | (5.1)                       | 21.3‡                         | (5.1)                        | 30.0‡                                       | (14.5) | 70.0‡                        | (14.5)  |  |
| Saskatchewan              | 86.7                       | (3.1)                       | 13.3‡                         | (3.1)                        | 0.0   | (0.0)  | 100.0‡                       | (0.0)   |  |
| Manitoba                  | 93.4                       | (2.2)                       | 6.6‡                          | (2.2)                        | 100.0‡                                      | (0.0)  | 0.0                          | (0.0)   |  |
| Ontario                   | 98.5                       | (1.0)                       | 1.5‡                          | (1.0)                        | 95.0  | (2.2)  | 5.0‡                         | (2.2)   |  |
| Quebec                    | 39.6‡                      | (6.7)                       | 60.4                          | (6.7)                        | 3.7‡  | (1.8)  | 96.3                         | (1.8)   |  |
| New Brunswick             | 98.7                       | (1.3)                       | 1.3‡                          | (1.3)                        | 94.6  | (3.0)  | 5.4‡                         | (3.0)   |  |
| Nova Scotia               | 98.1                       | (1.4)                       | 1.9‡                          | (1.4)                        | 100.0‡                                      | (0.0)  | 0.0                          | (0.0)   |  |
| Prince Edward Island      | 90.5‡                      | (6.4)                       | 9.5‡                          | (6.4)                        |   |        |                              |   |  |
| Newfoundland and Labrador | 89.4                       | (3.2)                       | 10.6‡                         | (3.2)                        |   |        |                              |   |  |
| Canada                    | 89.1                       | (1.4)                       | 10.9                          | (1.4)                        | 37.8  | (3.5)  | 62.2                         | (3.5)   |  |

#### TABLE A.4.10.2 Percentage of schools by daily mathematics instruction and language of school system

‡ There are fewer than 30 observations.

*Note:* Due to small sample sizes, results for the francophone school systems are not reported for Prince Edward Island and Newfoundland and Labrador; however they are included in the calculations for the overall Canadian and provincial means.

#### TABLE A.4.10.3 Achievement in mathematics by daily mathematics instruction, Canada

|   | Mean score | SE    |
|---|------------|-------|
| Mathematics instruction offered daily     | 497        | (3.3) |
| Mathematics instruction not offered daily | 521*       | (4.4) |

\* Significant difference compared to the mathematics instruction offered daily category

|  | Activities<br>not provided |       | Activ<br>provi | ities<br>ided |
|--|----------------------------|-------|----------------|---------------|
|  | %                          | SE    | %              | SE            |
| Enrichment activities                                      |                            |       |                |               |
| Mathematics interventions                                  | 22.5                       | (2.4) | 77.5           | (2.4)         |
| Tutoring for students who have difficulty with mathematics | 44.2                       | (3.2) | 55.8           | (3.2)         |
| Enrichment mathematics                                     | 62.5                       | (2.8) | 37.5           | (2.8)         |
| Clubs  |                            |       |                |               |
| Robotics/coding club                                       | 54.6                       | (3.0) | 45.4           | (3.0)         |
| Chess club   | 65.8                       | (2.9) | 34.2           | (2.9)         |
| Academic club (other than mathematics)                     | 70.0                       | (2.5) | 30.0           | (2.5)         |
| Mathematics club   | 79.7                       | (2.2) | 20.3           | (2.2)         |
| Debating club or debating activities                       | 87.1                       | (1.5) | 12.9           | (1.5)         |
| Other clubs  | 16.2                       | (2.5) | 83.8           | (2.5)         |
| Other extracurricular activities                           |                            |       |                |               |
| Volunteering or service activities                         | 30.2                       | (3.0) | 69.8           | (3.0)         |
| Band, orchestra, or choir                                  | 35.7                       | (3.1) | 64.3           | (3.1)         |
| Play, improv, or school musical                            | 41.5                       | (3.1) | 58.5           | (3.1)         |
| Yearbook, newspaper, or magazine                           | 48.9                       | (3.1) | 51.1           | (3.1)         |
| Mathematics competitions                                   | 58.8                       | (3.0) | 41.2           | (3.0)         |

### TABLE A.4.11.1 Percentage of schools by enrichment and extracurricular activities, Canada

# TABLE A.4.11.2 Relationship between the enrichment and extracurricular activities index and mathematics achievement, Canada

|                               |               | •                    |               |                      |                            |       |               |        |
|-------------------------------|---------------|----------------------|---------------|----------------------|----------------------------|-------|---------------|--------|
|                               | Bottom        | quarter <sup>1</sup> | Second        | quarter <sup>2</sup> | Third quarter <sup>3</sup> |       | Top qu        | arter⁴ |
|                               | Mean<br>score | SE                   | Mean<br>score | SE                   | Mean<br>score              | SE    | Mean<br>score | SE     |
| Canada overall                | 486*          | (5.1)                | 495           | (6.4)                | 498                        | (7.4) | 522*          | (4.2)  |
| Language of the school system |               |                      |               |                      |                            |       |               |        |
| Anglophone                    | 484*          | (5.5)                | 495           | (6.8)                | 494                        | (8.6) | 520*          | (5.1)  |
| Francophone                   | 506*          | (9.7)                | 504           | (12.9)               | 520                        | (6.3) | 532           | (4.6)  |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

|  | Never |       | Ra   | rely  | Som  | Sometimes |      | ten   |
|--|-------|-------|------|-------|------|-----------|------|-------|
|  | %     | SE    | %    | SE    | %    | SE        | %    | SE    |
| Instructional materials (e.g., textbooks)                | 38.1  | (3.1) | 36.6 | (2.9) | 23.6 | (2.5)     | 1.7‡ | (0.6) |
| Budget for supplies                                      | 29.9  | (2.8) | 36.4 | (3.0) | 27.7 | (2.9)     | 5.9  | (1.2) |
| School building and grounds                              | 52.9  | (3.1) | 32.2 | (3.0) | 13.7 | (2.0)     | 1.2‡ | (0.3) |
| Instructional space                                      | 48.7  | (3.1) | 31.1 | (3.0) | 17.0 | (2.1)     | 3.2  | (0.9) |
| Computers for mathematics instruction                    | 40.4  | (3.2) | 30.8 | (2.6) | 24.2 | (2.7)     | 4.6  | (1.2) |
| Sufficient Internet access (e.g., bandwidth)             | 50.4  | (3.1) | 31.4 | (2.7) | 13.2 | (1.7)     | 5.0  | (1.2) |
| Library materials relevant to mathematics instruction    | 38.9  | (3.1) | 40.9 | (3.1) | 18.7 | (2.2)     | 1.5  | (0.4) |
| Digital resources/software for mathematics               | 35.3  | (3.2) | 37.3 | (2.8) | 24.3 | (2.7)     | 3.1  | (0.9) |
| Qualified mathematics teachers                           | 36.7  | (3.0) | 31.9 | (2.7) | 27.1 | (3.0)     | 4.3  | (1.1) |
| Qualified education assistants                           | 28.6  | (2.9) | 25.7 | (2.4) | 35.1 | (3.1)     | 10.6 | (1.6) |
| Mathematics specialist to support mathematics teachers   | 24.3  | (2.3) | 30.1 | (2.6) | 32.3 | (3.1)     | 13.3 | (2.6) |
| Availability of qualified substitute/occasional teachers | 10.5  | (2.2) | 19.2 | (2.0) | 38.0 | (2.9)     | 32.3 | (3.1) |

# TABLE A.4.12 Percentage of schools by challenges to providing instruction due to shortage or inadequacy of resources, Canada

<sup>‡</sup> There are fewer than 30 observations.

### TABLE A.4.13.1 Percentage of schools by challenges to learning index, Canada

|  | Not at all |       | A little |       | Mor<br>a l | More than<br>a little |      | lot   |
|--|------------|-------|----------|-------|------------|-----------------------|------|-------|
|  | %          | SE    | %        | SE    | %          | SE                    | %    | SE    |
| Challenges to learning index                               |            |       |          |       |            |                       |      |       |
| Student absenteeism (all excused absences)                 | 11.3       | (2.2) | 48.8     | (3.1) | 27.1       | (3.2)                 | 12.9 | (1.8) |
| Class composition  | 23.6       | (2.9) | 52.3     | (3.1) | 19.4       | (2.6)                 | 4.8  | (0.9) |
| Students intimidating or bullying other students           | 39.4       | (3.2) | 51.6     | (3.1) | 7.7        | (1.6)                 | 1.3‡ | (0.6) |
| Students skipping classes                                  | 62.2       | (2.9) | 29.8     | (2.7) | 5.7        | (1.1)                 | 2.4  | (0.7) |
| Student use of alcohol or illegal drugs                    | 78.5       | (2.0) | 19.0     | (1.8) | 1.6        | (0.5)                 | 0.9‡ | (0.5) |
| Other factors  |            |       |          |       |            |                       |      |       |
| Teachers' low expectations of students                     | 53.1       | (3.1) | 36.3     | (2.9) | 8.8        | (1.5)                 | 1.8‡ | (0.6) |
| Poor student-teacher relations                             | 42.1       | (3.2) | 48.4     | (3.1) | 8.4        | (1.6)                 | 1.1  | (0.3) |
| Disruption of classes by students                          | 16.2       | (2.4) | 56.9     | (3.0) | 18.5       | (2.1)                 | 8.5  | (1.8) |
| Teachers not meeting individual students' needs            | 26.7       | (2.8) | 56.7     | (3.1) | 14.8       | (2.0)                 | 1.8  | (0.4) |
| Teacher absenteeism  | 46.6       | (3.1) | 38.5     | (2.9) | 12.1       | (2.7)                 | 2.9‡ | (0.9) |
| Students lacking respect for teachers                      | 35.0       | (2.9) | 51.8     | (3.1) | 9.8        | (1.6)                 | 3.5  | (1.3) |
| Staff resisting change                                     | 25.0       | (2.6) | 55.9     | (3.0) | 14.3       | (1.9)                 | 4.8  | (1.1) |
| Teachers being too strict with students                    | 67.0       | (2.8) | 30.4     | (2.7) | 2.1        | (0.5)                 | 0.5‡ | (0.4) |
| Lack of time to provide instructional leadership for staff | 22.2       | (2.6) | 39.4     | (3.0) | 28.2       | (3.0)                 | 10.2 | (1.7) |

### TABLE A.4.13.2 Relationship between the challenges to learning index and mathematics achievement, Canada

|                               | Bottom quarter <sup>1</sup> |       | Second quarter <sup>2</sup> |       | Third quarter <sup>3</sup> |       | Top quarter <sup>₄</sup> |       |
|-------------------------------|-----------------------------|-------|-----------------------------|-------|----------------------------|-------|--------------------------|-------|
|                               | Mean<br>score               | SE    | Mean<br>score               | SE    | Mean<br>score              | SE    | Mean<br>score            | SE    |
| Canada overall                | 515*                        | (4.4) | 518                         | (4.8) | 490*                       | (7.8) | 483                      | (5.1) |
| Language of the school system |                             |       |                             |       |                            |       |                          |       |
| Anglophone                    | 511*                        | (4.7) | 520                         | (5.1) | 487*                       | (8.7) | 474                      | (5.7) |
| Francophone                   | 542*                        | (6.2) | 503*                        | (9.9) | 515                        | (5.2) | 522                      | (6.2) |

\* Significant difference between:

<sup>1</sup> Bottom quarter and top quarter

<sup>2</sup> Second quarter and bottom quarter

<sup>3</sup> Third quarter and second quarter

<sup>4</sup> Top quarter and third quarter

| other than school-sp      | onsored activities |        |       |        |         |         |
|---------------------------|--------------------|--------|-------|--------|---------|---------|
|                           | Less t             | han 5% | 5-:   | 10%    | More th | nan 10% |
| Canada and provinces      | %                  | SE     | %     | SE     | %       | SE      |
| British Columbia          | 45.1               | (8.1)  | 44.8  | (7.6)  | 10.2‡   | (3.1)   |
| Alberta                   | 57.4               | (7.3)  | 30.3  | (6.6)  | 12.3‡   | (5.8)   |
| Saskatchewan              | 48.2               | (5.1)  | 41.0  | (5.1)  | 10.7‡   | (3.2)   |
| Manitoba                  | 49.4               | (5.2)  | 43.7  | (5.3)  | 6.9‡    | (2.7)   |
| Ontario                   | 40.9               | (5.3)  | 54.1  | (5.4)  | 5.0‡    | (1.8)   |
| Quebec                    | 67.7               | (5.7)  | 29.2  | (5.7)  | 3.0‡    | (1.1)   |
| New Brunswick             | 49.8               | (4.3)  | 41.2  | (4.3)  | 9.0‡    | (2.5)   |
| Nova Scotia               | 37.2               | (4.6)  | 57.4  | (4.7)  | 5.4‡    | (2.2)   |
| Prince Edward Island      | 36.2‡              | (10.2) | 63.8‡ | (10.2) | 0.0     | (0.0)   |
| Newfoundland and Labrador | 36.1               | (5.0)  | 51.1  | (5.2)  | 12.8‡   | (3.4)   |
| Canada                    | 46.5               | (3.1)  | 46.6  | (3.2)  | 6.8     | (1.3)   |

# TABLE A.4.14.1 Percentage of schools by proportion of students absent on a typical day for reasons other than school-sponsored activities

|                           |        |        | Anglophone school systems |        |         |
|---------------------------|--------|--------|---------------------------|--------|---------|
|                           | Less t | han 5% | 5–10%                     | More t | 10% nan |
| Canada and provinces      | %      | SE     | % SE                      | %      | SE      |
| British Columbia          | 44.2   | (8.3)  | 45.3 (7.9)                | 10.5‡  | (3.2)   |
| Alberta                   | 57.2   | (7.5)  | 30.2 (6.7)                | 12.6‡  | (5.9)   |
| Saskatchewan              | 47.7   | (5.1)  | 41.4 (5.1)                | 10.9‡  | (3.2)   |
| Manitoba                  | 47.2   | (5.4)  | 45.5 (5.5)                | 7.3‡   | (2.9)   |
| Ontario                   | 39.3   | (5.5)  | 55.6 (5.7)                | 5.1‡   | (1.9)   |
| Quebec                    | 51.0‡  | (7.0)  | 39.2‡ (6.8)               | 9.8‡   | (4.2)   |
| New Brunswick             | 32.5‡  | (5.3)  | 54.5 (5.7)                | 13.0‡  | (3.8)   |
| Nova Scotia               | 34.3   | (4.7)  | 60.8 (4.8)                | 4.9‡   | (2.1)   |
| Prince Edward Island      | 35.0‡  | (10.7) | 65.0‡ (10.7)              | 0.0    | (0.0)   |
| Newfoundland and Labrador | 35.5   | (5.0)  | 51.6 (5.2)                | 12.9‡  | (3.5)   |
| Canada                    | 43.1   | (3.4)  | 49.4 (3.5)                | 7.5    | (1.5)   |

# TABLE A.4.14.2 Percentage of schools by proportion of students absent on a typical day for reasons other than school-sponsored activities and language of school system

|                           |         |        | Francophone school systems |        |         |
|---------------------------|---------|--------|----------------------------|--------|---------|
|                           | Less tl | nan 5% | 5–10%                      | More t | han 10% |
| Canada and provinces      | %       | SE     | % SE                       | %      | SE      |
| British Columbia          | 72.7‡   | (13.4) | 27.3‡ (13.4)               | 0.0    | (0.0)   |
| Alberta                   | 66.7‡   | (15.7) | 33.3‡ (15.7)               | 0.0    | (0.0)   |
| Saskatchewan              | 83.3‡   | (15.2) | 16.7‡ (15.2)               | 0.0    | (0.0)   |
| Manitoba                  | 86.7‡   | (8.8)  | 13.3‡ (8.8)                | 0.0    | (0.0)   |
| Ontario                   | 68.4    | (4.7)  | 28.6‡ (4.6)                | 3.1‡   | (1.7)   |
| Quebec                    | 71.0    | (6.7)  | 27.3‡ (6.8)                | 1.7‡   | (1.0)   |
| New Brunswick             | 73.5    | (5.9)  | 23.0‡ (5.6)                | 3.5‡   | (2.5)   |
| Nova Scotia               | 66.7‡   | (15.7) | 22.2‡ (13.9)               | 11.1‡  | (10.5)  |
| Prince Edward Island      |         |        |                            |        |         |
| Newfoundland and Labrador |         |        |                            |        |         |
| Canada                    | 70.8    | (4.2)  | 27.0 (4.2)                 | 2.2‡   | (0.8)   |

**‡** There are fewer than 30 observations.